

MONTHLY BULLETIN JANUARY











In 2006 all ECB publications will feature a motif taken from the €5 banknote.

# MONTHLY BULLETIN JANUARY 2006

#### © European Central Bank, 2006

#### Address

Kaiserstrasse 29 60311 Frankfurt am Main Germany

#### Postal address

Postfach 16 03 19 60066 Frankfurt am Main Germany

**Telephone** +49 69 1344 0

Website http://www.ecb.int

Fax +49 69 1344 6000

**Telex** 411 144 ecb d

This Bulletin was produced under the responsibility of the Executive Board of the ECB. Translations are prepared and published by the national central banks.

All rights reserved. Reproduction for educational and non-commercial purposes is permitted provided that the source is acknowledged.

The cut-off date for the statistics included in this issue was 11 January 2006.

ISSN 1561-0136 (print) ISSN 1725-2822 (online)



5

7

# CONTENTS

# EDITORIAL

## ECONOMIC AND MONETARY DEVELOPMENTS

Th	ne external environment of the euro area	7
M	onetary and financial developments	14
Pr	ices and costs	30
Οı	utput, demand and the labour market	37
	cchange rate and balance of payments velopments	46
Bo	oxes:	
1	Factors accounting for the rise in oil prices	11
2	Stock options, share buy-backs and their effects on equity markets: evidence from the United States	27
3	Assessing the reliability of Eurostat's euro area HICP flash estimate	31
4	The 2005-2008 Lisbon National Reform Programmes of the euro area countries	37
5	Some country-specific factors behind recent euro area employment developments	43
AR	TICLES	
	ne predictability of the ECB's onetary policy	5 I
	edge funds: developments and licy implications	63
EU	IRO AREA STATISTICS	S I
AN	INEXES	
	nronology of monetary policy easures of the Eurosystem	1
Do	ocuments published by the	
Εu	ropean Central Bank since 2005	
Gl	ossary	IX



# **ABBREVIATIONS**

# COUNTRIES

COONTRIES			
BE	Belgium	HU	Hungary
CZ	Czech Republic	MT	Malta
DK	Denmark	NL	Netherlands
DE	Germany	AT	Austria
EE	Estonia	PL	Poland
GR	Greece	РТ	Portugal
ES	Spain	SI	Slovenia
FR	France	SK	Slovakia
IE	Ireland	FI	Finland
IT	Italy	SE	Sweden
CY	Cyprus	UK	United Kingdom
LV	Latvia	JP	Japan
LT	Lithuania	US	United States
LU	Luxembourg		

# **OTHERS**

BIS	Bank for International Settlements
b.o.p.	balance of payments
BPM5	IMF Balance of Payments Manual (5th edition)
CD	certificate of deposit
c.i.f.	cost, insurance and freight at the importer's border
CPI	Consumer Price Index
ECB	European Central Bank
EER	effective exchange rate
EMI	European Monetary Institute
EMU	Economic and Monetary Union
ESA 95	European System of Accounts 1995
ESCB	European System of Central Banks
EU	European Union
EUR	euro
f.o.b.	free on board at the exporter's border
GDP	gross domestic product
HICP	Harmonised Index of Consumer Prices
HWWA	Hamburg Institute of International Economics
ILO	International Labour Organization
IMF	International Monetary Fund
MFI	monetary financial institution
NACE Rev. 1	Statistical classification of economic activities in the European Community
NCB	national central bank
PPI	Producer Price Index
SITC Rev. 3	Standard International Trade Classification (revision 3)
ULCM	unit labour costs in manufacturing
ULCT	unit labour costs in the total economy

In accordance with Community practice, the EU countries are listed in this Bulletin using the alphabetical order of the country names in the national languages.

4



# **EDITORIAL**

At its meeting on 12 January 2006, the Governing Council of the ECB decided, on the basis of its regular economic and monetary analyses, to keep the key ECB interest rates unchanged, following the increase of 25 basis points on 1 December 2005. The information which has become available since then supports the assessment that an adjustment of the very accommodative monetary policy stance was warranted. It remains essential to keep medium to long-term inflation expectations in the euro area solidly anchored at levels consistent with price stability. Such anchoring of inflation expectations is a prerequisite for monetary policy to make an ongoing contribution towards supporting economic growth and job creation in the euro area. Maintaining price stability over the medium term is the ECB's guiding principle, and the Governing Council will consistently apply it when examining new information, making judgements and taking decisions. With interest rates across the whole maturity spectrum remaining historically low in both nominal and real terms, and with the monetary policy stance remaining accommodative, the Governing Council will continue to monitor very closely all developments with respect to risks to price stability over the medium term.

Starting with the economic analysis underlying the Governing Council's assessment, the information available confirms that, as expected, real GDP growth improved in the second half of 2005. According to Eurostat's first estimate, real GDP grew at a quarter-on-quarter rate of 0.6% in the third quarter of 2005, compared with 0.4% in the second quarter. The breakdown of GDP data for the third quarter of 2005 confirmed a stronger contribution from domestic demand. Moreover, recent economic indicators and survey data support the view that the expansion of economic activity broadly maintained its momentum in the fourth quarter of 2005 and will continue to do so in the first months of 2006, notwithstanding the impact of high oil prices.

Looking further ahead, the conditions remain in place for sustained growth of economic activity,

in line with the Eurosystem staff projections and other available forecasts. On the external side, the continued strength of global demand should support euro area exports. On the domestic side, investment should further benefit from continued very favourable financing conditions, robust corporate earnings and gains in corporate efficiency. Consumption growth should gradually rise, broadly in line with expected developments in disposable income.

Risks to this outlook for economic growth continue to lie on the downside and relate to high and volatile oil prices, concerns about global imbalances and the level of consumer confidence in the euro area, although the latter is improving.

Turning to price developments, annual HICP inflation was 2.2% in December, according to Eurostat's flash estimate, compared with 2.3% in November and 2.5% in October. This decline was the result of some relaxation of earlier tensions in oil and petrol markets. Nevertheless, annual HICP inflation rates are expected to remain at elevated levels over the short term, mainly on account of the most recent increases in oil prices and some adverse base effects. Beyond the short term, indirect effects of past oil price rises on other components of the price index may gradually materialise, and already announced changes to administered prices and indirect taxes can be expected to have an upward impact. Meanwhile, wage increases have remained moderate over recent quarters. All in all, the information available remains consistent with the scenario for price developments reflected in the December staff projections.

Risks to this scenario remain on the upside and include further rises in oil prices, additional increases in administered prices and indirect taxes, as well as – more fundamentally – potential second-round effects in wage and price-setting behaviour. It is therefore crucial that the social partners continue to meet their responsibilities, also in the context of a more favourable economic environment.



Turning to the monetary analysis, the annual growth rate of M3 moderated somewhat in November, but remained very robust, mainly owing to the stimulative impact of the prevailing low level of interest rates. The strong growth of M3 continues to be driven by significant contributions from its most liquid components. The growth of loans to the private sector – and, in particular, mortgage borrowing - has strengthened further over recent months, from already rapid rates of growth. Against this background, price dynamics in the housing markets need to be monitored closely. Liquidity in the euro area remains ample by all plausible measures. Strong monetary and credit growth in a context of already ample liquidity in the euro area points to upside risks to price stability over medium to longer horizons.

To sum up, the economic analysis suggests that some upward impact on HICP inflation will result from the indirect effects of recent oil price rises and already announced changes to administered prices and indirect taxes. It also indicates that risks to price stability over the medium term remain on the upside. This assessment is confirmed by cross-checking the economic analysis with the monetary analysis. It is essential that such risks do not affect medium-term inflation expectations, which need to remain firmly anchored at levels consistent with price stability. Monetary policy can thereby effectively contribute to sustainable economic growth and job creation. Accordingly, the Governing Council will continue to monitor very closely all developments with respect to risks to price stability over the medium term.

As regards fiscal policy, most euro area countries have submitted their updated stability programmes, which include their medium-term budget plans. The upcoming assessment of these programmes by the ECOFIN Council and their subsequent implementation provide an opportunity to forcefully underpin the commitment to sound fiscal policies and the rigorous implementation of the Stability and Growth Pact, which would have an important positive effect on confidence. This effect is likely to be considerable if triggered by welldefined and credible consolidation measures, restraint in expenditure commitments, and the incorporation of fiscal measures into a comprehensive and growth-friendly reform agenda.

With respect to structural reforms, the Governing Council welcomes the ECOFIN Council's Conclusions of 6 December 2005 on the Lisbon National Reform Programmes and the intended response to the challenges of globalisation. In fact, the process of international economic and financial integration, characterised by strong growth in trade and capital flows, has been one of the driving factors behind the rise in Europe's prosperity over the past decades. The ongoing transformation of the world economy, reflecting technological advances and the entrance of new economies into the world market, again offers great opportunities in terms of higher living standards. In order to translate these chances into achievements, Europe would greatly benefit from more flexible labour and product markets so as to speed up the necessary changeover from contracting to expanding activities and to minimise adjustment costs. It would also profit considerably from a more stimulative business environment which fosters the ability to innovate, invest and create new firms. Moreover, a fully operational EU internal market, including for services, offers great opportunities. The initiatives taken to relaunch the Lisbon strategy are a welcome step in the right direction.

This issue of the Monthly Bulletin contains two articles. The first article reviews the main conceptual issues relating to the predictability of the ECB's monetary policy and discusses how a transparent monetary policy strategy and constant communication have allowed the ECB to achieve a high level of overall predictability. The second article describes the main features of the hedge fund industry, with an emphasis on the European dimension, and provides an overview of recent developments in the industry and the current policy debate on regulation.



# I THE EXTERNAL ENVIRONMENT OF THE EURO AREA

The global economy continues to expand at a relatively robust pace, with particularly strong activity in the United States and Asia. In late 2005, consumer price inflation fell in a number of countries. Overall, the outlook for the external environment and euro area external demand remains favourable.

#### **I.I DEVELOPMENTS IN THE WORLD ECONOMY**

The global economy continues to expand at a relatively robust pace, with particularly strong activity in the United States and Asia. Survey evidence suggests that, at the moment, global activity is being supported by the services sector in particular, with a renewed moderation in the manufacturing sector of some countries, most notably the United States. Partly reflecting the decline in oil prices following the spike caused by the hurricanes in the United States, annual consumer price inflation moderated somewhat in a number of countries in late 2005 (see Chart 1). For the OECD countries, annual CPI inflation receded to 2.6% in November, after a high of 3.3% in September. Excluding food and energy, however, it increased slightly to 1.9% in November.



Chart | Price developments in the OECD

#### **UNITED STATES**

In the United States, economic activity expanded solidly in the third quarter. Real GDP growth increased to an annualised rate of 4.1% from 3.3% in the second quarter. Personal consumption, business investment in equipment and software, and federal government spending were the main contributors to this increase. A small negative contribution stemmed from net trade due to a deceleration in real exports and a return to positive real import growth.

Recent activity indicators show that the momentum of economic expansion remains strong. According to the latest available information, industrial production has fully recovered from the output losses suffered as a result of the hurricanes. Likewise, manufacturing activity continued to be fairly robust, although it may have slowed down slightly recently. Employment conditions seem to be improving fairly steadily after a short period of weakness, primarily as a result of job losses in hurricane-affected regions.

Private consumption spending continued to grow rapidly, partly due to improving employment conditions, while business fixed investment, especially in equipment and software, also remained strong. Annual headline inflation declined further to 3.5% in November because of a sharp decline in energy prices, while annual CPI excluding food and energy stood at 2.1%, as in October.

The outlook for real GDP growth remains positive, although there are signs that the brisk pace of economic activity decelerated at the end of 2005. In particular, higher interest rates and a possible moderation in housing prices could restrain private consumption spending in the context of high household indebtedness and a negative personal saving rate. With inflation expectations remaining

#### ECONOMIC AND MONETARY DEVELOPMENTS

The external environment of the euro area contained, price pressures are likely to continue to retreat provided that increasing resource utilisation does not lead to a pick-up in wages and an acceleration in unit labour costs.

As regards monetary policy, the US Federal Open Market Committee decided on 13 December to raise its target for the federal funds rate by 25 basis points for the thirteenth consecutive meeting, bringing the rate to 4.25%. Moreover, the statement released after the meeting mentioned that "the Committee judges that some further measured policy firming is likely to be needed to keep the risks to the attainment of both sustainable economic growth and price stability roughly in balance".

#### JAPAN

In Japan, the economy continues along a path of gradual recovery while consumer price deflation is abating. Nevertheless, revised national accounts data confirm that growth in economic activity was rather weak in the third quarter of 2005, with real GDP growth at 0.2% quarter on quarter. This figure represents a downward revision from the initial estimate (0.4% on a quarterly basis) and was largely determined by lower contributions to GDP growth from public spending and private inventories.

More recently, however, the Bank of Japan's Tankan survey for December 2005 showed a further improvement in Japanese firms' assessment of business conditions. Although relatively small, this improvement was rather broadly based across firm sizes and sectors. In particular, the net percentage of large manufacturers reporting a favourable assessment of business conditions increased, which is a

# Chart 2 Main developments in major industrialised economies



Statistical data and pipe, Eurostat and PCB cardinations.
 Eurostat data are used for the euro area and the United Kingdom; for the United States and Japan, national data are used. GDP figures have been seasonally adjusted.
 HICP for the euro area and the United Kingdom; CPI for the United States and Japan.

positive sign, given that their assessment of business conditions has historically provided a reliable indication of the state of the general business cycle.

As regards price developments, the annual rate of change in CPI excluding fresh food turned marginally positive (+0.1%) in November for the first time in two years. By contrast, headline CPI continued to decline (-0.8% on an annual basis) over the same period. Consumer prices are expected to pick up over the next few months as deflationary pressures abate.

At its meeting on 16 December 2005, the Bank of Japan decided to maintain its target for the outstanding balance of current accounts at around 30 to 35 trillion yen. At the same time, it

The external environment of the euro area

reiterated that when liquidity demand is judged to be exceptionally weak due to technical factors, the balance may be allowed to fall below the lower bound of the target.

## UNITED KINGDOM

Compared with 2004, economic activity in the United Kingdom in 2005 remained relatively subdued, with quarterly real GDP growing between 0.3% and 0.5% over the first three quarters. Household consumption accelerated moderately in the third quarter and retail sales grew at a robust pace in October and November. Business investment in the third quarter, however, was weak compared with the previous quarter, while sustained growth in imports and a contraction in exports led to a negative contribution of net exports to GDP. Looking ahead, economic activity is likely to regain momentum, though the pace of the recovery remains uncertain.

In November, annual HICP inflation decreased for the second consecutive month to 2.1%. This decline was driven primarily by a fall in fuel and lubricant prices. Furthermore, following an extended period of moderation in the annual growth rate of residential property prices, there are some indications that the housing market has strengthened again.

## **OTHER EUROPEAN COUNTRIES**

In most of the other non-euro area EU countries, output growth picked up or remained robust in the third quarter, and the economic outlook continues to be favourable. HICP inflation in November generally decreased somewhat as pressures stemming from energy prices eased.

In Denmark and Sweden, the quarterly rate of real GDP growth strengthened further to 1.4% and 1.0% respectively in the third quarter of 2005. The composition of output growth seems to be well balanced in both countries, with domestic and net foreign demand contributing to growth. Despite rising import prices, inflation remained subdued in both countries, reflecting intensive competition in retailing, moderate wage increases and strong productivity growth. Annual HICP inflation in November remained at 1.9% in Denmark, but picked up somewhat to 1.2% in Sweden.

In the three largest new EU Member States, output growth remained strong in the third quarter, mainly bolstered by a favourable export performance and a gradual improvement in domestic demand. In the Czech Republic, real GDP grew by 4.9% year on year, favoured in particular by a positive development in the external sector. In Hungary, real GDP growth was 4.4% year on year, driven mainly by investment, while in Poland annual GDP growth increased to 3.7%, largely due to improvements in domestic demand. In the Czech Republic and Poland, annual HICP inflation declined in November to 2.2% and 1.1% respectively, partly reflecting positive developments in food prices and lower growth in energy prices. In Hungary, however, it increased slightly to 3.3%, driven by increases in food and services prices.

## **NON-JAPAN ASIA**

In non-Japan Asia, economic growth remained robust at the end of 2005. Export growth has shown further upward momentum in most major economies in the region, except for China. At the same time, domestic demand has continued to expand steadily in most countries, despite high oil prices and monetary policy tightening. In November, inflationary pressures were generally rather moderate in the region.

In China, the economy continues to expand rapidly, driven increasingly by domestic demand, while exports have been decelerating. In November, retail sales and urban fixed asset investment rose by about 12% and 29% respectively on an annual basis. In both November and December, year-

on-year growth in exports declined, falling below 20% for the first time in nearly two years. In 2005, the trade surplus reached a record level of about USD 102 billion. With regard to price developments, inflationary pressures continue to be muted, with annual CPI inflation rising marginally to 1.3% in November. On 20 December 2005, following China's first nationwide comprehensive economic survey, China's National Bureau of Statistics revised upwards nominal GDP for 2004 by CNY 2.3 trillion, or 16.8% of that previously reported. This revision was prompted mainly by the improved measurement of China's services sector, which turned out to be almost 50% larger than previously calculated.

Economic prospects for non-Japan Asia remain favourable, underpinned by steady growth in domestic demand and a continued revival in export growth. High oil prices and the resultant pressures on headline inflation remain a downward risk to the region's benign outlook.

#### LATIN AMERICA

While the prospects for Latin America as a whole are generally positive, there appears to be more differentiation in the pace of economic activity in individual countries. In Brazil, third-quarter real GDP declined by 1.2% quarter on quarter, marking the first GDP contraction in two years against the background of subdued industrial activity. In mid-December, the central bank cut its key reference rate by another 50 basis points to 18%. By contrast, real GDP growth in Mexico has been rather buoyant, expanding by 3.3% year on year in the third quarter of 2005, partly as a result of robust consumption and investment demand. In December, the Bank of Mexico also cut interest rates by another 50 basis points to 8.25%. Finally, economic momentum in Argentina remained strong, with real GDP growing by 9.2% year on year in the third quarter of 2005. There is some early evidence that this trend was also maintained thereafter, with industrial production rising between 9% and 10% year on year both in October and November. Overall, prospects for the region remain favourable, with the gradual strengthening of domestic demand expected to compensate for an anticipated moderation in external demand.

### **I.2 COMMODITY MARKETS**

Oil prices rebounded in December and early January, with the price of Brent crude reaching USD 62.2 on 11 January 2006. The price of Brent now stands at around only 8% below the peak levels reached in the immediate aftermath of the hurricanes in the US Gulf of Mexico in early September, but is still 56% higher than at the start of 2005. Given the temporary nature of the relatively weak demand in the second half of 2005, the International Energy Agency (IEA) expects oil demand to rebound in 2006 and continue supporting prices. Global oil supply increased considerably in October and November, led by a partial recovery in US production. Meanwhile, OPEC crude oil production also increased, but only marginally. At its meeting on 12 December 2005, OPEC delegates indicated that they were content with



Monthly Bulletin January 2006

The external environment of the euro area

the current oil price and supply/demand balance and therefore decided to leave output quotas unchanged. The final communiqué said that the ceiling of 28 million barrels per day agreed in June 2005 "will be adequate to balance the market for the first quarter of the year". However, it was also mentioned that, in view of the supply/demand outlook for the second and third quarters of 2006, when demand is seasonally lower, OPEC may consider reducing production in the near future. In early January, concerns about the reliability of Russian gas supplies to Europe have put – possibly only temporarily – additional upward pressure on oil prices. Limited spare capacity throughout the oil supply chain, and therefore high sensitivity to unanticipated changes in the supply/demand balance, are bound to keep oil prices both relatively high and volatile in the near term (see Box 1 on "Factors accounting for the rise in oil prices").

Having remained broadly stable over the previous six months, non-energy commodity prices rose in November and December as the prices of both industrial raw materials and food increased. Expressed in US dollar terms, non-energy commodity prices were approximately 15% higher in December than one year earlier.

#### Box I

#### FACTORS ACCOUNTING FOR THE RISE IN OIL PRICES

Oil prices have soared since early 2004, reflecting the increased tightness in the oil markets that have become highly sensitive to shocks. The price of Brent crude oil (which serves as a

benchmark) increased from around USD 30 in January 2004 to an all-time high of USD 67.5 in September 2005. Thereafter, it has decreased only slightly, standing at USD 62.2 on 11 January 2006. This strong rise in oil prices is largely attributable to the erosion of spare capacity and the emergence of bottlenecks throughout the oil supply chain following unexpectedly strong demand. This box highlights that, in this context, some of the historical relationships between oil market fundamentals (i.e. supply, demand, inventories, capacity) and prices appear to have changed since 2004.

Notably, the recent rise in oil prices seems to reflect non-linearities in the relationship

# Chart A OPEC capacity utilisation and oil prices (quarterly data; Q1 1995-Q3 2005)



between oil prices and the quantities supplied to the market. Indeed, marginal increases in the volumes of oil produced and consumed are now associated with larger rises in the corresponding prices, as production capacities are reaching their limits. Since it often takes years to build additional extraction and refining capacity, the price of oil has become more sensitive to the quantity supplied as it approaches production capacity.

OPEC spare capacity acts as a buffer against unexpected supply disruptions or unexpected surges in demand. As shown in Chart A, the slope of the relationship between OPEC capacity utilisation and oil prices tends to become much steeper as the utilisation rate approaches 100%. The unexpected strength of demand, which in 2004 grew at its fastest pace in three decades, eroded spare capacity along the entire oil supply chain, as investments by oil companies had been driven by projections of a "normal" pace of demand growth.

#### Chart B Total oil supply and prices

(monthly data; January 1997-October 2005)





When total quantities of oil supplied are plotted against corresponding prices (see Chart B), the period 1997-2005 can be split into two sub-samples: from 1997 to the first half of 2004 and from the second half of 2004 to September 2005. The slope of the interpolation line for the most recent sub-sample has steepened significantly compared with the past. This is consistent with the hypothesis that capacity constraints in the oil industry have led to the emergence of non-linearities in the relationship between quantity and prices.

Moreover, the unexpected strong demand has eroded not only spare production capacity but also spare refining capacity. The oil industry's flexibility to switch between refining different types of oil has consequently decreased and the premium on light and sweet grades of crude oil (e.g. Brent) compared with heavy and sour grades (e.g. Dubai) has increased markedly since mid-2004 (see Chart C).

A change in the functioning of oil markets can also be illustrated by a break in the relationship between oil inventories and prices. Prior to 2004, higher oil prices were generally associated with lower industry-held oil inventories, as the higher the oil prices the

#### Chart C Premium of Brent crude oil over Dubai Fateh

(USD per barrel; 30-day moving average)



The external environment of the euro area



greater the cost of holding oil inventories. The historical negative relationship between levels of industry-held oil inventories and oil prices has broken down since 2004. It may be that heightened concerns over the security of oil supplies have been an incentive to hold higher levels of inventories as a buffer against possible future supply disruptions, despite rising prices (see Chart D). In this context, rising prices may have been interpreted as a signal of future price increases.

The capacity constraints that have emerged throughout the oil supply chain need to ease in order for oil markets to become less sensitive to shocks. Additional investments would allow capacity utilisation to return to levels consistent with a normal functioning of oil markets. Initiatives to foster a more efficient use of energy sources should also be encouraged.<sup>1</sup>

1 See also statement by G7 Finance Ministers and Central Bank Governors, London, 2-3 December 2005, and the press release of the Economic and Financial Affairs Council Meeting, Brussels, 6 December 2005.

# **1.3 OUTLOOK FOR THE EXTERNAL ENVIRONMENT**

Overall, the outlook for the external environment and thus for euro area external demand remains positive. The robust profit situation and favourable financing conditions should continue to support corporate investment spending. This encouraging outlook is supported by the fact that the six-month rate of change in the OECD composite leading indicator increased further in November, the seventh consecutive month of increase. The renewed increase in oil prices in response to concerns about the reliability of Russian energy supplies highlights the ongoing risk to the global economy stemming from oil markets. Moreover, the persistence of economic imbalances at the global level remains a significant risk for the world economy.



# **2 MONETARY AND FINANCIAL DEVELOPMENTS**

## 2.1 MONEY AND MFI CREDIT

While remaining at a strong rate of 7.6%, annual M3 growth moderated somewhat further in November 2005, as signs detected in October of a resumption of the unwinding of past portfolio shifts became more evident. Nonetheless, underlying monetary and credit dynamics remained strong, with the prevailing low level of interest rates continuing to be the main driver of monetary expansion. The ongoing strong growth of money and credit in an environment of already ample liquidity points to increasing upside risks to price stability over medium- to longer-term horizons. Moreover, these developments imply a need to monitor asset price dynamics closely, given the potential for price misalignments to emerge.

#### THE BROAD MONETARY AGGREGATE M3

The annual growth rate of the broad monetary aggregate M3 declined to 7.6% in November 2005, from 8.0% in October. The three-month average of the annual M3 growth rates in the period between September and November 2005 decreased to 8.0%, from 8.2% in the period between August and October. The decline in the annual rate of growth in November is a reflection of two consecutive subdued month-on-month growth rates, after a succession of relatively strong monthly increases over the summer. As a result, the annualised six-month rate of growth of M3, which in September had reached its highest level since the start of Stage Three of EMU, moderated further in November but nonetheless stayed above the annual growth rate (see Chart 4).

The November data suggest that the prevailing low level of interest rates has remained the main driver of strong M3 growth. This view is supported by the continued strong contribution to the annual growth rate of M3 stemming from the most liquid components of M3 contained in the narrow aggregate M1 and by the further strengthening of the robust demand for MFI loans by the private sector. At the same time, developments reflecting a possible unwinding of past portfolio shifts, last observed in the period between mid-2003 and mid-2004, exerted a dampening effect on M3 growth.<sup>1</sup> The signs of a resumption of this unwinding, which had already been detected in October, became more evident with the most recent available data.

Given the robust growth in M3 over the most recent quarters, liquidity in the euro area remains ample. This implies increasing risks to price stability over the medium to longer term, especially if a significant part of the ample liquidity were to be transformed into transaction

1 See the box entitled "Approaches to identifying and estimating portfolio shifts into and out of M3" in the January 2005 issue of the Monthly Bulletin for further information.



Monetary and financial developments

balances in an environment of strengthening confidence and real economic activity. In addition, strong monetary and credit growth in the context of ample liquidity implies a need to monitor asset price dynamics carefully, given the potential for price misalignments to emerge.

## MAIN COMPONENTS OF M3

The modest decline in the annual rate of growth of M3 in November was broadly based across its components. As regards M1, while this aggregate continued to expand at a robust pace, the annual growth rates of both currency in circulation and overnight deposits moderated in November. Nonetheless, M1 continued to make the largest overall contribution to the level of annual M3 growth. The annual rate of growth of short-term deposits other than overnight deposits remained broadly unchanged in November (see Table 1). The current strength of demand for short-term deposits is likely to reflect the low opportunity cost of holding these assets in an environment of low interest rates.

The annual growth rate of marketable instruments included in M3 declined somewhat in November, largely on account of a fall in the annual growth rate of money market fund shares/units. The moderation of the growth rate of money market fund shares/units – assets which are held by households and firms to park liquidity at times of heightened uncertainty – supports the view that the unwinding of past portfolio shifts may have regained some strength in recent months.

The annual growth rate of short-term deposits and repurchase agreements held by the private sector with MFIs (excluding the Eurosystem) – the broadest aggregation of M3 components for which information by holding sector is available – decreased further in November. After having risen substantially from mid-2004, the contribution to the development of this aggregation of assets stemming from financial intermediaries other than MFIs declined in October and November. By contrast, the upward trend in the corresponding contribution from non-financial corporations continued.

## Table I Summary table of monetary variables

(quarterly figures are averages; adjusted for seasonal and calendar effects)							
	Outstanding amount						
	as a percentage of M3 <sup>1)</sup>	2004 Q4	2005 Q1	2005 Q2	2005 Q3	2005 Oct.	2005 Nov.
M1	48.2	9.3	9.6	9.8	11.3	11.2	10.6
Currency in circulation	7.4	19.1	18.0	17.3	16.0	15.3	14.6
Overnight deposits	40.8	7.7	8.2	8.5	10.4	10.4	9.9
M2 - M1 (= other short-term deposits)	37.6	3.5	4.5	5.0	5.5	6.0	5.9
Deposits with an agreed maturity of up to							
two years	15.5	-2.4	0.5	2.6	4.5	6.1	6.7
Deposits redeemable at notice of up to							
three months	22.0	7.4	7.1	6.6	6.0	5.7	5.2
M2	85.7	6.4	7.1	7.5	8.5	8.6	8.3
M3 - M2 (= marketable instruments)	14.3	3.9	4.0	4.4	5.6	4.0	3.5
M3	100.0	6.1	6.7	7.1	8.0	8.0	7.6
Credit to euro area residents		6.0	6.5	6.6	7.0	7.8	8.1
Credit to general government		3.7	3.4	2.2	1.2	1.9	3.4
Loans to general government		0.4	-0.3	-0.7	-0.9	0.1	-0.2
Credit to the private sector		6.7	7.3	7.8	8.6	9.3	9.4
Loans to the private sector		6.9	7.3	7.6	8.4	8.9	9.0
Longer-term financial liabilities							
(excluding capital and reserves)		8.9	9.5	9.7	10.0	9.7	9.4

Source: ECB.

1) As at the end of the last month available. Figures may not add up due to rounding.



#### MAIN COUNTERPARTS OF M3

On the counterparts side, the annual growth rate of MFI loans to the private sector edged up further, to 9.0% in November, from 8.9% in October. The continued robust growth of MFI loans was broadly based across sectors, reflecting the stimulative impact of the low level of interest rates and possibly also improved confidence in some sectors of the economy.

MFI loans to households continued to be driven mainly by the dynamism of loans for house purchase, which grew at an annual rate of 11.2% in November, after 10.9% in October (see Table 2). The strong borrowing for house purchase reflects the environment of low mortgage lending rates in the euro area as a whole and the strength of housing market developments in many regions. In addition, consumer credit continued to accelerate and reached an annual rate of 8.0% in November. Growth in MFI loans to non-financial corporations, which has strengthened since early 2004, increased further in November, reflecting increases in the demand for loans with a medium- to long-term maturity.

In November, both main components of MFI credit to euro area residents strengthened further. The annual growth rate of credit to general government increased markedly, driven by the demand for securities other than shares, and growth in credit to the private sector remained robust.

Among the other counterparts of M3, the annual growth rate of MFI longer-term financial liabilities (excluding capital and reserves) declined slightly to 9.4% in November, from 9.7% in October, but nevertheless remained strong. The robust demand for this asset class is evident in the dynamics of both debt securities issued with a maturity of over two years and deposits with an agreed maturity of over two years. The ongoing robust demand for MFI longer-term financial liabilities supports the existence of an inclination in the euro area money-holding sector to invest in longer-term euro area financial instruments.

#### Table 2 MFI loans to the private sector

(quarterly figures are averages; not adjusted for seasonal and calendar effects)

	Outstanding amount		Ann	ual growth	rates		
	as a percentage	2004	2005	2005	2005	2005	2005
	of the total <sup>1)</sup>	Q4	Q1	Q2	Q3	Oct.	Nov.
Non-financial corporations	41.1	5.1	5.7	6.1	6.9	7.3	7.5
Up to one year	30.6	1.6	3.2	4.6	5.8	5.9	5.2
Over one and up to five years	17.4	5.8	6.7	6.4	6.3	7.4	7.7
Over five years	52.0	7.0	6.9	7.0	7.9	8.1	8.8
Households <sup>2)</sup>	50.5	7.9	8.1	8.2	8.6	9.0	9.2
Consumer credit <sup>3)</sup>	13.2	6.2	6.4	6.7	6.8	7.7	8.0
Lending for house purchase3)	69.3	9.9	10.1	10.2	10.7	10.9	11.2
Other lending	17.4	1.9	2.2	2.1	2.2	3.1	3.0
Insurance corporations and pension funds	0.9	14.4	23.0	14.4	16.5	24.0	37.3
Other non-monetary financial intermediaries	7.5	10.2	10.6	11.4	15.7	15.6	12.9

#### Source: ECB.

Notes: MFI sector including the Eurosystem; sectoral classification based on the ESA 95. For further details, see the relevant technical notes. 1) As at the end of the last month available. Sector loans as a percentage of total MFI loans to the private sector; maturity breakdown and breakdown by purpose as a percentage of MFI loans to the respective sector. Figures may not add up due to rounding. 2) As defined in the ESA 95.

3) The definitions of consumer credit and lending for house purchase are not fully consistent across the euro area



Monetary and financial developments

The annual flow in the net external asset position of MFIs turned negative in November, for the first time since December 2001. A net outflow of €2 billion was recorded for the euro area over the twelve months to November, compared with a net inflow of €51 billion in the year to October (see Chart 5). The decline in the annual flow reflected a series of net monthly outflows since August 2005. According to the latest evidence, although foreign investors have been the main driving force behind developments in the net external asset position of MFIs, investments by euro area residents abroad seem to have regained some momentum over recent months.

Summing up the information from the counterparts, the prevailing low level of interest rates fostered the increasing dynamism of MFI loans to the private sector, which continued to support strong M3 growth. At the same time, the strong demand for MFI longer-term financial liabilities and the recent developments in the MFI net external asset position have contributed to the moderation in monetary dynamics.



Notes: M3 is shown for reference only (M3 = 1+2+3-4+5). Longer-term financial liabilities (excluding capital and reserves) are shown with an inverted sign, since they are liabilities of the MFI sector.

## **2.2 SECURITIES ISSUANCE**

In October 2005 the annual growth rate of debt securities issued by euro area residents remained strong. The annual growth rate of debt securities issued by non-financial corporations rose somewhat from the low level observed in the previous month. At the same time, the growth rate of debt securities issued by MFIs was unchanged at robust levels. The annual growth rate of quoted shares issued by euro area residents remained unchanged at the high levels recorded since last July. However, this high level is to a large extent due to a revision of statistical data.

#### **DEBT SECURITIES**

The annual growth rate of debt securities issued by euro area residents increased slightly in October, and stood at 7.4%. This increase continued to be largely driven by a robust growth in long-term debt securities – around 8% – while the annual growth rate of short-term debt securities issued by euro area residents remained relatively low at 2.4%, even though the latter was up on the level recorded in September (see Table 3). The relatively high annual growth rate of long-term debt securities was mainly driven by strong issuance at variable rates, whereas issuance at fixed rates was more moderate.

Looking at issuance by sectors, the annual growth rate of debt securities issued by non-financial corporations increased somewhat in October, to 3.9%, up from 2.8% in September, although it remained low, overall, in comparison with the levels seen in 2003 (see Chart 6). This upward trend is confirmed by the analysis of seasonally adjusted data. The three-month growth rate showed a

### Table 3 Securities issued by euro area residents

	Amount outstanding (EUR billions)	Annual growth rates <sup>1)</sup>					
	2005	2004	2005	2005	2005	2005	2005
Issuing sector	Oct.	Q4	Q1	Q2	Q3	Sep.	Oct.
Debt securities:	10,160	7.2	7.6	7.7	7.6	7.3	7.4
MFIs	4,087	9.9	10.1	9.8	10.0	9.4	9.4
Non-monetary financial corporations	852	9.9	11.6	17.2	19.3	20.6	21.0
Non-financial corporations	627	3.1	2.9	4.6	2.2	2.8	3.9
General government of which:	4,594	5.3	5.5	4.9	4.5	4.2	4.1
Central government	4,321	4.8	5.1	4.5	4.0	3.8	3.6
Other general government	273	14.5	13.7	12.0	11.9	11.5	11.8
Quoted shares:	4,651	1.1	1.1	1.0	2.7	3.1	3.1
MFIs	752	2.1	2.7	2.2	2.7	3.2	3.2
Non-monetary financial corporations	473	1.5	1.0	1.0	1.2	1.3	1.3
Non-financial corporations	3,426	0.8	0.8	0.7	2.9	3.3	3.3

Source: ECB.

1) For details, see the technical notes for Sections 4.3 and 4.4 of the "Euro area statistics" section.

significant increase in October and reached the highest level recorded since April 2005. These developments seem to point to a stronger use of debt securities as a form of financing by non-financial corporations in the second half of 2005. This notwithstanding, the still relatively low annual growth appears to be related to substitution effects with loan financing, and to the relatively limited number of mergers and acquisitions involving non-financial corporations in recent months, even though – overall in 2005 – the latter have picked up significantly. In line with the overall trend, non-financial corporations increased their issuance, in particular of long-term securities financed at variable rates.

In October the annual growth rate of debt securities issued by MFIs remained unchanged from the previous month, at 9.4%. This robust level of growth was mainly the result of a continued strong issuance of both short-term and long-term securities at variable rates – which represent around 56% of total issuance – while issuance of securities at fixed rates remained subdued. The strong growth in MFI issuance was related to the financing needs of MFIs resulting from the rather robust growth of MFI loans to the private sector and to some securitisation of mortgage loans.

Non-financial corporations and MFIs also use non-monetary financial corporations to raise external funds indirectly. The annual growth rate of debt securities issued by non-monetary financial corporations increased slightly to reach 21.0% in October (see Chart 6). Within this sector, the annual growth rate was particularly high for the issuance of long-term debt securities at variable rates, which stood at 41.6% in October.

The annual growth rate of debt securities issued by the general government remained broadly unchanged at 4.1% in October.

#### **QUOTED SHARES**

The annual growth rate of quoted shares issued by euro area residents remained unchanged at 3.1% in October. The annual growth rate also remained unchanged at 3.3% when looking only at shares issued by non-financial corporations (see Chart 7). However, the growth rate has been revised



Monetary and financial developments





Note: Growth rates are calculated on the basis of financial transactions.



upwards since July 2005, owing to an instance of significant corporate restructuring involving an entity resident in the euro area and a non-resident entity. Without this statistical effect, the growth rate of quoted shares issued would have been only slightly higher than the annual growth rates recorded in the past few years, reflecting the moderate use of this form of financing by euro area non-financial corporations.

## 2.3 MONEY MARKET INTEREST RATES

From the beginning of December 2005 to early January 2006, money market interest rates with a maturity of one month decreased slightly, while longer-term money market rates increased. As a result, the slope of the money market yield curve steepened over that period.

Money market rates with a maturity of one month fell by 1 basis point, while longer-term money market rates, represented by the three- and twelve-month maturities, rose by 3 and 8 basis points respectively over the period from the beginning of December 2005 to 11 January 2006. On the latter date, the one-, three-, six- and twelve-month EURIBOR rates stood at 2.39%, 2.50%, 2.64% and 2.83% respectively. Consequently, the slope of the money market yield curve steepened over the same period. The difference between the twelve- and the one-month EURIBOR increased from 35 basis points at the beginning of December to 44 basis points on 11 January.



The interest rates implied by the prices of three-month EURIBOR futures contracts maturing in March, June and September 2006 stood on 11 January at 2.70%, 2.87% and 2.99% respectively, representing increases of 4 basis points for the March contract, 7 basis points for the June contract and 13 basis points for the September contract, compared with the levels observed at the beginning of December 2005.

On 1 December 2005 the Governing Council decided to raise key ECB interest rates by 25 basis points, with the minimum bid rate in the Eurosystem's main refinancing operations being set at 2.25% as from 6 December 2005. Towards the end of the maintenance period ending on 5 December 2005, the EONIA reached values close to or below 2%, on market participants' expectations of a loose end to the maintenance period. On 5 December, despite the conduct of a fine-tuning operation which absorbed the prevailing excess liquidity, the EONIA fell to 1.85%. The EONIA rose to 2.32% at the beginning of the new maintenance period, reflecting the increase in key ECB interest rates. Rates remained at around this level until Christmas week, when they rose somewhat. On the last trading day of the year the overnight rate rose to 2.42%, reflecting the usual end-of-year effect. The magnitude of this effect was however very modest compared with previous years, which may be related to the fact that the ECB, as in 2004, provided banks with ample liquidity in December to counteract liquidity uncertainty related to the holiday period.

During December 2005 the marginal and average rates in the Eurosystem's main refinancing operations rose, reflecting the increase in key ECB interest rates. Liquidity was provided at a marginal rate of 2.29-2.30% and an average rate of 2.30-2.31%, except in the tender operation covering the end of the year. Then, the marginal rate fell to 2.25%, while the average rate rose to



Monetary and financial developments

2.42%, with the wide spread between the two rates reflecting heightened end-of-year uncertainty. At the beginning of January 2006, the EONIA rate adjusted slowly downwards to levels slightly above those observed before Christmas, stabilising at 2.34%. Hence, the spread between the EONIA and the minimum bid rate increased slightly compared with that observed before Christmas.

In the Eurosystem's longer-term refinancing operation settled on 1 December, the marginal and the weighted average rates rose to 2.40% and 2.41% respectively, i.e. 23 and 22 basis points higher than in the previous operation, pricing in an increase in key ECB interest rates. Compared with the three-month EURIBOR prevailing on that date, tender rates were lower by 7 and 6 basis points respectively. In the longer-term refinancing operation settled on 22 December 2005, the marginal and the weighted average rates increased to 2.45% and 2.46% respectively, in part due to a technical incident caused by one inaccurate bid from a private bank which led to the provision of less liquidity than intended. This error required the conduct of a special operation on the next day to inject liquidity.

## **2.4 BOND MARKETS**

Long-term nominal and real government bond yields dropped slightly between the end of November and early January both in the euro area and in the United States. Shorter-term nominal and real bond yields in the euro area, however, remained broadly unchanged in the same period, suggesting that market participants did not revise their expectations regarding the growth prospects for the euro area economy over the next few years to any major extent. Long-term break-even inflation rates, which reflect market participants' inflation expectations over a long horizon, remained broadly unchanged in the euro area, but continued to moderate somewhat in the United States.

Bond market volatility, as extracted from option prices, continued to decline in both economies, reaching extremely low levels.

Ten-year government bond yields in the euro area and in the United States declined slightly in December and early January, after having risen in the period from September to mid-November (see Chart 10). The yield curve in the euro area shifted slightly downwards for all maturities longer than two years (see Chart 11), while interest rates in the United States fell for all maturities, except those at the very short end of the curve. Overall, ten-year government bond yields in the euro area and in the United States fell by about 15 and 5 basis points respectively between the end of November 2005 and 11 January 2006. Consequently, the differential between US and euro area ten-year government bond yields increased to approximately 115 basis points at the end of the review period. Japanese ten-year government bond yields remained more or less flat at around 1.5% in



Note: Long-term government bond yields refer to ten-year bonds or to the closest available bond maturity.



December and early January, despite survey data indicating a sustained improvement in business conditions. Market participants' uncertainty about near-term developments in the ten-year segment of the bond market, as indicated by the implied volatility extracted from bond options, declined in both the euro area and the United States in December, while it remained unchanged in Japan.

In the United States, after a parallel upward shift over the three preceding months, the movements of interest rates across different maturities became less correlated. Yields on bonds with a very short maturity rose further, pushed up by the latest 25 basis point increase in the federal funds target rate on 13 December. At longer maturities, interest rates dropped, in part possibly because market participants started to perceive a lower need for further monetary policy tightening in the medium term, as also reflected in the statement released by the Federal Open Market Committee (FOMC) after its meeting on 13 December.

Lower long-term rates resulted in a further flattening of the yield curve, despite macroeconomic releases and survey data that pointed towards robust economic growth. The differential between nominal ten-year government bond yields and three-month money market interest rates, which had been around 180 basis points at the beginning of 2005, has become negative since mid-December and stood at -10 basis points by the end of the review period. In the United States, an inverted yield curve has typically preceded periods of low economic growth. This may have somewhat increased market concerns about the short-term growth prospects for the US economy. In the current environment, however, the negative slope of the yield curve appears to mainly reflect low long-term bond yields, induced by a number of factors that have contributed to driving and holding

### Chart II Implied forward euro area overnight interest rates (percentages per annum; daily data) 11 January 2006 30 November 2005 4.5 4.5 4.0 4.0 3.5 3.5 3.0 3.0 2.5 2.5 2.0 2.0 1.5 1.5 2005 2006 2007 2008 2009 2010 2011 2012 2013 2014 2015

#### Source: ECB estimate.

Notes: The implied forward yield curve, which is derived from the term structure of interest rates observed in the market, reflects the market expectation of future levels for short-term interest rates. The method used to calculate these implied forward yield curves was outlined in Box 4 of the January 1999 issue of the Monthly Bulletin. The data used in the estimate are derived from swap contracts.

# Chart 12 Break-even inflation rates in the euro area and in the United States



Sources: Reuters and ECB calculations.

ECB Monthly Bulletin January 2006

Monetary and financial developments

down risk premia at the long end of the curve.<sup>2</sup> The decline in nominal US ten-year bond yields reflected entirely lower ten-year index-linked bond yields, while ten-year break-even inflation rates remained broadly unchanged over the review period (see Chart 12).

As in the United States, break-even inflation rates in the euro area remained broadly stable throughout the period under review, so that the movements of the nominal yield curve mirrored those of the real yields. Although slipping at the ten-year maturity, index-linked bond yields rose marginally over shorter maturities (see Chart 12), in line with overall better-than-expected releases of macroeconomic data and the positive signals embodied in survey results. This would suggest that market participants did not much revise the growth prospects they anticipate for the euro area economy over the next few years. Together with the increase in key ECB interest rates at the beginning of the period under review, this explains the rise in the implied forward overnight interest rate curve up to the two-year maturity range (see Chart 11). Overall in 2005, a noticeable decline was recorded in the steepness of the yield curve for the euro area. For example, the differential between ten-year government bond yields and the three-month EURIBOR has fallen by nearly 70 basis points since end-December 2004, but it nevertheless remained clearly positive at around 80 basis points on 11 January 2006. Finally, the slight drop in market participants' uncertainty over the review period, to close to the lowest levels recorded in 2005, may have led to a further reduction of the risk premia included in nominal and real yields.

# 2.5 INTEREST RATES ON LOANS AND DEPOSITS

In October 2005 MFI deposit and lending rates to households and non-financial corporations developed along mixed lines, but changed only little overall.

In October 2005 short-term interest rates on MFI loans to both households and non-financial corporations changed little in comparison with the preceding month, the only exception being the rates on loans to households for consumption, which declined by more than 20 basis points (see Table 4). Generally, however, short-term movements in the latter interest rate category have to be interpreted with particular caution as the levels are relatively volatile and as the decline in October was probably related to an unwinding of the increases recorded in the summer. Short-term MFI deposit rates followed a very similar pattern and remained broadly unchanged in October 2005.

The stability of short-term MFI interest rates was accompanied by some increases in short-term money market rates over the same period. This was possibly due to the fact that these increases were recorded mainly in the last days of the month and had thus not, as yet, been transmitted to MFI interest rates.

Taking a longer perspective, most short-term deposit rates have remained broadly unchanged since October 2004, which was in line with the stability of money market rates in the same period. Conversely, most short-term lending rates declined by at least 10 basis points in the same period, except for short-term lending rates to households for consumption purposes, which remained unchanged.

2 See the box entitled "Recent developments in long-term real interest rates" in the April 2005 issue of the Monthly Bulletin.

## Table 4 MFI interest rates on new business

(percentages per annum; basis points; weight-adjusted<sup>1), 2)</sup>) Change in basis points up to Oct. 2005 2004 2005 2005 2004 2004 2005 2005 2005 2005 03 04 01 02 Sep. Oct. Oct. July Sep. MFI interest rates on deposits Deposits from households with an agreed maturity of up to one year 1 90 1.95 1.92 1 94 1 97 1 97 5 6 1 10with an agreed maturity of over two years 2.48 2.31 2.38 2.21 2.06 2.16 -37 -6 redeemable at notice of up to three months 2.00 2.01 1.96 2.17 2.02 1.96 -5 -20 -6 2.29 2.27 -25 -2 redeemable at notice of over three months 2.52 2.52 2.47 -7 2.38 7 Overnight deposits from non-financial corporations 0.89 0.91 0.94 0.92 0.96 0.97 2 1 Deposits from non-financial corporations with an agreed maturity of up to one year 2.00 2.08 2.00 2.01 2.04 2.04 0 2 0 3.52 3.46 3.34 3.63 2.98 3.37 40 39 with an agreed maturity of over two years -15 MFI interest rates on loans Loans to households for consumption 19 -22 with a floating rate and an initial rate fixation of up to one year 6.89 6.74 6.62 6.61 6.96 6.74 3 Loans to households for house purchase with a floating rate and an initial rate fixation of up to one year 3.50 3.44 3.42 3.35 3.32 3.32 1 -16 1 with an initial rate fixation of over five and up to ten years 4 82 4 50 4.35 4.15 3.99 3.99 -77 -9 0 Bank overdrafts to non-financial corporations 5.38 5.27 5.26 5.13 5.13 5.09 -31 -3 -4 Loans to non-financial corporations of up to €1 million 3.99 3.98 3.91 2 7 with a floating rate and an initial rate fixation of up to one year 3.88 3.81 3.88 -13 4.20 -11 0 with an initial rate fixation of over five years 4.70 4.44 4.33 4.04 4.04 -61 Loans to non-financial corporations of over €1 million with a floating rate and an initial rate fixation of up to one year 3 00 3.04 3 01 2.94 2.94 2.91 -8 -5 -3 with an initial rate fixation of over five years 4.31 4.06 4.04 3.89 3.87 3.78 -41 -7 -9 Memo items 2.12 2.17 2.14 2.11 2.14 2.20 8 6 Three-month money market interest rate 5 Two-year government bond yield 2.60 2.36 2.49 2.07 2 21 2 45 -2 26 24 19 Five-year government bond yield 3.35 2.93 3.08 2.58 2.60 2.85 -33 25

Source: ECB.

1) The weight-adjusted MFI interest rates are calculated using country weights constructed from a 12-month moving average of new business volumes. For further information, see the box entitled "Analysing MFI interest rates at the euro area level" in the August 2004 issue of the Monthly Bulletin.

2) Quarterly data refer to the end of the quarter.

In October 2005 most long-term interest rates for households remained broadly unchanged, the sole exception being the rates on households' deposits with an agreed maturity of over two years, which increased by around 10 basis points. At the same time, long-term rates for non-financial corporations developed along disparate lines. The rates on deposits with an agreed maturity of over two years rose sharply, by about 40 basis points, which was due mainly to a very marked increase in one country. Long-term rates on loans to non-financial corporations, by contrast, declined by a few basis points (see Chart 14). Yields on two-year and five-year government bonds rose by around 25 basis points over the same period.

Since October 2004, long-term lending rates have fallen markedly in a range between 30 and 80 basis points. The most significant declines were recorded for long-term rates on housing loans and on loans to non-financial corporations of up to  $\notin 1$  million with an initial rate fixation of over five years. Over the same period, long-term deposit rates decreased by between 15 and 40 basis points, with the largest decrease being seen for long-term deposits from households (see Table 4).

Monetary and financial developments

# Chart 13 Short-term MFI interest rates and a short-term market rate

(percentages per annum; rates on new business; weight-adjusted<sup>1)</sup>)

- three-month money market rate
- loans to non-financial corporations of over €1 million with a floating rate and an initial rate fixation of up to one year
- loans to households for consumption with a floating
- rate and an initial rate fixation of up to one year overnight deposits from non-financial corporations
- deposits from households redeemable at notice of up to three months
- deposits from households with an agreed maturity of up to one year
- loans to households for house purchase with a floating rate and an initial rate fixation of up to one year



Source: ECB. 1) For the period from December 2003 onwards, the weight-adjusted MFI interest rates are calculated using country weights adjusted MF1 interest rates are calculated using country weights constructed from a 12-month moving average of new business volumes. For the preceding period, from January to November 2003, the weight-adjusted MF1 interest rates are calculated using country weights constructed from the average of new business volumes in 2003. For further information, see the box entitled "Analysing MF1 interest rates at the euro area level" in the August 2004 issue of the Monthly Bulletin.



(percentages per annum; rates on new business; weight-adjusted<sup>1)</sup>)



Source: ECB. 1) For the period from December 2003 onwards, the weight-adjusted MFI interest rates are calculated using country weights constructed from a 12-month moving average of new business volumes. For the preceding period, from January to November 2003, the weight-adjusted MFI interest rates are calculated using country weights constructed from the average of new business volumes in 2003. For further information, see the box entitled "Analysing MFI interest rates at the euro area level" in the August 2004 issue of the Monthly Bulletin

the August 2004 issue of the Monthly Bulletin.

These developments were consistent with the trends observed in five-year government bond yields, which fell by around 35 basis points over the same period.

# **2.6 EQUITY MARKETS**

Stock prices rose in the euro area and in the United States between end-November and January. They continued to grow particularly strongly in Japan over the same period. All in all, stock market uncertainty was unchanged in both the United States and the euro area, but rose in Japan. Profitability, both current and expected, continued to support equity prices in the major markets.

Stock prices increased in the euro area and in the United States between the end of November and 11 January (see Chart 15). In the same period, the upward trend recorded in Japanese stock prices in preceding months became even steeper. In the euro area, the Dow Jones EURO STOXX index



23.0

20.0

17.0

14.0

11.0

8.0

rose by nearly 7%, while the Standard & Poor's 500 index in the United States was up by 4%. At the same time, Japanese stock prices, as measured by the Nikkei 225 index, increased by around 10%. Overall, across 2005, broad equity indices rose by 40% in Japan, by 22% in the euro area and by only 4% in the United States. Box 2 briefly reviews recent changes in corporate accounting requirements and their potential impact on stock prices.

Stock market uncertainty, as measured by the implied volatility extracted from equity options, was almost unchanged in the euro area and in the United States between end-November and 11 January. However, it rose sharply in Japan (see Chart 15), remaining slightly above 20% in early January. The rise in perceived market uncertainty about future developments in Japanese equities may be linked to concerns about the sustainability of the gains of nearly 10% posted in three of the last four months of the year. Some uncertainty may also be related to market concerns about the robustness of the current recovery. In 2005 there were no major fluctuations in stock market uncertainty in the major markets, at least if the short-lived peak recorded after the two summer hurricanes in the United States and the sudden rise recorded in Japan in December are excluded.

In the United States, the rise in stock prices was in line with the expected robust profitability, both in the short run and over longer horizons, although the rate of growth in earnings per share declined somewhat in the last quarter of 2005. Looking at other determinants, equity risk premia do not seem to have played a significant role in December, given the stability of the implied stock market volatility derived from options on the Standard & Poor's 500 index (see Chart 16). The underperformance of US equities relative to the euro area may, to some extent, reflect the smaller decline in US long-term real interest rates. In addition, although inflation remained relatively low



Monetary and financial developments

and inflation expectations contained, the FOMC statement of 13 January pointed towards upside inflation risks emanating from resource utilisation and elevated energy prices, which may have contributed to depressing equity valuations.

The increase in euro area stock prices over the review period may have reflected lower long-term real interest rates and, more importantly, corporate profit growth – both actual and expected – which remained quite solid. The increase in equity prices was broad-based, but some sectors tended to post higher gains, especially the industrial, the basic materials and the financial sectors. The oil and gas sector, which has recently recorded strong gains, was the worst performer in December and early January, despite the renewed increase in oil prices in that month. Stock prices also recovered in the consumer goods sector, although such companies' outlook for earnings tended to remain comparatively weaker.

#### Box 2

# STOCK OPTIONS, SHARE BUY-BACKS AND THEIR EFFECTS ON EQUITY MARKETS: EVIDENCE FROM THE UNITED STATES

Changes in corporate accounting requirements can affect the valuations of companies and, consequently, the prices of stocks. Useful information on this topic can be gained by looking at recent changes in accounting requirements in the United States. After years of debate between regulators and company representatives (especially those of companies belonging to the so-called growth sectors), new accounting rules have been approved on the treatment of stock options – staff compensation in the form of shares in the company. The new accounting rules set out in the revised Financial Accounting Standards Board Statement No 123 require companies to recognise the value of stock options entirely as expenses on the date of issuance and for the entire vesting period of the option, starting in 2006 for the majority of the companies. Thus, the impact of these new requirements on corporate earnings is likely to become visible in releases over the next few quarters. Although the near-term impact of these new regulations on stock prices is likely to be significant for certain sectors, the overall effect on the broad index will probably be moderate.

The systematic exclusion of stock options from the cost account was a general characteristic of so-called pro-forma earnings.<sup>1</sup> These pro-forma earnings, and the distorted image they tended to give of company profitability, were mentioned as one of the reasons for the stock market bubble that built up at the end of the 1990s and then burst in 2000, especially in the case of companies in the technology sector. In addition, the release of pro-forma earnings helped to hide cases of fraud which were eventually revealed in the accounting scandals involving US corporations – Enron, Tyco and WorldCom. Pressured by concerned investors in the aftermath of the scandals, more companies have started to treat stock options as an expense, but their number remain low.

Analysts at Standard & Poor's started to include stock options as expenses in the calculation of earnings in 2001, and a new series, the core earnings series for companies included in the Standard & Poor's 500 index, was published (see Chart A). The difference between released

<sup>1</sup> These are projected earnings based on a set of assumptions which exclude non-recurring items. Items sometimes excluded from figures on pro-forma earnings include write-downs, goodwill, amortisation, depreciation, restructuring and merger costs, interest, taxes, stock-based employee pay and other expenses.



earnings and core earnings was sizable in the period from 2000 to 2001, but has declined noticeably in the most recent years, when a larger group of companies started to take stock options into account and several companies changed their compensation policies, thereby reducing the use of stock options altogether. The difference between the values of the earnings in the two series at the end of 2004 – the latest date for which official balance sheets data are available – was around 6%. Thus, the effect of the new regulations on earnings releases, and on the overall stock market index, in the next few quarters is not expected to be significant. However, the aggregate series hide large differences in the importance of stock options among sectors. Time series data on the earnings per share at a sectoral level are not readily available; however, option costs range from around 5-6% of the total operating costs in sectors such as "consumer discretionary" and "health care" to almost 18% in the case of companies in the information technology sector (see Chart B).



The impact of the new accounting rules is also evident in another important trend characterising recent developments in the stock market, especially in the United States: the rise in share buybacks. When a stock option is "in-the-money", the holder of the option is likely to exercise it. At this point, a firm really has a choice between two alternatives: it can issue new shares and give them to the holder of the option, or it can buy outstanding shares in the market and sell them to the holder of the option at the price implied by the option contract, which is lower than the market price. If a company issues new shares, the earnings per share decline because of the dilution effect – there are simply more shares outstanding. Conversely, if the company buys shares at the market price and then sells them at a lower price, it will incur a loss which will decrease its overall earnings. The monetary impact the two alternatives have on profits will not be the same.

Over the past few years, US companies seem to have increasingly chosen the second option, as reflected in the rising trend of share buy-backs in the United States in the last few quarters.<sup>2</sup>

2 See the box entitled "The recent surge in US share buy-backs: causes and possible financial stability implications," in the ECB Financial Stability Review of December 2005.



Monetary and financial developments

Under US accounting rules, share buy-backs allow a more "dynamic" management of equity liabilities since shares retired from the market through a buy-back can later be re-utilised where necessary. Another reason why companies buy back their shares is that the funds they have available are higher than their near-term investment needs, which seems to have been true in an environment of comparatively low investment by US companies over the past few years. All in all, the increase in the use of share buy-backs, and thus an increased demand for corporate shares, is likely to have supported stock prices.

In Europe, except in the case of the United Kingdom, both evidence on the use of stock options as a form of employees' compensation and data on share buy-back programmes are fragmented. Nevertheless, there are signs that buying back shares is on the rise in the euro area as well. Under the new international financial reporting standards, which all listed companies in the EU have to fulfil as from the 2005 fiscal year, companies are required to recognise the value of share-based payments in their income statements. Whether this will translate into a one-off impact on earnings or whether it will have a longer-lasting effect depends on the extent to which stock options are used as a form of compensation in the euro area. Anecdotal evidence suggests that the use of this form of compensation is limited in the euro area, although it is significant in specific industries (in the technology sector, in particular).





# **3 PRICES AND COSTS**

Largely reflecting developments in oil prices, annual HICP inflation in the euro area continued to decline and stood at 2.2% in December as reported by Eurostat's flash estimate. Domestic price pressures remained moderate as no clear sign that higher energy prices have been passed through to non-energy producer prices or to labour costs has emerged so far. Overall, the inflation outlook is broadly unchanged with upside risks to price stability relating mainly to oil prices and the risk of potential second-round effects in wage and price setting.

#### **3.1 CONSUMER PRICES**

#### FLASH ESTIMATE FOR DECEMBER 2005

According to Eurostat's flash estimate, HICP inflation eased slightly to 2.2% in December (see Table 5), continuing the pattern of declining inflation rates seen since September 2005. Although no breakdown is available so far, this further fall in December could be related to a base effect in food prices.

Eurostat's flash estimate is calculated on the basis of preliminary data, and as such is surrounded by uncertainty. However, as discussed in the box entitled "Assessing the reliability of Eurostat's euro area HICP flash estimate", the flash estimate has proved to be a very reliable indicator of HICP developments. In addition, its forecasting performance has exceeded that of alternative methods.

## **HICP INFLATION UP TO NOVEMBER 2005**

After a pick-up in September, overall HICP inflation decreased somewhat in recent months and stood at 2.3% in November. These movements in inflation were primarily due to changes in the contribution of energy prices.

Following the decline in oil prices in October and November, the annual growth rate of energy prices decreased notably, although it still remained elevated at a rate of 10% in November. The decline in energy price growth was mainly driven by transport and heating fuel, while gas prices, which tend to respond more slowly to oil price changes, continued to edge up. The price pressure

(annual percentage changes, unless otherwise indicated	)							
	2004	2005	2005	2005	2005	2005	2005	2005
			July	Aug.	Sep.	Oct.	Nov.	Dec.
HICP and its components								
Overall index 1)	2.1		2.2	2.2	2.6	2.5	2.3	2.2
Energy	4.5		11.7	11.5	15.0	12.1	10.0	
Unprocessed food	0.6		0.3	1.0	1.0	1.1	1.5	
Processed food	3.4		1.6	1.7	2.3	2.4	2.6	
Non-energy industrial goods	0.8		0.0	0.0	0.2	0.3	0.4	
Services	2.6		2.3	2.2	2.2	2.2	2.1	
Other price indicators								
Industrial producer prices	2.3		4.1	4.0	4.4	4.2	4.2	
Oil prices (EUR per barrel)	30.5	44.6	48.3	52.0	52.2	49.3	47.9	48.5
Non-energy commodity prices	10.8	9.4	9.6	11.9	13.2	17.4	22.5	29.8

Table 5 Price developments

Sources: Eurostat, HWWA and ECB calculations based on Thomson Financial Datastream. 1) HICP inflation in December 2005 refers to Eurostat's flash estimate.

1) HICP inflation in December 2005 refers to Eurostat's flash estimate.



Prices and costs

from unprocessed food increased, partly dampening the effect of developments in energy prices in November.

In contrast to overall inflation the annual growth rate of the HICP excluding energy and unprocessed food prices remained broadly unchanged in recent months, and stood at a more moderate rate of 1.5% in November. This relative stability has, however, been a result of conflicting movements in its components (see Chart 17). Following a pick-up in September due to an increase in German tobacco taxes, the annual growth rate of processed food prices continued to rise and stood at 2.6% in November. The annual growth in non-energy industrial goods prices edged up to 0.4% in November due to an increase in clothing prices following the relatively sharp seasonal discounting observed during the summer sales period. However, the annual rate of growth of non-energy industrial goods prices remains well below its historical average, suggesting that there are still no significant indirect effects from previous increases in commodity prices. In contrast to processed food and non-energy industrial goods prices, services price inflation decreased slightly to 2.1% in November.





#### Box 3

#### ASSESSING THE RELIABILITY OF EUROSTAT'S EURO AREA HICP FLASH ESTIMATE

Eurostat's euro area HICP flash estimate provides a very timely indication of euro area inflation developments. As the first flash estimate was released for the October 2001 HICP, there are now 50 months of official HICP figures against which to assess its performance.

The flash estimate is computed by Eurostat using econometric techniques that combine the available provisional national HICP figures with early information available from the European Commission's weekly Oil Bulletin about the prices of some energy products. In addition, where available, information regarding the effects on prices of one-off events, such as changes in indirect taxes, is also taken into account. Eurostat's estimate is generally released on the last working day of the reference month. Only the annual growth rate of the overall HICP for the euro area is published. The HICP flash estimate currently incorporates provisional national



# Table A Comparison of forecast performancestatistics

(percentage points, unless otherwise indicated)

	Eurostat's flash estimate	Random walk	AR model
Mean error	0.00	0.00	-0.01
RMSE	0.09	0.21	0.20
Maximum deviation	0.2	0.5	0.4
Minimum deviation	-0.2	-0.5	-0.6
Absolute size of deviation	Number of occasions	Number of occasions	Number of occasions
0.0	22	11	18
0.1	25	15	16
>0.1	3	24	16

Source: Eurostat.

Source: ECB calculations.

estimates covering around 65% of the euro area HICP whereas, in the first two years of its publication, only HICP data for Germany and Italy and CPI data for Belgium were available – these countries accounting for a little over 50% of the euro area HICP. Greece and Spain began to produce provisional inflation data sufficiently early for inclusion in the euro area HICP flash estimate in 2003 and 2004 respectively.

#### Summary of performance to date

The chart plots the flash estimate alongside the actual HICP outcome. Generally the flash estimate appears to have performed well. Of the 50 releases to date, 22 were entirely accurate, 25 were 0.1 percentage point out and three were 0.2 percentage point out. In addition, the flash estimate has never suggested that the inflation rate declined when it actually increased, or vice versa. Lastly, the average deviation between the flash estimate and the actual HICP outcome – the mean error reported in Table A – is exactly 0.0 percentage point. This indicates that there has to date been no tendency toward bias in the flash estimate as a predictor of the official HICP outcome.

Another measure that can be used to assess the reliability of the flash estimate is the root mean square error (RMSE). The RMSE measure, which provides an indication of the average size and variability of the deviation between the flash estimate and the actual HICP outcome, shows that the flash estimate has clearly outperformed both a random walk forecast (i.e. assuming an unchanged annual inflation rate) and a forecast based on a simple Autoregressive (AR) model of inflation (see Table A). This indicates the usefulness of the flash estimate in providing information about the latest inflation developments compared with other simple benchmark indicators.

#### Possible sources of deviation between the flash estimate and the HICP outcome

Deviation of the flash estimate from the HICP outcome may arise from a number of sources. First, the HICP is a revisable index and countries that have provided provisional information may subsequently revise their data based on more complete information. Second, inflation developments in countries which have not provided provisional information may move differently to developments in countries for which provisional information is available. Third, volatility in



Prices and costs

the seasonal pattern or other atypical developments in some HICP components may be difficult to capture using econometric techniques. Lastly, deviation between the flash estimate and the HICP outcome may simply reflect the rounding of the price indices employed in the calculation of inflation.

Analysis suggests that differences between inflation developments in countries which have provided provisional information and those in 
 Table B Correlation coefficients between monthly changes in HICP components and deviation of the flash estimate from the HICP outcome

HICP component	Correlation
Unprocessed food	0.23
Processed food	0.33
Unprocessed and processed food	0.34
Non-energy industrial goods	0.06
Energy	0.18
Services	0.02

Source: ECB calculations

countries which have not may explain a small part of the deviation between the flash estimate and the HICP outcome. The correlation coefficient is approximately 0.2 between deviations of the flash estimate from the HICP outcome and differences in inflation developments between the countries providing data and those that do not. Furthermore, it is interesting to consider whether developments in particular components have been associated with deviation from the flash estimate. Table B shows the correlation of (seasonally adjusted) month-on-month changes in the components of the HICP with deviations between the flash estimate and the HICP outcome.<sup>1</sup> The results indicate that the highest correlation is to be found with movements in the two food components (unprocessed and processed food), at 0.23 and 0.33 respectively. Given the high volatility of oil price developments and their significant contribution to inflation over the last four years, it is not surprising that some correlation also exists with the energy component. However, the information available from the European Commission's weekly Oil Bulletin may have served to limit the extent of deviations emanating from energy price developments. The relatively low correlation with developments in both the non-energy industrial goods and the services components suggests that such developments are captured fairly well through available country data when the flash estimate is compiled and by the econometric techniques used by Eurostat.

Overall, it appears that some of the deviations may relate to differences in inflation patterns between the countries providing data and those that do not, and to component-specific developments. Other factors such as the rounding conventions employed to calculate the HICP<sup>2</sup> and data revisions may also have played a role. However, this analysis can only be regarded as indicative given that the flash estimate incorporates information from a number of sources and some judgement.

## Forthcoming improvements in the flash estimate

In conclusion, Eurostat's euro area HICP flash estimate has significantly improved the timeliness of a key economic indicator for the euro area and has performed very satisfactorily over the past four years. Looking forward, the performance of the flash estimate is expected to improve further as a result of work by Eurostat and the national statistical institutes on the Principal European Economic Indicators.<sup>3</sup> The national statistical institutes of France, Austria and Finland



<sup>1</sup> As the flash estimate is presented in terms of the annual rate of change, deviations from the actual HICP outcome should not be due to "normal" seasonal variations. Thus, the seasonally adjusted month-on-month rates of change are used to capture atypical developments in the individual components.

<sup>2</sup> At present the overall euro area HICP is calculated on the basis of country indices rounded to one decimal place and is then rounded to one decimal place in turn. Hence, small country-specific developments can sometimes give rise to 0.1 percentage point deviations from the outcome, both in the estimates provided by Member States and in the euro area flash estimate

<sup>3</sup> See also Box 6 in the December 2005 issue of the ECB Monthly Bulletin entitled "Further progress on the Principal European Economic Indicators"

have indicated that in 2006 they will start providing provisional information to Eurostat for inclusion in the euro area flash estimate. Moreover, work on a national HICP flash estimate is also being undertaken in the Netherlands. As a result, the overall country coverage of the euro area flash estimate may increase to more than 95%, which would further enhance its reliability.

#### 3.2 INDUSTRIAL PRODUCER PRICES

In November producer price inflation in the industrial sector was broadly stable. The annual rate of change in industrial producer prices excluding construction remained at 4.2% despite a slight decrease in the contribution of energy prices compared with previous months (see Chart 18). The relatively low annual growth rate of industrial producer prices excluding construction and energy also persisted in November, supporting the view that there had been only a limited transmission of higher energy costs to prices in other sectors of the economy.

In line with the above assessment, although the annual growth rate of consumer goods producer prices has risen since August 2005, reaching 1.4% in November, the process has mainly been driven by tobacco and food prices. Moreover, the pace of the increase is below the peak levels associated with commodity price shocks in the past. Meanwhile, the annual growth rate of intermediate goods prices may have levelled off after a sharp drop in the first half of 2005. In addition, the annual growth of capital goods prices has also remained muted in recent months.



ECB Monthly Bulletin January 2006

Prices and costs

Reflecting developments in producer prices, price-related survey indicators suggest that there is ongoing pressure on firms' input prices in both the manufacturing and the services sectors, although they indicate only a moderate increase in selling prices (see Chart 19). The Eurozone Manufacturing Input Price Index remains high at 62.2 and firms reported that the high prices of energy and other raw materials were the main contributing factors. The rise of the selling price index was, however, still moderate in November, indicating that the surge in input costs was not passing through fully to selling prices, notably due to strong competition and weak consumer demand. The picture in the services sector is broadly similar: the gap between input and selling prices widened in the period, although to a lesser extent than in manufacturing.

# **3.3 LABOUR COST INDICATORS**

Recent developments in labour cost indicators suggest that wage pressures remained subdued in the euro area up to the third quarter of 2005 (see Chart 20 and Table 6). In particular, the annual growth rate of negotiated wages was 2.1% in the second and third quarters. Following a peak in the first quarter, which mainly resulted from statistical factors, the annual growth rate of total hourly labour costs declined gradually during the year to reach 2.2% in the third quarter. The decline was significant in the non-wage components while the decline in wage costs remained small. Following an increase at the end of 2004, the annual rate of change in compensation per employee also decreased in the first half of 2005. The fall was more marked in the industrial sector, but wage growth declined in the services sector as well (see Chart 21). According to available country information, the moderate pace of growth in compensation per employee continued in the third quarter.

Considering that productivity growth picked up by 0.3 percentage point in the third quarter, and compensation per employee growth is likely to have remained moderate, unit labour costs growth is estimated to have remained subdued in that quarter.


# Table 6 Labour cost indicators

(annual percentage changes, unless otherwise indicated)

(annuar percentage enanges, antess other wi	2003	2004	2004 Q3	2004 Q4	2005 Q1	2005 Q2	2005 Q3
Negotiated wages	2.4	2.1	2.0	2.0	2.2	2.1	2.1
Total hourly labour costs	3.0	2.5	2.4	2.4	3.2	2.5	2.2
Compensation per employee	2.3	2.0	1.5	1.7	1.5	1.5	
Memo items:							
Labour productivity	0.5	1.1	1.1	0.7	0.4	0.5	0.8
Unit labour costs	1.8	0.9	0.4	1.1	1.1	1.0	

Sources: Eurostat, national data and ECB calculations.

#### 3.4 THE OUTLOOK FOR INFLATION

Overall, taking account of the latest information, the inflation outlook has remained broadly unchanged from the Eurosystem staff projections published in early December 2005. Following the decline since September, inflation rates could exhibit some volatility in the early months of 2006 and could rise somewhat, partly reflecting unfavourable base effects from the unprocessed food component. Moreover, the short-term outlook continues to depend crucially on developments in oil prices. More generally, reflecting the subdued labour markets, wage increases are expected to remain contained and thus contribute to continued overall moderate pressures on domestic prices. However, the inflation outlook continues to be subject to a number of significant upside risks, in particular, relating to oil prices and the risk of potential second-round effects in wage and price setting. In this regard, the effects of the most recent rise in oil prices and oil price futures need to be monitored closely. In addition, further increases in administered prices and indirect taxes could also put upward pressure on euro area inflation.



Output, demand and the labour market

# **4 OUTPUT, DEMAND AND THE LABOUR MARKET**

The latest data releases confirm the picture of stronger economic activity in the second half of 2005, indicating that the euro area economy demonstrated considerable resilience in the face of higher oil prices. Overall, real economy indicators and survey data continue to support the expectation of sustained economic growth in 2006, and there are also signs of improvement in labour market conditions. Downside risks to the economic outlook relate mainly to prevailing global imbalances and uncertainties surrounding oil price developments.

#### 4.1 OUTPUT AND DEMAND DEVELOPMENTS

### REAL GDP AND EXPENDITURE COMPONENTS

The first release of national accounts data for the third quarter of 2005, reported in the December 2005 issue of the Monthly Bulletin, confirmed that euro area real GDP rose by 0.6% quarter on quarter, in line with the flash estimate. Taking into account the quarter-onquarter growth rates of 0.4% in the second quarter and 0.3% in the first quarter of 2005 (see Chart 22), this confirms the picture of a strengthening in activity. The euro area economy has therefore shown considerable resilience to the marked increase in oil prices.

The contributions from domestic demand and net exports were both positive. The domestic demand contribution stemmed from slightly higher private consumption growth and from strong investment growth, which recorded its highest contribution to GDP growth since the first quarter of 2000. Both net exports and investment have been contributing substantially





to economic activity since the beginning of the recovery in the third quarter of 2003.

While growth dynamics have strengthened, however, potential growth in the euro area remains disappointingly low. EU Member States recently submitted National Reform Programmes aimed at achieving higher labour participation and employment growth, and at boosting labour productivity growth and competitiveness. Box 4 reviews the main structural measures planned by the euro area countries in their programmes.

#### Box 4

#### THE 2005-2008 LISBON NATIONAL REFORM PROGRAMMES OF THE EURO AREA COUNTRIES

The March 2005 European Council meeting relaunched the Lisbon strategy by focusing on growth and employment.<sup>1</sup> At the same time, in order to improve the implementation of structural reforms, a number of changes were made to the governance framework of the strategy.

1 See the article entitled "The Lisbon strategy - five years on" in the July 2005 issue of the Monthly Bulletin.

Specifically, the European Commission put forward an Integrated Guidelines package, consisting of a new set of Broad Economic Policy Guidelines (BEPGs) and Employment Guidelines. The ECOFIN Council adopted the 2005-2008 Integrated Guidelines in July 2005.<sup>2</sup> The Integrated Guidelines will remain valid for three years, thus providing policy continuity for this period.

In response to the Integrated Guidelines, EU Member States submitted in autumn 2005 National Reform Programmes (NRPs), in consultation with key stakeholders in the implementation of the Lisbon strategy, such as national parliaments, regional and local authorities, and social partners.<sup>3</sup> The NRPs set out a comprehensive three-year strategy to implement the Integrated Guidelines at the national level. They present key country-specific challenges and an outline of policy initiatives and concrete measures in response to the challenges identified at the EU level. As part of the existing multilateral surveillance arrangements, the ECOFIN Council, assisted by the Economic Policy Committee of the EU, undertook a preliminary review of the NRPs.<sup>4</sup> In addition, the Commission will present its own assessment of the NRPs in the form of a first annual progress report on the Lisbon strategy in late January 2006.

This box provides an overview of the main structural measures decided or planned by the euro area countries in their NRPs for the period 2005-2008. The focus of the NRPs is on reforms aimed at achieving higher labour participation and employment growth, and at boosting labour productivity growth and competitiveness. To achieve this, the NRPs include measures to boost innovation, continue the completion of the EU internal market, create the right climate for entrepreneurs and ensure the long-term sustainability and quality of public finances.

With regard to labour market reforms, stimulating labour participation and employment is a core challenge for policy. It is essential to provide stronger incentives for people to stay in or enter the labour market and to increase the employment rate, in particular for women, the young and older workers. Therefore, the labour market measures in the NRPs target these groups as a matter of priority. Measures have been announced in some countries to develop childcare facilities in order to support an increase in the female employment rate. Measures aimed at raising older workers' employment rates mainly consist in increasing financial incentives to remain in the labour market (e.g. by providing the possibility to draw a pension and a salary at the same time), in limiting the possibilities for early retirement (e.g. by withdrawing tax incentives for early retirement schemes) or in adapting the incentive structure for companies to hire older workers by introducing a more flexible employment contract. Finally, measures aimed at reducing youth unemployment mainly include tax cuts at the lower end of the wage scale in order to reduce labour costs and/or measures to develop human capital by improving the educational and training system. By contrast, little or no action aimed at easing employment protection legislation or increasing wage flexibility has been taken in most euro area countries. These remain important issues that need to be addressed, especially with regard to the employment prospects for unskilled workers.

Turning to measures aimed at boosting labour productivity growth and competitiveness, most euro area countries have decided to significantly enhance investment in research and

<sup>4</sup> See the ECOFIN Council conclusions of 6 December 2005 and the Economic Policy Committee's November 2005 "Report on the Lisbon National Reform Programmes 2005".



<sup>2</sup> See the box entitled "The Integrated Guidelines for Growth and Jobs 2005-2008" in the August 2005 issue of the Monthly Bulletin.

<sup>3</sup> A parallel Community Lisbon Programme was adopted by the Commission in July 2005 to complement the NRPs with action to be taken at the European level.

Output, demand and the labour market

development. However, while specific actions to boost public R&D investment have been announced, measures aimed at increasing private R&D investment focus often on strengthening fiscal incentives. In addition, some countries intend to step up cooperation with the private sector to achieve this goal.

With regard to reforms aimed at increasing product market competition, the NRPs would have benefited from focusing more on concrete measures decided or envisaged. Even though progress has been recorded over the past few years, more still needs to be done in this field, particularly in the services sector, including network industries. In June 2005 nine euro area countries had not reached the EU target of reducing the transposition deficit for internal market directives to below 1.5%, with three of them being the worst performers of the 25 EU Member States in this respect.

With respect to measures on the fiscal side, the NRPs address two broad areas for structural reforms, namely: (i) the sustainability of public finances, especially in view of the ageing of the EU population; and (ii) the composition of public expenditure and the economic efficiency of taxation and public spending. In the area of fiscal sustainability, demographic ageing has been projected to have a pronounced effect on fiscal balances, in particular through its impact on public expenditure on pensions as well as health and long-term care. In light of such projections, several euro area countries have implemented pension reforms in the past few years with a view to making their public pension systems financially sustainable. In particular, the financial soundness of public pay-as-you-go pension systems has been strengthened, partly through measures to raise the effective retirement age and to tighten benefits. At the same time, the role of funded pension pillars has been increased. Intentions for future pension reforms tend to go in the same direction. Overall, however, it appears that more needs to be done to ensure the financial viability of public pension systems in several countries.

A number of programmes elaborate on ongoing or planned reforms to tackle ageing-induced fiscal pressures in public healthcare systems. While such reforms are generally targeted at addressing current inefficiencies, they would also contribute to preparing the systems for future challenges.

Some NRPs present concrete plans to raise the efficiency of taxation and public spending and to strengthen institutions of budget management and control. The design of tax-benefit systems represents a direct link between the need to ensure fiscal sustainability and the incentive structure for the entire economy, in particular in the area of labour supply and demand. Several programmes acknowledge the disincentive effects arising from high effective marginal tax rates due to the loss of social transfers for workers re-entering employment after a period of unemployment. In addition to lowering overall taxation on labour, some countries have implemented or envisage a reduction in the tax rates on low incomes to make work pay. Further measures in this regard consist in granting low-wage employees continued entitlement to some social benefits and introducing negative income taxes. Such efforts are targeted in particular at raising the work incentives for recipients of disability benefits.

Overall, the NRPs appear to demonstrate a stronger political commitment to the reform process in the context of the Lisbon strategy. All euro area governments acknowledge the need for further reforms and the benefits arising from implementing such reforms. However, despite important progress in some areas, the NRPs would have benefited from focusing more on concrete measures to meet the identified challenges.

Looking ahead, it will be important to communicate, monitor and evaluate the implementation of the reform process effectively. It is likely that in some areas euro area countries will need to go beyond the reforms presented in their NRPs in order to meet the objectives of the Lisbon strategy by 2010. The submission of the updated NRPs in autumn 2006, which will include an assessment of the previous year's implementation efforts, should also provide the opportunity to review to a certain degree national strategies and commitments while maintaining policy continuity over the period 2005-2008.

# SECTORAL OUTPUT AND INDUSTRIAL PRODUCTION

In terms of the sectoral composition of growth in the euro area, the data for real value added in the third quarter of 2005 (which were reported in the December 2005 issue of the Monthly Bulletin) showed a positive contribution to growth from both the industrial and the services sector. Value added in the industrial sector increased at a marginally higher pace than in the services sector.

Euro area industrial production (excluding construction) fell by 0.8% month on month in October 2005; on a three-month moving average basis, however, the latest results show a rate of expansion in the industrial sector of 0.5% in September. The three-month moving average growth rate has remained broadly stable since the second quarter. The three-month moving average for industrial production excluding energy as well as construction continued the upward trend it had followed throughout 2005 (see Chart 23). In terms of three-month moving averages, the strongest growth in the industrial

# Chart 23 Industrial production growth and contributions



sector in September was recorded in intermediate goods (at 1.5%).

#### SURVEY DATA FOR THE MANUFACTURING AND SERVICES SECTORS

Survey data for the manufacturing and services sectors in the euro area continue to show signs of improvement. For the manufacturing sector, the European Commission's industrial confidence indicator increased by two points in December 2005, having registered no change in November and a slight increase in October. The Purchasing Managers' Index (PMI) for the manufacturing sector rose in December for the fourth successive month, to reach its highest level since August 2004. The increase in the PMI was widely spread across components and countries, confirming that the improvement which began in May 2005 was sustained in the fourth quarter. Overall, the



Output, demand and the labour market



increase in both the industrial confidence indicator and the PMI in December provides a positive signal for the fourth quarter and the start of 2006 (see Chart 24).

As regards the services sector, the European Commission's services confidence indicator declined slightly in December 2005, although it remained close to its highest value of 2005. The business activity index of the PMI survey for the services sector rose in December. Both the employment and outstanding business indices also rose, reaching their highest values in more than four years. Overall, the available survey information for the services sector provides a positive signal for the last quarter of 2005 and the start of 2006.

# INDICATORS OF HOUSEHOLD SPENDING

Growth in euro area private consumption continued to improve in the third quarter of 2005, while remaining moderate. Private consumption increased by 0.3% quarter on quarter, following an increase of 0.2% in the second quarter and 0.1% in the first. The contributions from both retail sales and new car registrations were larger in the third quarter.

Retail sales volumes decreased by 0.1% month on month in November, following an increase of 0.2% in October, thus again recording approximately the same rate of change as in May and June. The decline in November was attributable to lower growth in sales of food products. From a medium-term perspective, retail sales growth remained subdued. As regards other indicators of household spending, new passenger car registrations fell by 0.9% month on month in November, following a fall of 1.1% in October. On a three-month moving average basis, the latest results show a contraction in new passenger car registrations of 0.8% in October.



The picture of a moderate recovery in consumption growth is consistent with recent developments in the European Commission's consumer confidence indicator. In December 2005, following a gradual improvement over the previous few months, consumer confidence reached its long-term average (see Chart 25).

# 4.2 LABOUR MARKET

The latest indicators suggest that the improvement in labour market conditions which started in the first half of 2005 has been sustained. Employment expectations for both the industrial and services sectors improved in December, signalling a positive outlook for underlying labour market conditions in the fourth quarter of 2005 and the start of 2006.

#### **UNEMPLOYMENT**

The euro area unemployment rate remained unchanged at 8.3% in November 2005 for the third consecutive month (see Chart 26). The number of unemployed increased for the first

time since March 2005. However, this appears to reflect primarily the developments in Germany, where recent methodological and institutional changes make the underlying trend difficult to assess (see Box 5). The total number of unemployed in all euro area countries for which data are available excluding Germany declined in November. While the new data alone do not signal an expansion in employment, taken together with other information on actual and expected employment developments, they suggest an improvement in labour market conditions.

#### Table 7 Employment growth

(percentage changes compared with the previous period; seasonally adjusted)

	Annual rates		Quarterly rates					
	2003	2004	2004	2004	2005	2005 Q2	2005 Q3	
			Q3	Q4	Q1			
Whole economy	0.3	0.7	0.3	0.2	0.0	0.2	0.3	
of which:								
Agriculture and fishing	-2.0	-0.8	0.1	-0.4	-1.1	-0.2	-0.7	
Industry	-1.0	-0.9	-0.1	0.0	-0.6	0.0	0.0	
Excluding construction	-1.5	-1.6	-0.5	0.1	-0.8	-0.1	-0.	
Construction	0.2	1.0	1.1	-0.3	0.0	0.3	0.4	
Services	0.9	1.4	0.4	0.4	0.4	0.2	0.4	
Trade and transport	0.3	0.9	0.4	0.2	0.1	0.2	0.2	
Finance and business	1.3	2.6	0.6	0.5	0.7	0.3	0.	
Public administration	1.3	1.2	0.3	0.5	0.4	0.2	0.5	

Sources: Eurostat and ECB calculations.







Output, demand and the labour market

# **EMPLOYMENT**

Employment increased by 0.3% quarter on quarter in the third quarter of 2005, following an increase of 0.2% in the second quarter and no change in the first (see Table 7). The increase in the third quarter compared with the second quarter was driven mainly by employment growth in the services sector. The main contribution to the strong performance of service-sector employment in the first three quarters of 2005 came from the finance and business and the public administration sub-sectors.

Survey data support the picture of gradually improving labour market conditions. The employment index of the PMI survey for the services sector rose in December to its highest value in more than four years. Positive developments were also recorded in the employment index of the PMI survey for the manufacturing sector. Employment expectations from the European Commission's confidence indicator improved further for both the industrial and the services sector in December. Employment expectations in industry are at their highest level since mid-2001.

Developments in euro area employment over recent years partly reflect the implementation of labour market reforms. These reforms may also have led to changes in the composition of employment, which should be taken into account when interpreting employment figures. Furthermore, some important methodological and statistical changes to employment data have been introduced in a number of euro area countries. Box 5 discusses these issues, concluding that while caution is needed in interpreting recent developments in euro area labour markets, recent figures point to improving conditions.

#### Box 5

## SOME COUNTRY-SPECIFIC FACTORS BEHIND RECENT EURO AREA EMPLOYMENT DEVELOPMENTS

Euro area employment data point to a gradual improvement in labour market conditions. According to national accounts data, euro area employment, measured in persons, rose by an average of 0.2% quarter on quarter from the beginning of 2004 to the third quarter of 2005, compared with just 0.1% during the years 2002 and 2003. However, when interpreting recent employment data, certain important country-specific factors should be taken into account. This box highlights the potential impact of various labour market reforms, as well as methodological and statistical changes, for some of the larger euro area countries. It concludes with an assessment of recent employment developments in the euro area as a whole against the background of these factors.

Labour market reforms have affected employment developments in a number of euro area countries. In Germany, while contracts in the form of full-time employment for an unlimited period have become less prevalent in the past few years, self-employment and part-time employment have been gaining importance.<sup>1</sup> These changes reflect the need for flexibility in enterprises and households, and have recently been further strengthened by measures introduced in the context of the Acts promoting modern labour market services (also known as the Hartz reforms). These include i) the introduction of financial assistance for unemployed persons setting up new businesses; ii) the promotion of part-time jobs paying up to €400 a month; and

1 For a comprehensive overview see the Deutsche Bundesbank, "Rapid change in paid employment", Monthly Report, July 2005, pp. 13-25.





iii) the promotion since the beginning of 2005 of "one-euro" jobs, which allow the long-term unemployed to take up specifically allocated work and earn up to  $\notin 2$  per hour on top of their unemployment benefits. As a result of these measures, the employment figures measured in terms of persons have improved considerably more than employment measured in hours worked. In France, a special two-year contract, entailing no costs for dismissing new recruits for enterprises with less than 20 employees, was introduced in August 2005 to boost employment in the private sector. In Italy, new legislation implemented in 2003 (known as the Biagi Law) introduced measures to enhance flexibility in the labour market, mainly in the form of more flexible part-time contracts and non-standard labour contracts for temporary workers. In the Netherlands, the Act on the extension of continued payment of salary in case of illness provided that from January 2005 firms must pay an extra year of sick-leave before employees can qualify for the disability scheme. This has led to an increase in measured employment, as workers are counted as employed as long as they receive sick-leave payment from their employers.

Statistical and methodological changes have also been implemented in some euro area countries. In Spain, revised employment data for the period from 1996 to 2004, published in March 2005, showed both higher levels and growth rates of employment. These upward revisions are due to the incorporation of a higher number of immigrant workers than previously recorded. In particular, the level of employment in the fourth quarter of 2004 was increased by 965,000 persons as a result of this factor. The revised annual growth rates reflect the fact that the immigrant population has been rising more strongly over the last few years than initially recorded. Moreover, the national statistical institute introduced a second set of important methodological changes in April 2005, aimed at adjusting its labour force survey questionnaire to EU requirements.<sup>2</sup> In Italy, the budget laws for 2003 and 2004 introduced measures to legalise immigrant workers in the grey economy and an opportunity for retired persons who have continued to work to register as employed. As the employment data in the national accounts already include an estimate of employment in the informal economy, these measures amount only to a redistribution between official and grey economy workers, without affecting the overall employment figures.

Overall, recent employment figures in some of the largest euro area countries have been affected by labour market reforms and by methodological and statistical changes, influencing the data for the euro area as a whole. Although the impact on the euro area data is very difficult to quantify, the effects in some countries appear to be non-negligible, suggesting that caution is needed when interpreting recent euro area employment developments. More detailed employment data, for instance on hours worked, would help to better evaluate the impact of such countryspecific factors. Nonetheless, even when these factors are taken into account, there are signs of improving labour market conditions, as indicated by a rising number of job vacancies and rising employment expectations in various surveys.

2 For a more detailed explanation see the Banco de España, "La revisión de las cifras de la EPA", Boletín Económico, April 2005, pp.14-15 and "EPA 2005: Resumen de los cambios metodológicos", Boletín Económico, May 2005, p. 32.



Output, demand and the labour market

# 4.3 THE OUTLOOK FOR ECONOMIC ACTIVITY

The information that has become available since the beginning of December 2005 tends to confirm the scenario of a strengthening and broadening recovery in euro area economic activity. New data show that employment has continued to strengthen in the euro area, and there have also been signs of a further improvement in consumer and industrial confidence. This scenario of sustained growth is in line with the Eurosystem staff macroeconomic projections published in early December as well as with forecasts by international and private sector organisations. Downside risks to this outlook mainly relate to prevailing global imbalances and uncertainties surrounding oil price developments.





# 5 EXCHANGE RATE AND BALANCE OF PAYMENTS DEVELOPMENTS

#### **5.I EXCHANGE RATES**

Following a period of weakness in November, the euro was relatively stable in December and appreciated moderately in early January. The strengthening of the euro in nominal effective terms over this period largely reflected its appreciation against the US dollar and, to a lesser extent, against the Chinese renminbi and the pound sterling, which was partially offset by its fall against the Japanese yen and some of the new Member States' currencies.

## **US DOLLAR/EURO**

Following a considerable decline against the US dollar in November, the euro appreciated moderately against the US currency in December and more strongly in early January (see Chart 27). Developments in the euro-dollar exchange rate were notably affected by market expectations about the future course of monetary policy in the United States and in the euro area. Favourable news about the euro area economy combined with a change in the Federal Reserve's communication on monetary policy decisions tended to weaken market expectations about a further widening of the US-euro area interest rate differential. Concerns about a new record high in the US trade deficit were mitigated by reports of very high portfolio investment inflows. Overall, on 11 January, the euro stood at USD 1.21, i.e. 2.7% above its end-November level and 2.8% lower than its 2005 average.

#### **JAPANESE YEN/EURO**

Following a substantial decline against the major currencies in recent months, the Japanese yen showed a relatively strong and broad-based appreciation in the second half of December. The recovery may have been related to technical factors, as investors reportedly closed sizeable short positions cumulated earlier against the currency. On 11 January the euro traded 1.8% lower than its end-November level and 1.1% above its 2005 average (see Chart 27).

#### **EU MEMBER STATES' CURRENCIES**

In December and early January, most currencies in ERM II were trading at or close to their respective central parity (see Chart 28). Together with some other currencies in central and eastern Europe, the Slovak koruna appreciated



vis-à-vis the euro (1.4%) to trade 2.8% above parity at the stronger end of the ERM II band on 11 January. With regard to the currencies of other EU Member States, the euro appreciated vis-à-



Exchange rate and balance of payments developments



±2.25%; for all other currencies the standard fluctuation band

Chart 29 Euro effective exchange rate and its decomposition 1)



**Contributions to EER changes <sup>2</sup>)** From 30 November 2005 to 11 January 2006 (in percentage points)



Source: ECB.

 An upward movement of the index represents an appreciation of the euro against the currencies of the most important trading partners of the euro area and all non-euro area EU Member States.

2) Contributions to EER-23 changes are displayed individually for the currencies of the six main trading partners of the euro area. The category 'NMS' refers to the aggregate contribution of the currencies of the ten new Member States which joined the EU on 1 May 2004. The category 'Other' refers to the aggregate contribution of the remaining seven trading partners of the euro area in the EER-23 index. Changes are calculated using the corresponding overall trade weights in the EER-23 index.

vis the pound sterling to stand at GBP 0.69 on 11 January, 0.9% stronger than its end-November level and 0.7% stronger than its 2005 average. The euro depreciated against the Swedish krona (1.9%), which experienced a rather strong recovery in the period under review, and against the Czech koruna, the Polish zloty and the Hungarian forint (by 0.6%, 3.4% and 1.1% respectively).

## **OTHER CURRENCIES**

of ±15% applies.

Since end-November the euro has been stable vis-à-vis the Swiss franc. It appreciated by 0.7% against the Norwegian krone, by 2.4% against the Canadian dollar, by 0.9% against the Australian dollar and by 2.6% against the Chinese renminbi. At the same time, it depreciated by 2.4% against the South Korean won.



#### **EFFECTIVE EXCHANGE RATE OF THE EURO**

On 11 January the nominal effective exchange rate of the euro – as measured against the currencies of 23 of the euro area's important trading partners – was 0.6% above its end-November level and 1.6% weaker than its average in 2005 (see Chart 29). The moderate strengthening of the euro in nominal effective terms in December and early January largely reflected a rise against the US dollar and, to a lesser extent, against the Chinese renminbi and the pound sterling, which was partially offset by its fall against the Japanese yen and some of the new Member States' currencies.

#### **5.2 BALANCE OF PAYMENTS**

Balance of payments data up to October 2005 indicate that values of extra-euro area imports and exports remain on a strong upward trend. While the rapid rise in the value of imports is mainly explained by rising import prices, the robust rise in exports largely reflects strong export volumes. The stronger growth in import values relative to export values is the main factor explaining why the 12-month euro area current account turned into a deficit of almost  $\notin 9$  billion in October 2005, compared with a  $\notin 43$  billion surplus a year earlier. In the financial account, the euro area recorded net inflows of  $\notin 60$  billion in combined direct and portfolio investment in the 12-month period up to October, compared with virtually balanced net flows until May 2005. This development reflects strong investment in euro area equities.

## TRADE AND THE CURRENT ACCOUNT

The latest balance of payments data for October 2005 confirm that values of extra-euro area imports and exports of goods and services remain on a strong upward trend, albeit one that is decelerating to some extent in comparison with previous months (see Chart 30). The three-month moving average

#### Table 8 Main items of the euro area balance of payments

(EUR billions; seasonally adjusted, unless otherwise indicated)

			Three-month moving average figures ending				12-month cumulated figures ending	
	2005	2005	2005	2005	2005	2005	2004	2005
	Sep.	Oct.	Jan.	Apr.	July	Oct.	Oct.	Oct.
Current account	-3.3	-9.0	1.9	1.8	-0.8	-5.8	42.6	-8.7
Goods balance	5.3	0.2	6.8	6.7	5.6	2.6	110.1	65.4
Exports	105.8	103.0	96.0	97.0	100.6	104.6	1,109.4	1,194.6
Imports	100.5	102.7	89.1	90.3	95.0	102.0	999.3	1,129.2
Services balance	2.5	2.6	2.2	2.5	2.0	2.4	29.1	27.0
Exports	31.7	32.1	30.3	31.1	31.4	31.7	356.3	373.4
Imports	29.2	29.6	28.1	28.6	29.4	29.3	327.2	346.4
Income balance	-4.9	-7.4	-1.4	-2.9	-4.0	-5.5	-39.8	-41.4
Current transfers balance	-6.2	-4.4	-5.7	-4.5	-4.4	-5.3	-56.8	-59.7
Financial account <sup>1)</sup>	26.0	-6.6	12.3	-1.3	18.5	4.7	-23.3	102.8
Combined direct and portfolio investment	16.5	-6.1	-5.5	-4.9	38.4	-8.0	-1.5	60.3
Direct investment	2.2	-0.9	-6.7	-7.0	-27.1	-4.6	-38.6	-136.1
Portfolio investment	14.3	-5.1	1.2	2.2	65.5	-3.4	37.1	196.4
Equities	5.9	-12.5	14.7	-16.1	53.4	4.5	-4.1	169.6
Debt instruments	8.5	7.4	-13.5	18.3	12.1	-7.9	41.2	26.8

Source: ECB.

Note: Figures may not add up due to rounding.

1) Figures refer to balances (net flows). A positive (negative) sign indicates a net inflow (outflow). Not seasonally adjusted.



Exchange rate and balance of payments developments



of exports increased by 3.2% in October compared with the corresponding figure for July (see Table 8), due to a strong rise in goods exports (3.9%) and a subdued growth in the exports of services (1.1%). Over the same period, imports continued to grow faster than exports (5.5%) owing to the rapid rise of imports of goods, while imports of services remained virtually flat.

The breakdown of extra-euro area trade in goods into volumes and prices provided by Eurostat (up to September 2005) suggests that the ongoing increase in the value of goods exports mostly stems from strong growth in export volumes. In the third quarter of 2005, export volumes to Asia (especially China) and the new EU Member States grew robustly. This reflects both the sustained import growth in these countries together with gains in the price competitiveness of euro area exports, partly due to the depreciation of the euro in the second and third quarters of 2005. In terms of product categories, export volumes of capital goods registered much stronger growth in the third quarter than intermediate and consumer goods.

Meanwhile, the recent increase in goods imports is mainly explained by the strong growth in import prices, although import volumes continued to accelerate as well. The rise in import prices recorded in the third quarter of 2005 was mainly related to the increase in the price of oil and, to a lesser extent, to the pick-up in manufacturing import prices. The latter might reflect both the depreciation of the euro as well as higher production costs for euro area import suppliers following the hike in oil prices.

The acceleration in import volumes which started in the second quarter of 2005 continued in the third quarter. This partly reflected the pick-up in euro area GDP growth over the period as well as the strong growth in import-intensive categories of expenditure, such as exports and investment.

In this context, the recovery in euro area imports can mostly be ascribed to significant increases in imports of capital goods (see Chart 31), while since the beginning of 2005, import volumes of consumer goods grew at a moderate pace and volumes of intermediate goods remained almost flat.

As a result of the above trends, the 12-month cumulated goods surplus fell to about  $\notin 65$  billion in October. Also, taking into account the reduction in the services surplus and the rise in the deficit for income and current transfers, the 12-month cumulated current account turned into a deficit of around  $\notin 9$  billion in October 2005 (or less than 0.1% of GDP), which compares with a  $\notin 43$  billion surplus a year earlier.

#### **FINANCIAL ACCOUNT**

The euro area combined direct and portfolio investment recorded average monthly net

outflows of  $\notin$ 8 billion over the three-month period up to October 2005 (see Table 8), as net outflows in direct investment ( $\notin$ 4.6 billion) and debt instruments ( $\notin$ 7.9 billion) more than offset net inflows in equity portfolio investment ( $\notin$ 4.5 billion). Net outflows in debt instruments recorded in September and October were the result of large net purchases of foreign debt by euro area residents, possibly related to the wider interest rate differentials between the United States and the euro area.

In 12-month cumulated terms the euro area recorded net inflows of  $\in 60$  billion in combined direct and portfolio investment in October (see Chart 32 and Table 8). In particular, net investment in euro area equities rose to  $\in 170$  billion from a broadly balanced position the previous year, partly reflecting the financing of one large transaction involving euro area direct investment abroad in July 2005. Nonetheless, even when adjusted by the above transaction, net investment in euro area equity securities still accounted for most of the net inflows in combined direct and portfolio investment in the 12-month cumulated period. Against this background, market survey data also indicated an interest in euro area equity securities among foreign investors, as in relative terms these were considered to be more attractively priced.

# Chart 32 Net direct and portfolio investment flows



Note: A positive (negative) number indicates a net inflow (outflow) into (out of) the euro area.



# THE PREDICTABILITY OF THE ECB'S MONETARY POLICY

Current best practice in central banking views a high level of monetary policy predictability as desirable. A clear distinction, however, is made between short-term and longer-term predictability. While short-term predictability can be narrowly defined as the ability of the public to anticipate monetary policy decisions correctly over short horizons, the broader and more meaningful concept of longer-term predictability would encompass the ability to understand the monetary policy framework of a central bank, i.e. its objectives and systematic behaviour in reacting to different circumstances and contingencies. In this broader sense, longer-term predictability is also closely related to the credibility of the central bank. This article reviews the main conceptual issues relating to predictability, in both its short and longer-term dimensions, and discusses how a transparent monetary policy strategy and constant communication have allowed the ECB to achieve a high level of overall predictability. Moreover, it presents descriptive evidence indicating that financial markets have been able to anticipate the ECB's monetary policy decisions with a degree of precision similar to that with which they anticipate the decisions of other large central banks. The ECB remains committed to fostering a genuine understanding of its monetary policy among the public and in financial markets, thereby contributing further to its already high level of predictability.

# I INTRODUCTION

Current best practice in monetary policymaking, as embodied in the monetary policy framework of the ECB, emphasises the desirability of a high level of predictability in central bank decisions. A distinction can be made between the notions of short-term and longer-term predictability. Short-term predictability is achieved when the public is in a position to anticipate correctly the central bank's next monetary policy decisions.<sup>1</sup> A more fundamental aspect of monetary policy predictability relates to its longer-term dimension, which requires that the public has a genuine understanding of the central bank's monetary policy framework and its behaviour over time. A high degree of predictability of interest rate decisions is the result of monetary policy being conducted in a credible and transparent manner that is well explained to the public. Hence, while predictability broadly understood is not an objective per se, it enhances the effectiveness of monetary policy and contributes to accountability vis-à-vis the public at large.

This article reviews the main conceptual and empirical issues related to predictability and illustrates how the ECB's monetary policy framework contributes to enhancing both the

short and longer-term predictability of the single monetary policy. Section 2 sets out a general discussion of predictability and its role in practical monetary policy-making. Section 3 presents the ECB's approach to enhancing the predictability of its monetary policy, while Section 4 provides empirical evidence focused on the short-term predictability of the ECB relative to that of other major central banks. Section 5 offers some concluding remarks.

# **2 PREDICTABILITY AND THE CONDUCT OF MONETARY POLICY**

The predictability of monetary policy is often understood in its narrower definition as the ability of financial markets to correctly anticipate the next monetary policy decisions of a central bank. Predictability of central bank decisions should not be restricted to this shortterm notion, however, as it does not adequately reflect the appropriateness of monetary policy decisions as regards the achievement of the objective of price stability. A more meaningful definition therefore relates to a longer-term dimension and centres on the central bank's close adherence to its institutional objectives

1 See the article entitled "Transparency in the monetary policy of the ECB" in the November 2002 issue of the Monthly Bulletin.







as well as its consistent and transparent use of the instruments available to achieve these objectives. A central bank is predictable in the longer term if its objectives are transparent and credible, and if these are consistently pursued in monetary policy decisions. This in turn will normally result in a genuine understanding on the part of the public of the behaviour of the central bank and, in particular, its systematic reactions to different circumstances and contingencies (see Chart 1).

short-term Against this background, predictability should not be taken mechanically as an indicator of monetary policy transparency. Rather, a high level of short-term predictability should be seen as the natural outcome of a central bank's consistent pursuit of its monetary policy strategy combined with communication that explains its objectives and economic assessment. As such, the short-term predictability of interest rate decisions is an observable reflection of the public's overall understanding of a central bank's monetary policy framework.

Longer-term predictability enhances the effectiveness of monetary policy mainly through its contribution to the formation of expectations on the likely future path of the economy. This expectation formation process is a crucial element in the transmission of monetary policy. In a market-based economy, the central bank can directly influence only short-term interest rates. However, consumption and investment decisions, and therefore also medium-term price developments, are to a large extent influenced by longer-term interest rates, which in turn depend on private expectations regarding future central bank decisions and inflation. As a result, it is important that the private sector is in a position to anticipate correctly the broad direction of monetary policy over the medium term.

Predictability reduces uncertainty about interest rates and thereby facilitates the pricing of assets and lowers risk premia, which in turn contributes to the efficiency of market allocation. It therefore allows firms to better manage their balance sheets, reduces their vulnerability to economic shocks and lowers risk management costs, thus creating the right conditions for investment decisions.

In addition, understanding of the monetary policy strategy among the public helps to guide price and wage-setting behaviour in a fashion that is consistent with the objectives of the central bank. This can only be achieved through consistent and credible implementation of the central bank's monetary policy strategy. In a setting where the private sector has no clear understanding of the central bank's reaction to economic developments, a perceived lack of commitment to maintaining price stability over the medium term may result in poor predictability and in inflation expectations hence not being anchored in a manner that is consistent with the central bank's objectives. Short-term changes in inflation and output might then become more protracted via wage and price-setting behaviour, possibly resulting in unwarranted economic fluctuations and welfare losses.

While a deeper understanding of the systematic behaviour of monetary policy will normally result in a high level of short-term predictability, perfect short-term predictability may not be attainable for a monetary policy geared towards the attainment of price stability over the medium term. First, perfect short-term predictability

The predictability of the ECB's monetary policy

could be trivially achieved if the central bank always mechanically executed the market's expectations, but this approach would not be appropriate. Since financial market expectations of future short-term interest rates largely reflect current market views about the forthcoming policy decision, a mechanical execution of market expectations by the central bank would result in these expectations becoming selffulfilling even though they do not necessarily reflect an adequate monetary policy stance to maintain price stability. By contrast, a central bank following a price stability-oriented monetary policy is able to underpin interest rate expectations by frequently providing an independent and comprehensive assessment of the current and prospective economic and monetary situation. Second, a central bank may achieve perfect predictability by systematically pre-announcing changes in interest rates and then implementing them under any contingency. However, an unconditional commitment by the central bank regarding the future path of policy rates would restrict the flexibility of its monetary policy framework by limiting its ability to react swiftly to rapid changes in the economic situation. The need to react quickly, on occasion, may also limit the opportunity to fully prepare markets prior to a monetary policy decision. As a result, while central banks have no intention to surprise markets, any indications about the monetary policy stance must be seen as conditional on new information relevant for assessing the risks to price stability.

A coherent track record of reliable policymaking is clearly indispensable for ensuring that the public understands the behaviour of the central bank. However, it is useful to note that this goal cannot be achieved by mechanically implementing a simple policy rule linking monetary policy rates to a small set of indicators in a perfectly predictable manner.<sup>2</sup> Changes in key variables affecting the monetary policy stance are often subject to substantial revisions and to uncertain structural relationships. Moreover, a monetary policy guided by a small set of indicators would not necessarily guarantee the attainment of the price stability objective. When reacting to risks to price stability and shocks, a central bank has to weigh up a number of factors, such as the nature and type of the shock, the current business cycle position, accumulated imbalances in the economy, the stability of the financial system and asset price developments.

Finally, communication plays a key role in enhancing predictability by allowing the public to understand monetary policy decisions. Given that monetary policy decisions are necessarily based on judgement and cannot be taken mechanically, there are limits to the extent to which these may be self-explanatory to the public. Therefore, a central bank has to be open and transparent in communicating its policy objectives and the underlying rationale of its decisions. It should not provide unconditional predictions of policy rates in the medium to long term, but allow the public to understand how monetary policy reacts systematically to different economic and monetary conditions and forthcoming developments. The combination of transparent objectives and consistent decisionmaking credibly explained through convincing communication to the public thus establishes a sound track record that forms the basis for a high level of monetary policy predictability.

#### 3 THE ECB'S MONETARY POLICY FRAMEWORK AND PREDICTABILITY

Each central bank has to find its own approach to matching "words and deeds" so as to enhance the predictability of its monetary policy. The approach chosen is inextricably linked to the history of the central bank and to the surrounding economic and institutional environment. Although in 1999 the ECB, as a newly created central bank, was able to build upon the experience of the NCBs in the Eurosystem, it did not have its own track record of consistent monetary policy. At the same time, the ECB was faced with a high degree of

<sup>2</sup> For a detailed discussion, see the article entitled "Issues related to monetary policy rules" in the October 2001 issue of the Monthly Bulletin.

uncertainty stemming from data, structural parameters and the need to identify the appropriate economic model for analysing the euro area economy. These conditions constituted a considerable challenge to the task of conducting predictable monetary policy.

From the outset, the ECB gave high priority to establishing a monetary policy strategy that could underpin a credible and predictable policy course.<sup>3</sup> First, when announcing its strategy in late 1998 the ECB's Governing Council quantified the objective of price stability in line with its mandate. This quantified objective provides a benchmark for the ECB's monetary policy, increasing its public accountability and providing a focal point for long-term inflation expectations.<sup>4</sup> While the empirical measurement of such expectations has to be carried out with some caution, existing measures derived from index-linked bonds and surveys suggest that the ECB has been successful in anchoring inflation expectations in the euro area.5

Second, the ECB has maintained a medium-term orientation in its monetary policy strategy. While a central bank sets its monetary policy in response to shocks hitting the economy and the existing structures and expectations, it also needs to take account of the fact that financial markets and the public at large try for their part to understand the general pattern of monetary policy when forming their expectations. Such a pattern may be easier to detect if the central bank operates in a rather gradualist manner and does not aim to fine-tune economic developments, in particular given the problems of measuring the state of the economy and the long and variable time lags.<sup>6</sup> The mediumterm orientation of the policy course pursued by the ECB helps the public to understand the reaction of the central bank to the course of broad economic trends, thereby contributing to lower overall volatility and avoiding the disturbances caused by an erratic interest rate-setting policy.

The ECB has also been proactive in being transparent and continuously communicating with the public. In this context, transparency can be defined as an environment in which the central bank provides in an open, clear and timely manner all relevant information on its mandate, strategy, assessments and policy decisions to the general public and the markets.7 The ECB sees transparency not only as an obligation to ensure democratic accountability, but also as an opportunity to communicate information facilitating the processes of learning and expectation formation in the private sector.<sup>8</sup> In particular, its systematic approach to monetary policy has been further clarified in a number of speeches, studies and Monthly Bulletin articles. At the same time it should be noted that the quest for transparency and predictability implies that information made available to the public should be well structured and organised. If it is not, it could be confusing and counterproductive to the process of efficient expectation formation.

The collegial manner of decision-making in the Governing Council is reflected in the introductory statement given at the press conferences held on Governing Council meeting days, at which monetary policy decisions are explained to the public almost in real time. Moreover, empirical research supports the view that the Eurosystem's communication about the economic situation and monetary policy stance has also been generally homogeneous.<sup>9</sup>

- 4 See Levin, A. T., F. M. Natalucci and J. M. Piger (2004), "Explicit inflation objectives and macroeconomic outcomes", ECB Working Paper No 383.
- 5 See in particular the intervention by Jean-Claude Trichet, President of the ECB, entitled "Monetary policy and 'credible alertness", at the panel discussion at the Jackson Hole Conference on 27 August 2005.
- 6 See Issing, O. et al. (2003), Background studies for the ECB's evaluation of its monetary policy strategy, ECB.
- 7 See the article entitled "Transparency in the monetary policy of the ECB" in the November 2002 issue of the Monthly Bulletin. See also Issing, O. (2005), "Communication, transparency, accountability: monetary policy in the twenty-first century", Federal Reserve Bank of St. Louis Review 87 (2, Part 1), pp. 65-83.
- 8 See the article entitled "The external communication of the European Central Bank" in the February 2001 issue of the Monthly Bulletin.
- 9 See Ehrmann, M. and M. Fratzscher (2005), "Communication and decision-making by central bank committees: different strategies, same effectiveness?", ECB Working Paper No 488.

<sup>3</sup> See ECB (2004): The Monetary Policy of the ECB.

The predictability of the ECB's monetary policy

#### 4 THE PREDICTABILITY OF THE ECB'S MONETARY POLICY: EMPIRICAL EVIDENCE

Despite the relatively short history of the euro area, a number of empirical studies have assessed the predictability of the ECB's monetary policy. Mainly for reasons related to measurability and the feasibility of empirical analyses, most of these studies have focused on the narrower notion of short-term predictability. Despite using somewhat different approaches and data, these studies have generally concluded that financial markets have predicted the ECB's monetary policy decisions well. This section presents and updates some of these results.<sup>10</sup>

There is no single approach to measuring shortterm predictability. Consequently, results from empirical studies using different approaches can vary to some degree. For example, measures of predictability can be based on information derived from different money market asset prices or surveys of financial market participants. The time horizon also matters: a shorter horizon focuses the empirical analysis towards the monetary policy decision on a given day and includes the information available to the central bank at the time of the decision, whereas a longer time horizon may incorporate additional information about the future path of monetary policy.

Short-term predictability is most commonly measured using changes in money market interest rates around the time of monetary policy decisions.11 Financial markets follow central bank decisions very closely and market interest rates therefore contain all the information available to the markets about the economic outlook and monetary policy stance at a given point in time. As a result, changes in market interest rates around the time of monetary policy decisions can be interpreted as a measure of the "surprise" element contained in the announced policy decision. In this context, both unexpected changes in the policy rate and no action when a change in the policy rate was expected constitute a surprise. From these data, "hit rates" – the number of monetary policy decision days when the surprise element was smaller than a given threshold value, divided by the number of all monetary policy decision days – can be calculated (see Chart 2). Higher hit rates indicate a higher degree of predictability. Monetary policy decision days include all days with scheduled meetings of the decision-making bodies, as well as those with unscheduled meetings at which interest rate decisions were taken. In the case of the ECB, this includes the monthly meeting of the Governing Council at which monetary policy decisions are normally discussed.<sup>12</sup>

In the exercise considered in this article, two threshold values were used to calculate different hit rates. They are defined as a 12.5 basis point daily change, corresponding to a 50% probability of a 25 basis point change in the policy interest rate (hit rate 1), and twice the normal volatility of daily changes (hit rate 2). While the threshold values and the consequent hit rates are to some extent arbitrary, they are a useful tool for comparing short-term predictability across major central banks. The hit rates are calculated using money market interest rates for assets with three different maturities (one month, three months and twelve months). As shown in Chart 2, the hit rates for different maturities and threshold

- 10 The analysis presented in this section closely follows the approach in Wilhelmsen, B.-R. and A. Zaghini (2005), "Monetary policy predictability in the euro area: an international comparison", ECB Working Paper No 504, including data up to 12 December 2005. Examples of other recent studies that also include an international comparison are Connolly, E. and M. Kohler (2004), "News and interest rate expectations: a study of six central banks", Reserve Bank of Australia Research Discussion Paper, and Ehrmann and Fratzscher (2005). See the references in these publications for a more complete list of studies.
- 11 In this article, the focus is on relatively simple, illustrative indicators of short-term predictability that can be easily compared across currency areas. These indicators were first used to measure the predictability of the ECB in Sicilia, J. and G. Pérez-Quirós (2002), "Is the European Central Bank (and the United States Federal Reserve) predictable?", ECB Working Paper No 192, and they have been more recently applied to data for several other central banks in Wilhelmsen and Zaghini (2005).
- 12 Until November 2001 these meetings took place twice a month. Monetary policy decision days also include the decision to lower interest rates taken at an unscheduled meeting on 17 September 2001.



Solites' ECBal Financial Data. Notes: Bars indicate hit rates for the ECB and lines represent the range of hit rates for a group of major central banks, i.e. the Federal Reserve System of the United States, the Bank of England, the Bank of Canada, the Reserve Bank of Australia, the Swiss National Bank and the Reserve Bank of Australia, the Swiss National Bank and the Reserve Bank of New Zealand. For details on the methodology, see Wilhelmsen and Zaghini (2005). The sample period is 1 January 1999 to 12 December 2005 (owing to the unavailability of data, the sample length for some assets in the international benchmark is slightly shorter). The underlying data are based on interbank rates of different maturities (EURIBOR for the euro area).

values indicate a high level of predictability for ECB decisions. The ECB hit rates are high in absolute terms, ranging from a low of 84% to a high of 96%, and in all cases are close to the upper bound of the range of hit rates for a group major central banks. The two hit rates provide similar information, with hit rate 2 providing a somewhat more stringent test of short-term predictability.

Chart 3 shows developments in the minimum bid rate for the main refinancing operations (MROs) of the Eurosystem together with daily changes in the one-month EURIBOR. The light circles close to zero (on the right-hand scale) correspond to days on which the absolute daily change in market rates was smaller than 12.5 basis points, i.e. days on which the financial markets forecast the ECB's monetary policy decisions well. The dark circles outside the band reflect days on which financial markets were surprised by the decision.

The chart further illustrates the high predictability of the ECB's decisions in the short term, as on most days financial markets anticipated the monetary policy decision. These results show that, out of a total of 120 days on which Governing Council meetings were held, financial markets were surprised –



Sources: Reuters and ECB calculations.

Notes: The thin lines represent  $\pm 12.5$  basis point threshold values. The circles are daily changes in the one-month EURIBOR around the time of the monetary policy decision day.



The predictability of the ECB's monetary policy

according to this definition - on only eight occasions. The greatest surprise occurred on 17 September 2001 when the ECB lowered interest rates at an unscheduled meeting as a response to the exceptional events of 11 September 2001. The surprises are roughly evenly split between days when the policy rate was changed and days when it was not. All surprises that occurred on days when no changes in policy rates were made were followed by a change in policy rates a month later, suggesting that these surprises were related to the precise timing of the decisions. It is also likely that some of the surprises were related to the size of the change in policy rates. This is particularly true for surprises that occurred within longer periods of gradual tightening or loosening of policy rates (such as in early 2000 or early 2003 respectively). Finally, the largest surprises occurred within the first three years of Monetary Union, indicating that the short-term predictability of the ECB may have increased over time. This evidence may reflect the fact that financial markets have gradually learned about the ECB's monetary policy framework and communication.<sup>13</sup>

The announcement of the ECB's policy decision is followed by a press conference at which the President provides a detailed explanation of the economic outlook for the euro area and the risks

to price stability. This communication is aimed at improving the public's understanding of the current decision and the possible future course of policy interest rates. Thus, a separate analysis of the volatility of market interest rates in short time windows around the time of the announcement and the press conference provides a useful starting point for measuring the impact of these events on financial markets. Evidence presented in Box 1 suggests that the volatility of long-term bond futures prices increases around the time that the ECB makes its monetary policy announcements and holds its press conferences, suggesting that both of these events contain information that is relevant to bond markets. However, the increase in volatility is relatively muted and short-lived, which is consistent with the interpretation that the ECB's decisions and its communication have, on the whole, been predictable.<sup>14</sup>

#### Box I

# THE EFFECTS OF THE ECB'S MONETARY POLICY ANNOUNCEMENTS AND COMMUNICATION ON LONG-TERM BOND MARKETS

Long-term interest rates reflect, among other factors, the views of market participants about the future path of monetary policy rates. As a result, through its monetary policy decisions and related communication, a central bank can have a significant impact on long-term interest rates. This box examines the extent to which monetary policy announcements and communications by the ECB tend to move the market for long-term bonds in the euro area. For this purpose, an assessment is made of how the price volatility of futures contracts on German ten-year government bonds (Bunds) behaves in short intraday time windows around the ECB's statement on its monetary policy decision and the press conference at which the President of the ECB elaborates on the decisions taken.



<sup>13</sup> The BIS documents a general improvement in the predictability of major central banks since the mid-1980s (see BIS (2004), Annual Report). In the case of the ECB, the increase in predictability may also be related to the fact that, from November 2001 to December 2005, monetary policy decisions were taken only once a month.

<sup>14</sup> For a discussion of the separate impact of policy announcements and communication in the case of the United States Federal Reserve System, see Gürkaynak, R., B. Sack and E. T. Swanson (2005), "Do actions speak louder than words? The response of asset prices to monetary policy actions and statements", International Journal of Central Banking 1(1), pp. 55-93. Complementary evidence based on the communication of the ECB between meetings is presented in Ehrmann and Fratzscher (2005).

The price volatility of Bund futures indicates whether and to what extent these policy events contain "news" for market participants that leads them to revise their expectations of the future course of monetary policy. If such events have an impact on market expectations, volatility should increase when compared with a comparatively "eventless" period of time. Even if monetary policy is largely predictable, some market reactions to monetary policy events would still be expected, resulting in increased volatility. First of all, central banks may have conveyed messages to the public that differ in tone from those of previous events. Second, even if a monetary policy decision and communication is fully anticipated and understood by market participants, the events may trigger portfolio adjustments by those individuals who deviated from the on-average correct anticipations. Nevertheless, any

# Bond price volatility around the ECB's monetary policy decision announcement and press conference





increase in market volatility should be relatively moderate and only short-lived. This means that if monetary policy decisions and communication tend to be largely anticipated and well understood, they should not trigger persistently higher uncertainty in financial markets.

To examine how volatility behaves around such times, the average absolute percentage price change in five consecutive ten-minute windows around each event taking place between January 1999 and August 2005 has been calculated. The first window covers the ten-minute period immediately before the announcement, the second covers the ten-minute period immediately after the announcement and the last three windows display average volatility up to 40 minutes after the announcement. These average volatilities are then used to construct a ratio between volatility on monetary policy meeting days and volatility on corresponding days when there was no meeting. A value larger than 1 indicates that monetary policy announcements and communication induced stronger market movements than could be deemed "normal" volatility had the announcements not been made. The greater the news content of an announcement, the stronger the financial market reaction should be. The results for the overall sample period are displayed in the chart.

The market impact following the ECB's announcement of its monetary policy decisions (see the bar corresponding to the ten-minute interval after the announcement) is reflected in an increase in volatility, which remains at a higher-than-normal level in the third interval (between 10 and 20 minutes after the announcement). However, volatility tends to fall back to normal quickly thereafter. This suggests that monetary policy announcements in the sample period tended to contain some news for the market, and that investors needed some time to fully adjust to the news. However, while the immediate increase in volatility is significant in statistical terms, its impact can still be regarded as relatively muted and it tends to be short-lived.

In a similar manner, higher-than-average volatility can be observed after the press conferences, with volatility also remaining elevated in the third time interval and gradually declining



The predictability of the ECB's monetary policy

thereafter. Part of the jump in volatility surrounding the press conference could also stem from the macroeconomic data releases which are published at the same time.

Overall, higher-than-average bond market volatility accompanied the ECB's monetary policy decisions and press conferences. This should reflect investors incorporating into prices the new pieces of information contained in the ECB's communication. However, the increase in volatility is moderate and short-lived, which is consistent with the view that the monetary policy actions taken by the ECB have been quite predictable.

More generally, while asset price volatility is influenced by a number of factors, it also provides a measure of the overall level of persistent uncertainty in financial markets when evaluated over a longer period. This uncertainty may partly originate from a poor understanding of the monetary policy framework. In this respect changes in market volatility over time may also provide information about possible changes in the predictability of monetary policy in a broader, longer-term sense. Box 2 presents evidence of a recent decline in market uncertainty and tentatively suggests that this decline is indicative of an increase in the predictability of the ECB.

#### Box 2

# CENTRAL BANK PREDICTABILITY AND IMPLIED VOLATILITY DERIVED FROM OPTIONS ON SHORT-TERM INTEREST RATE FUTURES

A central bank can reduce uncertainty regarding future levels of short-term interest rates by increasing the predictability of its future actions. A widely used measure of uncertainty is implied volatility derived from options on futures contracts on the three-month money

market interest rate. This box analyses the developments in the implied volatility derived from three-month EURIBOR futures.

Given appropriate assumptions, implied volatility is normally calculated using option pricing models to obtain an estimate of the expected dispersion of future changes in short-term interest rates measured in percentages per annum. However, this direct estimate can hide very different levels of volatility in the futures interest rates, as it depends on the level of the implied interest rate itself. This is addressed by weighting the implied volatility measured in percentages per annum by the level of the implied interest rate. For instance, a value of implied volatility equal to 20% is equivalent to an annualised expected deviation of 40 basis points in



(basis points; ten-day moving average)





interest rate changes if the interest rate implied by the futures rate is 2%. This box uses a derivation of a constant maturity measure, obtained on the basis of an interpolation of an implied volatility term curve weighted with a corresponding measure of the implied interest rate.<sup>1</sup>

The implied volatility with six months to maturity as derived from EURIBOR futures was, on average, 58 basis points in the period from February 1999 to December 2005 (see Chart). Between the second half of 1999 and the end of 2002 implied volatility fluctuated at around 70 basis points. In the first half of 2003 it fell to a level of around 50 basis points. It should be noted that the level of the ECB's key interest rates was unchanged between June 2003 and November 2005. In 2005 implied volatility averaged 33 basis points.

Overall, high volatility is often associated with periods of significant change in the slope of the money market yield curve as indicated by the spread between twelve-month and three-month interest rates. In particular, the large changes in the slope of the yield curve in 2002 were accompanied by high levels of implied volatility. The changes in the slope of the yield curve observed in 2004 and 2005 are comparable with previous episodes in most of 2001 and 2003. However, the level of implied volatility was significantly below the levels observed in those periods. The increase in implied volatility observed since October 2005 largely reflects uncertainty about the future path of interest rates. However, compared with the previous interest rate increase in late 1999, the level of implied volatility remains very low. Altogether, this supports the view that the ECB may have become more predictable over time. At the same time, this decline in volatility may also have been affected by the macroeconomic environment.

1 For a more comprehensive discussion on the analysis of implied volatility over longer periods, see the box entitled "Measures of implied volatility derived from options on short-term interest rate futures" in the May 2002 issue of the Monthly Bulletin.

In summary, the evidence presented in this section regarding short-term predictability supports the conclusion that financial markets have generally predicted the ECB's monetary policy decisions well in the shorter term. These results also show that, beyond the impact of the policy announcement in question, the ECB's communication on the day of the decision influences financial markets by providing them with additional information about the ECB's current and future policy decisions. Finally, lower market volatility in more recent years supports the view that understanding of the ECB's monetary policy framework has improved over time.

#### **5** CONCLUSION

In 1999 the ECB, as a newly created central bank, faced a considerable challenge in

establishing a monetary policy framework and communication that would foster predictability. It therefore announced, in late 1998, a medium-term-oriented monetary policy strategy, including a quantification of its objective of price stability and an analytical framework for guiding its decisions. This has been coupled with regular communication which aims to explain the ECB's assessment of the economic situation, the risks to mediumterm price stability and the way in which they systematically influence policy decisions.

After seven years of conducting monetary policy for the euro area, empirical evidence confirms that, in applying this approach, the ECB has achieved a high degree of short-term predictability. This is a natural outcome of a monetary policy strategy that emphasises a high level of predictability of central bank behaviour, underpinned

The predictability of the ECB's monetary policy

by transparent, comprehensive and timely communication. In the years ahead, the ECB will remain committed to fostering a genuine understanding of its monetary policy. This is essential for the effectiveness of its monetary policy and contributes to its accountability vis-à-vis the public at large. However, it should be recognised that short-term predictability is not an objective of monetary policy per se and that there are limits to further increases in the ability to anticipate the next policy decisions.



# HEDGE FUNDS: DEVELOPMENTS AND POLICY IMPLICATIONS

The near-default of Long-Term Capital Management (LTCM) in 1998 highlighted the significance of the hedge fund industry for the global financial system at large. Since then the hedge fund industry has continued to grow and develop unabated so that it has remained a focus of attention for authorities and the financial community. With an emphasis on the European dimension, this article describes the main features of the hedge fund industry and discusses recent developments. It also provides an overview of the current policy debate on regulation, which is centred on the implications of the increasing role of hedge funds for the financial system and the possible public and private sector initiatives to address them.

## **I** INTRODUCTION

The near-default of Long-Term Capital Management (LTCM) in September 1998 and the fall-out effects on the global financial system brought hedge funds firmly to the attention of both authorities and the financial community. The LTCM episode, however, proved to be only a temporary set-back for the long-term growth of this industry, which has continued unabated since then. Between December 1998 and September 2005, total hedge fund capital under management grew at an annual rate of 28%, with estimates exceeding USD 1 trillion.1 Whereas hedge funds were reserved initially for very wealthy investors, they have now developed into an important alternative investment vehicle accessible to both institutional investors and, directly or indirectly, retail investors.

The purpose of this article is to provide an overview of the hedge fund industry's development and the policy debate it has triggered. The industry is essentially global in nature but wherever possible its specific European dimension is also addressed. To this end, the article is divided into seven sections, with the following two providing facts on the hedge fund industry and the subsequent three focusing on the current policy debate.

Section 2 looks at the typical features of hedge funds and how they differ from traditional investment funds. Section 3 reviews the main developments in the industry over recent years. The implications for the financial system at large, mainly from a stability angle, are addressed in Section 4. Section 5 reviews the debate about the possible regulation of hedge funds, either directly or through their interactions with banks. Section 6 follows up on this issue by reviewing the supervisory implications for banks in the field of risk management practices and capital requirements. The final section draws some conclusions on the policy debate.

#### 2 CHARACTERISTICS OF HEDGE FUNDS

#### DEFINITION

The origin of the term "hedge fund" is related to the activities of the first institutions of this kind in the beginning of the second half of the last century.<sup>2</sup> These institutions were involved in buying and short-selling equities with the aim of eliminating (hedging) the risk of marketwide fluctuations. Since then hedge funds have become increasingly sophisticated in using a wide variety of other investment strategies that do not necessarily involve hedging. As a result, there is at present no generally accepted definition of what exactly a hedge fund is. Moreover, alternative terms have also occasionally been used - such as "leveraged investment funds", "highly leveraged

## ARTICLES

Hedge funds: Developments and policy implications



<sup>1</sup> This figure does not include private managed accounts accepted by hedge fund managers and managed using hedge fund-like strategies. According to Tremont Capital Management, total assets in such accounts were USD 325 billion at the end of June 2005.

<sup>2</sup> Alfred Winslow Jones is often credited with having started one of the first hedge funds as a private partnership in 1949. His hedge fund combined short-selling and leverage to hedge against stock market movements. Short-selling is the sale of borrowed assets that a seller does not own. Leverage refers to debt financing or the making of investments on margin.

institutions" and "sophisticated alternative investment vehicles" – which have the same definitional problems.

One possible way of defining hedge funds is to exclude various types of pooled investment vehicles from the funds' universe, rather than to try to single out their truly distinctive features. By following this approach, one would separate traditional investment funds (e.g. UCITS<sup>3</sup>) and other alternative funds (e.g. real estate, venture capital, private equity funds). The remaining funds could then be labelled as "hedge funds". However, such an approach would not be satisfactory for statistical or legal purposes and would in the end not add much clarity.

An examination of typical hedge fund characteristics (see Table 1) allows for a better understanding of some of the differences in relation to other investment pools. It tends to support the view that hedge funds represent a flexible business model rather than an alternative asset class. The key differences between hedge funds and other investment pools that emerge from such an analysis are that hedge funds generally have broad investment mandates, no or very limited regulatory restrictions on the type of instruments or strategies and that they make extensive use of short-selling, leverage and derivatives. The ability to pursue unconstrained and leveraged investment strategies lies at the core of hedge fund activities and should be an enduring feature, whereas other second-tier characteristics – including regulation, investor base and disclosure – will probably evolve.

For the purpose of this article, a hedge fund can therefore be described as a fund whose managers receive performance-related fees and can freely use various active investment strategies to achieve positive absolute returns, involving any combination of leverage, derivatives, long and short positions in securities or any other assets in a wide range of markets.

3 Undertakings of Collective Investments in Transferable Securities.

Investment strategies	Position-taking in a wide range of markets. Free to choose various investment techniques and instruments, including short-selling, leverage and derivatives.
Return objective	Positive absolute returns under all market conditions. Usually managers also commit their own money, hence preservation of capital is important.
Incentive structure	Typically a 2% management fee and a 20% performance fee. Quite often high "watermarks" apply (i.e. performance fees are paid only if cumulative performance recovers any past shortfalls) and/or a certain hurdle rate must be exceeded before managers receive any performance fees. Moral hazard stemming from asymmetric performance fees is to some extent curtailed by high watermarks and managers co-investing their own money.
Subscription/ Redemption	Predefined schedule with quarterly or monthly subscriptions and redemptions. Lock-up periods for up to several years until first redemption. Some hedge funds retain the right to suspend redemptions under exceptional circumstances.
Domicile	Offshore financial centres with low tax and a "light touch" regulatory regime, as well as some onshore financial centres.
Legal structure	Private investment partnership that provides pass-through tax treatment or offshore investment corporation.
Managers	May or may not be registered or regulated by financial supervisors. Managers serve as general partners in private partnership agreements.
Investor base	High net worth individuals and institutional investors. Not widely available to the public. Securities issued take the form of private placements.
Regulation	Generally minimal or no regulatory oversight due to their offshore residence or "light touch" approach by onshore regulators; exempted from many investor protection requirements.
Disclosure	Voluntary or very limited disclosure requirements.

#### Table I Typical hedge fund characteristics



Hedge funds: Developments and policy implications

#### **TYPES OF STRATEGY**

The investment style of a hedge fund is more important to its risk-return profile than its asset class selection or sector/geographic orientation. In general, four major groups of strategies can be distinguished:

- Directional hedge funds generally try to anticipate market movements and offer returns commensurate to the high risks and leverage involved. Macro hedge funds are the most prominent example of this investment style. These funds follow a "top-down" approach and try to take advantage of major economic trends or events. By contrast, emerging markets and other directional hedge funds with a regional focus favour a "bottom-up" approach, i.e. they tend to be asset-pickers in certain markets and look for inefficiencies in developing markets.
- Market neutral hedge funds (also referred to as "arbitrage" or "relative value" funds) search for relative value or arbitrage opportunities to exploit various price discrepancies and try to avoid exposure to market-wide movements. Returns from such strategies usually exhibit lower volatility, but their implementation requires medium to high leverage in order to benefit from small pricing distortions, particularly in bond and other credit markets.
- Event driven strategies try to take advantage of "special situations" in a company's life, such as mergers and acquisitions, reorganisations or bankruptcies. These strategies lie somewhere in the middle of the volatility spectrum, with corresponding medium volatility and low to medium leverage. Some event driven hedge funds, specialising in securities of distressed companies, try to exploit the fact that it is difficult to value such securities and that institutional investors are prohibited from investing in them.
- Funds of hedge funds (FOHFs) invest in a number of other hedge funds and are

expected to have lower volatility and attractive risk-adjusted returns due to diversification benefits.

## **PARTIES INVOLVED**

Hedge funds are predominantly domiciled offshore, meaning that they generally have minimal regulatory intervention and a favourable tax treatment, although their managers generally conduct their operations from major financial centres. Most of the European hedge funds, for instance, are managed from London.

Hedge fund managers prefer to concentrate on their proprietary trading strategies (where their strengths are) and typically outsource support services to fund administrators. Administrators provide a variety of services, including the valuation of positions and the calculation of the fund's net asset value, legal counselling, assistance in reporting and the processing of investor transactions. Position valuation and net asset value calculation are particularly important for ensuring that investors have adequate information on a hedge fund's performance and its investment portfolio.

Hedge fund investment strategies involve substantial trading and thus require extensive operational support, brokerage and financing services from "prime brokers", i.e. banks or securities firms offering brokerage and other professional services to hedge funds and other large institutional clients. Prime brokerage platforms facilitate the financing, risk management, execution, clearance and settlement of transactions. Other services include custody of assets, access to research, consulting and the introduction of managers to potential investors. The major share of prime brokers' income comes from trading commissions and collateralised cash or securities-lending to facilitate short-selling.

Sometimes the assets of a hedge fund are deposited with a custodian bank instead of a prime broker. For hedge fund investors, this arrangement serves as an additional safeguard as the custodian bank is subject to fiduciary duties vis-à-vis them, whereas a prime broker holds assets largely as a principal and as a security against its underlying fund positions.

#### 3 DEVELOPMENTS IN THE HEDGE FUND INDUSTRY

#### **CAPITAL UNDER MANAGEMENT**

Persistently low interest rates and ample liquidity led to a global search for yield that began in 2003.<sup>4</sup> Faced with the unsatisfactory performance of traditional assets, such as bonds and equities, many investors turned to hedge funds to improve their risk-adjusted returns. Investors were particularly attracted by the performance profile of hedge funds (see



Sources: Datastream, Bloomberg and ECB calculations. Notes: CSFB/Tremont Index and sub-indices began in January 1994, except Multi-Strategy, which began in April 1994, and Investable, which began in January 2000. Global Bond Index (GBI) EMU \$ began in January 1995. Chart 1), which is largely uncorrelated with that of other assets.

Inflows into the hedge fund industry have been particularly strong since 2002 (see Charts 2 and 3). The European segment, comprising funds either domiciled or managed from Europe, has been growing faster than the whole industry and is estimated to account for at least 20% of capital under management globally.<sup>5</sup> Another development is the growing share of multi-strategy funds, as both managers and investors appear to prefer the ability to switch among investment strategies depending on market conditions (see Charts 4 and 5).

However, there are some signs that inflows have been decelerating following recent mediocre returns. This has raised questions about whether there are capacity constraints for the hedge fund industry to continue delivering high absolute returns as both the number of market imperfections and resulting arbitrage opportunities may eventually decline. If capacity limits are reached, this would probably induce more pressure on hedge fund fees and attrition rates would increase, especially as some hedge funds are increasingly found to be taking exposures towards general market risk rather than providing extra returns resulting from active management. Expansion beyond capacity limits could also lead to the process of redistribution of capital among hedge funds themselves, as active hedge fund trading would itself create opportunities for other hedge funds. According to another scenario, the differences between the traditional fund management industry and hedge funds could become more blurred as conventional funds start using investment techniques similar to hedge funds and the latter are compelled to lower their fees.

4 See ECB (2004), Financial Stability Review, December; ECB (2005), Financial Stability Review, June; ECB (2005), Financial Stability Review, December.

<sup>5</sup> See Garbaravicius, T. and F. Dierick (2005), "Hedge funds and their implications for financial stability", ECB Occasional Paper No 34, August.

Hedge funds: Developments and policy implications



#### **INVESTOR BASE**

Throughout the 1990s, high net worth individuals were the dominant investors in hedge funds (see Chart 6). This fact, notwithstanding the LTCM episode, diluted somewhat the systemic concerns of such funds. However, the growing interest from institutional investors has, over time, changed the investor profile, as even moderate absolute hedge fund returns can enhance the overall risk-return profile of institutional portfolios thanks to the low correlation of hedge fund returns with traditional investments.

The growing role of FOHFs is another noticeable trend (see Chart 7), as even institutional investors often prefer to rely on their expertise and diversification benefits despite the second layer of fees charged on top of the fees of the underlying single hedge funds. Futhermore, the attrition rate is rather high among single hedge funds – another reason why some investors prefer FOHFs.



# Moreover, institutional investors often have a minimum amount to allocate in absolute terms or in relation to the capital under management of a target hedge fund owing to the high costs associated with the due diligence and monitoring of a large number of funds. This makes FOHFs the key source of funds for smaller hedge funds, because the latter are usually too small for institutional investors. At the same time, FOHFs are an important vehicle for retail investors acquiring access to hedge funds owing to the lower minimum investment requirements or lower restrictions on public offering in some countries.

As more institutional investors consider investing in hedge funds, they also bring requests for stronger governance and better risk management. This may lead to some consolidation in the industry as the costs of running a hedge fund have been increasing, and many funds are rather small with less than USD 100 million under management. Consolidation could take place in tandem with the growing institutionalisation of the hedge fund industry, as banks increase their participation by acquiring or setting up their own funds in response to investors' demand for a broader spectrum of alternative investments.

### Chart 7 The importance of FOHFs

(FOHFs as a percentage of single hedge funds; end-of-year data)



#### **4 FINANCIAL STABILITY IMPLICATIONS**

#### **POSSIBLE POSITIVE EFFECTS**

There is often more discussion about the risks posed by hedge funds than discussion about the positive aspects of their activities. However, hedge funds can also have beneficial effects. They may contribute to market liquidity, as they tend to be more willing to put their capital at risk in volatile market conditions so that market shocks can be absorbed. The presence of hedge funds as active risk-takers may also contribute to the development of fledgling and sophisticated over-the-counter markets, such as the credit derivatives market, and enhance the spreading of risks among market participants. In their quest for excess returns, hedge funds arbitrage away price differences for the same risk across markets, which is beneficial to the price discovery process. It may be argued that in this way hedge funds also contribute to the integration of financial markets. Their activity may also enhance the disciplinary force exercised by markets. Furthermore, hedge funds offer more possibilities for diversifying portfolios, thereby increasing the completeness of financial markets and ultimately leading to

8 Monthly Bulletin January 2006

Hedge funds: Developments and policy implications

greater social welfare. Finally, all these features, taken together, suggest that hedge funds can even contribute positively to the stability of the global financial system.

## **POSSIBLE NEGATIVE EFFECTS**

The rapid growth of the hedge fund industry also raises important questions about possible negative implications for financial stability. Hedge funds can cause financial instability through their potential impact on financial markets and banks. These two channels are closely related and a hedge fund-related triggering event associated with either of them may be further escalated through these mutual links.

#### (A) THROUGH FINANCIAL MARKETS

The near-collapse of LTCM in September 1998 provides the most vivid example of how hedge funds have the potential to disrupt the functioning of global financial markets. The prevention of a similar event occurring depends critically on the application of prudent risk management practices by both hedge funds and banks. In this respect, it is particularly important that both take into account the interaction between leverage, credit risk and liquidity risk. If not supported by adequate liquidity reserves or borrowing capacity, leveraged and possibly undiversified market risk associated with high return objectives can force a fund to default on its margin calls and other obligations. The situation can be further exacerbated by asset illiquidity in stressed markets, as hedge funds may not be able to unwind their positions at reasonable prices and banks may encounter difficulties in liquidating collateral.

Since the LTCM event, most hedge funds have seemed to make more cautious use of leverage, though a comprehensive assessment is hampered by the limited disclosure of the industry. Many of the largest hedge funds are also diversified among several strategies, which also reduces somewhat the concerns. Nonetheless, there are some indications that lower profit opportunities in the current environment of low interest rates, low volatility and high perceived liquidity have prompted some hedge funds to increase their exposures to illiquid instruments. At the same time, hedge funds seem to be seeking more stable funding facilities from banks or imposing longer lock-up periods to protect themselves from sudden withdrawals by investors. However, lock-ups alone do not provide a complete picture of redemption risk, as redemption frequency, notice periods and early redemption possibilities (after paying the "gate" fees) have to be considered as well.

Another often debated issue is the impact of hedge fund trading on market volatility. Hedge funds are often blamed for their aggressive short-term-oriented strategies, which may cause excessive volatility and destabilise financial markets. However, it is not clear whether hedge fund managers generally tend to momentum or contrarian traders. be Momentum or positive feedback trading refers to the buying of financial instruments after price increases and selling after decreases, which can amplify price swings or lead to bubbles. In this respect, the study by the US Commodity Futures and Trading Commission (CFTC) is particularly interesting. On the basis of micro trading data, it found that managed futures hedge funds<sup>6</sup> can dampen, rather than increase, volatility in energy markets by liquidity providing to other market participants.7

A further concern is related to the "crowding" of hedge fund trades. As an increasing number of funds attempt to exploit profitable opportunities from similar strategies, the positioning of individual hedge funds can become more similar or crowded. If market participants try to liquidate their positions simultaneously, this could leave hedge funds,



<sup>6</sup> Managed futures hedge funds invest in financial and commodity futures markets, and are reportedly cited as adopting trend-following (i.e. momentum) strategies.

<sup>7</sup> Haigh, M. S., J. Hranaiova and J. A. Overdahl (2005), "Price Dynamics, Price Discovery and Large Futures Trader Interactions in the Energy Complex", CFTC, April.





capital under management (excluding FOHFs) at the end of December 2004, as reported by Tremont Capital Management.

investment banks with hedge fund-like strategies, and affected markets vulnerable to adverse market dynamics. This concern is partly validated by the fact that correlations of hedge fund returns within strategies have recently been increasing for some strategies (see Chart 8). Moreover, the correlations are the highest for convertible arbitrage and credit strategies, which usually have the highest leverage. The unwinding of leveraged positions could be disruptive for affected markets, especially if the degree of liquidity in these markets was to prove low.

#### (B) THROUGH BANKS

Bank exposures to hedge funds can be divided into direct and indirect exposures. Direct exposures include financing, trading, investment and income exposures. Indirect risks arise from exposures to counterparties that in turn have exposures to hedge funds and to financial markets affected by hedge funds. Among direct risks, financing and trading links with hedge funds constitute the greatest source of risk, given the complexities associated with the management of such exposures.

In 2005 the ESCB's Banking Supervision Committee conducted a survey on large EU banks' exposures to hedge funds.<sup>8</sup> The survey provided some evidence that exposures of large EU banks varied significantly across countries. Generally, they were not large in relation to banks' balance sheets or total income and were mostly in the form of investments. This is at least partially due to the fact that the global prime brokerage market is largely dominated by US financial institutions. But even the limited data showed that exposures were growing rapidly and are likely to continue doing so in line with the further expansion of the hedge fund industry and its European segment.

Generally speaking, the banks surveyed had stringent requirements for exposures to hedge funds, with a strong emphasis on collateralisation. Most banks reported the use of stress tests for the evaluation of potential effects of volatile or illiquid markets on their exposures. At the same time, the survey highlighted scope for further improvement in a number of areas. Stress testing, for example, seemed to be less common for collateral and was mostly limited to individual exposures.

8 ECB (2005), "Large EU banks' exposures to hedge funds", November.

CB Monthly Bulletin January 2006

Hedge funds: Developments and policy implications

Some banks had difficulties in aggregating - on a firm-wide basis - their exposures to individual hedge funds and groups of hedge funds with similar strategies. Banks also experienced difficulties in assessing the risk profile of a fund as a whole, particularly in the case of large funds with financing and trading relationships involving several counterparties. Furthermore, certain shortcomings regarding the quantity, quality and timeliness of information provided by hedge funds to banks were identified. Moreover, counterparty discipline, as applied by banks, was found to be under pressure owing to competitive market conditions. Hedge funds, particularly the larger ones, were successful in negotiating less rigorous credit terms, including, for example, lower lending spreads, higher net asset value decline triggers or trading on a variation margin only.

In conclusion, the survey indicated that most of the recommendations made after the neardefault of LTCM (see Section 6) remain relevant. Banks should therefore continue to strengthen their risk management further and keep exerting pressure on hedge funds to increase transparency.

#### **5 REGULATORY ISSUES**

The strong development of hedge funds has raised a number of concerns that are central to the debate about whether hedge funds should be regulated, and if so how. There are three main reasons why one might consider regulation: for financial stability, to protect investors and for market integrity. Another, though less frequently quoted reason, is that a common (light) regulatory regime might benefit market integration. This argument was raised in the debate about a possible EU regime for hedge funds that would benefit from the "single European passport", as is currently the case for UCITS.

The financial stability reason is probably the most relevant, but, as mentioned in

Section 4, it is very difficult to make a definitive judgement about the net result of positive and negative aspects. The investor protection reason is very much tied to the question of to what extent retail investors should be permitted to invest in hedge funds and be protected against practices such as inappropriate selling and insufficient disclosure. In some countries hedge funds have become more accessible to retail investors. This "retailisation" can occur directly, but very often it takes place in an indirect way through FOHFs or financial instruments whose performance is linked to that of hedge funds (e.g. certain types of structured notes or unitlinked insurance policies). Finally, the market integrity reason is based on the argument that the international and unregulated character of the hedge funds business makes it particularly vulnerable to illicit activities, such as fraud, market abuse and money laundering. However, there is no conclusive evidence that such abuses occur more frequently in the case of hedge funds than for other types of unregulated intermediaries.

As regards the possible regulatory approaches, a wide spectrum is available ranging from no regulation at all to full direct regulation, with various alternatives in between. These alternatives include self-regulation (e.g. through codes of conduct adopted by the asset management industry), indirect regulation (through the interaction of hedge funds with regulated counterparties) and soft direct regulation (e.g. by regulating certain aspects, such as disclosure or the interaction with retail investors). Since the hedge fund business is very international in nature and can easily evade national regulations, the emphasis up to now has been on indirect regulation, which is ultimately more in the field of banking supervision. Recently, however, the Securities and Exchange Commission (SEC) in the United States adopted a regulation targeted at hedge fund advisors (managers), who will be required to register with the SEC before February 2006 and will henceforth be subject to the provisions of the Investment Advisers Act.

ECB
### **CURRENT EU FRAMEWORK**

The current regulatory framework for investment funds in the EU is based on the UCITS regime.9 UCITS are collective investment schemes that are dedicated to the investment of funds raised from retail investors. They benefit from the "European passport", meaning that once they have been authorised in one Member State they can also be offered to retail investors in all other EU Member States, subject only to a simple notification. As a corollary to this greater ease of cross-border commercialisation, risks to retail investors are limited through strict rules as regards the UCITS' investment policy, capital and disclosures, asset-safekeeping and oversight by an independent depository. The UCITS regulation is quite restrictive in terms of financial products to invest in and the risk diversification rules that apply, so that the typical hedge fund would be excluded. However, as a result of the "UCITS III" regime, investment restrictions have been loosened. For example, UCITS are now allowed to make greater use of derivatives and leverage so that they have more possibilities to engage in hedge fund-like strategies.

At present, there is no common regulatory regime in the EU that specifically addresses hedge funds or their managers. However, several countries have adopted domestic legislation. For example, France, Germany, Ireland, Italy, Luxembourg and Spain are countries in the euro area that have introduced national legislation for single hedge funds or FOHFs. The focus of these national rules seems to be mainly on investor protection, though they differ in various aspects, such as the way hedge funds can be distributed, subscription restrictions, rules regarding the management of the fund and disclosure requirements.

Finally, though not specifically targeted at hedge funds or the asset management business, there are a number of more general EU rules that may also affect the hedge fund industry. These include the Markets in Financial Instruments Directive (MIFID)<sup>10</sup>, the Prospectus Directive<sup>11</sup> and the Market Abuse Directive<sup>12</sup>. For example, a hedge fund manager provides investment advice and portfolio management services, activities which are covered by the MIFID. Hence, the manager also needs to comply with the ensuing obligations in areas such as order execution, conflicts of interest and risk management.

## **RECENT EU INITIATIVES**

National differences regarding nonharmonised asset management products, such as hedge funds, result in fragmentation that may hamper the development of a single market. This was an important consideration in the resolution adopted by the European Parliament in January 2004 with the proposal to adopt a light regulatory regime for "sophisticated alternative investment vehicles" (including hedge funds). The purpose of this proposal was to bring funds onshore that are presently offshore and to provide them with the benefit of the European passport.

The question of hedge fund regulation at the EU level is also being addressed within the current discussions on the overall financial services strategy for the next five years, now that the Financial Services Action Plan (FSAP) is nearing completion. An expert group that was set up to provide input in the area of asset management recommended in May 2004 that the European Commission review the framework for non-harmonised products, such as hedge funds, with the aim of developing a pan-European market. In this respect, the current UCITS framework was seen as a useful reference point that could be adapted to the specific nature of such products.

- 10 Directive 2004/39/EC.
- 11 Directive 2003/71/EC.
- 12 Directive 2003/6/EC.

<sup>9</sup> See in particular Directive 85/61/EEC, as amended, inter alia, by Directive 2001/107/EC ("Management Company Directive") and Directive 2001/108/EC ("Product Directive"). These latter two Directives are generally known as "UCITS III".

## ARTICLES

Hedge funds: Developments and policy implications

Basel Committee on Banking Supervision, Banks' Interactions with Highly Leveraged Institutions (HLIs)
Basel Committee on Banking Supervision, Sound Practices for Banks' Interactions with HLIs
Counterparty Risk Management Policy Group I, Improving Counterparty Risk Management Practices
International Organization of Securities Commissions, Report on Hedge Funds and Other HLIs
Basel Committee on Banking Supervision, Banks' Interactions with HLIs: Implementation of the Basel Committee's Sound Practices Paper
Financial Stability Forum, Report of the Working Group on HLIs
Basel Committee on Banking Supervision/International Organization of Securities Commissions, Review of issues relating to HLIs
Alternative Investment Management Association, Guide to Sound Practices for European Hedge Fund Managers
Counterparty Risk Management Policy Group II, Toward Greater Financial Stability: A Private Sector Perspective
Managed Funds Association, Sound Practices for Hedge Fund Managers (update of the practices papers of 2000 and 2003)

Table 2 Major initiatives of international organisations related to hedge funds

In May 2005 the Commission published a Green Paper outlining its preliminary views on financial services policy for the next five years. However, the communication did not indicate any particular Commission initiatives as regards hedge funds. In July of the same year, the Commission launched a public consultation on the enhancement of the EU framework for investment funds in which it stated that there was currently no compelling evidence for EU legislation on hedge funds. It nevertheless confirmed that further attention should be paid to the growing "retailisation" of hedge funds, the impact of this on financial markets and the exposures of investment banks. To investigate these issues in greater detail, it announced the creation of an industry-wide working group on alternative investment strategies.

## **6** SUPERVISORY ISSUES

Any direct regulation of hedge funds is confronted with the problem that the industry is global in nature and that hedge funds can easily relocate their domicile, thus evading national regulation. This is basically why all international initiatives regarding hedge funds taken since the LTCM episode have tried to influence the activity of hedge funds through their interactions with regulated firms, and in particular banks. Most of these initiatives relate to sound risk management practices. Although banking supervisors were the first to publish specific guidance in this area, the private sector has also taken several initiatives (see Table 2). Another important supervisory dimension is how hedge fund exposures are dealt with under the capital rules for banks.

## **RISK MANAGEMENT PRACTICES**

## (A) PUBLIC SECTOR INITIATIVES

In 1999, when reviewing banks' dealings with hedge funds in the aftermath of the LTCM episode, the Basel Committee on Banking Supervision (BCBS) identified a number of weaknesses in banks' credit risk management. These included an insufficient weight placed on in-depth credit analyses of counterparties, an over-reliance on financial collateral to limit credit risk, and deficiencies in the measurement and management of exposures. To address these weaknesses, the Committee developed specific guidance for banks' interactions with "highly leveraged institutions" (HLIs), which are primarily hedge funds. These recommendations are complementary to the BCBS' general risk management guidance, such as in the area of credit risk.<sup>13</sup> The Financial

13 Basel Committee on Banking Supervision (2000), "Principles for the Management of Credit Risk", September. Stability Forum (FSF) later stressed that strong counterparty risk management and enhanced oversight of HLI credit providers are key elements with which to provide an adequate response to the systemic risk concerns posed by HLIs.

The BCBS guidance requires banks to establish clear policies and procedures for their interactions with HLIs as part of the general credit risk management. This has to include adequate information gathering, due diligence and satisfactory credit analysis. Credit exposures should be correctly measured and closely monitored. As the measures have to be adapted to the nature of the dealings with HLIs, which often include trading and derivatives transactions, the use of "potential future credit exposure" measures 14 and stress testing would normally be part of such a process. To bound potential losses, exposures need to be subject to formal limits. Another way to limit losses is to better align collateral and contractual provisions with the features of HLIs, for example by requiring initial margins and by using covenants that allow an early termination in the event of a material deterioration in a HLI's credit quality. The BCBS and the International Organization of Securities Commissions (IOSCO) also reviewed to what extent the sound practices had actually been implemented. They concluded that although banks had improved their risk management practices, further improvements were possible in the due diligence process, exposure measurement and stress testing.

## **(B) PRIVATE SECTOR INITIATIVES**

The various public sector initiatives are complemented by the efforts of the financial industry to improve risk management standards, something that was also called for by the authorities in the aftermath of the LTCM crisis.<sup>15</sup> One of the most significant initiatives in this respect are the recommendations of the Counterparty Risk Management Policy Group (CRMPG), a group of major, internationally active commercial and investment banks. The first report, released in 1999, aimed at improving internal counterparty credit and market risk management practices. It was followed in 2005 by an update, driven by developments such as the increased proliferation of hedge funds and complex financial instruments. Some of the CRMPG recommendations are targeted at individual firms (in particular for risk management practices), others at the industry (e.g. in the area of netting and master agreements) and a third category at authorities (e.g. to investigate the reporting of large exposures by regulated firms).

Finally, in line with the policy recommendations of the FSF, the asset management industry also took several initiatives to improve sound risk management practices for hedge fund managers. For Investment example, the Alternative Management Association (AIMA) and Managed Funds Association (MFA) both issued recommendations in this respect.

## **CAPITAL REQUIREMENTS**

Banks are required by supervisors to hold regulatory capital in relation to the risks they take on. At present, the capital requirements are based on crude credit risk measures as laid down by the BCBS in its Capital Accord of 1988 ("Basel I"). In 1996 the Accord was extended to cover market risk. A fundamental change in the capital rules came about in 2004 when a new, and much more developed and risk-sensitive framework was adopted ("Basel II"), which countries will implement in the coming years.<sup>16</sup> Whereas Basel I only covered minimum capital requirements, the Basel II

<sup>14</sup> Current credit exposure is equal to the value of credit outstanding or the replacement cost of trading positions. Potential future credit exposure, by contrast, takes into account the possible variations in the value of the current credit exposure over the life of the trading positions.

<sup>15</sup> See, for example, The President's Working Group on Financial Markets (1999), "Hedge Funds, Leverage, and the Lessons of Long-Term Capital Management", April.

<sup>16</sup> Basel Committee on Banking Supervision (2004), "International Convergence of Capital Measurement and Capital Standards", June.

framework is based on three complementary pillars:

- Pillar I minimum capital requirements,
- Pillar II the supervisory review process, and
- Pillar III market discipline.

An important innovative feature of the new framework is that, in order to calculate the minimum capital requirements for credit risk, banks can now rely for their risk assessment either on ratings provided by rating agencies (the "standardised approach") or on ratings that result from their own risk assessment models (the "internal ratings-based approach").

The capital requirements for banks do not provide for a specific treatment of exposures to hedge funds. Hence, they have to be fitted into the general solvency framework. In this respect, Basel II is much better suited to deal with the risks that hedge funds pose.

Whereas Basel I does not provide much differentiation of capital requirements in terms of risk levels, this is much more the case under Basel II, both under the standardised approach and the internal ratings-based approach. Hence, Basel II can better accommodate the increased risk that hedge funds generally demonstrate as a result of their high leverage, relative opacity and dynamic risk profile.

But even under Basel II, banks may face considerable challenges when using their own models to estimate the risk of hedge funds exposures. Potential difficulties include the suitability of credit risk models originally developed for corporate clients to hedge funds that have a very different and more complex risk profile. For market risk models, difficulties may relate to the low frequency of net asset value figures (needed to calculate volatilities) and the skewed distribution of returns that result from certain investment strategies.

Furthermore, under the supervisory review process of Basel II, the bank's management has to make sure that the institution has adequate capital to support its risks; supervisors should take appropriate action when this is not the case. Such action can, for example, include requiring the bank to strengthen its risk management, improve internal controls, increase provisions and, ultimately, even increase capital. Pillar II therefore provides a useful framework for ensuring that the bank adequately addresses its risks, including those resulting from its interactions with hedge funds.

### 7 CONCLUSION

Hedge funds have shown very impressive growth over recent years and have developed into an important alternative investment instrument that has also become increasingly available to retail investors. This development, as well as the international character of the hedge funds business and its largely unregulated nature, poses considerable challenges to authorities. These challenges can be subsumed under two basic policy questions: the possible implications of hedge funds for the stability of the financial system, and the way in which the public and private sectors can reduce risks associated with the increasing role of hedge funds in the financial system.

Although hedge funds are often associated with negative market events, a balanced assessment should also take into account their beneficial effects on the financial system, for example through their contribution to the price discovery process, market liquidity, market discipline, risk diversification and financial integration. Nevertheless, forming an unambiguous assessment about the systemic impact of hedge funds still remains a challenge, not least because such an assessment is hampered by the lack of high-quality information and the continued opaqueness of the industry. It is important, therefore, that both the industry and authorities continue with their efforts to make further progress in this area and improve their understanding of the

## ARTICLES

Hedge funds: Developments and policy implications



implications of hedge fund developments for the financial system at large.

The main arguments advanced in the debate on a possible direct regulation of hedge funds are in the area of financial stability and investor protection. Although hedge funds and hedgerelated products have fund become increasingly available to retail investors, the extent to which they have already developed into a significant investment alternative for households remains unclear. Efforts to gain a better insight into this issue, such as the recent survey organised by IOSCO, should therefore be welcomed. At the same time, if one comes to the conclusion that hedge funds should be regulated for financial stability reasons, it seems that this can only be done effectively in a strongly coordinated manner at the international level because of the very nature of the business.

The indirect regulation of hedge funds (i.e. the control of risks through banks) has been addressed through various public and private initiatives that pertain mainly to the domain of risk management practices. Moreover, the capital adequacy regime for banks, and in particular the supervisory review process of Basel II, provides an appropriate and flexible framework for addressing risks, also in relation to hedge funds. Thus, it is important that these best practices and recommendations are put in place effectively and are not eroded as a result of competitive pressures.

76 ECB Monthly Bulletin January 2006

EURO AREA STATISTICS

# EURO AREA STATISTICS

ECB Monthly Bulletin January 2006



# **CONTENTS**<sup>1</sup>

	EUR	D AREA OVERVIEW	
	Sum	mary of economic indicators for the euro area	<b>S5</b>
I.		ETARY POLICY STATISTICS	
	1.1	Consolidated financial statement of the Eurosystem	<b>S6</b>
		Key ECB interest rates	<b>S7</b>
		Eurosystem monetary policy operations allotted through tenders	<b>S8</b>
	1.4	Minimum reserve and liquidity statistics	<b>S9</b>
2	MON	EY, BANKING AND INVESTMENT FUNDS	
	2.1	Aggregated balance sheet of euro area MFIs	<b>SIO</b>
	2.2	Consolidated balance sheet of euro area MFIs	<b>S    </b>
		Monetary statistics	<b>SI2</b>
		MFI loans, breakdown	<b>SI4</b>
		Deposits held with MFIs, breakdown	S17
		MFI holdings of securities, breakdown	<b>S20</b>
		Revaluation of selected MFI balance sheet items	<b>S2</b> I
		Currency breakdown of selected MFI balance sheet items	<b>S22</b>
		Aggregated balance sheet of euro area investment funds	<b>S24</b>
	2.10	Assets of euro area investment funds broken down by investment policy and type of investor	<b>\$25</b>
3		NCIAL AND NON-FINANCIAL ACCOUNTS	
		Main financial assets of non-financial sectors	<b>S26</b>
		Main liabilities of non-financial sectors	<b>S27</b>
		Main financial assets and liabilities of insurance corporations and pension funds	<b>S28</b>
	3.4	Annual saving, investment and financing	<b>S29</b>
4	FINA	NCIAL MARKETS	
	4.1	Securities, other than shares, by original maturity, residency of the issuer and currency	<b>S30</b>
	4.2		<b>S3</b> I
	4.3	Growth rates of securities, other than shares, issued by euro area residents	<b>S33</b>
	4.4		<b>S35</b>
		MFI interest rates on euro-denominated deposits and loans by euro area residents	<b>S37</b>
		Money market interest rates	\$39
		Government bond yields	<b>S40</b>
	4.8	Stock market indices	<b>S4</b> I
5		ES, OUTPUT, DEMAND AND LABOUR MARKETS	
	5.1	HICP, other prices and costs	<b>S42</b>
	5.2	Output and demand	S45

- 5.2 Output and demand
- 5.3 Labour markets

1) For further information, please contact us at: statistics@ecb.int. See the ECB's website (www.ecb.int) for longer runs and more detailed data.



**S49** 

6	GOV	ERNMENT FINANCE	
	6.1	Revenue, expenditure and deficit/surplus	\$50
	6.2	Debt	S5 I
	6.3	Change in debt	\$52
	6.4	Quarterly revenue, expenditure and deficit/surplus	\$53
	6.5	Quarterly debt and change in debt	<b>\$54</b>
7	EXT	ERNAL TRANSACTIONS AND POSITIONS	
	7.1	Balance of payments	\$55
	7.2	Monetary presentation of the balance of payments	<b>S60</b>
	7.3	Geographical breakdown of the balance of payments and international investment position	S6 I
	7.4	International investment position (including international reserves)	<b>S63</b>
	7.5	Trade in goods	<b>S65</b>
8	EXC	HANGE RATES	
	8.1	Effective exchange rates	S67
	8.2	Bilateral exchange rates	<b>S68</b>
9	DEV	ELOPMENTS OUTSIDE THE EURO AREA	
	9.1	In other EU Member States	<b>S69</b>
	9.2	In the United States and Japan	<b>S70</b>
LIST O	F CHA	RTS	\$72
TECHN	IICAL	NOTES	\$73
GENER	RAL NO	DTES	\$77

# WHAT'S NEW

The statistics in Sections 2.9 and 2.10, which have not previously included Ireland, will from now on cover all euro area countries.

# **Conventions used in the tables**

۰۰_٬٬	data do not exist/data are not applicable
·· ."	data are not yet available
·· ··	nil or negligible
"billion"	109
(p)	provisional
s.a.	seasonally adjusted
n.s.a.	non-seasonally adjusted





# EURO AREA OVERVIEW

# Summary of economic indicators for the euro area

## 1. Monetary developments and interest rates

	M1 <sup>1)</sup>	M2 <sup>1)</sup>	M3 <sup>1), 2)</sup>	M3 <sup>1),2)</sup> 3-month moving average (centred)	MFI loans to euro area residents excluding MFIs and general government <sup>1)</sup>	Securities other than shares issued in euro by non- financial and non- monetary financial corporations <sup>1)</sup>	3-month interest rate (EURIBOR, % per annum, period averages)	10-year government bond yield (% per annum, period averages)
	1	2	3	4	5	6	7	8
2004	10.0	6.3	5.8	-	6.1	9.9	2.11	4.14
2005				-			2.19	3.44
2005 Q1	9.6	7.1	6.7		7.3	9.0	2.14	3.67
				-				
Q2	9.8	7.5	7.1	-	7.6	13.2	2.12	3.41
Q3	11.3	8.5	8.0	-	8.4	13.2	2.13	3.26
Q2 Q3 Q4				-			2.34	3.42
2005 July	11.1	8.3	7.9	7.9	8.3	12.7	2.12	3.32
Aug.	11.6	8.6	8.2	8.2	8.5	13.1	2.13	3.32
Sep.	11.1	8.9	8.5	8.2	8.7	13.9	2.14	3.16
Oct.	11.1	8.6	8.0	8.0	8.9	13.9	2.14	3.32
				8.0		14.0		
Nov.	10.6	8.3	7.6		9.0	•	2.36	3.53
Dec.		•		•			2.47	3.41

# 2. Prices, output, demand and labour markets

	HICP	Industrial producer prices	Hourly labour costs		Industrial production excluding construction	utilisation in	Employment	Unemployment (% of labour force)
	1	2	3	4	5	6	7	8
2004 2005	2.1	2.3	2.5	2.1	2.0	81.6	0.7	8.9
2005 Q1 Q2 Q3 Q4	2.0 2.0 2.3	4.1 3.9 4.2	3.2 2.5 2.2	1.2 1.2 1.6	0.6 0.7 1.4	81.5 81.0 81.1	0.9 0.8 0.8	8.8 8.6 8.4
2005 July Aug. Sep. Oct. Nov.	2.2 2.2 2.6 2.5 2.3	4.1 4.0 4.4 4.2 4.2			0.6 2.6 1.2 0.2	80.9 - 81.2	- - - -	8.5 8.4 8.3 8.3 8.3
Dec.	2.2		-	-		-	-	

# 3. Balance of payments, reserve assets and exchange rates

(EUR billions, unless otherwise indicated)

	Balar	ce of payments (n	et transactions)		Reserve assets (end-of-period	Effective exchang the euro: EER		USD/EUR exchange rate
	Current and		Direct Portfolio		positions)	(index, 1999 Q1	= 100)	
	capital	Goods	investment	investment				
	accounts					Nominal	Real (CPI)	
	1	2	3	4	5	6	7	8
2004	63.0	106.7	-46.8	71.2	280.7	103.8	105.9	1.2439
2005						103.0	105.2	1.2441
2005 Q1	4.6	15.6	-20.0	3.8	285.0	105.7	107.9	1.3113
Q2	-9.4	18.8	-11.6	106.7	302.3	103.4	105.6	1.2594
Q3 Q4	-1.0	16.7	-94.0	70.2	311.6	101.9	104.2	1.2199
Q4						100.9	103.2	1.1884
2005 July	2.2	9.9	-81.2	75.3	296.3	101.7	103.9	1.2037
Aug.	-0.9	2.1	-15.0	-19.5	295.7	102.3	104.6	1.2292
Sep.	-2.4	4.7	2.2	14.3	311.6	101.8	104.1	1.2256
Oct.	-8.4	2.3	-0.9	-5.1	310.5	101.4	103.7	1.2015
Nov.					322.7	100.7	103.0	1.1786
Dec.				•	•	100.7	103.1	1.1856

Sources: ECB, European Commission (Eurostat and Economic and Financial Affairs DG) and Reuters.

Note: For more information on the data, see the relevant tables later in this section.
1) Annual percentage changes of monthly data refer to the end of the month, whereas those of quarterly and yearly data refer to the annual change in the period average of the series. See the Technical notes for details.

2) M3 and its components exclude holdings by non-euro area residents of money market fund shares/units and debt securities with a maturity of up to two years.

3) For the definition of the trading partner groups and other information, please refer to the General notes.





# MONETARY POLICY STATISTICS

# 1.1 Consolidated financial statement of the Eurosystem (EUR millions)

### 1. Assets

	2005 16 Dec.	2005 23 Dec.	2005 30 Dec.	2006 6 Jan.
Gold and gold receivables	148,298	148,133	163,881	163,804
Claims on non-euro area residents in foreign currency	156,711	152,014	154,141	155,332
Claims on euro area residents in foreign currency	22,606	23,960	23,694	22,780
Claims on non-euro area residents in euro	9,456	9,289	9,185	8,895
Lending to euro area credit institutions in euro	398,516	404,019	405,967	406,239
Main refinancing operations	308,500	314,002	315,001	316,000
Longer-term refinancing operations	90,000	90,017	90,017	90,017
Fine-tuning reverse operations	0	0	0	0
Structural reverse operations	0	0	0	0
Marginal lending facility	0	0	949	211
Credits related to margin calls	16	0	0	11
Other claims on euro area credit institutions in euro	3,203	3,500	3,635	3,735
Securities of euro area residents in euro	93,859	94,435	92,367	92,987
General government debt in euro	40,662	40,274	40,113	40,113
Other assets	143,632	144,125	145,169	144,729
Total assets	1,016,943	1,019,749	1,038,152	1,038,614

## 2. Liabilities

	2005 16 Dec.	2005 23 Dec.	2005 30 Dec.	2006 6 Jan.
Banknotes in circulation	558,596	567,962	565,216	560,253
Liabilities to euro area credit institutions in euro	158,390	150,732	155,535	153,127
Current accounts (covering the minimum reserve system)	158,283	150,714	155,283	153,097
Deposit facility	107	18	252	29
Fixed-term deposits	0	0	0	0
Fine-tuning reverse operations	0	0	0	0
Deposits related to margin calls	0	0	0	1
Other liabilities to euro area credit institutions in euro	219	207	207	207
Debt certificates issued	0	0	0	0
Liabilities to other euro area residents in euro	42,389	43,476	41,767	49,468
Liabilities to non-euro area residents in euro	12,065	12,364	13,224	12,609
Liabilities to euro area residents in foreign currency	410	480	367	402
Liabilities to non-euro area residents in foreign currency	8,441	8,424	8,405	10,133
Counterpart of special drawing rights allocated by the IMF	5,885	5,885	5,920	5,920
Other liabilities	68,373	68,041	70,043	68,969
Revaluation accounts	103,749	103,749	119,113	119,113
Capital and reserves	58,426	58,429	58,355	58,413
Total liabilities	1,016,943	1,019,749	1,038,152	1,038,614

Source: ECB.



With effect from <sup>1)</sup>	Deposit facility Main refinancing operations					Marginal lendi	ng facility
			Fixed rate tenders	Variable rate tenders			
			Fixed rate	Minimum bid rate			
	Level	Change	Level	Level	Change	Level	Change
	1	2	3	4	5	6	7
1999 1 Jan.	2.00	-	3.00	-	-	4.50	-
4 2)	2.75	0.75	3.00	-		3.25	-1.25
22	2.00	-0.75	3.00	-		4.50	1.25
9 Apr.	1.50	-0.50	2.50	-	-0.50	3.50	-1.00
5 Nov.	2.00	0.50	3.00	-	0.50	4.00	0.50
2000 4 Feb.	2.25	0.25	3.25	-	0.25	4.25	0.25
17 Mar.	2.50	0.25	3.50	-	0.25	4.50	0.25
28 Apr.	2.75	0.25	3.75	-	0.25	4.75	0.25
9 June	3.25	0.50	4.25	-	0.50	5.25	0.50
28 <sup>3)</sup>	3.25		-	4.25		5.25	
1 Sep.	3.50	0.25	-	4.50	0.25	5.50	0.25
6 Oct.	3.75	0.25	-	4.75	0.25	5.75	0.25
2001 11 May	3.50	-0.25	-	4.50	-0.25	5.50	-0.25
31 Aug.	3.25	-0.25	-	4.25	-0.25	5.25	-0.25
18 Sep.	2.75	-0.50	-	3.75	-0.50	4.75	-0.50
9 Nov.	2.25	-0.50	-	3.25	-0.50	4.25	-0.50
2002 6 Dec.	1.75	-0.50	-	2.75	-0.50	3.75	-0.50
2003 7 Mar.	1.50	-0.25	-	2.50	-0.25	3.50	-0.25
6 June	1.00	-0.50	-	2.00	-0.50	3.00	-0.50
2005 6 Dec.	1.25	0.25	-	2.25	0.25	3.25	0.25

Source: ECB.

1) From 1 January 1999 to 9 March 2004, the date refers to the deposit and marginal lending facilities. For main refinancing operations, changes in the rate are effective from the first operation following the date indicated. The change on 18 September 2001 was effective on that same day. From 10 March 2004 onwards, the date refers to the deposit and marginal lending facilities. marginal lending facilities and to the main refinancing operations (changes effective from the first main refinancing operation following the Governing Council discussion),

On 22 December 1998 the ECB announced that, starting from the operation to be settled on 28 June 2000, the main refinancing operations of the Eurosystem would be conducted as 2)

3) variable rate tenders. The minimum bid rate refers to the minimum interest rate at which counterparties may place their bids.



# 1.3 Eurosystem monetary policy operations allotted through tenders $^{(1), (2)}$

### 1. Main and longer-term refinancing operations<sup>3)</sup>

Date of settlement	Bids (amount)	Number of participants	Allotment (amount)	•	Variable rate tenders		Running for () days
	(uniounio)	participanto	("""""")	Minimum bid rate	Marginal rate <sup>4)</sup>	Weighted average rate	() uujo
	1	2	3	4	5	6	7
			Main refinar	cing operations			
2005 14 Sep.	435,111	356	279,500	2.00	2.06	2.06	7
21	396,280	402	287,500	2.00	2.06	2.06	7
28	364,417	361	293,500	2.00	2.06	2.07	7
5 Oct.	382,662	348	288,000	2.00	2.06	2.07	7
12	380,299	372	281,000	2.00	2.07	2.07	7
19	383,949	397	296,000	2.00	2.06	2.07	7
26 3 Nov.	371,445 354,563	384 340	301,500 294,000	2.00 2.00	2.06 2.05	2.07 2.06	8
9	366,131	350	294,000	2.00	2.05	2.00	6 7
16	401,859	393	293,500	2.00	2.05	2.06	7
23	403,121	434	311,000	2.00	2.08	2.10	, 7
30	361,548	379	306,500	2.00	2.09	2.11	
6 Dec.	354,476	300	333,500	2.25	2.29	2.31	6 8 7
14	378,799	345	308,500	2.25	2.29	2.30	
21	391,591	393	314,000	2.25	2.30	2.31	8
29	315,797	386	315,000	2.25	2.25	2.42	6
2006 4 Jan.	359,312	353	316,000	2.25	2.30	2.31	7
11	378,353	368	309,000	2.25	2.30	2.31	7
			Longer-term ref	inancing operations			
2005 27 Jan.	58,133	164	30,000	-	2.09	2.10	91
24 Feb.	40,340	145	30,000	-	2.08	2.09	91
31 Mar.	38,462	148	30,000	-	2.09	2.10	91
28 Apr.	47,958	148	30,000	-	2.08	2.09	91
26 May	48,282	140	30,000	-	2.08	2.08	98
30 June	47,181	141	30,000	-	2.06	2.07	91
28 July	46,758	166	30,000	-	2.07	2.08	92
1 Sep. 29	62,563 52,795	153 142	30,000 30,000	-	2.08 2.09	2.09 2.10	91 84
29 28 Oct.	52,795	142	30,000	-	2.09	2.10	84 90
28 Oct. 1 Dec.	52,369	152	30,000		2.17	2.19	90 84
$22^{5}$	89,877	165	12,500		2.40	2.46	98
23 5)	45,003	127	17,500	-	2.44	2.45	97

### 2. Other tender operations

Date of settlement	Type of operation	Bids (amount)	Number of participants	Allotment (amount)	Fixed rate tenders	Vari	able rate tend	ders	Running for () days
	-				Fixed rate	Minimum	Marginal	Weighted	
						bid rate	rate <sup>4)</sup>	average rate	
	1	2	3	4	5	6	7	8	9
2002 4 Jan.	n	57 (44		25.000	5	2.25	3.30	3.32	
	Reverse transaction	57,644	61	25,000	-	3.25			3
10 18 Dec	Reverse transaction	59,377	63	40,000	-	3.25	3.28	3.30	1
18 Dec.	Reverse transaction	28,480	50	10,000	-	2.75	2.80	2.82	6
2003 23 May	Collection of fixed-term deposits	3,850	12	3,850	2.50	-	-	-	3
2004 11 May	Collection of fixed-term deposits	16,200	24	13,000	2.00	-	-	-	1
8 Nov.	Reverse transaction	33,175	42	6,500	-	2.00	2.06	2.07	1
7 Dec.	Collection of fixed-term deposits	18,185	16	15,000	2.00	-	-	-	1
2005 18 Jan.	Reverse transaction	33,065	28	8,000	-	2.00	2.05	2.05	1
7 Feb.	Reverse transaction	17,715	24	2,500	-	2.00	2.05	2.05	1
8 Mar.	Collection of fixed-term deposits	4,300	5	3,500	2.00	-	-	-	1
7 June	Collection of fixed-term deposits	3,708	6	3,708	2.00	-	-	-	1
12 July	Collection of fixed-term deposits	9,605	11	9,605	2.00	-	-	-	1
9 Aug.	Collection of fixed-term deposits	500	1	500	2.00	-	-	-	1
6 Sep.	Reverse transaction	51,060	41	9,500	_	2.00	2.09	2.10	1
11 Oct.	Collection of fixed-term deposits	23,995	22	8,500	2.00	-	-	_	1
5 Dec.	Collection of fixed-term deposits	21,240	18	7,500	2.00	-	-	-	1

Source: ECB.

The amounts shown may differ slightly from those in Section 1.1 due to operations allotted but not settled. 1)

With effect from April 2002, split tender operations, i.e. operations with one-week maturity conducted as standard tenders in parallel with a main refinancing operation, are classified as main refinancing operations. For split tender operations conducted before this month, see Table 2 in Section 1.3. On 8 June 2000 the ECB announced that, starting from the operation to be settled on 28 June 2000, the main refinancing operations of the Eurosystem would be conducted as 2)

3) variable rate tenders. The minimum bid rate refers to the minimum interest rate at which counterparties may place their bids.

4) In liquidity-providing (absorbing) operations, the marginal rate refers to the lowest (highest) rate at which bids were accepted.

An exceptional operation based on longer-term refinancing operation (LTRO) procedures was carried out because an erroneous bid had prevented the ECB from executing its LTRO in the full amount on the previous day. 5)



**1.4** Minimum reserve and liquidity statistics (EUR billions; period averages of daily positions, unless otherwise indicated; interest rates as percentages per annum)

# 1. Reserve base of credit institutions subject to reserve requirements

Reserve base	Total	Liabilities to which a 2% res	erve coefficient is applied	Liabilities to which	cient is applied	
as at ":		Deposits (overnight, up to 2 years' agreed maturity and notice period)	Debt securities up to 2 years' agreed maturity	Deposits (over 2 years' agreed maturity and notice period)	Repos	Debt securities over 2 years' agreed maturity
	1	2	3	4	5	6
2003	11,538.7	6,283.8	412.9	1,459.1	759.5	2,623.5
2004	12,415.9	6,593.7	458.1	1,565.2	913.7	2,885.3
2005 Q1	12,866.9	6,783.2	472.3	1,599.3	1,010.8	3,001.1
Q2	13,328.1	7,021.1	488.2	1,676.0	1,027.9	3,114.9
2005 July	13,431.5	7,064.6	496.2	1,682.8	1,068.0	3,119.9
Aug.	13,388.1	7,003.0	499.9	1,689.2	1,066.0	3,130.1
Sep.	13,562.1	7,125.7	498.5	1,697.7	1,085.4	3,154.9
Oct.	13,712.7	7,184.5	503.4	1,712.0	1,127.0	3,185.8

## 2. Reserve maintenance

Maintenance period ending on:	Required reserves 1	Credit institutions' current accounts	Excess reserves 3	Deficiencies 4	Interest rate on minimum reserves 5
2003	131.8	132.6	0.8	0.0	2.00
2004	137.9	138.5	0.6	0.0	2.05
2005 Q1	140.5	141.3	0.8	0.0	2.05
Q2	144.6	145.5	0.9	0.0	2.05
Q3	149.7	150.2	0.6	0.0	2.05
2005 11 Oct.	150.7	151.4	0.7	0.0	2.05
8 Nov.	149.5	150.2	0.7	0.0	2.06
5 Dec.	152.0	153.0	1.0	0.0	2.07
2006 17 Jan.	153.3				

# 3. Liquidity

Maintenance period ending on:		Liquidity	-providing fact Monetary po		ns of the Euro	system	Liquidi		Credit institutions' current accounts	Base money		
	Eurosystem's net assets in gold and foreign currency	Main refinancing operations	Longer-term refinancing operations	Marginal lending facility	Other liquidity- providing operations	Deposit facility	Other liquidity- absorbing operations	Banknotes in circulation	Central government deposits with the Eurosystem	Other factors (net)		
	1	2	3	4	5	6	7	8	9	10	11	12
2003 2004	320.1 298.0	235.5 265.7	45.0 75.0	0.6 0.1	$\begin{array}{c} 0.0\\ 0.0\end{array}$	0.1 0.1	0.0 0.5	416.1 475.4	57.0 60.2	-4.5 -36.0	132.6 138.5	548.7 614.1
2005 Q1 Q2	280.2 286.8	277.8 273.1	82.2 90.0	0.1 0.1	0.0 0.0	0.1 0.2	0.1 0.1	489.5 512.8	68.5 53.5	-59.2 -62.0	141.3 145.5	630.9 658.5
2005 12 July 9 Aug. 6 Sep. 11 Oct. 8 Nov. 5 Dec.	293.3 305.5 304.8 307.9 315.1 313.2	297.6 309.5 303.5 288.6 293.4 301.3	90.0 90.0 90.0 90.0 90.0 90.0	$\begin{array}{c} 0.1 \\ 0.0 \\ 0.0 \\ 0.1 \\ 0.1 \\ 0.0 \end{array}$	$\begin{array}{c} 0.0 \\ 0.0 \\ 0.3 \\ 0.0 \\ 0.0 \\ 0.0 \end{array}$	0.2 0.3 0.1 0.1 0.1 0.1	0.3 0.0 0.0 0.2 0.0 0.3	522.6 532.6 531.5 531.6 535.6 539.8	67.4 67.4 63.1 47.9 50.4 51.0	-57.3 -45.0 -46.2 -44.6 -37.9 -39.6	147.9 149.8 150.2 151.4 150.2 153.0	670.6 682.7 681.8 683.1 686.0 692.9

Source: ECB. 1) End of period.





# MONEY, BANKING AND INVESTMENT FUNDS

# 2.1 Aggregated balance sheet of euro area MFIs (EUR billions; outstanding amounts at end of period)

# 1. Assets

	Total	Lo	ans to euro a	irea resident	s		igs of secur isued by eu			Money market fund	Holdings of shares/ other equity	External assets	Fixed assets	Remaining assets
		Total	General government	Other euro area residents	MFIs	Total £	General government	Other euro area residents	MFIs	shares/ units <sup>1)</sup>	issued by euro area residents			
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
							Eurosystem							
2003	1,086.8	471.3	22.6	0.6	448.0	133.6	121.5	1.3	10.8	-	12.8	317.9	12.4	138.8
2004	1,197.3	546.5	21.5	0.6	524.3	154.8	140.0	1.7	13.1	-	14.2	294.1	14.0	173.8
2005 Q1	1,274.5	599.9	21.5	0.6	577.8	167.8	151.9	1.6	14.4	-	14.0	301.0	12.5	179.3
Q2	1,353.6	638.4	21.2	0.6	616.6	176.9	158.8	2.0	16.1	-	14.1	319.6	13.3	191.3
2005 July	1,353.4	642.4	21.2	0.6	620.6	176.8	159.6	1.8	15.5	-	14.4	312.8	13.3	193.7
Aug.	1,327.9	610.7	21.2	0.6	588.9	180.6	162.8	1.9	15.9	-	14.4	313.4	13.3	195.5
Sep.	1,351.1	609.8	21.2	0.6	588.0	183.8	165.6	1.9	16.3	-	14.9	328.9	13.3	200.4
Oct.	1,370.1	630.5	21.2	0.6	608.6	184.2	166.4	2.1	15.7	-	14.4	326.9	13.3	200.8
Nov. (p)	1,386.9	630.8	21.2	0.6	609.0	184.3	167.7	2.0	14.6	-	14.4	339.0	13.4	204.9
						MFIs exclu	uding the Eu	irosystem						
2003	19,795.4	12,113.1	817.5	7,101.8	4,193.9	2,944.0	1,242.6	427.7	1,273.6	67.3	894.9	2,567.8	161.8	1,046.4
2004	21,351.8	12,825.3	811.4	7,556.1	4,457.8	3,187.7	1,299.9	465.2	1,422.6	72.5	942.9	2,942.9	159.6	1,220.9
2005 Q1	22.027.0	13.051.3	806.6	7,669.3	4.575.4	3,295.2	1.358.5	481.2	1.455.5	73.1	970.5	3,182.3	156.5	1,298.3
Q2	22,770.4	13,256.3	808.4	7,918.5	4,529.4	3,394.2	1,383.9	506.9	1,503.5	75.1	999.3	3,404.2	163.1	1,478.1
2005 July	22,874.4	13.326.7	808.9	7,976.8	4.541.0	3,396.6	1,378.7	506.5	1,511.5	75.1	999.9	3,444,4	164.3	1,467.6
Aug.	22,826.6	13,305.3	809.9	7,979.7	4,515.6	3,386.5	1,371.1	505.1	1,510.3	80.6	999.5	3,436.8	164.3	1,453.7
Sep.	23,058.6	13,434.5	816.0	8,067.1	4,551.4	3,375.2	1,360.4	504.5	1,510.3	81.4	1,012.6	3,517.6	164.5	1,472.9
Oct.	23,300.4	13,595.1	812.2	8,132.6	4,650.4	3,435.6	1,400.7	520.9	1,514.1	83.7	989.5	3,577.1	165.3	1,454.0
Nov. <sup>(p)</sup>	23,783.0	13,708.3	805.6	8,216.5	4,686.2	3,552.3	1,484.6	542.4	1,525.3	86.9	1,008.8	3,721.9	165.5	1,539.3

## 2. Liabilities

Total Currency Deposits of in	of euro area residents		Money market	Debt securities	Capital and	External liabilities	Remaining liabilities
	entral Other general government/ other euro area residents		fund shares/ units <sup>2)</sup>	issued <sup>3)</sup>	reserves		
1 2 3	4 5	6	7	8	9	10	11
	Eurosystem						
	21.3 16.9		-	1.6	143.8	27.5	139.4
	24.7 15.0		-	0.5	138.4	27.2	167.4
	61.1 17.6		-	0.5	149.9	24.9	171.3
	76.4 18.7	338.5	-	0.6	173.6	24.4	180.5
	68.5 13.8		-	0.6	170.0	23.9	182.0
	56.6 13.4 47.4 15.3	332.6 333.9	-	0.6 0.6	171.6 186.6	25.2 27.4	183.5 189.7
	46.3 17.5	349.7		0.6	180.0	27.4	189.8
	47.9 17.9		-	0.6	194.1	28.1	195.7
	MFIs excluding the Eu	ırosystem					
2003 19,795.4 - 10,774.8 1	34.4 6,275.5	4,364.9	648.8	3,161.4	1,145.0	2,606.4	1,458.9
2004 21,351.8 - 11,487.5 1	37.7 6,640.9	4,709.0	677.4	3,496.9	1,199.5	2,815.0	1,675.6
	26.3 6,706.7	4,820.7	687.6	3,614.8	1,213.5	3,085.6	1,771.7
Q2 22,770.4 - 11,849.5 1	35.1 6,921.1	4,793.3	696.4	3,761.9	1,258.7	3,228.0	1,976.0
	42.3 6,950.5	4,802.7	710.1	3,775.9	1,268.7	3,264.1	1,960.1
	10.2 6,929.8	4,773.9	720.3	3,788.7	1,274.0	3,254.4	1,975.3
	34.6 6,987.7	4,791.3	712.9 712.6	3,807.5	1,271.8	3,350.6	2,002.2 1,990.4
	33.0 7,029.8 36.7 7,062.1	4,900.0 4,947.1	712.6	3,844.7 3,873.8	1,271.1 1,289.4	3,418.9 3,608.7	2,148.8

Source: ECB.

Amounts issued by euro area residents. Amounts issued by non-euro area residents are included in external assets.
 Amounts held by euro area residents.
 Amounts issued with maturity up to two years held by non-euro area residents are included in external liabilities.



# 2.2 Consolidated balance sheet of euro area MFIs (EUR billions; outstanding amounts at end of period; transactions during period)

## 1. Assets

	Total	Loans to	euro area resi	dents		ecurities other y euro area re		Holdings of shares/ other equity	External assets	Fixed assets	Remaining assets
		Total	General government	Other euro area residents	Total	General government	residents	issued by other euro area residents			
	1	2	3	4	5 Outstand	6 ing amounts	7	8	9	10	11_
2003 2004	14,551.8 15,719.1	7,942.6 8,389.6	840.1 832.9	7,102.5 7,556.8	1,793.1 1,906.8	1,364.1 1,439.9	429.0 466.9	623.6 666.2	2,885.7 3,236.9	174.2 173.6	1,132.6 1,345.9
2005 Q1	16,260.4	8,498.1	828.1	7,669.9	1,993.2	1,510.4	482.8	683.9	3,483.3	169.0	1,432.9
Q2	17,039.4	8,748.7	829.6	7,919.2	2,051.5	1,542.7	508.9	713.7	3,723.8	176.4	1,625.2
2005 July	17,112.3	8,807.5	830.0	7,977.5	2,046.5	1,538.3	508.2	706.7	3,757.1	177.5	1,617.0
Aug.	17,091.5	8,811.5	831.1	7,980.4	2,040.9	1,533.9	507.0	705.7	3,750.1	177.6	1,605.7
Sep.	17,317.6	8,904.9	837.2	8,067.7	2,032.4	1,526.0	506.4	726.0	3,846.5	177.8	1,630.1
Oct. Nov. <sup>(p)</sup>	17,452.5 17.888.4	8,966.6 9.044.0	833.4 826.8	8,133.2 8,217.2	2,090.0 2,196.8	1,567.0 1,652.4	523.0 544.4	702.5 708.1	3,904.0 4.061.0	178.7 178.8	1,610.6 1,699.8
1107.	17,000.4	9,044.0	820.8	0,217.2	-	· · · ·	544.4	/00.1	4,001.0	178.8	1,099.0
						sactions					
2003	766.2	384.0	12.1	371.8	170.4	116.3	54.1	19.3	224.8	-3.6	-28.6
2004	1,268.0	499.7	-7.0	506.7	91.9	58.1	33.8	34.6	437.6	2.7	201.5
2005 Q1	448.9	107.9	-5.5	113.4	82.5	66.2	16.4	15.7	187.2	-4.1	59.7
Q2	533.0	207.8	0.9	206.9	39.7	15.8	23.8	25.2	126.4	1.0	133.0
2005 July	88.6	61.9	0.5	61.4	-1.5	-0.7	-0.8	-8.3	44.6	1.1	-9.2
Aug.	-16.3	6.2	1.2	5.0	-7.4	-5.7	-1.8	-1.6	-1.3	0.1	-12.1
Sep.	182.7	91.1	6.2	84.9	-7.2	-6.4	-0.8	11.9	64.1	0.1	22.7
Oct.	135.6	61.6	-3.8 -6.7	65.4 82.7	33.2 61.2	17.1 44.3	16.1 16.9	8.4 1.5	56.1 23.9	0.9 0.2	-24.6 48.8
Nov. <sup>(p)</sup>	211.6	76.1	-0./	82.7	01.2	44.5	16.9	1.5	23.9	0.2	48.8

## 2. Liabilities

	Total	Currency in circulation	Deposits of central government	Deposits of other general government/ other euro area residents 4	Money market fund shares/ units <sup>1)</sup>	Debt securities issued <sup>2)</sup>	Capital and reserves 7	External liabilities	Remaining liabilities 9	Excess of inter- MFI liabilities
	-			0	utstanding amour					
2003	14,551.8	397.9	155.7	6,292.3	581.5	1,878.5	1,004.7	2,634.0	1,598.3	8.9
2004	15,719.1	468.4	162.4	6,655.9	604.9	2,061.7	1,047.0	2,842.2	1,842.9	33.6
2005 Q1	16,260.4	471.8	187.4	6,724.4	614.5	2,145.5	1,062.9	3,110.5	1,943.0	0.4
Q2	17,039.4	496.6	211.5	6,939.8	621.3	2,242.9	1,132.6	3,252.4	2,156.5	-14.2
2005 July	17,112.3	506.4	210.8	6,964.3	635.1	2,249.4	1,131.1	3,288.0	2,142.1	-14.9
Aug.	17,091.5	500.9	166.8	6,943.2	639.7	2,263.1	1,137.5	3,279.6	2,158.8	1.9
Sep.	17,317.6	507.1	181.9	7,002.9	631.4	2,281.5	1,157.0	3,378.0	2,192.0	-14.2
Oct.	17,452.5	510.5	179.3	7,047.3	628.9	2,315.5	1,153.9	3,446.2	2,180.2	-9.3
Nov. <sup>(p)</sup>	17,888.4	514.5	184.6	7,080.0	629.5	2,334.5	1,168.4	3,636.8	2,344.5	-4.3
					Transactions					
2003	766.2	79.0	15.1	313.7	56.7	133.5	38.6	130.8	-60.8	59.8
2004	1,268.0	70.5	6.1	377.4	22.3	197.1	50.5	276.8	229.4	37.7
2005 Q1	448.9	3.3	25.0	57.8	9.7	65.3	13.1	212.0	107.1	-44.4
Q2	533.0	24.8	24.1	175.3	6.7	80.3	24.6	61.4	169.0	-33.2
2005 July	88.6	9.9	-0.7	28.2	12.9	6.8	2.9	43.9	-16.9	1.7
Aug.	-16.3	-5.5	-44.3	-19.4	4.1	14.8	4.3	-3.0	15.4	17.4
Sep.	182.7	6.2	15.2	57.6	-7.0	13.4	5.6	83.6	27.1	-19.1
Oct.	135.6	3.4	-2.6	44.1	-6.0	33.0	2.9	65.6	-11.3	6.4
Nov. <sup>(p)</sup>	211.6	4.0	5.2	30.3	-9.7	7.3	4.5	67.7	102.3	0.1

Source: ECB.

Amounts held by euro area residents.
 Amounts issued with maturity up to two years held by non-euro area residents are included in external liabilities.



# 1. Monetary aggregates 1) and counterparts

	M1	M2-M1	M2	M3-M2	M3	M3 3-month moving average (centred)	Longer-term financial liabilities	Credit to general government	Credit to o euro area re:		Net external assets <sup>2)</sup>
	1	2	3	4	5	6	7	8	9	10	11
				(	Outstanding a	imounts					
2003	2,680.6	2,553.3	5,233.9	907.2	6,141.1	-	4,133.3	2,226.1	8,149.6	7,093.4	230.6
2004	2,912.7	2,661.0	5,573.6	960.6	6,534.2		4,454.5	2,297.0	8,681.9	7,545.2	375.8
2005 Q1	3,007.2	2,675.4	5,682.7	944.9	6,627.5	-	4,579.5	2,328.5	8,830.0	7,673.6	387.5
Q2	3,258.0	2,557.9	5,815.9	981.1	6,797.0		4,797.2	2,353.5	9,114.6	7,893.9	469.8
2005 July	3,302.6	2,572.2	5,874.8	994.6	6,869.3		4,801.4	2,364.5	9,185.5	7,960.6	472.5
Aug.	3,329.9	2,584.9	5,914.8	1,005.2	6,920.0		4,827.8	2,375.9	9,240.0	8,010.5	466.6
Sep.	3,349.4	2,629.4	5,978.8	1,002.0	6,980.8		4,859.5	2,370.1	9,333.0	8,083.1	457.5
Oct.	3,369.7	2,635.6	6,005.3	990.1	6,995.3		4,896.9	2,407.8	9,389.5	8,147.6	437.3
Nov. <sup>(p)</sup>	3,385.6	2,639.5	6,025.1	1,002.1	7,027.2		4,938.0	2,478.1	9,474.8	8,212.4	403.7
					Transacti	ons					
2003	259.4	113.4	372.9	32.2	405.1	-	236.1	131.9	442.9	370.1	96.0
2004	240.4	111.8	352.2	56.0	408.3		340.8	53.5	572.5	504.2	163.0
2005 Q1	91.8	8.2	99.9	-20.5	79.5	-	107.1	26.4	146.8	128.6	8.9
Q2	86.3	38.7	125.0	36.5	161.5		126.0	8.2	235.6	178.1	48.6
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	45.1 28.4 18.8 20.7 14.9	14.7 13.1 43.5 5.4 3.1	59.8 41.5 62.2 26.2 18.0	12.5 10.1 -1.2 -15.8 -5.3	72.3 51.7 61.1 10.3 12.6	- - - -	11.8 25.5 11.8 42.9 25.9	14.7 10.1 -4.2 13.7 29.3	72.6 55.4 81.8 88.6 75.3	69.8 52.0 70.2 64.4 63.6	5.6 -5.6 -26.6 -19.0 -43.6
					Growth r	ates					
2003 Dec.	10.6	4.6	7.6	3.8	7.1	7.0	6.0	6.3	5.7	5.5	96.0
2004 Dec.	9.0	4.4	6.7	6.2	6.6	6.5	8.2	2.4	7.0	7.1	163.0
2005 Mar.	9.3	4.8	7.1	2.8	6.5	6.6	8.7	2.5	7.4	7.5	98.2
June	10.9	5.1	8.1	5.1	7.6	7.6	9.6	1.2	8.2	8.1	160.3
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	11.1 11.6 11.1 11.2 10.6	5.3 5.3 6.5 6.0 5.9	8.3 8.6 8.9 8.6 8.3	5.5 5.8 6.2 4.0 3.5	7.9 8.2 8.5 8.0 7.6	7.9 8.2 8.2 8.0	9.2 9.2 8.5 8.8 8.6	1.2 1.2 1.4 1.9 3.4	8.4 8.7 9.0 9.3 9.4	8.3 8.5 8.7 8.9 9.0	162.8 134.4 81.1 50.6 -2.1

C1 Monetary aggregates

# C2 Counterparts (annual growth rates; seas



Source: ECB.

1) Monetary liabilities of MFIs and central government (post office, treasury) vis-à-vis non-MFI euro area residents excluding central government (M1, M2, M3: see glossary). 2)

Values in the section "growth rates" are sums of the transactions during the 12 months ending in the period indicated.



2.3 Monetary statistics (EUR billions and annual growth rates; seasonally adjusted; outstanding amounts and growth rates at end of period, transactions during period)

# 2. Components of monetary aggregates and longer-term financial liabilities

-	•	, 00 0	L. L.	,							
	Currency in circulation	Overnight deposits	Deposits with agreed maturity up to 2 years	Deposits redeemable at notice up to 3 months	Repos	Money market fund shares/units	Debt securities up to 2 years	Debt securities over 2 years	Deposits redeemable at notice over 3 months	Deposits with agreed maturity over 2 years	Capital and reserves
	1	2	3	4	5	6	7	8	9	10	11
					Outstanding a	mounts					
2003	386.9	2,293.7	1,031.0	1,522.3	218.4	596.1	92.7	1,789.6	90.7	1,250.0	1,003.1
2004	452.7	2,460.0	1,026.5	1,634.5	239.4	618.9	102.3	1,962.6	89.6	1,357.3	1,044.9
2005 Q1	477.6	2,529.7	1,019.3	1,656.1	225.8	613.1	106.1	2,038.7	90.9	1,387.1	1,062.8
Q2	493.7	2,764.3	1,039.2	1,518.7	239.7	622.9	118.5	2,122.6	91.6	1,449.2	1,133.8
2005 July	494.7	2,807.9	1,045.1	1,527.0	243.0	632.3	119.2	2,126.1	90.8	1,453.9	1,130.6
Aug.	501.5	2,828.4	1,047.8	1,537.1	252.6	631.8	120.7	2,146.2	89.8	1,455.4	1,136.4
Sep.	507.4	2,842.0	1,083.5	1,545.9	244.9	636.6	120.5	2,155.2	87.8	1,465.4	1,151.1
Oct.	514.9	2,854.8	1,086.9	1,548.7	237.2	631.2	121.6	2,186.0	87.0	1,473.9	1,150.0
Nov. <sup>(p)</sup>	519.6	2,866.0	1,090.0	1,549.5	237.8	633.5	130.8	2,196.3	86.7	1,484.5	1,170.5
					Transacti	ons					
2003	77.5 65.8	181.9	-29.7	143.1	-10.3	57.6	-15.1	149.2	-13.2	61.9	38.1
2004		174.7	-0.8	112.6	23.1	21.7	11.2	185.2	-1.1	106.6	50.1
2005 Q1	24.9	66.9	-13.2	21.4	-13.7	-5.7	-1.0	62.4	0.8	28.9	15.0
Q2	16.2	70.2	14.3	24.4	13.7	9.7	13.0	66.2	0.0	34.0	25.8
2005 July	1.0	44.1	6.3	8.4	3.3	8.5	0.7	3.8	-0.8	7.6	$     \begin{array}{r}       1.2 \\       3.7 \\       0.8 \\       4.8 \\       10.5     \end{array} $
Aug.	6.8	21.6	3.0	10.1	9.6	-1.1	1.6	21.2	-1.0	1.6	
Sep.	5.9	12.8	34.7	8.8	-7.8	6.2	0.5	3.2	-2.0	9.8	
Oct.	7.5	13.2	2.6	2.8	-7.6	-8.8	0.6	30.3	-0.7	8.4	
Nov. <sup>(p)</sup>	4.7	10.2	2.3	0.7	0.5	-8.0	2.2	5.7	-0.3	10.1	
					Growth ra	ates					
2003 Dec.	24.9	8.6	-2.8	10.4	-4.6	11.0	-14.9	8.9	-12.7	5.2	3.9
2004 Dec.	17.0	7.6	-0.1	7.4	10.7	3.6	12.3	10.3	-1.2	8.5	5.0
2005 Mar.	17.8	7.8	1.7	6.8	4.0	1.8	6.5	10.6	0.5	9.1	5.4
June	17.2	9.8	3.4	6.1	10.3	2.3	10.5	11.7	1.4	9.4	6.7
2005 July	16.0	10.3	4.0	6.0	6.7	3.7	13.5	11.2	0.3	9.3	6.4
Aug.	15.6	10.9	4.3	6.0	11.0	2.6	13.5	11.3	-0.9	8.8	6.5
Sep.	15.3	10.4	7.1	5.9	8.8	3.7	15.0	10.1	-2.3	8.9	5.9
Oct.	15.3	10.4	6.1	5.7	5.0	1.4	17.6	10.9	-3.8	8.7	6.0
Nov. <sup>(p)</sup>	14.6	9.9	6.7	5.2	6.6	0.5	14.9	10.8	-4.5	8.3	5.8
										11 11171	

# C3 Components of monetary aggregates



debt securities over 2 years . . . . deposits with agreed maturity over 2 years



Source: ECB.



2.4 MFI loans, breakdown <sup>1</sup>) (EUR billions and annual growth rates; outstanding amounts and growth rates at end of period, transactions during period)

# 1. Loans to financial intermediaries and non-financial corporations

		corporations ion funds	Other f			Non-financia	l corporations	
	Total	Up to 1 year	Total	Up to 1 year	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years
	1	2	3	4	5	6	7	8
			Οι	itstanding amounts				
2003	35.4	22.1	511.4	325.0	3,034.3	961.5	524.1	1,548.8
2004	48.6	31.4	546.3	334.4	3,152.7	973.8	547.8	1,631.2
2005 Q1	58.2	39.7	561.3	351.1	3,189.5	983.8	554.7	1,650.9
Q2	63.8	43.9	581.5	362.9	3,281.9	1,025.1	564.1	1,692.7
2005 July	63.7	43.3	579.5	357.1	3,308.7	1,024.9	573.1	1,710.7
Aug.	64.4	42.7	572.6	349.2	3,294.3	1,001.1	571.0	1,722.3
Sep.	65.2	42.6	601.7	373.1	3,320.8	1,014.4	572.6	1,733.8
Oct.	68.9	45.3	601.8	368.9	3,341.8	1,022.1	582.6	1,737.1
Nov. <sup>(p)</sup>	75.4	50.4	616.9	377.2	3,373.2	1,031.6	586.2	1,755.5
				Transactions				
2003	4.2	2.2	53.4	26.2	102.6	-7.9	15.9	94.7
2004	13.1	9.1	52.1	27.7	162.9	23.2	31.2	108.5
2005 Q1	8.6	7.9	11.3	14.9	37.2	8.0	7.7	21.5
Q2	5.5	4.2	16.4	9.5	82.4	35.0	9.8	37.6
2005 July	-0.1	-0.6	-1.4	-5.5	27.7	0.3	9.1	18.3
Aug.	0.7	-0.5	-7.1	-7.8	-12.4	-22.8	-2.0	12.4
Sep.	0.7	-0.2	28.4	23.5	24.8	12.3	1.4	11.1
Oct.	3.7	2.7	-2.5	-5.9	23.0	8.8	10.2	3.9
Nov. <sup>(p)</sup>	6.4	5.1	13.0	7.7	31.7	9.7	4.1	17.9
				Growth rates				
2003 Dec.	11.8	11.6	11.6	8.8	3.5	-0.8	3.1	6.5
2004 Dec.	36.9	41.5	10.5	9.1	5.4	2.4	6.0	7.0
2005 Mar.	23.7	21.8	12.6	17.5	5.9	3.8	6.6	6.9
June	17.5	9.8	14.0	18.8	6.5	5.6	6.1	7.2
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	11.3 18.3 22.7 24.0 37.3	3.1 9.1 12.6 15.0 32.1	15.0 16.4 17.5 15.6 12.9	19.1 21.4 23.0 19.0 12.6	7.1 6.9 7.2 7.3 7.5	6.0 5.5 5.9 5.9 5.9 5.2	7.0 5.8 5.9 7.4 7.7	7.7 8.0 8.5 8.1 8.8

# C5 Loans to financial intermediaries and non-financial corporations



Source: ECB.

MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
 This category includes investment funds.



2.4 MFI loans, breakdown <sup>1</sup>) (EUR billions and annual growth rates; outstanding amounts and growth rates at end of period, transactions during period)

## 2. Loans to households<sup>2)</sup>

	Total		Consume	er credit		Le	ending for h	ouse purchas	e		Other l	ending	
		Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years
	1	2	3	4	5	6	7	8	9	10	11	12	13
					0	utstanding ar	nounts						
2003	3,520.6	484.5	112.0	181.0	191.5	2,360.5	14.4	63.3	2,282.7	675.6	145.0	95.5	435.1
2004	3,808.4	515.4	120.3	189.6	205.6	2,591.5	14.6	65.8	2,511.1	701.5	144.1	99.2	458.2
2005 Q1	3,860.4	519.3	120.2	191.1	208.0	2,640.3	14.3	67.1	2,558.9	700.9	144.4	98.7	457.8
Q2	3,991.3	537.3	124.4	197.3	215.6	2,737.2	14.8	66.2	2,656.1	716.8	149.9	101.0	465.9
2005 July	4,024.9	539.1	122.3	199.1	217.7	2,772.9	14.8	66.8	2,691.2	712.9	145.2	101.2	466.5
Aug.	4,048.3	540.6	123.2	198.8	218.6	2,795.1	14.6	67.2	2,713.3	712.6	143.6	101.3	467.8
Sep.	4,079.4	544.3	126.1	198.6	219.6	2,820.2	15.0	67.6	2,737.5	714.9	145.7	101.1	468.0
Oct.	4,120.1	548.3	127.2	199.6	221.5	2,851.1	14.7	67.9	2,768.4	720.7	145.7	101.6	473.5
Nov. <sup>(p)</sup>	4,151.1	549.7	126.7	200.3	222.7	2,877.4	14.8	68.4	2,794.2	724.0	148.2	102.2	473.7
						Transactio	ons						
2003	211.6	13.3	8.4	6.3	-1.4	177.3	-5.9	1.7	181.4	21.0	-6.3	-4.9	32.2
2004	278.6	29.0	7.1	8.6	13.3	236.9	0.9	2.9	233.1	12.7	-0.8	2.0	11.6
2005 Q1	56.2	4.4	-0.3	1.6	3.1	49.5	-0.2	1.3	48.4	2.2	1.3	-0.4	1.3
Q2	102.5	15.7	4.0	6.0	5.7	75.6	0.5	0.5	74.5	11.3	5.6	0.5	5.1
2005 July	35.2	2.4	-1.9	1.9	2.3	35.9	0.0	0.5	35.3	-3.1	-4.5	0.2	1.2
Aug.	23.8	2.1	1.0	-0.3	1.3	22.1	-0.2	0.4	21.9	-0.3	-1.5	0.1	1.1
Sep.	31.1	3.9	2.7	0.1	1.1	25.4	0.4	0.5	24.5	1.7	1.8	-0.2	0.1
Oct.	41.2	4.4	1.4	1.1	1.9	31.0	-0.2	0.3	31.0	5.8	-0.1	0.5	5.4
Nov. <sup>(p)</sup>	31.6	2.7	-0.5	0.7	2.5	26.2	0.1	0.5	25.7	2.7	2.6	0.6	-0.6
						Growth ra	tes						
2003 Dec.	6.4	2.8	8.1	3.6	-0.2	8.1	-26.3	2.6	8.6	3.3	-4.1	-5.0	8.5
2004 Dec.	7.9	6.0	6.3	4.8	6.9	10.0	6.0	4.6	10.2	1.9	-0.5	2.0	2.6
2005 Mar.	8.0	6.4	7.7	4.6	7.5	10.1	5.0	8.1	10.1	2.0	2.0	1.1	2.2
June	8.4	6.7	6.6	5.8	7.5	10.4	4.1	4.6	10.6	2.4	3.8	0.6	2.3
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	8.5 8.6 9.0 9.2	6.7 6.9 7.2 7.7 8.0	6.1 6.4 8.0 9.4 9.2	6.0 6.1 6.0 6.2 6.4	7.8 8.0 7.8 8.1 8.9	10.7 10.8 10.6 10.9 11.2	5.4 4.3 5.7 0.9 4.1	4.7 4.7 4.9 4.8 6.5	10.9 11.0 10.8 11.2 11.3	2.1 2.2 2.3 3.1 3.0	3.1 3.1 2.7 3.2 2.8	0.9 1.2 1.1 0.7 1.5	2.0 2.1 2.4 3.6 3.4

C6 Loans to households (annual growth rates)



Source: ECB.

MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
 Including non-profit institutions serving households.



# 2.4 MFI Ioans, breakdown <sup>1)</sup>

# 3. Loans to government and non-euro area residents

		G	eneral governme	nt			Non-	euro area reside	ents	
	Total	Central government	Other	general governm	ent	Total	Banks <sup>2)</sup>		Non-banks	
		g	State government	Local government	Social security funds			Total	General government	Other
	1	2	3	4	5	6	7	8	9	10
				Outstar	nding amounts					
2003 2004	817.5 811.4	128.4 129.6	265.1 252.3	388.9 405.7	35.0 23.8	1,757.9 1,974.7	1,182.2 1,342.2	575.7 632.5	59.3 61.3	516.4 571.1
2005 Q1 Q2 Q3 <sup>(p)</sup>	806.6 808.4 816.0	129.3 125.0 124.8	248.1 247.5 247.1	406.6 407.3 414.5	22.5 28.6 29.6	2,136.5 2,292.5 2,375.7	1,463.8 1,582.4 1,638.7	672.7 710.1 736.9	62.0 62.1 64.2	610.7 648.0 672.7
				Tra	ansactions	,				
2003 2004	13.7 -5.9	-5.9 2.0	-12.2 -13.9	16.6 17.3	15.3 -11.2	159.4 275.6	109.2 194.9	50.1 80.4	-5.0 1.8	55.0 78.6
2005 Q1 Q2 Q3 <sup>(p)</sup>	-5.5 1.2 7.8	-0.5 -4.7 -0.1	-4.2 -0.8 -0.5	0.5 0.6 7.3	-1.3 6.0 1.1	124.8 93.9 85.2	98.6 81.1 57.5	26.2 12.9 27.9	0.6 0.2 2.1	25.5 12.7 25.7
				Gre	owth rates					
2003 Dec. 2004 Dec.	1.7 -0.7	-4.4 1.5	-4.4 -5.2	4.4 4.4	77.5 -32.1	9.3 15.6	9.6 16.4	8.8 13.9	-7.7 3.1	11.0 15.2
2005 Mar. June Sep. <sup>(p)</sup>	-1.8 -0.8 0.8	-1.6 -0.4 1.5	-5.5 -2.6 -2.4	4.6 4.0 5.0	-42.1 -34.7 -23.5	12.4 17.0 19.9	14.6 19.9 23.4	8.0 11.0 12.8	1.1 2.1 5.5	8.8 12.0 13.6

# C7 Loans to government and non-euro area residents



Source: ECB.
MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
The term "banks" is used in this table to indicate institutions of a similar type to MFIs resident outside the euro area.



# 2.5 Deposits held with MFIs, breakdown <sup>1)</sup> (EUR billions and annual growth rates; outstanding amounts and growth rates at end of period, transactions during period)

# 1. Deposits by financial intermediaries

		Insu	rance corpoi	rations and	d pension fu	inds				Other finan	cial intern	nediaries <sup>2)</sup>		
	Total	Overnight	With agreed	maturity	Redeemab	e at notice	Repos	Total	Overnight	With agree	d maturity	Redeemable	e at notice	Repos
			Up to 2 years	Over 2 years	Up to 3 months	Over 3 months				Up to 2 years	Over 2 years	Up to 3 months	Over 3 months	
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
						Outstand	ing amounts							
2003 2004	542.4 583.2	58.9 59.2	41.7 51.4	420.5 449.4	1.3 1.2	0.8 1.3	19.1 20.8	565.6 636.6	180.9 180.3	130.8 139.0	143.3 187.3	6.1 10.1	0.1 0.1	104.4 119.8
2005 Q1 Q2	597.0 595.7	65.7 61.2	48.5 48.3	460.3 463.0	1.3 1.1	1.3 1.6	19.8 20.5	692.9 792.2	213.3 226.7	134.2 149.1	205.2 264.3	11.5 11.1	0.1 0.1	128.7 140.7
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	603.7 604.1 602.9 609.9 605.3	64.3 59.1 60.0 66.7 67.9	51.9 50.8 50.8 48.0 42.0	464.8 466.1 466.9 468.5 469.7	1.1 1.1 1.1 1.1 1.2	1.6 1.6 1.5 1.5	20.1 25.4 22.4 24.1 23.1	799.4 797.4 834.4 847.4 853.8	223.5 217.2 242.7 232.4 227.1	155.8 157.5 169.9 176.7 180.3	267.7 268.1 276.8 285.5 298.3	11.6 11.4 10.6 11.0 11.1	$\begin{array}{c} 0.1 \\ 0.1 \\ 0.1 \\ 0.1 \\ 0.1 \\ 0.1 \end{array}$	140.5 143.2 134.4 141.7 136.9
						Tran	sactions							
2003 2004	19.0 39.9	1.6 0.7	3.9 10.3	11.8 27.7	0.3 -0.1	0.4 -0.1	1.1 1.5	82.8 72.2	25.3 0.9	-0.2 5.8	38.5 43.7	3.2 4.1	0.1 0.0	16.0 17.7
2005 Q1 Q2	12.5 -2.1	6.4 -5.3	-3.1 -0.5	10.0 2.7	0.1 0.2	0.0 0.0	-1.0 0.8	48.8 66.3	32.1 10.7	-9.4 11.9	16.0 31.0	1.3 0.8	$\begin{array}{c} 0.0 \\ 0.0 \end{array}$	8.7 11.9
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	8.1 0.5 -1.4 6.9 -4.7	3.2 -5.2 0.9 6.7 1.1	3.6 -1.0 -0.1 -2.9 -6.1	1.8 1.3 0.7 1.5 1.2	-0.1 0.0 0.0 0.0 0.0	0.0 0.1 0.0 0.0 0.0	-0.4 5.3 -2.9 1.6 -0.9	10.4 -2.1 36.4 8.8 5.8	-3.0 -6.3 25.2 -10.3 -5.7	6.8 1.5 12.2 6.6 3.7	6.2 0.4 8.6 4.9 12.4	0.5 -0.3 -0.8 0.4 0.2	0.0 0.0 0.0 0.0 0.0	-0.2 2.7 -8.8 7.2 -4.8
						Grov	wth rates							
2003 Dec. 2004 Dec.	3.6 7.4	2.8 1.2	9.9 24.6	2.9 6.6	41.3 -8.0	58.8 -43.1	6.0 7.9	17.0 12.7	16.3 0.5	-0.2 4.3	36.4 30.4	70.4 67.6	-	17.1 17.1
2005 Mar. June	6.8 5.1	2.3 1.8	16.4 15.3	7.5 4.8	1.7 18.1	-51.5 31.3	-10.4 -3.0	17.6 26.8	8.5 16.5	9.3 15.1	40.1 52.5	50.0 50.1	-	11.4 21.0
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	6.2 6.3 4.8 5.0 3.2	15.1 10.0 -2.7 13.3 6.0	13.2 9.1 7.8 -2.1 -12.3	4.9 5.1 5.1 4.6 4.1	13.7 11.3 26.3 22.3 18.2	32.1 32.6 33.0 2.8 2.9	-8.3 13.8 12.0 5.5 10.9	28.6 29.6 33.6 29.7 28.0	19.8 23.1 27.3 22.7 13.8	19.6 16.7 33.5 25.2 34.0	51.8 48.4 50.1 51.0 47.9	56.2 55.3 46.3 45.9 27.1	- - - -	18.6 25.2 18.8 14.6 13.8



Source: ECB.MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.This category includes investment funds.



# 2.5 Deposits held with MFIs, breakdown $^{1)}$

# 2. Deposits by non-financial corporations and households

			Non-finar	icial corpo	orations					н	ouseholds <sup>2</sup>	i -		
	Total	Overnight	With agreed	maturity			Repos	Total	Overnight	With agree	d maturity	Redeemabl	e at notice	Repos
			Up to 2 years	Over 2 years	Up to 3 months	Over 3 months				Up to 2 years	Over 2 years	Up to 3 months	Over 3 months	
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
						Outstand	ing amounts	8						
2003	1,050.1	633.3	280.2	67.6	38.1	1.0	30.0	3,978.6	1,311.8	544.0	600.8	1,379.2	89.9	52.9
2004	1,114.6	674.7	291.1	73.8	44.2	1.1	29.7	4,162.0	1,403.1	515.0	634.3	1,466.1	88.0	55.6
2005 Q1	1,096.3	675.3	280.2	72.1	44.0	1.1	23.6	4,177.0	1,409.1	513.2	632.8	1,481.9	88.7	51.2
Q2	1,133.5	722.2	274.1	71.9	41.6	1.5	22.1	4,246.6	1,629.9	511.2	630.8	1,336.3	87.2	51.2
2005 July	1,134.0	719.8	278.9	70.8	42.2	1.4	21.1	4,265.4	1,643.0	512.8	630.0	1,339.5	86.4	53.6
Aug.	1,138.6	715.3	287.2	70.7	41.8	1.3	22.3	4,238.4	1,610.4	515.6	630.2	1,342.3	85.5	54.4
Sep.	1,151.8	719.1	296.5	68.7	43.9	1.2	22.4	4,246.3	1,626.9	515.0	627.1	1,341.9	83.9	51.6
Oct.	1,168.9	734.0	302.8	65.6	44.5	1.2	20.8	4,247.2	1,629.9	517.3	625.6	1,339.8	83.4	51.2
Nov. <sup>(p)</sup>	1,181.2	746.3	299.0	66.6	44.8	1.2	23.3	4,260.4	1,642.4	520.8	623.7	1,337.4	83.7	52.5
						Tran	sactions							
2003	70.4	40.9	20.3	3.3	10.2	0.0	-4.2	141.9	95.3	-45.4	10.0	117.4	-13.7	-21.8
2004	80.8	48.5	17.1	6.7	8.0	0.7	-0.2	178.1	90.5	-29.6	31.1	85.2	-1.9	2.7
2005 Q1	-20.0	-0.5	-12.1	-1.1	-0.2	0.0	-6.1	14.8	5.4	-2.0	-0.1	15.8	0.2	-4.5
Q2	33.3	41.3	-7.7	0.2	1.1	-0.1	-1.5	67.1	63.4	-3.6	-2.3	11.1	-1.4	0.0
2005 July	0.9	-2.3	4.9	-1.1	0.6	-0.2	-1.0	19.0	13.2	1.8	-0.8	3.2	-0.8	2.5
Aug.	6.1	-3.4	8.7	0.0	-0.4	-0.1	1.3	-26.7	-32.6	2.8	0.2	2.9	-1.0	0.8
Sep.	12.4	3.4	8.9	-2.0	2.0	0.0	0.1	7.2	16.3	-0.9	-3.2	-0.5	-1.5	-2.9
Oct.	21.0	15.0	6.3	0.8	0.6	0.0	-1.7	0.9	3.5	1.9	-1.5	-2.1	-0.5	-0.3
Nov. <sup>(p)</sup>	11.2	11.9	-4.3	0.9	0.3	0.0	2.5	12.7	12.4	3.2	-2.0	-2.5	0.2	1.3
						Grov	vth rates							
2003 Dec.	7.2	6.7	7.7	5.3	41.5	-3.9	-12.4	3.7	7.9	-7.7	1.7	9.3	-13.2	-29.2
2004 Dec.	7.8	7.7	6.2	9.9	21.2	72.2	-0.8	4.5	6.9	-5.4	5.2	6.2	-2.1	5.2
2005 Mar.	7.4	9.3	3.7	4.4	15.2	68.0	-8.3	4.4	6.6	-2.7	3.8	5.6	0.1	-1.3
June	8.1	10.6	4.4	3.3	14.9	-5.8	-13.4	4.6	7.8	-1.3	2.9	4.8	1.1	1.5
2005 July	8.3	11.7	$\begin{array}{c} 4.1 \\ 7.0 \\ 10.6 \\ 7.1 \\ 6.8 \end{array}$	1.1	14.9	-17.5	-20.5	4.8	8.3	-0.9	2.6	4.7	0.1	2.0
Aug.	8.8	11.7		1.0	12.5	-24.5	-22.5	4.5	7.8	0.0	2.3	4.7	-0.9	-0.3
Sep.	8.4	9.0		-1.7	16.5	-26.5	-12.6	4.5	7.9	0.4	1.8	4.6	-2.4	-4.0
Oct.	9.7	12.6		0.4	18.2	-28.2	-19.9	4.0	7.1	1.1	1.1	4.2	-3.5	-5.8
Nov. <sup>(p)</sup>	9.7	12.2		1.5	16.2	-29.8	-8.9	4.1	7.5	1.7	0.7	3.8	-4.1	-3.5

# C9 Deposits by non-financial corporations and households (annual growth rates)



### Source: ECB.

MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
 Including non-profit institutions serving households.



2.5 Deposits held with MFIs, breakdown <sup>1)</sup> (EUR billions and annual growth rates; outstanding amounts and growth rates at end of period, transactions during period)

# 3. Deposits by government and non-euro area residents

		Ge	neral governme	nt			Non-	euro area reside	nts	
	Total	Central government	Other	general governr	nent	Total	Banks <sup>2)</sup>		Non-banks	
		<u> </u>	State government	Local government	Social security funds			Total	General government	Other
	1	2	3	4	5	6	7	8	9	10
				Out	standing amount	s				
2003 2004	273.3 282.2	134.4 137.7	31.1 30.5	66.9 69.6	40.9 44.3	2,245.1 2,428.9	1,580.8 1,748.0	664.3 680.9	96.1 103.4	568.2 577.5
2005 Q1 Q2 Q3 <sup>(p)</sup>	269.9 288.3 286.8	126.3 135.1 134.6	33.4 35.1 35.9	67.5 69.7 71.4	42.7 48.4 44.9	2,669.1 2,784.9 2,904.9	1,935.7 2,034.1 2,109.8	733.4 750.8 795.1	105.4 118.6 124.3	628.0 632.3 670.8
					Transactions					
2003 2004	21.5 11.0	23.3 2.7	-0.5 1.8	-2.3 2.8	1.0 3.8	138.6 247.1	117.5 214.8	21.1 32.0	-1.3 6.9	22.4 25.1
2005 Q1 Q2 Q3 <sup>(p)</sup>	-12.2 18.3 -1.6	-11.4 8.8 -0.8	2.8 1.7 0.9	-2.1 2.2 1.8	-1.6 5.7 -3.5	188.2 42.2 120.3	147.1 42.7 76.3	41.0 -0.5 44.0	2.0 13.2 5.7	39.1 -13.7 38.3
					Growth rates					
2003 Dec. 2004 Dec.	8.6 4.0	21.3 2.0	-1.5 5.6	-3.4 4.1	2.6 9.2	6.2 11.0	7.6 13.5	3.0 4.8	-1.3 7.2	3.7 4.4
2005 Mar. June Sep. <sup>(p)</sup>	-0.3 -1.2 -0.1	-10.5 -13.8 -8.3	19.6 19.3 14.0	8.3 8.3 8.1	8.6 16.9 5.2	11.8 12.6 17.1	13.7 13.5 18.0	7.0 10.1 14.6	4.1 15.8 17.8	7.5 9.0 14.0

# C10 Deposits by government and non-euro area residents



Source: ECB.

1) MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.

2) The term "banks" is used in this table to indicate institutions of a similar type to MFIs resident outside the euro area.



2.6 MFI holdings of securities, breakdown <sup>1</sup>) (EUR billions and annual growth rates; outstanding amounts and growth rates at end of period, transactions during period)

			5	Securities o	ther than sh	ares				Shares and	l other equity	Ŷ
	Total	MI	FIs	Gen gover		Other area res		Non-euro area residents	Total	MFIs	Non-MFIs	Non-euro area residents
		Euro	Non-euro	Euro	Non-euro	Euro	Non-euro					
	1	2	3	4	5	6	7	8	9	10	11	12
					Out	standing am	ounts					
2003	3,576.3	1,216.2	57.4	1,227.1	15.6	409.1	18.6	632.3	1,071.4	279.7	615.3	176.4
2004	3,939.5	1,362.7	59.9	1,284.1	15.8	449.0	16.3	751.7	1,158.1	286.4	656.4	215.2
2005 Q1	4,093.1	1,388.9	66.6	1,342.8	15.8	464.9	16.3	797.9	1,217.0	296.1	674.4	246.5
Q2	4,269.0	1,435.8	67.7	1,368.1	15.8	488.0	18.9	874.7	1,234.8	295.3	704.1	235.5
2005 July	4,272.4	1,444.3	67.2	1,362.8	15.9	486.8	19.7	875.8	1,241.2	303.2	696.7	241.3
Aug.	4,281.2	1,443.0	67.3	1,355.0	16.1	484.7	20.4	894.6	1,241.5	303.7	695.7	242.0
Sep.	4,271.4	1,442.5	67.8	1,343.7	16.7	484.9	19.6	896.2	1,256.9	297.1	715.5	244.4
Oct.	4,346.3	1,444.4	69.7	1,383.8	16.9	498.5	22.3	910.7	1,227.8	296.9	692.6	238.3
Nov. <sup>(p)</sup>	4,471.0	1,454.1	71.2	1,467.3	17.4	518.0	24.4	918.7	1,249.7	310.7	698.1	240.9
			1,454.1 71.2 1,467.3 17.4				IS					
2003	324.6	90.8	4.1	79.0	0.8	52.3	1.7	95.9	18.8	7.2	19.3	-7.8
2004	368.4	148.1	4.9	40.3	1.3	34.8	-1.3	140.4	67.6	2.2	34.5	30.8
2005 Q1	137.7	29.1	4.7	55.3	-0.5	17.0	-0.5	32.5	57.1	9.4	16.1	31.6
Q2	128.9	46.2	-1.8	11.6	-1.0	21.9	1.6	50.3	14.5	5.0	25.2	-15.7
2005 July	9.6	8.3	$\begin{array}{c} 0.1 \\ 0.1 \\ 0.0 \\ 1.7 \\ 0.9 \end{array}$	-2.3	0.4	-1.6	1.0	3.7	-0.1	3.9	-8.3	4.3
Aug.	7.0	-2.1		-8.7	0.2	-2.5	0.6	19.5	-1.9	0.1	-1.8	-0.3
Sep.	-14.0	-1.2		-9.8	0.4	0.0	-0.8	-2.5	8.1	-2.9	11.7	-0.7
Oct.	49.0	3.1		14.8	0.3	13.3	2.6	13.2	2.4	0.2	8.4	-6.3
Nov. <sup>(p)</sup>	74.4	7.8		42.7	-0.1	15.1	1.8	6.2	20.1	13.1	1.6	5.3
						Growth rate	es					
2003 Dec.	9.9	8.1	8.7	6.9	5.0	14.8	8.2	17.2	1.9	2.7	3.4	-4.2
2004 Dec.	10.2	12.2	8.4	3.3	7.7	8.4	-7.3	22.0	6.3	0.8	5.6	17.2
2005 Mar.	9.3	9.1	14.1	3.9	-4.1	11.1	-4.6	19.1	7.3	1.9	4.0	26.4
June	10.2	11.4	8.1	1.4	-9.9	12.7	4.2	24.2	6.1	1.4	6.5	11.6
2005 July	9.2	10.5	9.4	1.5	-6.7	12.2	15.9	19.7	6.2	3.5	6.7	8.6
Aug.	8.9	9.9	7.3	0.9	-2.9	11.7	20.9	19.9	6.7	3.9	8.0	6.3
Sep.	8.3	9.1	5.8	-0.1	2.9	12.3	10.3	20.3	9.1	4.7	10.1	12.4
Oct.	8.6	7.7	8.3	1.6	-19.2	14.7	22.5	19.8	8.8	4.7	10.9	8.2
Nov. <sup>(p)</sup>	9.3	7.5	8.9	4.4	-14.2	16.6	31.1	16.9	6.9	7.1	9.1	0.6



Source: ECB. 1) MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.



# 2.7 Revaluation of selected MFI balance sheet items <sup>1)</sup> (EUR billions)

## 1. Write-offs/write-downs of loans to households<sup>2)</sup>

		Consum	er credit		L	ending for h	ouse purchase			Other l	ending	
	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years
	1	2	3	4	5	6	7	8	9	10	11	12
2003 2004	-2.7 -3.2	-1.1 -1.3	-0.5 -0.7	-1.1 -1.3	-3.2 -3.4	-0.3 -0.3	-0.1 -0.1	-2.8 -3.0	-7.2 -6.7	-2.8 -2.3	-0.3 -0.3	-4.1 -4.1
2005 Q1 Q2	-1.3 -0.8	-0.6 -0.3	-0.2 -0.2	-0.5 -0.3	-1.2 -0.8	-0.1 0.0	$\begin{array}{c} 0.0\\ 0.0\end{array}$	-1.1 -0.7	-2.7 -1.6	-1.1 -0.8	-0.1 -0.1	-1.6 -0.8
2005 July Aug. Sep. Oct. Nov. <sup>(p)</sup>	-0.4 -0.2 -0.3 -0.3 -0.2	-0.2 -0.1 -0.1 -0.2 0.0	0.0 0.0 -0.1 -0.1 -0.1	-0.1 -0.1 -0.1 -0.1 -0.1	-0.2 -0.1 -0.2 -0.1 -0.2	$\begin{array}{c} 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \\ 0.0 \end{array}$	$\begin{array}{c} 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\end{array}$	-0.2 -0.1 -0.2 -0.1 -0.2	-0.3 -0.3 -0.4 -0.3 -0.6	-0.1 -0.1 -0.2 -0.1 -0.2	$\begin{array}{c} 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\end{array}$	-0.1 -0.1 -0.2 -0.2 -0.4

# 2. Write-offs/write-downs of loans to non-financial corporations and non-euro area residents

		Non-financial corp	oorations		Non-euro	area residents	
	Total	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Total	Up to 1 year	Over 1 year
	1	2	3	4	5	6	7
2003 2004	-17.5 -16.1	-8.8 -8.8	-1.3 -0.8	-7.4 -6.5	-1.1 -1.6	-0.3 -0.5	-0.7 -1.1
2005 Q1 Q2	-5.1 -3.8	-2.5 -1.9	-0.7 -0.2	-1.9 -1.8	-0.3 -0.3	-0.1 0.0	-0.3 -0.3
2005 July Aug.	-0.5 -0.5 -0.8	-0.3 -0.2 -0.4	-0.1 -0.1 -0.1	-0.2 -0.2 -0.3	-0.1 -0.1 -0.1	0.0 0.0 -0.1	-0.1 0.0 0.0
Sep. Oct. Nov. <sup>(p)</sup>	-0.8 -1.4 -0.5	-0.4 -1.1 -0.4	-0.1 -0.1 0.0	-0.3 -0.2 0.0	-0.1 0.0 0.0	-0.1 0.0 0.0	0.0 0.0 0.0

## 3. Revaluation of securities held by MFIs

			5	Securities of	ther than sh	ares				Shares and	l other equity	7
	Total	MF	FIs	Gen govern		Other area res		Non-euro area residents	Total	MFIs	Non-MFIs	Non-euro area residents
	1	Euro 2	Non-euro	Euro 4	Non-euro 5	Euro 6	Non-euro 7	8	9	10	11	12
2003	-1.2	-0.8	-0.3	3.0	0.0	-1.1	-0.1	-1.9	19.4	8.0	5.0	6.4
2004	13.5	1.5	-0.1	10.8		0.9	-0.1	0.6	8.1	1.3	3.4	3.5
2005 Q1	5.9	1.0	0.1	3.8	0.1	-0.7	0.1	1.6	4.6	0.5	2.7	1.4
Q2	17.2	2.9	0.2	7.8	0.2	1.6	0.1	4.4	9.8	0.9	4.3	4.6
2005 July	-3.1	-0.1	0.0	-3.0	-0.1	0.0	0.0	0.2	5.0	1.0	2.4	1.6
Aug.	1.0	0.2	0.1	0.9	0.0	0.1	0.1	-0.5	2.3	0.2	1.0	1.0
Sep.	-1.0	0.1	0.1	-1.5	0.0	0.2	-0.1	0.3	6.9	0.6	3.5	2.7
Oct.	-3.1	-0.6	-0.1	-1.4	0.0	-0.4	$\begin{array}{c} 0.1\\ 0.0 \end{array}$	-0.7	-0.9	-0.3	-0.8	0.2
Nov. <sup>(p)</sup>	4.7	-0.6	0.0	0.7	0.5	0.0		4.0	7.6	1.0	3.4	3.2

Source: ECB.

MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
 Including non-profit institutions serving households.



# 2.8 Currency breakdown of selected MFI balance sheet items <sup>1</sup>) (percentages of total; outstanding amounts in EUR billions; end of period)

# 1. Deposits

			MFI	S <sup>2)</sup>						Non-N	MFIs			
	All	Euro <sup>3)</sup>		Non-euro	o currencie	s		All	Euro <sup>3)</sup>		Non-euro	o currencies		
	(outstanding amount)		Total					(outstanding amount)		Total				
				USD	JPY	CHF	GBP		0	10	USD	JPY	CHF	GBP
	1	2	3	4	5	By euro ar	/	nts	9	10	11	12	13	14
2003	4,364.9	91.3	8.7	5.4	0.5	1.5	0.9	6,409.9	97.3	2.7	1.7	0.3	0.1	0.3
2004	4,709.0	91.4	8.6	5.0	0.5	1.5	1.1	6,778.5	97.2	2.8	1.7	0.3	0.1	0.4
2005 Q1	4,820.7	91.0	9.0	5.4	0.5	1.4	1.1	6,833.0	97.0	3.0	1.9	0.3	0.1	0.4
Q2	4,793.3	90.9	9.1	5.5	0.4	1.4	1.1	7,056.2	96.9	3.1	1.9	0.3	0.1	0.4
Q3 (p)	4,791.3	90.6	9.4	5.7	0.5	1.5	1.1	7,122.2	96.7	3.3	2.0	0.3	0.1	0.4
					В	y non-euro	area resid	dents						
2003	1,580.8	46.9	53.1	35.6	1.8	3.6	9.4	664.3	51.0	49.0	32.1	2.1	2.2	9.6
2004	1,748.0	46.7	53.3	35.8	2.1	3.2	9.5	680.9	55.4	44.6	28.9	1.5	2.2	9.3
2005 Q1	1,935.7	46.9	53.1	35.2	2.4	2.9	9.7	733.4	54.6	45.4	29.4	1.5	2.0	9.2
Q2	2,034.1	45.8	54.2	36.0	2.4	3.1	9.5	750.8	52.5	47.5	30.6	1.5	2.3	9.9
Q3 <sup>(p)</sup>	2,109.8	46.7	53.3	35.0	2.5	3.0	9.5	795.1	52.3	47.7	30.6	1.8	2.2	10.0

# 2. Debt securities issued by euro area MFIs

	All currencies	Euro <sup>3)</sup>		Non-eu	iro currencies		
	(outstanding amount)		Total				
				USD	JPY	CHF	GBP
	1	2	3	4	5	6	7
2003 2004	3,304.0 3,653.9	85.4 84.6	14.6 15.4	7.9 7.6	1.5 1.7	1.7 1.9	2.3 2.7
$2005 \underset{Q3}{Q1} \underset{Q3}{Q2} \underset{(p)}{Q2}$	3,794.9 3,942.7 3,994.0	83.4 82.4 81.8	16.6 17.6 18.2	8.2 9.0 9.2	1.7 1.8 1.9	1.9 1.9 1.9	2.9 3.0 3.1

1) MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
2) For non-euro area residents, the term "MFIs" refers to institutions of a similar type to euro area MFIs.
3) Including items expressed in the national denominations of the euro.



# 2.8 Currency breakdown of selected MFI balance sheet items <sup>1</sup>) (percentages of total; outstanding amounts in EUR billions; end of period)

# 3. Loans

			MF	Is <sup>2)</sup>						Non-l	MFIs			
	All	Euro <sup>3)</sup>		Non-eu	ro currencie	s		All	Euro <sup>3)</sup>		Non-eur	o currencies	5	
	(outstanding amount)		Total					outstanding amount)		Total				
	1	2	3	USD	JPY 5	CHF	GBP 7	8	0	10	USD 11	JPY 12	CHF 13	GBP 14
	1	2			5	To euro ai	rea resider	U U		10]	11	12	15	
2003 2004	4,193.9 4,457.8	-	-	-	-	-	- -	7,919.3 8,367.5	96.5 96.6	3.5 3.4	1.6 1.4	0.3 0.2	1.2 1.3	0.3 0.4
2005 Q1 Q2	4,575.4 4,529.4	-	-	-	-	-	-	8,475.9 8,726.9	96.5 96.4	3.5 3.6	1.5 1.6	0.2 0.2	1.3 1.3	0.4 0.4
Q3 <sup>(p)</sup>	4,551.4	-	-	-	- T	- `o non-euro	-	8,883.1	96.3	3.7	1.7	0.2	1.2	0.4
			10.0											
2003 2004	1,182.2 1,342.2	50.2 51.4	49.8 48.6	29.3 29.9	4.7 3.7	2.5 2.2	9.2 8.7	575.7 632.5	38.8 42.2	61.2 57.8	43.6 40.1	2.4 2.6	4.6 4.5	7.0 7.2
2005 Q1 Q2 Q3 <sup>(p)</sup>	1,463.8 1,582.4 1,638.7	51.8 49.3 49.3	48.2 50.7 50.7	29.2 31.0 30.1	3.4 4.2 4.3	2.1 2.0 2.0	9.2 9.0 9.6	672.7 710.1 736.9	41.8 41.0 39.8	58.2 59.0 60.2	42.1 43.1 43.0	1.4 1.1 1.6	4.3 4.4 3.9	7.1 7.2 8.3

# 4. Holdings of securities other than shares

			Issued by	WFIs <sup>2)</sup>						Issued by	non-MFIs			
	All	Euro <sup>3)</sup>		Non-euro	currencies	5		All	Euro <sup>3)</sup>		Non-eur	o currencies	3	
	(outstanding amount)		Total					(outstanding amount)		Total				
	uniounty			USD	JPY	CHF	GBP	unount)			USD	JPY	CHF	GBP
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
					Iss	ued by euro	area resi	idents						
2003	1,273.6	95.5	4.5	1.7	0.3	0.9	1.3	1,670.3	98.0	2.0	1.0	0.5	0.3	0.2
2004	1,422.6	95.8	4.2	1.8	0.3	0.5	1.3	1,765.1	98.2	1.8	0.9	0.5	0.1	0.3
2005 Q1	1,455.5	95.4	4.6	2.1	0.4	0.4	1.5	1,839.7	98.3	1.7	0.9	0.4	0.1	0.3
Q2	1,503.5	95.5	4.5	2.1	0.3	0.4	1.5	1,890.8	98.2	1.8	1.0	0.4	0.1	0.3
Q3 (p)	1,510.3	95.5	4.5	2.0	0.3	0.4	1.5	1,864.8	98.1	1.9	1.0	0.4	0.1	0.4
					Issue	d by non-eu	iro area r	esidents						
2003	276.9	45.1	54.9	30.6	1.2	4.9	15.4	355.5	45.8	54.2	31.1	5.8	5.8	6.4
2004	341.3	50.3	49.7	28.6	1.0	0.5	17.0	410.4	44.8	55.2	30.5	8.6	0.7	9.2
2005 Q1	359.8	48.9	51.1	30.3	1.0	0.5	16.5	438.0	43.7	56.3	32.7	7.2	0.8	9.1
Q2	397.4	47.9	52.1	30.3	0.8	0.5	17.8	477.3	41.1	58.9	34.0	7.9	0.8	9.9
Q3 <sup>(p)</sup>	404.9	49.6	50.4	29.1	0.8	0.5	17.1	491.3	40.2	59.8	35.4	7.3	0.8	10.6
Source: ECP														

Source: ECB.

MFI sector excluding the Eurosystem; sectoral classification is based on ESA 95.
 For non-euro area residents, the term "MFIs" refers to institutions of a similar type to euro area MFIs.
 Including items expressed in the national denominations of the euro.



# 2.9 Aggregated balance sheet of euro area investment funds <sup>1</sup>) (EUR billions; outstanding amounts at end of period)

# 1. Assets

	Total	Deposits		lings of securitie her than shares	s	Holdings of shares/ other	Holdings of investment fund shares	Fixed assets	Other assets
			Total	Up to	Over	equity			
				l year	1 year				
	1	2	3	4	5	6	7	8	9
2004 Q2	3,631.6	263.7	1,540.5	75.7	1,464.7	1,206.9	299.8	151.0	169.7
Q3	3,652.8	265.6	1,585.6	78.5	1,507.1	1,179.2	302.5	155.5	164.3
Q4	3,790.0	259.4	1,617.6	78.1	1,539.5	1,250.5	317.3	158.6	186.7
2005 Q1	4,013.0	286.9	1,687.3	79.2	1,608.1	1,324.7	342.4	163.3	208.5
Q2	4,263.4	294.9	1,778.7	91.3	1,687.4	1,404.9	379.1	167.7	238.1
Q3 <sup>(p)</sup>	4,572.2	301.4	1,856.0	100.7	1,755.2	1,556.6	417.0	170.4	270.8

# 2. Liabilities

	Total	Deposits and loans taken		Other liabilities
	1	2	3	4
2004 Q2	3,631.6	54.2	3,441.1	136.3
Q3	3,652.8	53.3	3,463.1	136.4
Q4	3,790.0	52.3	3,588.4	149.2
2005 Q1	4,013.0	60.5	3,764.0	188.5
Q2	4,263.4	57.8	3,996.9	208.6
Q3 <sup>(p)</sup>	4,572.2	59.5	4,306.0	206.7

## 3. Total assets/liabilities broken down by investment policy and type of investor

	Total		Fun	ds by investment po	olicy		Funds by type of investor		
		Equity funds	Bond funds	Mixed funds	Real estate funds	Other funds	General public funds	Special investors' funds	
	1	2	3	4	5	6	7	8	
2004 Q2 Q3 Q4	3,631.6 3,652.8 3,790.0	814.5 796.8 834.3	1,178.1 1,204.8 1,229.8	893.6 889.1 912.0	193.5 196.4 196.9	552.0 565.8 617.0	2,669.4 2,686.6 2,795.5	962.2 966.2 994.4	
2005 Q1 Q2 Q3 <sup>(p)</sup>	4,013.0 4,263.4 4,572.2	861.9 1,094.4 1,221.8	1,274.8 1,498.4 1,568.9	951.7 974.6 1,032.2	201.2 207.2 211.8	723.4 488.8 537.6	2,981.1 3,179.6 3,448.0	1,032.0 1,083.8 1,124.2	



Source: ECB.

1) Other than money market funds. For further details, see the General notes.



# 2.10 Assets of euro area investment funds broken down by investment policy and type of investor (EUR billions; outstanding amounts at end of period)

## 1. Funds by investment policy

	Total	Deposits		ngs of securities r than shares		Holdings of shares/ other	Holdings of investment fund shares	Fixed assets	Other assets
			Total	Up to 1 year	Over 1 year	equity			
	1	2	3	4	5	6	7	8	9
				Equity funds					
2004 Q2	814.5	33.9	34.0	3.5	30.5	692.1	27.2	-	27.2
Q3	796.8	33.9	35.5	4.0	31.5	673.3	27.0		27.1
Q4	834.3	30.8	36.7	4.0	32.6	705.8	30.2	-	30.8
2005 Q1	861.9	33.7	36.7	4.0	32.7	729.8	31.3		30.5
Q2	1,094.4	44.8	41.2	4.5	36.8	936.6	37.9		33.9
Q3 <sup>(p)</sup>	1,221.8	48.0	42.8	4.9	37.9	1,045.1	50.2		35.8
				Bond funds					
2004 Q2	1,178.1	85.2	979.7	39.5	940.2	35.5	23.5	-	54.2
Q3	1,204.8	87.0	1,003.8	42.1	961.8	34.4	25.2	-	54.4
Q4	1,229.8	83.7	1,016.9	43.3	973.6	39.9	25.1	-	64.2
2005 Q1	1,274.8	97.5	1,042.1	44.7	997.4	39.4	28.1	-	67.7
Q2	1,498.4	110.2	1,225.8	58.4	1,167.4	38.4	32.6	-	91.3
Q3 <sup>(p)</sup>	1,568.9	110.0	1,285.7	67.0	1,218.7	38.4	35.0	-	99.8
				Mixed funds					
2004 Q2	893.6	56.3	366.1	24.0	342.1	300.3	123.7	0.3	46.8
Q3	889.1	56.3	374.5	23.7	350.8	291.2	124.4	0.3	42.4
Q4	912.0	54.5	374.7	21.7	353.0	304.1	131.0	0.3	47.4
2005 Q1	951.7	60.4	387.6	22.4	365.2	314.1	134.8	0.2	54.7
Q2	974.6	64.9	417.3	21.2	396.2	276.6	146.5	0.2	69.0
Q3 <sup>(p)</sup>	1,032.2	66.3	425.1	21.6	403.5	300.0	160.2	0.2	80.4
				Real estate fund	ls				
2004 Q2	193.5	16.1	9.2	0.7	8.6	0.7	8.3	149.8	9.3
Q3	196.4	15.5	9.2	0.7	8.5	0.8	8.1	154.1	8.7
Q4	196.9	15.7	7.6	0.7	6.9	1.0	7.5	156.4	8.7
2005 Q1	201.2	14.3	8.4	0.7	7.7	1.1	7.5	160.9	9.0
Q2	207.2	14.0	8.2	0.8	7.5	1.1	7.6	167.2	9.0
Q3 <sup>(p)</sup>	211.8	15.2	8.7	1.2	7.6	1.3	8.1	169.8	8.7

# 2. Funds by type of investor

	Total	Deposits	Holdings of securities other than shares	Holdings of shares/ other equity	Holdings of investment fund shares	Fixed assets	<b>Other</b> assets
	1	2	3	4	5	6	7
			General pul	blic funds			
2004 Q2	2,669.4	217.6	1,018.3	958.4	227.2	129.7	118.2
Q3	2,686.6	221.5	1,049.0	939.5	229.6	133.5	113.6
Q4	2,795.5	217.3	1,072.4	1,000.1	239.2	137.6	128.9
2005 Q1	2,981.1	241.3	1,129.5	1,058.7	259.5	141.2	150.7
Q2	3,179.6	247.2	1,202.2	1,124.9	284.0	144.9	176.3
Q3 <sup>(p)</sup>	3,448.0	250.8	1,256.1	1,257.8	320.9	145.2	217.3
			Special inves	stors' funds			
2004 Q2	962.2	46.1	522.2	248.5	72.6	21.3	51.5
Q3	966.2	44.1	536.6	239.7	72.9	22.0	50.8
Q4	994.4	42.0	545.2	250.3	78.1	21.0	57.8
2005 Q1	1,032.0	45.5	557.7	266.0	82.9	22.0	57.8
Q2	1,083.8	47.6	576.5	280.0	95.1	22.8	61.8
Q3 <sup>(p)</sup>	1,124.2	50.6	599.9	298.9	96.2	25.2	53.4

Source: ECB.





# FINANCIAL AND NON-FINANCIAL ACCOUNTS

Total **Currency and deposits** Memo: deposits of Deposits of non-financial sectors other than central government with euro area MFIs Total Currency Deposits of Deposits with non-banks non-MFIs<sup>1</sup> with banks central government outside the Total Overnight With agreed Repos Redeemable with euro euro area maturity at notice area MFIs 9 11 Outstanding amounts 2004 Q1 Q2 5,180.6 5,263.9 2,020.6 2,101.2 1,545.0 1,529.5 1,533.9 1,553.9 183.8 223.7 396.9 397.2 15,701.6 350.8 372.3 81.2 79.4 5 920 1 204.8 15,963.4 6,056.9 197.0 Q3 Q4 16.026.8 6,081.9 383.5 413.7 5,284.3 5,435.0 2,104.2 2,165.2 1,532.2 1,577.9 1,565.1 1,603.7 82.8 88.2 204.1 162.4 210.0 230.7 394.9 16,361.8 6,241.7 385.4 2005 Q1 Q2 5,433.3 5,550.7 16 670 2 6 257 7 408.4 2 174 8 1 620 0 78.5 429.8 1 560 0 187.4 228 5 17,136.0 6,424.3 430.8 2,449.1 1,552.8 1,471.1 77.7 211.5 231.3 446.0 Transactions -15.8 -14.7 5.5 6.3 -7.8 13.0 2004 Q1 27.8 -5.0 -7.6 22.4 -3.9 28.1 44.4 146.5 -1.6 82.0 4.6 295.2 118.5 139.2 29.4 21.4 11.3 86.1 24.8 20.7 11.3 -1.9 3.4 39.4 -19.7 -0.2 1.7 Q2 Q3 Õ4 152.0 168.5 30.2 159.4 65.2 49.9 38.9 5.4 -41.7 20.7 2.0 -5.2 2005 Q1 154.1 15.4 -3.5 7.9 -17.4 15.7 -9.7 25.0 -0.9 37.7 111.0 Q2 298.1 160.2 111.0 -10.1 11.0 -0.9 24.1 2.8 5.5 Growth rates 2004 Q1 Q2 5.9 12.8 10.7 3.9 8.0 3.9 10.7 16.2 4.4 5.3 5.3 20.9 4.3 4.2 8.8 8.3 -1.3 -1.7 6.5 6.3 -23.0 26.2 21.9 4.7 19.5 -15.7 4.6 4.6 5.6 6.2 18.8 17.4 4.4 5.1 8.1 7.1 6.2 6.2 -8.5 3.6 16.5 13.8 Q3 Q4 -1.1 1.6 2005 Q1 Q2 4.6 4.5 6.0 6.2 5.2 5.5 7.9 9.0 1.5 1.8 5.6 4.9 -3.3 -2.1 1.7 -5.5 16.4 12.2 10.4 15.7 18.1 11.8 Securities other than shares Shares<sup>2)</sup> **Insurance technical reserves** Total Short-term Long-term Total Quoted Mutual fund Total Net equity of households in Prepayments shares shares Money of insurance life insurance premiums and reserves marke fund reserves and shares/units for outstanding pension fund reserves claims 19 20 21 12 13 14 16 17 18 Outstanding amounts 2004 Q1 1,894.8 157.0 1,737.8 3,917.4 1,957.7 1,959.7 419.6 3,969.3 3,595.4 373.9 Q2 Q3 1,918.4 1,931.5 174.2 172.8 1,744.2 1,758.7 3,967.6 3,926.8 2,014.3 1,977.7 1,953.3 1,949.2 423.1 423.7 4,020.4 4,086.5 3,642.0 3,704.4 378.4 382.2 Õ4 1,904.7 160.5 1,744.2 4,056.2 2,094.3 1,961.9 406.6 4,159.2 3,774.6 384.6 2,191.6 2,250.2 2,012.5 2005 Q1 1,938.4 1,776.8 4,204.1 414.9 4,270.1 3,876.6 393.5 161.6 Ò2 2.016.1 167.9 1 848 2 4 321 9 4162 4 373 7 3 976 7 397.0 Transactions 68.2 57.7 62.1 8.5 17.5 6.7 2004 Q1 13.1 4.6 37.4 30.8 15.1 58.6 9.6 4.4 3.8 2.5 Q2 Q3 40.2  $\begin{array}{c} 22.6\\ 16.0 \end{array}$ 58.2 56.5 7.1 1.7 3.8 -0.4 -2.0 53 2 16.1 0.1 10.9 58.3 -45.1 -11.0-15.5 Õ4 -21.2 -16.1 -5.2 -56.2 60.9 58.4 2005 Q1 33.9 50.6 -0.3 8.8 5.2 73.7 -0.2 34.1 31.1 31.4 64.8 8.9

Ž2 Source: ECB

Q2

Q2 Ô3

Q4

2004 O1

2005 O1

Covering deposits with euro area central government (S.1311 in ESA 95), other financial intermediaries (S.123 in ESA 95) and insurance corporations and pension funds (S.125 1) in ESA 95).

2.5

18

3.6

2.8 1.3

0.9

-1.8

Growth rates

232

5.1 3.0

2.0

1.3

1.3

2.4

61.6

6.6 6.3 6.2

6.4

6.4

6.4

3.0 1.8 0.7 -0.7

-2.2

-0.8

578

6.7 6.4

6.4 6.5

6.5

6.6

3.8

5.3 5.1

5.0

5.6

5.2 5.0

2) Excluding unquoted shares

-1.2 2.5 2.6

2.6

3.6

4.1

53

-1.9

15.9

10.2

6.9

0.8

-6.3

453

-1.1 1.3 1.9 2.2

3.9 5.2

258

3.6 3.3 2.4

1.3

1.1

0.3



**3.2 Main liabilities of non-financial sectors** (EUR billions and annual growth rates; outstanding amounts at end of period, transactions during the period)

	Total		Loans taken from euro area MFIs and other financial corporations by											
		Total		G	eneral govern	iment	Non-fi	nancial corpo	orations		Households 1)		taken from banks	
			Taken from euro area MFIs	Total	Short-term	Long-term	Total	Short-term	Long-term	Total	Short-term	Long-term	outside the euro area by non-banks	
	1	2	3	4	5	6	7	8	9	10	11	12	13	
						Outstand	ling amounts							
2004 Q1 Q2 Q3 Q4	17,078.1 17,275.4 17,388.4 17,755.2	8,570.8 8,714.2 8,795.1 8,919.0	7,463.0 7,593.6 7,671.3 7,794.7	938.9 932.8 928.5 927.9	86.1 91.5 90.1 80.9	852.8 841.3 838.4 847.0	3,693.0 3,736.5 3,740.4 3,781.8	1,178.2 1,187.8 1,171.5 1,193.6	2,514.8 2,548.7 2,568.9 2,588.2	3,938.9 4,044.9 4,126.1 4,209.4	280.1 291.9 289.5 294.6	3,658.8 3,753.0 3,836.6 3,914.8	338.5 289.3 284.2 290.6	
2005 Q1 Q2	18,121.0 18,636.0	9,008.9 9,214.5	7,878.6 8,103.4	924.3 923.8	77.5 82.4	846.8 841.5	3,812.5 3,905.3	1,192.7 1,240.6	2,619.8 2,664.7	4,272.1 4,385.3	294.6 305.1	3,977.5 4,080.3	332.8 359.8	
						Trai	nsactions							
2004 Q1 Q2 Q3 Q4	205.5 266.1 153.9 114.7	51.5 150.2 78.5 141.7	74.3 134.5 86.1 139.6	3.9 -9.2 -5.2 1.8	3.9 5.4 -1.4 -9.2	0.0 -14.6 -3.8 11.0	-8.8 69.8 0.0 56.0	-3.7 16.7 -16.8 24.8	-5.1 53.0 16.8 31.2	56.4 89.6 83.8 83.9	-4.3 8.6 -1.9 6.3	60.7 81.1 85.7 77.6	66.7 -2.9 -1.9 15.8	
2005 Q1 Q2	232.6 315.3	89.6 195.1	87.9 185.8	-4.3 -1.1	-3.4 4.9	-0.9 -5.9	28.9 86.3	5.4 38.9	23.5 47.4	65.1 109.9	1.0 10.4	64.1 99.4	35.2 18.3	
						Gro	wth rates							
2004 Q1 Q2 Q3 Q4	4.4 4.4 4.5 4.4	4.5 4.9 4.8 4.9	4.7 5.3 5.7 5.9	1.7 1.7 0.9 -0.9	25.6 28.1 24.6 -1.6	-0.4 -0.7 -1.2 -0.9	2.0 2.4 2.2 3.2	-1.6 -2.3 -1.5 1.8	3.8 4.8 3.9 3.8	7.6 8.0 8.3 8.1	-0.8 0.8 1.9 3.0	8.3 8.7 8.8 8.5	35.4 33.6 22.0 29.2	
2005 Q1 Q2	4.5 4.7	5.4 5.8	6.0 6.6	-1.8 -0.9	-10.0 -10.0	-1.0 0.0	4.2 4.6	2.6 4.4	5.0 4.7	8.2 8.5	5.0 5.4	8.4 8.7	13.7 23.3	

			Securities of	ther than share		Quoted shares	Deposit liabilities of	Pension fund		
	Total	Ge	neral governmen	t	Non-	financial corpora	tions	issued by non-financial	central government	reserves of non-
		Total	Short-term	Long-term	Total	Short-term	Long-term	corporations	government	financial corporations
	14	15	16	17	18	19	20	21	22	23
					Outstanding am	ounts				
2004 Q1 Q2 Q3 Q4	5,193.5 5,242.9 5,338.8 5,342.4	4,566.5 4,599.4 4,688.3 4,698.1	594.7 618.5 620.1 587.6	3,971.9 3,980.9 4,068.2 4,110.5	627.0 643.5 650.4 644.3	214.3 228.3 224.7 216.4	412.7 415.2 425.8 427.9	2,834.5 2,843.1 2,763.8 2,980.4	189.0 181.9 194.0 213.5	290.3 293.4 296.6 299.9
2005 Q1 Q2	5,459.1 5,660.3	4,802.6 4,992.6	597.0 618.0	4,205.6 4,374.6	656.5 667.7	231.7 237.0	424.8 430.7	3,138.4 3,240.5	212.0 214.7	302.7 306.0
					Transaction	s				
2004 Q1 Q2 Q3 Q4	140.6 118.1 53.2 -52.7	145.3 98.4 45.0 -44.2	40.1 23.3 3.4 -32.7	105.2 75.1 41.6 -11.6	-4.7 19.7 8.1 -8.5	14.0 14.1 -2.6 -7.5	-18.7 5.6 10.8 -1.0	3.0 1.8 6.8 2.3	7.3 -7.1 12.1 19.4	3.0 3.1 3.2 4.0
2005 Q1 Q2	135.8 115.6	119.8 110.6	7.9 22.2	111.9 88.4	16.0 5.0	16.7 4.8	-0.8 0.2	4.7 -1.4	-0.2 2.6	2.8 3.4
					Growth rate	s				
2004 Q1 Q2 Q3 Q4 2005 Q1	5.5 5.4 5.7 5.2 4.9	5.6 5.7 5.8 5.6 4.8	10.4 7.2 9.0 6.2 0.3	4.9 5.5 5.3 5.5 5.5	4.8 3.5 5.2 2.3 5.6	13.8 15.9 12.9 9.1 9.7	0.8 -2.1 1.5 -0.8 3.5	1.0 0.4 0.5 0.5 0.6	7.5 4.8 11.3 17.5 12.8	4.9 4.8 4.8 4.7 4.5
2005 Q1 Q2	4.9	4.8 5.0	0.3	5.5 5.8	3.0 3.2	9.7 5.0	3.5 2.2	0.6	12.8 18.7	4.5 4.6

Source: ECB. 1) Including non-profit institutions serving households.



# **3.3** Main financial assets and liabilities of insurance corporations and pension funds (EUR billions and annual growth rates; outstanding amounts at end of period, transactions during the period)

	Main financial assets											
	Total		Deposit	s with euro are	a MFIs			Loans		Securitie	es other than s	shares
		Total	Overnight	With agreed maturity	Redeemable at notice	Repos	Total	Short-term	Long-term	Total	Short-term	Long-term
	1	2	3	4	5	6	7	8	9	10	11	12
					Outs	tanding amou	nts					
2004 Q1 Q2 Q3 Q4	4,017.2 4,038.2 4,100.9 4,195.6	557.3 565.4 573.6 583.2	64.7 59.9 61.5 59.2	468.3 482.0 489.8 500.8	2.3 2.3 2.3 2.5	22.0 21.2 20.0 20.8	354.2 347.6 353.2 332.7	63.5 61.8 64.6 57.4	290.7 285.8 288.6 275.3	1,607.6 1,612.8 1,656.5 1,715.4	64.7 65.7 65.3 66.0	1,543.0 1,547.1 1,591.1 1,649.4
2005 Q1 Q2	4,329.2 4,454.6	597.0 595.7	65.7 61.2	508.8 511.3	2.7 2.7	19.8 20.5	335.1 324.5	59.3 57.6	275.9 266.9	1,765.7 1,832.2	66.2 67.2	1,699.5 1,765.0
2004.01				5.0		Fransactions	1.0	1.0		(5.0	1.0	
2004 Q1 Q2 Q3 Q4	80.2 30.4 54.8 51.6	14.6 7.2 8.2 9.9	5.7 -4.9 1.6 -1.7	5.9 13.7 7.8 10.6	0.2 -0.6 -0.1 0.2	2.8 -0.9 -1.1 0.7	1.2 -6.6 5.6 -20.6	1.9 -1.7 2.8 -7.3	-0.8 -4.9 2.7 -13.3	47.2 26.2 28.3 50.3	1.0 1.0 -0.5 0.5	46.2 25.3 28.8 49.8
2005 Q1 Q2	94.9 50.1	12.5 -2.1	6.4 -5.3	6.9 2.2	0.2 0.2	-1.0 0.8	0.2 -10.8	1.9 -1.7	-1.7 -9.2	52.2 39.0	-0.1 0.1	52.2 38.9
					(	Growth rates						
2004 Q1 Q2 Q3 Q4 2005 Q1	7.1 6.1 6.6 5.6 5.8	3.9 4.8 7.5 7.4 6.8	5.0 -6.4 6.8 1.2 2.2	3.0 6.9 7.7 8.2 8.3	38.5 6.5 -12.8 -12.0 -11.6	18.7 -6.3 6.7 7.8 -10.5	3.3 0.6 2.1 -5.8 -6.0	6.8 0.3 7.4 -6.9 -6.8	2.5 0.6 1.0 -5.6 -5.9	10.5 10.4 10.2 9.8 9.8	8.2 4.4 1.6 3.2 1.4	10.6 10.6 10.5 10.1 10.1
2003 Q1 Q2	6.2	5.1	1.8	5.7	23.8	-3.0	-7.4	-6.9	-7.5	10.5	0.0	11.0

		Mai	in financial a	ssets					Mai	n liabilities			
		Share	es 1)		Prepayments of insurance	Total		aken from rea MFIs	Securities other than	Quoted shares	Insu	rance technical r	reserves
	Total	Quoted shares	Mutual fund shares	Money market fund shares/ units	premiums and reserves for outstanding claims		and othe	r financial prations Taken from euro area MFIs	shares		Total	Net equity of households in life insurance reserves and pension fund reserves	Prepayments of insurance premiums and reserves for outstanding claims
	13	14	15	16	17	18	19	20	21	22	23	24	25
						Outstandin	ng amounts	3					
2004 Q1 Q2 Q3	1,363.7 1,376.7 1,380.0	657.2 661.9 656.6	706.5 714.8 723.5	66.0 65.8 65.1	134.4 135.6 137.7	4,176.3 4,232.5 4,292.2	84.5 89.0 90.7	46.3 53.7 52.5	23.8 24.4 23.1	191.5 193.9 186.4	3,876.5 3,925.1 3,992.0	3,292.5 3,335.8 3,396.3	584.0 589.3 595.7
Q4	1,425.5	687.6	737.8	70.3	138.8	4,370.9	79.5	48.6	23.8	207.9	4,059.8	3,461.9	597.9
2005 Q1 Q2	1,488.9 1,557.6	715.0 745.6	773.9 812.0	70.3 90.4	142.4 144.6	4,512.3 4,626.7	90.0 92.7	58.2 63.8	23.7 23.9	220.3 223.7	4,178.3 4,286.4	3,564.1 3,674.2	614.2 612.2
						Transa	actions						
2004 Q1 Q2 Q3 Q4	13.5 2.4 10.5 10.7	-3.8 -3.2 5.2 3.4	17.3 5.6 5.4 7.3	-0.9 -0.5 -0.6 5.3	3.7 1.2 2.2 1.2	79.1 60.1 63.3 43.4	9.2 4.2 1.7 -10.9	10.8 7.0 -1.1 -3.6	0.6 0.6 -1.2 0.5	0.8 0.1 2.1 0.1	68.5 55.3 60.7 53.7	55.2 50.0 55.2 50.3	13.3 5.2 5.5 3.3
2005 Q1 Q2	26.5 21.3	6.4 1.9	20.1 19.3	0.1 6.8	3.5 2.8	84.0 63.7	9.6 2.7	8.6 5.5	0.4 0.1	0.0 0.5	74.0 60.3	61.2 56.0	12.8 4.3
						Growt	h rates						
2004 Q1 Q2 Q3 Q4	6.0 3.8 3.9 2.8	3.3 1.0 1.1 0.3	8.4 6.5 6.6 5.3	6.7 -2.0 3.7 5.0	0.1 -0.5 -0.1 6.3	6.7 6.1 6.1 6.1	8.5 3.3 6.1 5.6	8.5 18.8 17.5 36.9	27.1 25.7 12.4 2.3	8.3 3.5 4.7 1.6	6.5 6.2 6.2 6.3	6.8 6.5 6.5 6.6	4.9 4.5 4.4 4.8
2005 Q1 Q2	3.7 5.0	1.8 2.6	5.4 7.3	6.5 17.6	6.0 7.1	6.0 6.0	5.4 3.5	23.7 17.6	1.3 -0.8	1.2 1.4	6.3 6.3	6.6 6.7	4.6 4.4

. .

Source: ECB. 1) Excluding unquoted shares.



# 3.4 Annual saving, investment and financing (EUR billions, unless otherwise indicated)

### 1. All sectors in the euro area

		Net acquisit	tion of non-fina	ncial assets		Net acquisition of financial assets							
	Total	Gross fixed capital formation	Consumption of fixed capital (-)	Changes in inven- tories <sup>1)</sup>	Non- produced assets	Total	Monetary gold and SDRs	Currency and deposits	Securities other than shares <sup>2)</sup>	Loans	Shares and other equity	Insurance technical reserves	Other investment (net) <sup>3)</sup>
	1	2	3	4	5	6	7	8	9	10	11	12	13
1998	403.1	1,203.4	-823.6	23.2	0.3	2,812.4	10.5	479.9	487.6	516.4	1,050.4	219.6	48.0
1999	444.7	1,293.4	-863.7	14.8	0.2	3,360.5	-0.1	564.8	550.4	797.6	1,155.7	264.3	27.8
2000	492.4	1,396.5	-913.1	17.3	-8.2	3,341.8	-2.2	361.6	343.3	780.7	1,549.6	252.7	56.0
2001	461.8	1,452.1	-973.6	-18.8	2.1	2,893.7	1.7	588.0	574.1	694.5	809.5	257.0	-31.0
2002	407.2	1,442.1	-1,004.8	-31.3	1.1	2,591.5	-0.1	801.9	384.6	521.9	615.5	228.5	39.3
2003	431.5	1,471.3	-1,033.2	-7.1	0.5	2,835.6	-1.5	729.1	584.7	634.5	628.6	241.8	18.3
2004	492.0	1,538.9	-1,069.5	23.0	-0.5	3,087.4	-2.1	962.5	609.2	697.8	543.5	260.3	16.3

		Changes in n	et worth 4)		Net incurrence of liabilities							
	Total	Gross saving	Consumption of fixed capital (-)	Net capital transfers receivable	Total	Currency and deposits	Securities other than shares <sup>2)</sup>	Loans	Shares and other equity	Insurance technical reserves		
	14	15	16	17	18	19	20	21	22	23		
1998	497.3	1,299.1	-823.6	21.9	2,718.6	670.8	376.3	514.6	933.3	224.6		
1999	509.8	1,352.0	-863.7	21.5	3,295.9	836.9	557.3	760.8	874.1	267.6		
2000	527.7	1,419.4	-913.1	21.4	3,307.1	502.7	466.3	874.1	1,205.8	257.9		
2001	496.4	1,449.4	-973.6	20.6	2,859.7	616.4	493.8	651.1	822.0	263.2		
2002	496.2	1,480.9	-1,004.8	20.1	2,502.8	634.5	450.5	541.0	638.7	232.1		
2003	483.9	1,486.1	-1,033.2	31.1	2,783.4	676.7	574.0	590.9	690.2	251.4		
2004	550.0	1,592.2	-1,069.5	27.2	3,029.9	1,045.9	638.0	525.7	562.0	262.2		

## 2. Non-financial corporations

	Net acquisit	ion of non-fin	ancial assets						Changes in	net worth <sup>4)</sup>	Net incurrence of liabilities			ies
	Total			Total					Total		Total			
		Gross fixed capital	Consumption of fixed		Currency and	Securities other than	Loans	Shares and other		Gross saving		Securities other than	Loans	Shares and other
		formation	capital (-)		deposits	shares 2)		equity				shares 2)		equity
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
1998	184.5	632.5	-468.3	464.7	45.6	16.2	119.3	231.6	145.0	563.1	504.2	13.1	274.5	206.0
1999	207.6	684.0	-489.4	670.8	23.6	80.3	186.3	348.0	108.4	546.5	770.0	46.8	429.1	282.9
2000	310.7	756.1	-522.1	971.7	73.7	68.7	245.2	546.1	83.3	556.7	1,199.1	66.9	615.5	505.0
2001	214.8	784.8	-558.4	671.9	108.4	45.2	185.3	241.1	87.1	585.7	799.5	101.5	382.4	304.1
2002	151.7	765.0	-581.5	443.3	25.1	-15.7	66.5	253.8	90.1	614.6	504.9	18.3	260.2	213.9
2003	150.9	760.0	-598.4	449.5	89.7	-26.3	148.9	206.5	74.6	626.2	525.8	77.9	209.5	224.6
2004	180.9	771.5	-610.1	323.8	85.8	-32.7	88.4	167.1	134.5	702.9	370.2	21.9	157.9	181.5

# 3. Households <sup>5)</sup>

	Net acquisiti	on of non-fir	nancial assets	•					Changes in net worth <sup>4)</sup>		Net incurrence of liabilities		Memo:	
	Total			Total					Total		Total		Disposable	Gross
		Gross fixed	Consumption		Currency	Securities	Shares	Insurance		Gross		Loans	income	saving
		capital	of fixed		and	other than	and other	technical		saving				ratio 6)
		formation	capital (-)		deposits	shares 2)	equity	reserves						
		2	2	4	e	(	7	0	0	10	11	12	12	14
	1	2	3	4	5	6	/	8	9	10	11	12	13	14
1998	178.8	392.2	-217.2	462.7	93.4	-130.2	277.4	211.9	428.2	604.9	213.7	212.3	3,971.6	15.2
1999	190.3	419.8	-231.3	489.8	122.5	-30.1	201.2	249.7	412.3	587.6	268.2	266.5	4,116.9	14.3
2000	200.4	439.3	-240.3	441.0	67.0	45.3	124.7	246.9	418.9	608.4	223.1	221.1	4,337.4	14.0
2001	187.9	449.7	-257.8	431.1	178.7	92.4	48.8	236.7	440.8	652.6	178.9	177.2	4,630.2	14.1
2002	201.1	461.1	-260.7	483.5	223.0	71.5	5.8	218.5	472.2	695.0	212.8	210.6	4,789.7	14.5
2003	217.8	483.6	-268.2	537.1	207.8	13.4	90.7	240.8	507.0	737.2	248.1	245.9	4,953.9	14.9
2004	245.7	530.5	-287.1	564.3	227.8	76.3	19.3	248.7	522.0	751.8	288.2	285.8	5,112.5	14.7

Source: ECB.

Source: ECB.
Including net acquisition of valuables.
Excluding financial derivatives.
Financial derivatives, other accounts receivable/payable and statistical discrepancies.
Arising from saving and net capital transfers receivable, after allowance for consumption of fixed capital (-).
Including non-profit institutions serving households.
Gross saving as a percentage of disposable income.



# FINANCIAL MARKETS

# Securities, other than shares, by original maturity, residency of the issuer and currency (EUR billions and period growth rates; seasonally adjusted; transactions during the month and end-of-period outstanding amounts; nominal values)

	Total in euro <sup>1)</sup>			By euro area residents								
					In euro		In all currencies					
	Outstanding amounts	Gross issues	Net issues	Outstanding amounts	Gross issues	Net issues	Outstanding amounts	Gross issues	Net issues	Annual growth rates	Seasonally a	djusted <sup>2)</sup>
										8	Net issues	6-month
	1	2	3	4	5	6	7	8	9	10	11	12
	Total											
2004 Oct.	9,937.1	712.2	21.7	8,535.7	652.7	32.5	9,385.3	696.7	42.7	7.0	41.2	7.0
Nov. Dec.	10,028.3 10,032.5	703.7 706.6	84.8 4.8	8,597.6 8,575.8	635.9 648.1	56.6 -22.0	9,441.8 9,414.6	674.0 680.4	59.5 -23.2	7.1 7.6	62.8 76.5	6.9 6.9
2005 Jan.	10,097.9	763.3	52.8	8,653.6	712.5	74.3	9,527.1	759.6	90.4	7.5	52.5	7.2
Feb.	10,217.8	705.5	116.7	8,760.9	726.6	106.7	9.640.7	769.2	116.8	7.8	81.4	7.9
Mar.	10,326.1	798.6	107.3	8,806.4	704.4	44.3	9,709.1	750.4	54.5	7.4	35.9	7.6
Apr.	10,384.1	835.5	54.6	8,893.1	788.4	86.7	9,820.3	834.9	102.9	7.8	87.8	8.6
May	10,449.3	873.1	65.7	8,953.7	818.2	61.6	9,910.8	857.9	66.2	7.5	36.5	8.0
June	10,640.5	996.6	189.3	9,076.6	872.9	123.4	10,049.3	922.6	133.1	8.1	136.3	9.2
July	10,612.8	788.1	-28.3	9,079.6	737.3	2.8	10,055.9	778.6	5.8	7.6	2.9	8.1
Aug.	10,619.4	729.4	0.9	9,071.4	676.6	-11.7	10,056.6	715.8	-3.5	7.4	34.4	7.0
Sep.	10,714.8	860.3	96.0	9,102.5	751.9	32.7	10,105.8	793.1 757.2	41.6 52.9	7.3 7.4	40.9	7.1 6.2
Oct.		•	•	9,139.2	711.3	36.6	10,160.0	/5/.2	52.9	/.4	51.5	6.2
	Long-term											
2004 Oct.	9,011.1	174.0	33.8	7,694.1	139.2	17.7	8,440.6	158.0	28.2	7.4	36.6	7.4
Nov.	9,087.0	168.7	70.1	7,749.8	138.3	51.9	8,494.8	155.1	61.3	7.6	65.4	7.3
Dec.	9,112.1	148.5	24.8	7,766.6	124.6	16.2	8,502.5	136.0	16.2	7.8	62.5	7.3
2005 Jan.	9,181.9	198.5	59.8	7,827.9	174.9	57.1	8,589.9	195.0	67.7	8.0	68.3	8.0
Feb.	9,297.4	220.5	113.0	7,924.9	180.2	96.8	8,692.8	199.4	105.4	8.2	75.6	8.8
Mar.	9,372.3	203.0	74.3	7,975.2	163.4	50.1	8,759.9	183.8	58.0	8.2	47.9	8.7
Apr.	9,427.0	182.9	53.1	8,035.8	163.8	60.7	8,839.7	181.7	72.1	8.4	67.9	9.4
May	9,497.5	180.5	69.0	8,097.2	151.1	61.6	8,928.1	166.5	67.9	8.0	41.0	8.7
June	9,678.9	299.2	177.7	8,241.9	232.1	145.0	9,092.4	254.6	155.1	8.9	142.7	10.6
July	9,673.5 9,671.4	154.2 85.4	-5.8	8,235.1	130.4 62.2	-6.9	9,089.0 9,085.6	145.1 75.7	-2.3 -10.1	8.4 8.1	-2.6 21.1	8.8 7.4
Aug.	9,671.4	85.4 183.7	-5.1 65.3	8,221.1 8,259.3	137.3	-17.6 38.8	9,085.6	156.8	-10.1 50.2	8.1 7.9	43.1	7.4
Sep. Oct.	9,730.1	185.7	03.5	8,239.3	137.5	38.8 19.0	9,140.8	150.8	30.2	8.0	43.1	6.6
		•	•	0,277.0	150.5	19.0	,1,5.5	157.9	51.8	0.0	71.9	0.0

Total outstanding amounts and gross issues of securities, other than shares, issued by euro area residents C13

total gross issues (right-hand scale) . . . . total outstanding amounts (left-hand scale) outstanding amounts in euro (left-hand scale) WA 

Sources: ECB and BIS (for issues by non-euro area residents). 1) Total euro-denominated securities, other than shares, issued by euro area residents and non-euro area residents.

2) For the calculation of the growth rates, see the Technical notes. The 6-month growth rates have been annualised.



# 4.2 Securities, other than shares, issued by euro area residents, by sector of the issuer and instrument type (EUR billions ; transactions during the month and end-of-period outstanding amounts; nominal values)

	Outstanding amounts							Gross issues						
	Total	MFIs	Non-MFI corporations		General government		Total	MFIs (including	Non-MFI corporations		General government			
		(including Eurosystem)	Non-monetary financial corporations	Non-financial corporations	Central government	Other general government		Eurosystem)	Non-monetary financial corporations	Non-financial corporations	Central government	Other general government		
	1	2	3	4	5	6	7	8	9	10	11	12		
2002	0.751	2 2 5 2		501	2.022	Total	7 200	4 407	245	912	1 470	07		
2003 2004	8,751 9,415	3,353 3,714	665 735	591 595	3,923 4,120	219 250	7,209 8,049	4,487 5,252	245 222	1,028	1,479 1,464	87 83		
2004 Q4 2005 Q1	9,415 9,709	3,714 3,850	735 755	595 608	4,120 4,238	250 259	2,051 2,279	1,434 1,544	84 50	228 248	284 412	20 25		
Q2	10,049	3,994	830	619	4,341	266	2,615	1,803	109	281	399	25 23		
Q3 2005 July	10,106 10,056	4,045	837	617	4,337	270 268	2,287 779	1,649	44	251	323	21		
Aug.	10,057 10,106	4,031 4,045	826 837	619	4,313	268 270	716 793	546 553	8 22	81	77 126	7 4 9		
Sep. Oct.	10,100	4,043	852	617 627	4,337 4,321	270	793	531	22 26	82 85	120	9 7		
Short-term														
2003 2004	861 912	390 447	6 7	94 90	367 362	3 5	5,333 6,147	3,698 4,383	41 44	796 931	768 756	29 33		
2004 Q4	912	447	7	90	362	5	1,602	1,223	12	205	155	79		
2005 Q1 Q2	949 957	457 462	8 7	105 105	374 377	5 5	1,701 2,013	1,262 1,558	12 11	229 258	188 178	9 8 9		
Q3	965	475	7	99	379	5	1,910	1,480	12	234	175			
2005 July Aug.	967 971	475 481	7 7	103 102	376 376	5 5	634 640	490 501	4 4	81 77	56 56	3 3 3 3		
Sep. Oct.	965 985	475 491	7 7	99 102	379 380	5 5	636 599	490 457	4 4	76 75	63 61	3		
						Long-term <sup>1)</sup>								
2003 2004	7,890 8,502	2,963 3,266	659 728	497 505	3,556 3,758	216 245	1,876 1,902	788 869	203 179	115 97	711 708	58 49		
2004 Q4 2005 Q1	8,502 8,760	3,266 3,393	728 747	505 503	3,758 3,863	245 254	449 578	211 281	72 37	24 19	129 224	13 16		
Q2	9,092	3,531	822	514	3,965	260	603	245	98	24	221	14		
Q3 2005 July	9,141 9,089	3,570 3,540	830 822	518	3,957	265 263	378 145	168 60	33	17	148 63	12		
Aug.	9,086	3,550	819	517	3,937	262	76	45	4	4	22	1		
Sep. Oct.	9,141 9,175	3,570 3,597	830 845	518 525	3,957 3,941	265 268	157 158	63 74	19 23	6 10	63 47	6 4		
					Of whi	ch long-term fi	xed rate							
2003 2004	6,118 6,380	1,885 1,929	406 416	422 414	3,240 3,436	165 186	1,287 1,193	415 408	114 69	91 61	626 620	41 36		
2004 Q4 2005 Q1	6,380 6,517	1,929 1,968	416 426	414 409	3,436 3,517	186 196	259 387	93 137	26 21	15 15	117 199	9 15		
Q2 Q3	6,675 6,674	2,004	445	417 415	3,607	203	343	101	28	15	187	12		
2005 July	6,651	2,015	436	415	3,601	207 205	235 93	80	8	8	133	8		
Aug. Sep.	6,650 6,674	2,003 2,015	437 436	416 415	3,589 3,601	205 207	46 97	24	23	2	17 56	1		
Oct.	6,690	2,015 2,031	430	415	3,592	207	104	32 43	8	7	43	3		
	Of which long-term variable rate													
2003 2004	1,580 1,869	959 1,149	249 309	59 77	262 275	51 59	508 619	337 404	90 109	12 32	53 60	16 14		
2004 Q4 2005 Q1	1,869 1,957	1,149 1,212	309 317	77 79	275 291	59 58	175 167	105 129	47 16	7 3	12 17	42		
Q2 Q3	1,957 2,115 2,157	1,212 1,292 1,309	374 390	83 86	308 313	57 58	236 118	128 75	70 25	7	28 8	4 2 2 5		
2005 July	2,129	1,302	379	84	307	57	44	31	7	2	2	1		
Aug. Sep.	2,131 2,157	1,301 1,309	379 390	85 86	309 313	57 58	23 51	17 27	2 15	1 3	2 4	0 3		
Oct.	2,167	1,314	402	88	304	58	46	26	14	2	3	1		

## 1. Outstanding amounts and gross issues

Source: ECB.
1) The residual difference between total long-term debt securities and fixed and variable rate long-term debt securities consists of zero coupon bonds and revaluation effects.


#### 4.2 Securities, other than shares, issued by euro area residents, by sector of the issuer and instrument type

(EUR billions unless otherwise indicated; transactions during the period; nominal values)

2. Net iss	ues											
			Non-season	ally adjusted					Seasonally	y adjusted		
	Total	MFIs (including		-	General go	overnment	Total	MFIs (including			General go	overnment
		Eurosystem)	Non-monetary financial corporations		Central government	Other general government		Eurosystem)	Non-monetary financial corporations		Central government	Other general government
	1	2	3	4	5	6	7	8	9	10	11	12
						Total						
2003 2004	602.0 663.0	213.5 350.9	118.1 74.3	53.7 8.8	176.7 197.6	39.9 31.5	607.9 667.0	214.9 353.6	115.6 72.4	53.7 8.5	183.4 200.8	40.2 31.7
2004 Q4 2005 Q1	79.0 261.7	72.3 114.5	50.5 13.2	-4.5 13.7	-47.6 111.3	8.4 9.0	180.5 169.8	100.6 72.1	36.5 29.3	0.9 10.2	34.6 50.0	7.9 8.3
Q2 Q3	302.2 43.9	118.4 37.4	71.0 7.8	8.7 -0.9	98.1 -4.8	6.0 4.5	260.6 78.2	122.7 48.9	64.0 13.8	4.9 1.0	63.5 8.3	5.5 6.3
2005 July Aug.	5.8 -3.5	19.8 10.6	0.1 -3.2	-0.3 0.7	-16.3 -11.1	2.6 -0.5	2.9 34.4	10.4 28.6	-7.0 7.1	-4.9 2.7	1.1 -4.9	3.3 0.8
Sep. Oct.	41.6 52.9	7.0 41.3	11.0 15.2	-1.3 9.3	22.6 -15.4	2.3 2.6	40.9 51.5	9.9 31.7	13.7 18.2	3.1 7.2	12.1 -6.6	2.2 1.1
						Long-term						
2003 2004	546.3 615.7	202.4 298.2	118.8 72.9	51.2 12.4	133.3 202.3	40.5 29.8	547.4 618.8	203.1 299.2	116.4 71.1	51.2 12.3	135.9 206.1	40.8 30.1
2004 Q4 2005 Q1 Q2	105.7 231.1 295.2	47.6 111.5 113.1	49.2 12.8 71.3	5.2 -1.2 8.6	-5.4 99.3 96.0	9.1 8.7 6.1	164.5 191.8 251.7	74.4 85.3 111.2	35.3 29.1 63.9	4.0 3.1 3.4	41.8 66.8 67.3	8.9 7.5 5.8
Q2 Q3	37.8	26.9	8.1	5.9	-7.6	4.5	61.6	27.9	14.2	7.8	5.5	6.2
2005 July Aug. Sep.	-2.3 -10.1 50.2 31.8	9.0 2.0 15.9 23.6	0.1 -2.6 10.6 15.7	1.9 1.6 2.4 6.0	-15.9 -10.7 19.0 -16.3	2.7 -0.5 2.3 2.7	-2.6 21.1 43.1 41.9	1.8 13.0 13.1 23.5	-6.7 7.9 13.0 18.7	-0.9 4.1 4.6 5.2	-0.6 -4.4 10.5 -6.7	3.8 0.5 2.0 1.2
Oct.			15.7	0.0	-10.3	2.7	41.9	23.5	18./	5.2	-0./	1.2

C14 Net issues of securities, other than shares, seasonally adjusted and non-seasonally adjusted (EUR billions: transactions during the month: nominal values)





(per	centage chan	(203)										
		Annual	growth rates (r	on-seasonally	adjusted)			6-mon	th seasonally a	djusted growt	h rates	
	Total	MFIs (including	Non-MFI co	orporations	General g	overnment	Total	MFIs (including	Non-MFI co	orporations	General go	overnment
		Eurosystem)	Non-monetary financial corporations	Non-financial corporations	Central government	Other general government		Eurosystem)	Non-monetary financial corporations		Central government	Other general government
	1	2	3	4	5	6	7	8	9	10	11	12
						Total						
2004 Oct.	7.0	9.8	9.1	3.3	4.5	14.2	7.0	9.4	12.1	4.8	4.3	9.9
Nov.	7.1	9.6	10.3	3.3	4.7	14.5	6.9	8.8	17.0	4.1	3.8	13.0
Dec.	7.6	10.4	11.2	1.5	5.0	14.4	6.9	10.4	13.3	1.0	3.6	10.7
2005 Jan.	7.5	9.8	11.0	2.7	5.1	15.1	7.2	9.4	15.2	0.5	4.4	16.1
Feb.	7.8	10.5	11.1	3.0	5.3	13.0	7.9	10.2	15.9	2.7	4.9	15.6
Mar.	7.4	9.7	14.4	4.4	4.6	11.9	7.6	9.7	19.9	3.7	4.1	13.7
Apr.	7.8	9.9	15.9	5.4	4.7	12.7	8.6	10.4	19.6	6.1	5.3	15.6
May	7.5	9.3	18.4	4.8	4.2	11.7	8.0	9.8	19.8	5.4	4.6	10.5
June	8.1	10.5	20.1	3.1	4.6	11.1	9.2	10.6	27.1	5.1	5.5	11.3
July	7.6	10.0	18.8	1.5	4.3	12.7	8.1	10.7	22.5	2.5	4.1	9.4
Aug.	7.4	10.0	18.9	2.2	3.7	11.8	7.0	9.8	22.2	1.7	2.7	8.3
Sep.	7.3	9.4	20.6	2.8	3.8	11.5	7.1	9.1	21.5	1.9	3.4	9.3
Oct.	7.4	9.4	21.0	3.9	3.6	11.8	6.2	8.4	22.6	1.9	2.0	8.2
						Long-term						
2004 Oct.	7.4	10.1	9.2	4.4	4.9	13.5	7.4	9.4	12.0	9.4	4.6	9.2
Nov.	7.6	9.9	10.4	3.9	5.2	13.5	7.3	8.5	16.8	7.9	4.2	12.1
Dec.	7.8	10.0	11.1	2.5	5.7	13.8	7.3	9.7	12.8	6.1	4.2	10.9
2005 Jan.	8.0	10.1	10.9	2.9	6.1	14.8	8.0	9.8	14.7	2.8	5.5	16.0
Feb.	8.2	10.5	10.9	2.0	6.4	12.6	8.8	10.8	15.4	2.7	6.3	15.6
Mar.	8.2	10.0	14.2	4.6	5.8	11.5	8.7	10.2	19.7	2.9	5.9	14.3
Apr.	8.4	9.7	15.6	6.2	6.0	12.6	9.4	10.0	19.2	3.1	7.5	16.2
May	8.0	9.1	18.3	4.9	5.4	11.7	8.7	9.8	19.6	2.0	6.6	11.3
June	8.9	11.0	19.9	4.4	5.7	11.1	10.6	12.2	27.4	2.6	7.2	11.1
July	8.4	10.3	18.6	2.7	5.4	13.0	8.8	10.8	22.7	2.7	5.2	10.0
Aug.	8.1	10.0	19.0	3.5	4.8	12.1	7.4	9.2	22.8	4.3	3.4	8.8
Sep.	7.9	9.2	20.7	3.7	4.8	12.0	7.3	8.3	21.8	4.5	3.8	9.7
Oct.	8.0	9.2	21.1	4.2	4.7	12.2	6.6	8.3	23.1	5.3	2.0	8.4

### 4.3 Growth rates of securities, other than shares, issued by euro area residents 1)

C15 Annual growth rates of long-term debt securities, by sector of the issuer, in all currencies combined



Source: ECB. 1) For the calculation of the growth rates, see the Technical notes. The 6-month growth rates have been annualised.



### 4.3 Growth rates of securities, other than shares, issued by euro area residents $^{(1)}$ (cont'd)

			Long-tern	n fixed rate					Long-term	variable rate		
	Total	MFIs (including		orporations	General g	overnment	Total	MFIs (including	Non-MFI c	orporations	General go	overnment
		Eurosystem)	Non-monetary financial corporations		Central government	Other general government		Eurosystem)	Non-monetary financial corporations		Central government	Other general government
	13	14	15	16	17	18	19	20	21	22	23	24
					In all	currencies con	nbined					
2003	5.2	2.2	15.3	12.2	4.4	22.6	8.5	8.3	51.2	-13.4	-9.2	43.2
2004	5.1	3.1	6.4	3.4	5.8	14.7	16.3	18.5	27.2	8.5	0.6	26.4
2004 Q4	4.3	2.4	2.6	0.8	5.7	12.3	17.8	20.2	22.6	29.7	2.3	18.7
2005 Q1	4.7	2.8	3.9	-1.2	6.3	13.7	18.3	19.4	23.8	28.4	7.7	12.4
Q2	4.8	2.5	6.1	1.2	5.8	14.8	19.4	18.9	34.8	27.2	8.5	3.0
Q3	4.5	3.0	6.6	0.5	5.0	15.7	20.6	19.6	38.4	18.2	11.2	1.3
2005 May	4.6	2.2	6.2	0.9	5.6	15.1	18.9	17.7	38.0	26.8	7.7	1.5
June	4.8	3.1	7.5	0.7	5.5	14.7	22.0	21.5	39.0	26.9	10.8	0.0
July	4.6	3.0	6.3	0.2	5.2	16.5	20.6	20.4	37.1	15.7	10.1	1.8
Aug.	4.5	3.1	6.4	0.9	5.0	15.8	19.9	19.1	37.9	16.5	10.1	0.6
Sep.	4.2	3.0	6.7	0.3	4.5	14.7	20.5	17.3	41.4	18.9	16.2	3.2
Oct.	4.7	4.1	7.0	1.0	4.7	15.4	18.4	15.2	41.6	18.4	11.3	2.1
						In euro						
2003	4.6	0.1	20.5	12.5	4.2	21.3	8.6	7.6	51.2	-8.9	-9.3	43.7
2004	4.8	1.3	10.4	2.0	5.9	14.7	15.7	17.8	27.3	8.8	0.5	25.3
2004 Q4	4.0	0.3	6.3	-0.9	5.8	12.5	17.2	19.7	22.8	27.3	2.3	18.1
2005 Q1	4.3	0.5	7.8	-2.5	6.2	13.7	17.6	18.2	24.0	27.5	7.8	12.9
Q2	4.4	0.3	10.1	0.8	5.8	15.1	18.9	18.0	34.9	25.3	8.9	3.7
Q3	4.2	1.0	10.2	0.3	4.9	16.1	20.3	18.8	37.9	19.1	11.8	2.2
2005 May	4.3	0.1	10.3	0.4	5.6	15.5	18.6	17.0	38.1	24.5	8.1	2.6
June	4.5	0.9	11.6	0.5	5.4	15.2	21.9	21.2	38.8	24.8	11.4	1.0
July	4.2	0.9	10.0	-0.2	5.0	17.1	20.5	19.8	36.9	16.9	10.6	2.7
Aug.	4.1	1.0	9.9	0.7	4.8	16.3	19.7	18.3	37.7	17.9	10.6	1.4
Sep.	3.8	1.1	9.9	0.1	4.3	14.9	19.8	15.9	39.6	20.4	16.9	4.0
Oct.	4.3	2.1	9.7	1.3	4.6	15.7	17.7	14.0	39.1	19.9	11.8	2.2

C16 Annual growth rates of short-term debt securities, by sector of the issuer, in all currencies combined

. . .





Source: ECB. 1) For the calculation of the growth rates, see the Technical notes.



#### 1. Outstanding amounts and annual growth rates

(outstanding amounts as end-of-period)

		Total		MF	Is	Non-monetary finance	ial corporations	Non-financial c	orporations
	Total	Index Dec. 01 = 100	Annual growth rates (%)	Total	Annual growth rates (%)	Total	Annual growth rates (%)	Total	Annual growth rates (%)
	1	2	3	4	5	6	7	8	9
2003 Oct.	3,484.0	101.2	1.0	535.2	1.0	335.9	2.0	2,612.9	0.9
Nov.	3,546.9	101.3	1.0	549.5	1.6	340.0	3.0	2,657.4	0.6
Dec.	3,647.4	101.4	1.1	569.5	1.7	351.0	2.8	2,726.9	0.8
2004 Jan.	3,788.6	101.5	1.2	584.1	1.7	375.1	3.0	2,829.4	0.8
Feb.	3,852.1	101.5	1.2	587.9	2.0	377.1	3.2	2,887.1	0.8
Mar.	3,766.5	101.8	1.5	571.9	2.2	357.7	3.1	2,836.9	1.2
Apr.	3,748.5	102.0	1.0	579.4	2.3	363.7	1.3	2,805.4	0.7
May	3,687.9	101.9	1.0	568.1	2.4	353.0	1.3	2,766.8	0.7
June	3,790.1	102.0	1.0	582.5	2.7	364.4	1.4	2,843.2	0.6
July	3,679.8	102.1	0.9	562.3	1.8	356.2	1.9	2,761.3	0.6
Aug.	3,621.2	102.0	0.9	562.5	1.4	355.3	1.6	2,703.4	0.6
Sep.	3,707.9	102.1	0.9	579.6	1.3	364.2	2.1	2,764.1	0.7
Oct.	3,787.6	102.2	0.9	598.0	1.2	374.6	2.0	2,815.0	0.7
Nov.	3,906.5	102.5	1.2	623.9	2.8	388.6	0.9	2,894.1	0.9
Dec.	4,033.8	102.6	1.2	643.7	2.9	407.7	1.1	2,982.4	0.8
2005 Jan.	4,138.0	102.6	1.1	662.6	2.9	414.2	0.9	3,061.3	0.8
Feb.	4,254.5	102.7	1.1	681.1	2.6	434.1	1.0	3,139.2	0.8
Mar.	4,242.4	102.7	0.9	677.7	2.3	424.0	1.0	3,140.7	0.6
Apr.	4,094.7	102.9	0.9	656.0	2.1	403.9	0.9	3,034.8	0.7
May	4,272.6	103.0	1.0	678.1	2.1	417.1	0.9	3,177.4	0.8
June	4,381.7	103.1	1.1	698.0	2.4	434.0	1.6	3,249.7	0.8
July	4,631.7	105.1	3.0	727.9	2.3	460.1	1.1	3,443.8	3.3
Aug.	4,606.4	105.1	3.0	723.4	3.0	450.3	1.1	3,432.7	3.3
Sep.	4,818.7	105.2	3.1	764.1	3.2	475.8	1.3	3,578.8	3.3
Oct.	4,651.1	105.3	3.1	752.4	3.2	473.0	1.3	3,425.7	3.3

### C17 Annual growth rates for quoted shares issued by euro area residents



Source: ECB.
 For the calculation of the index and the growth rates, see the Technical notes.



### 4.4 Quoted shares issued by euro area residents <sup>1)</sup> (EUR billions; market values)

#### 2. Transactions during the month

	Total				Non-monetary financial corporations		orporations	-		ations		
	Gross issues	Redemptions	Net issues	Gross issues	Redemptions	Net issues	Gross issues	Redemptions	Net issues	Gross issues	Redemptions	Net issues
	1	2	3	4	5	6	7	8	9	10	11	12
2003 Oct.	5.5	3.9	1.6	0.4	0.0	0.4	0.2	0.0	0.1	4.9	3.9	1.0
Nov.	7.5	5.6	1.9	2.7	0.0	2.7	4.2	0.3	3.9	0.6	5.3	-4.6
Dec.	5.7	1.4	4.2	0.8	0.1	0.8	0.4	0.9	-0.5	4.5	0.5	4.0
2004 Jan.	2.9	1.0	1.8	0.1	0.0	0.1	0.9	0.0	0.9	1.8	1.0	0.8
Feb.	3.5	0.6	2.9	2.0	0.0	2.0	0.0	0.2	-0.2	1.4	0.3	1.1
Mar.	12.1	1.5	10.6	1.5	0.0	1.5	0.0	0.1	-0.1	10.6	1.3	9.3
Apr.	6.6	0.7	5.8	3.1	0.1	3.1	0.6	0.1	0.5	2.9	0.6	2.3
May	3.3	4.2	-0.9	0.3	0.0	0.3	0.0	0.0	0.0	2.9	4.2	-1.2
June	3.9	2.2	1.6	0.7	1.6	-0.9	0.3	0.0	0.2	2.9	0.6	2.3
July	6.4	3.9	2.5	0.4	0.0	0.4	2.2	0.0	2.2	3.8	3.9	0.0
Aug.	2.0	2.9	-0.9	0.1	2.2	-2.1	0.0	0.0	0.0	1.9	0.7	1.1
Sep.	5.0	2.3	2.7	0.1	0.9	-0.8	0.0	0.0	0.0	4.8	1.4	3.4
Oct.	3.5	0.8	2.8	0.1	0.0	0.1	0.0	0.0	0.0	3.5	0.8	2.7
Nov.	15.2	3.5	11.8 3.2	12.8	0.3	12.5	0.1	0.0	0.1	2.4 3.9	3.1 2.2	-0.8
Dec.	5.5	2.3		1.2	0.0	1.2	0.4	0.1	0.3			1.7
2005 Jan.	1.1	1.5	-0.4	0.1	0.0	0.1	0.2	0.0	0.2	0.8	1.5	-0.7
Feb.	4.0	0.7	3.4	0.1	0.0	0.1	0.2	0.1	0.1	3.8	0.6	3.2
Mar.	5.0	2.0	3.0	0.9	0.8	0.1	0.1	0.1	0.0	3.9	1.1	2.9
Apr.	10.3	2.3	8.1	2.5	0.0	2.5	0.2	0.0	0.2	7.6	2.2	5.3
May	3.7	2.5	1.2	0.0	0.0	0.0	0.0	0.2	-0.2	3.6	2.2	1.4
June	12.1	5.4	6.7	1.9	1.0	0.9	4.1	0.7	3.3	6.1	3.6	2.5
July	90.5	7.3	83.2	2.4	2.9	-0.4	0.5	0.0	0.5	87.5	4.4	83.1
Aug.	2.8	1.9	0.9	2.5	0.0	2.5	0.0	0.2	-0.1	0.4	1.8	-1.4
Sep.	8.2	1.8	6.5	0.4	0.0	0.4	1.1	0.0	1.1	6.7	1.7	5.0
Oct.	5.7	0.8	4.9	0.0	0.1	-0.1	0.1	0.0	0.1	5.6	0.7	4.9

#### C18 Gross issues of quoted shares by sector of the issuer (EUR billions; transactions during the month; market values)



#### Source: ECB.

1) For the calculation of the index and the growth rates, see the Technical notes.



#### 1. Interest rates on deposits (new business)

			Deposits fr	om household	8		Depos	ations	Repos		
	Overnight 1)	Wi	th agreed matur	ity	Redeemable a	t notice 1), 2)	Overnight 1)	Wit	h agreed matur	ity	
		Up to 1 year	Over 1 and up to 2 years	Over 2 years	Up to 3 months	Over 3 months		Up to 1 year	Over 1 and up to 2 years	Over 2 years	
	1	2	3	4	5	6	7	8	9	10	11
2004 Nov.	0.73	1.94	2.20	2.50	2.01	2.51	0.90	2.04	2.23	3.39	2.02
Dec.	0.73	1.95	2.19	2.31	2.00	2.52	0.90	2.08	2.70	3.51	2.02
2005 Jan.	0.74	1.95	2.29	2.54	1.98	2.49	0.93	2.04	2.25	3.26	2.05
Feb.	0.74	1.95	2.19	2.33	1.97	2.49	0.93	2.03	2.25	3.47	2.03
Mar.	0.74	1.93	2.16	2.40	1.96	2.47	0.94	2.00	2.35	3.15	1.99
Apr.	0.74	2.01	2.09	2.32	1.95	2.45	0.95	2.01	2.23	2.92	2.00
May	0.75	1.94	2.01	2.20	1.97	2.43	0.95	2.01	2.12	3.31	2.00
June	0.69	1.95	2.21	2.20	2.17	2.38	0.91	2.01	2.05	3.57	2.00
July	0.68	1.94	2.01	2.19	2.15	2.34	0.94	2.02	2.21	3.11	2.00
Aug.	0.69	1.95	2.07	2.32	2.03	2.31	0.96	2.02	2.22	2.90	2.01
Sep.	0.69	1.97	2.05	2.04	2.02	2.29	0.96	2.04	2.23	2.97	2.04
Oct.	0.69	1.98	2.28	2.16	1.96	2.27	0.97	2.04	2.58	3.44	2.02

#### 2. Interest rates on loans to households (new business)

	Bank overdrafts <sup>1)</sup>		Consumer	credit		Lending for house purchase					Other lending by initial rate fixation		
		By initi	al rate fixation	on	Annual percentage	I	By initial rate	e fixation		Annual percentage			
		Floating rate and up to 1 year	Over 1 and up to 5 years	Over 5 years	rate of charge <sup>3)</sup>	Floating rate and up to 1 year	Over 1 and up to 5 years	Over 5 and up to 10 years	Over 10 years	rate of charge <sup>3)</sup>	Floating rate and up to 1 year	Over 1 and up to 5 years	Over 5 years
	1	2	3	4	5	6	7	8	9	10	11	12	13
2004 Nov.	9.48	6.89	6.84	8.23	7.85	3.45	4.07	4.66	4.58	4.09	3.97	4.89	4.82
Dec.	9.53	6.73	6.60	7.67	7.59	3.43	3.95	4.49	4.41	4.07	3.82	4.59	4.65
2005 Jan.	9.60	6.97	6.83	8.33	8.01	3.44	3.97	4.43	4.45	4.07	3.96	4.64	4.62
Feb.	9.65	6.20	6.83	8.18	7.77	3.40	3.94	4.39	4.33	3.98	4.00	4.73	4.49
Mar.	9.60	6.62	6.72	8.12	7.83	3.40	3.89	4.35	4.27	3.97	3.84	4.60	4.57
Apr.	9.62	6.60	6.64	8.19	7.81	3.40	3.89	4.36	4.28	3.95	3.97	4.71	4.62
May	9.64	6.96	6.56	8.00	7.82	3.38	3.85	4.28	4.20	3.93	3.86	4.68	4.61
June	9.61	6.62	6.50	7.90	7.72	3.32	3.76	4.13	4.09	3.89	3.84	4.60	4.50
July	9.52	6.67	6.61	7.96	7.80	3.33	3.70	$4.06 \\ 4.00$	4.05	3.87	3.89	4.54	4.29
Aug.	9.58	6.99	6.70	8.10	7.99	3.32	3.72		3.99	3.89	3.80	4.59	4.41
Sep.	9.61	7.04	6.43	7.94	7.85	3.31	3.68	3.98	3.96	3.82	3.85	4.51	4.25
Oct.	9.64	6.82	6.36	7.99	7.75	3.33	3.67	3.99	3.95	3.82	3.89	4.50	4.28

#### 3. Interest rates on loans to non-financial corporations (new business)

	Bank overdrafts <sup>1)</sup>		is up to EUR 1 million itial rate fixation	1	Other loans over EUR 1 million by initial rate fixation			
		Floating rate and up to 1 year	Over 1 and up to 5 years	Over 5 years	Floating rate and up to 1 year	Over 1 and up to 5 years	Over 5 years	
	1	2	3	4	5	6	7	
2004 Nov.	5.37	4.02	4.79	4.55	2.95	3.41	4.31	
Dec.	5.26	3.97	4.67	4.46	3.05	3.55	4.10	
2005 Jan.	5.38	3.97	4.69	4.47	3.02	3.29	4.10	
Feb.	5.30	3.91	4.76	4.36	3.01	3.33	3.81	
Mar.	5.28	3.90	4.50	4.32	3.02	3.47	4.11	
Apr.	5.22	3.88	4.51	4.34	3.00	3.53	3.99	
May	5.14	3.91	4.45	4.24	2.99	3.60	3.80	
June	5.12	3.87	4.45	4.14	2.92	3.44	3.88	
July	5.12	3.86	4.40	4.11	2.96	3.57	3.77	
Aug.	5.04	3.91	4.45	4.13	2.87	3.52	3.81	
Sep.	5.14	3.81	4.36	4.03	2.90	3.39	3.87	
Oct.	5.10	3.88	4.43	4.01	2.88	3.58	3.80	

Source: ECB.

 For this instrument category, new business and outstanding amounts coincide. End-of-period.
 For this instrument category, households and non-financial corporations are merged and allocated to the household sector, since the outstanding amounts of non-financial corporations are negligible compared with those of the household sector in all participating Member States combined.
 The annual percentage rate of charge covers the total cost of a loan. The total cost comprises an interest rate component and a component of other (related) charges, such as the cost of inquiries, administration, preparation of documents, guarantees, etc.



#### 4.5 MFI interest rates on euro-denominated deposits and loans by euro area residents

#### 4. Interest rates on deposits (outstanding amounts)

		Depos	its from househ	olds		Deposits from	rporations	Repos	
	Overnight <sup>1)</sup>	With agreed	maturity	Redeemable	at notice 1),2)	Overnight <sup>1)</sup>	With agreed	maturity	
		Up to 2 years	Over 2 years	Up to 3 months	Over 3 months	-	Up to 2 years	Over 2 years	
	1	2	3	4	5	6	7	8	9
2004 Nov.	0.73	1.90	3.26	2.01	2.51	0.90	2.12	3.86	2.00
Dec.	0.73	1.92	3.24	2.00	2.52	0.90	2.16	3.77	2.02
2005 Jan.	0.74	1.91	3.23	1.98	2.49	0.93	2.12	3.73	2.01
Feb.	0.74	1.92	3.26	1.97	2.49	0.93	2.11	3.70	2.00
Mar.	0.74	1.92	3.22	1.96	2.47	0.94	2.09	3.71	1.99
Apr.	0.74	1.93	3.22	1.95	2.45	0.95	2.10	3.57	1.99
May	0.75	1.92	3.19	1.97	2.43	0.95	2.11	3.50	2.00
June	0.69	1.92	3.22	2.17	2.38	0.91	2.10	3.54	2.01
July	0.68	1.91	3.18	2.15	2.34	0.94	2.11	3.49	1.98
Aug.	0.69	1.92	3.18	2.03	2.31	0.96	2.10	3.51	2.00
Sep.	0.69	1.91	3.19	2.02	2.29	0.96	2.11	3.53	2.01
Oct.	0.69	1.93	3.17	1.96	2.27	0.97	2.12	3.48	2.03

#### 5. Interest rates on loans (outstanding amounts)

			Loans to h		Loans to n	on-financial corpo	orations		
	Lendi	ng for house purch with maturity	ase,	Consume	er credit and other with maturity	loans,		With maturity	
	Up to 1 year Over 1 and Up to 5 years 1 2 3			Up to 1 year	Over 1 and up to 5 years	Over 5 years	Up to 1 year	Over 1 and up to 5 years	Over 5 years
	1	2	3	4	5	6	7	8	9
2004 Nov.	4.67	4.52	4.86	7.94	6.98	5.82	4.40	3.96	4.48
Dec.	4.78	4.50	4.83	7.94	7.01	5.80	4.35	3.97	4.44
2005 Jan.	4.78	4.45	4.79	8.07	6.97	5.77	4.41	3.90	4.41
Feb.	4.74	4.45	4.76	8.06	7.03	5.76	4.39	3.92	4.46
Mar.	4.75	4.41	4.78	8.07	6.97	5.77	4.38	3.91	4.40
Apr.	4.69	4.38	4.74	8.02	6.94	5.76	4.34	3.86	4.37
May	4.63	4.36	4.71	8.00	6.87	5.74	4.33	3.85	4.35
June	4.62	4.33	4.67	7.92	6.93	5.72	4.32	3.85	4.35
July	4.57	4.29	4.63	7.89	6.86	5.70	4.30	3.82	4.29
Aug.	4.54	4.24	4.60	7.96	6.86	5.73	4.25	3.80	4.28
Sep.	4.52	4.23	4.59	7.94	6.85	5.70	4.25	3.78	4.26
Oct.	4.50	4.19	4.58	7.95	6.80	5.70	4.24	3.77	4.25

#### C19 New deposits with agreed maturity



by non-financial corporations, up to 1 year

by households, over 2 years



C20 New loans at floating rate and up to 1 year initial

- to households for consumption
- to households for house purchase







#### 4.6 Money market interest rates

			Euro area <sup>1)</sup>			United States	Japan
	Overnight	1-month	3-month	6-month	12-month	3-month	3-month
	deposits	deposits	deposits	deposits	deposits	deposits	deposits
	(EONIA)	(EURIBOR)	(EURIBOR)	(EURIBOR)	(EURIBOR)	(LIBOR)	(LIBOR)
	1	2	3	4	5	6	7_
2003	2.32	2.35	2.33	2.31	2.34	1.22	0.06
2004	2.05	2.08	2.11	2.15	2.27	1.62	0.05
2005	2.09	2.14	2.19	2.24	2.34	3.56	0.06
2004 Q4	2.08	2.12	2.16	2.20	2.32	2.30	$\begin{array}{c} 0.05 \\ 0.05 \\ 0.05 \\ 0.06 \\ 0.06 \end{array}$
2005 Q1	2.06	2.11	2.14	2.19	2.32	2.84	
Q2	2.07	2.10	2.12	2.14	2.19	3.28	
Q3	2.08	2.11	2.13	2.15	2.20	3.77	
Q4	2.14	2.25	2.34	2.46	2.63	4.34	
2004 Dec.	2.05	2.17	2.17	2.40	2.30	2.50	0.05
2005 Jan.	2.08	2.11	2.15	2.19	2.31	2.66	0.05
Feb.	2.06	2.10	2.14	2.18	2.31	2.82	0.05
Mar.	2.06	2.10	2.14	2.19	2.34	3.03	0.05
Apr.	2.08	2.10	2.14	2.17	2.27	3.15	0.05
May	2.07	2.10	2.13	2.14	2.19	3.27	0.05
June	2.06	2.10	2.11	2.11	2.10	3.43	0.05
July	2.07	2.11	2.12	2.13	2.17	3.61	0.06
Aug.	2.06	2.11	2.13	2.16	2.22	3.80	0.06
Sep.	2.09	2.12	2.14	2.17	2.22	3.91	0.06
Oct.	2.07	2.12	2.20	2.27	2.41	4.17	0.06
Nov.	2.09	2.22	2.36	2.50	2.68	4.35	0.06
Dec.	2.28	2.41	2.47	2.60	2.79	4.49	0.07



Source: ECB.
1) Before January 1999 synthetic euro area rates were calculated on the basis of national rates weighted by GDP. For further information, see the General notes.



### 4.7 Government bond yields

		Eı	iro area 1)			United States	Japan
	2 years	3 years	5 years	7 years	10 years	10 years	10 years
	1	2	3	4	5	6	7
2003	2.49	2.74	3.32	3.74	4.16	4.00	0.99
2004	2.47	2.77	3.29	3.70	4.14	4.26	1.50
2005	2.38	2.55	2.85	3.14	3.44	4.28	1.39
2004 Q4	2.41	2.62	3.06	3.51	3.84	4.17	1.45
2005 Q1	2.45	2.66	2.99	3.36	3.67	4.30	1.41
Q2	2.21	2.40	2.73	3.07	3.41	4.16	1.28
Q2 Q3 Q4	2.21	2.36	2.65	2.94	3.26	4.21	1.36
Q4	2.66	2.79	3.01	3.18	3.42	4.48	1.53
2004 Dec.	2.36	2.53	2.93	3.35	3.69	4.23	1.40
2005 Jan.	2.39	2.57	2.92	3.31	3.63	4.21	1.37
Feb.	2.45	2.67	2.97	3.32	3.62	4.16	1.40
Mar.	2.49	2.74	3.08	3.44	3.76	4.49	1.45
Apr.	2.34	2.55	2.89	3.25	3.57	4.34	1.32
May	2.22	2.41	2.74	3.05	3.41	4.14	1.27
June	2.07	2.24	2.58	2.93	3.25	4.00	1.24
July	2.19	2.34	2.66	2.99	3.32	4.16	1.26
Aug.	2.24	2.40	2.70	2.99	3.32	4.26	1.43
Sep.	2.21	2.34	2.60	2.84	3.16	4.19	1.38
Oct.	2.45	2.61	2.85	3.05	3.32	4.45	1.54
Nov. Dec.	2.73 2.80	2.86 2.88	3.10 3.07	3.28 3.21	3.53 3.41	4.53 4.46	1.52 1.54
Dec.	2.80	2.00	5.07	5.21	5.41	4.40	1.34

### C23 Euro area government bond yields

### C24 10-year government bond yields



Source: ECB.

To December 1998, euro area yields are calculated on the basis of harmonised national government bond yields weighted by GDP. Thereafter, the weights are the nominal outstanding amounts of government bonds in each maturity band.



#### 4.8 Stock market indices (index levels in points; period averages)

	Bench	ımark		United States	Japan									
	Broad	50	Basic materials	Consumer services	Consumer goods	Oil & gas	Financials	Industrials	Technology	Utilities	Telecom.	Health care	Standard & Poor's 500	Nikkei 225
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
2003 2004 2005	213.3 251.1 293.8	2,422.7 2,804.8 3,208.6	212.5 251.4 307.0	144.9 163.4 181.3	193.8 219.9 245.1	259.5 300.5 378.6	199.3 238.2 287.7	213.5 258.6 307.3	275.2 298.3 297.2	210.7 266.3 334.1	337.5 399.2 433.1	304.5 395.9 457.0	964.9 1,131.1 1,207.4	9,312.9 11,180.9 12,421.3
2004 Q4 2005 Q1 Q2 Q3 Q4	259.2 276.2 280.1 303.4 315.2	2,869.7 3,025.3 3,063.7 3,308.0 3,433.1	268.9 290.4 291.1 311.9 334.0	162.7 177.0 177.7 185.0 185.5	215.0 227.9 232.4 256.7 262.8	315.7 335.8 354.5 411.3 411.8	249.1 269.0 271.2 293.4 316.8	268.0 290.9 291.7 318.6 327.6	281.8 274.8 284.8 303.8 325.0	287.3 309.6 321.7 346.0 358.6	423.5 446.5 423.0 439.7 423.4	419.1 427.0 455.7 466.5 478.3	1,163.7 1,191.7 1,182.2 1,223.6 1,231.6	11,027.1 11,594.1 11,282.4 12,310.8 14,487.0
2004 Dec.	264.8	2,926.0	277.2	166.5	217.7	314.4	256.8	273.2	281.3	295.0	446.2	419.6	1,199.7	11,086.3
2005 Jan. Feb. Mar. Apr. May June July Aug. Sep. Oct. Nov. Dec.	269.4 279.0 279.8 275.9 276.1 288.2 298.4 303.1 308.4 306.8 312.7 325.7	2,957.0 3,050.4 3,065.8 3,013.7 3,023.5 3,151.7 3,267.1 3,303.3 3,351.8 3,340.1 3,404.9 3,550.1	277.0 294.2 299.4 290.0 285.7 297.7 302.0 311.5 321.7 322.4 330.8 348.4	172.0 179.5 179.3 176.7 175.4 181.0 184.9 185.7 184.4 182.4 183.2 190.8	221.6 230.0 227.9 228.7 240.4 249.5 257.1 263.0 260.6 259.3 268.4	318.1 338.5 349.5 345.5 344.1 373.4 398.3 405.8 405.8 405.3 405.3 411.2 418.5	262.8 270.1 273.7 269.0 267.1 277.4 288.2 293.4 298.5 302.6 316.4 330.8	284.2 295.1 293.5 287.6 285.2 302.0 313.8 318.9 322.9 317.3 322.3 342.7	270.4 277.4 276.5 268.5 208.5 301.5 308.6 297.6 305.7 312.4 322.9 339.2	302.9 317.5 308.7 314.2 319.4 331.2 336.8 343.9 357.0 347.7 354.0 373.5	450.6 453.8 436.3 426.1 421.3 421.7 437.5 444.7 436.5 434.0 418.2 418.5	423.8 428.7 428.6 443.1 460.5 462.8 463.4 473.0 462.5 466.8 471.6 496.1	1,181.6 1,199.7 1,193.9 1,164.4 1,179.2 1,202.3 1,220.9 1,224.3 1,225.6 1,192.0 1,238.7 1,262.4	11,401.1 11,545.7 11,812.4 11,377.2 11,071.4 11,402.7 11,718.9 12,205.0 12,986.6 13,384.9 14,362.0 15,664.0









# PRICES, OUTPUT, DEMAND AND LABOUR MARKETS

#### 5.1 HICP, other prices and costs (annual percentage changes, unless otherwise indicated)

#### 1. Harmonised Index of Consumer Prices

			Total			Total (s.a., percentage change on previous period)							
	Index 1996 = 100		Total Total excl. unprocessed food and energy	Goods	Services	Total	Processed food	Unprocessed food	Non-energy industrial goods	Energy (n.s.a.)	Services		
% of total 1)	100.0	100.0	83.9	59.2	40.8	100.0	12.0	7.5	31.0	8.6	40.8		
	1	2	3	4	5	6	7	8	9	10	11		
2001 2002 2003 2004	108.5 110.9 113.2 115.7	2.3 2.3 2.1 2.1	1.9 2.5 2.0 2.1	2.3 1.7 1.8 1.8	2.5 3.1 2.5 2.6	-	-		- - -		-		
2004 Q3 Q4 2005 Q1 Q2 Q3	115.9 116.6 116.7 118.1 118.6	2.2 2.3 2.0 2.0 2.3	2.1 2.0 1.7 1.5 1.4	2.0 2.1 1.8 1.8 2.4	2.6 2.7 2.4 2.3 2.2	0.5 0.5 0.4 0.7 0.8	0.3 0.4 0.7 0.3 0.5	-0.2 0.1 0.7 0.2 -0.2	0.2 0.1 -0.1 0.1 0.0	1.9 1.8 0.3 4.5 5.6	0.7 0.6 0.5 0.5 0.6		
2005 July Aug. Sep. Oct. Nov. Dec. <sup>2)</sup>	118.2 118.5 119.1 119.4 119.1	2.2 2.2 2.6 2.5 2.3 2.2	1.3 1.3 1.4 1.5 1.5	2.1 2.2 2.9 2.6 2.4	2.3 2.2 2.2 2.2 2.1	0.3 0.3 0.5 0.2 -0.2	$\begin{array}{c} 0.1 \\ 0.1 \\ 0.6 \\ 0.1 \\ 0.1 \end{array}$	-0.5 0.4 -0.1 0.0 0.5	-0.1 0.0 0.2 0.1 0.1	2.7 1.3 3.0 0.2 -3.0	0.2 0.2 0.1 0.2 0.1		

			Goods							Services		
	Food (incl. ald	coholic beverage	es and tobacco)		Industrial good	s	Hous	sing	Transport	Communication	Recreation	Miscellaneous
	Total	Processed food	Unprocessed food	od industrial goods				Rents			personal	
% of total 1)	19.6	12.0	7.5	39.6	31.0	8.6	10.3	6.4	6.3	2.9	14.6	6.6
	12	13	14	15	16	17	18	19	20	21	22	23
2001 2002 2003 2004	4.5 3.1 2.8 2.3	2.9 3.1 3.3 3.4	7.0 3.1 2.1 0.6	1.2 1.0 1.2 1.6	0.9 1.5 0.8 0.8	2.2 -0.6 3.0 4.5	1.8 2.4 2.3 2.4	1.4 2.0 2.0 1.9	3.6 3.2 2.9 2.8	-4.1 -0.3 -0.6 -2.0	3.6 4.2 2.7 2.4	2.7 3.4 3.4 5.1
2004 Q3 Q4 2005 Q1 Q2 Q3	2.0 1.4 1.6 1.2 1.4	3.6 2.8 2.4 1.6 1.8	-0.3 -0.7 0.5 0.8 0.8	2.0 2.4 1.9 2.1 2.8	0.8 0.8 0.3 0.3 0.1	6.3 8.5 7.6 8.8 12.7	2.5 2.6 2.6 2.7 2.5	2.0 2.1 2.1 2.1 2.1 2.1	2.8 3.0 3.1 2.4 2.6	-2.6 -2.6 -1.9 -2.0 -2.2	2.5 2.4 2.4 2.3 2.3	5.3 5.3 3.5 3.4 3.0
2005 June July Aug. Sep. Oct. Nov.	1.1 1.1 1.4 1.8 1.9 2.2	1.5 1.6 1.7 2.3 2.4 2.6	0.5 0.3 1.0 1.0 1.1 1.5	2.2 2.6 2.5 3.4 2.9 2.5	0.3 0.0 0.2 0.3 0.4	9.4 11.7 11.5 15.0 12.1 10.0	2.7 2.5 2.6 2.5 2.5 2.5 2.4	2.2 2.1 2.1 2.1 1.9 1.9	2.5 2.7 2.6 2.6 2.9 2.8	-2.1 -2.1 -2.1 -2.2 -2.8 -2.8	2.2 2.4 2.4 2.3 2.4 2.2	3.2 3.0 3.1 2.9 2.7 2.7

Sources: Eurostat and ECB calculations.

Referring to the index period 2005.
 Estimate based on first releases by Germany, Spain and Italy (and, when available, by other Member States), as well as on early information on energy prices.



Prices, output, demand and labour markets

### (annual percentage changes, unless otherwise indicated)

#### 2. Industry, construction, residential property and commodity prices

			Indus	trial pro	ducer prices e	xcluding	constru			Construct- ion <sup>1)</sup>	property	price	d market s of raw	Oil prices <sup>4)</sup> (EUR per	
	Total (index	Т	otal		Industry exc	luding co	nstructio	on and ener	rgy	Energy		prices <sup>2</sup> )	mat	erials <sup>3)</sup>	barrel)
	2000 = 100)	Manu- facturing         Total         Intermediate goods         Capital goods         Consumer goods           Total         Unreplied         Total         Durable         Non-durable						goods				Г	otal		
					Been	0	Total	Durable	Non-durable					Total excluding energy	
% of total 5)	100.0	100.0	89.5	82.5	31.6	21.3	29.5	4.0	25.5	17.5			100.0	32.8	
	1	1 2 3 4 5 6 7 8 9										12	13	14	15
2002 2003 2004 2005	101.9 103.4 105.7	-0.1 1.4 2.3	0.3 0.9 2.5	0.5 0.8 2.0	-0.3 0.8 3.5	0.9 0.3 0.7	1.0 1.1 1.3	1.3 0.6 0.7	1.0 1.2 1.4	-2.3 3.8 3.9	2.7 2.1 2.6	6.8 7.1 7.0	-4.1 -4.0 18.4 28.5	-0.9 -4.5 10.8 9.4	26.5 25.1 30.5 44.6
2004 Q4 2005 Q1 Q2 Q3 Q4	107.2 108.2 109.4 110.8	3.8 4.1 3.9 4.2	4.0 3.8 3.1 3.0	2.8 2.8 1.9 1.3	5.5 5.1 3.1 1.7	1.2 1.6 1.5 1.2	1.2 1.2 0.9 0.9	1.1 1.4 1.4 1.2	1.2 1.1 0.8 0.9	8.5 10.0 12.1 15.7	3.5 3.5 3.1	7.26	22.9 22.9 22.4 33.5 34.2	1.3 1.9 2.2 11.6 23.2	34.5 36.6 42.2 50.9 48.6
2005 July Aug. Sep. Oct. Nov. Dec.	110.3 110.8 111.3 112.0 111.7	4.1 4.0 4.4 4.2 4.2	3.0 2.9 3.2 2.8 2.7	1.3 1.3 1.3 1.4 1.4	1.9 1.7 1.6 1.5 1.7	1.2 1.1 1.2 1.2 1.0	0.8 0.9 1.1 1.3 1.4	1.2 1.2 1.2 1.3 1.2	0.7 0.9 1.1 1.3 1.4	15.1 15.2 16.6 15.2 14.8	- - - - -	- - - - -	34.1 32.4 33.9 23.1 33.0 48.6	9.6 11.9 13.2 17.4 22.5 29.8	48.3 52.0 52.2 49.3 47.9 48.5

#### 3. Hourly labour costs 7)

	Total (s.a. index	(s.a. index	Вус	component	By selec	vity	Memo: indicator	
	2000 = 100)		Wages and salaries	Employers' social contributions	Mining, manufacturing and energy	Construction	Services	of negotiated wages
% of total <sup>5</sup>	100.0	100.0	73.3	26.7	36.8	8.9	54.4	
	1	2	3	4	5	6	7	8
2001	103.8	3.9	4.0	3.7	3.7	3.9	4.0	2.6
2002	107.5	3.5	3.3	4.5	3.2	4.3	3.6	2.7
2003	110.7	3.0	2.8	3.8	3.0	3.9	2.8	2.4
2004	113.6	2.5	2.3	3.0	2.8	3.3	2.3	2.1
2004 Q3	114.0	2.4	2.3	2.7	2.4	3.3	2.3	2.0
Q4	114.7	2.4	1.9	4.0	2.8	2.9	2.1	2.0
2005 Q1	115.5	3.2	2.6	4.6	3.3	3.1	3.2	2.2
Q2	116.1	2.5	2.2	3.2	2.7	2.2	2.4	2.1
Q3	116.8	2.2	2.1	2.7	2.5	1.5	2.1	2.1

Sources: Eurostat, HWWA (columns 13 and 14 in Table 2 in Section 5.1), ECB calculations based on Thomson Financial Datastream data (column 15 in Table 2 in Section 5.1), ECB calculations based on Eurostat data (column 6 in Table 2 in Section 5.1 and column 7 in Table 3 in Section 5.1) and ECB calculations (column 12 in Table 2 in Section 5.1 and column 8 in Table 3 in Section 5.1).

1) Residential buildings, based on non-harmonised data.

Residential property price indicator for the euro area, based on non-harmonised sources. 2)

Refers to the prices expressed in euro. Brent Blend (for one-month forward delivery). 3)

4)

5) In 2000.

6) The quarterly data for the second (fourth) quarter refer to semi-annual averages of the first (second) half of the year, respectively. Since some national data are only available at annual frequency, the semi-annual estimate is partially derived from annual results; therefore, the accuracy of semi-annual data is lower than the accuracy of annual data.
7) Hourly labour costs for the whole economy, excluding agriculture, public administration, education, health and services not elsewhere classified. Owing to differences in coverage, the estimates for the components may not be consistent with the total.



### 5.1 HICP, other prices and costs

## **4.** Unit labour costs, compensation per employee and labour productivity *(seasonally adjusted)*

	Total (index	Total											
	2000 = 100)		Agriculture, hunting, forestry and fishing	Mining, manufacturing and energy	Construction	Trade, repairs, hotels and restaurants, transport and communication	Financial, real estate, renting and business services	Public administration, education, health and other services					
	1	2	3	4	5	6	7	8					
				τ	Jnit labour costs	1)							
2001	102.3	2.3	1.7	1.3	2.3	1.4	4.0	2.6					
2002	104.5	2.2	1.6	1.1	3.1	1.6	3.1	2.9					
2003 2004	106.4 107.4	1.8 0.9	3.0 -7.3	0.9 -0.3	1.6 2.4	1.7 0.5	1.9 2.1	2.7 1.6					
2004 Q2	107.3 107.3	0.8 0.4	-8.3 -8.8	-1.4 -1.4	0.9 3.6	-0.1 0.8	1.9 2.3	2.6 0.4					
Q3 04	107.3	0.4	-8.8 -5.5	-1.4	3.0	0.8	2.5 2.0	0.4					
2005 Q1	108.2	1.1	0.8	-0.1	4.0	0.7	1.9	1.6					
Q2	108.4	1.0	3.3	-0.2	2.6	1.1	2.6	1.3					
				Comp	ensation per em	ployee							
2001	102.7	2.7	1.3	2.5	3.1	2.5	2.6	3.0 2.9					
2002	105.3	2.5	3.1	2.6	3.2	2.2	1.9	2.9					
2003 2004	107.7 109.8	2.3 2.0	1.0 0.5	2.7 3.0	2.5 3.3	2.0 1.5	1.8 1.4	2.5 2.0					
	109.8	2.4	0.3	3.2	2.7	1.3	1.4	3.3					
2004 Q2 Q3	109.8	2.4 1.5	0.3	2.4	3.0	1.5	1.5 1.6	5.5 0.7					
ŏ4	110.3	1.7	1.7	2.4	3.2	1.0	1.0	1.5					
2005 Q1	111.0	1.5	2.8	1.7	2.6	1.9	1.7	1.1					
Q2	111.5	1.5	2.6	1.8	3.2	2.1	2.2	0.3					
				La	bour productivit	ty <sup>2)</sup>							
2001	100.5	0.5	-0.3	1.2	0.8	1.1	-1.4	0.4					
2002	100.7 101.2	0.2	1.5	1.5	0.1	0.6	-1.1	0.0					
2003 2004	101.2 102.3	0.5 1.1	-1.9 8.4	1.8 3.3	0.9 0.9	0.3 1.0	-0.1 -0.8	-0.2 0.4					
2004 Q3	102.4	1.1	10.0	3.9	-0.5	0.9	-0.7	0.3					
2004 Q3 Q4	102.4	0.7	7.7	1.7	-0.5	1.3	-0.7	0.5					
2005 Q1	102.6	0.4	2.0	1.8	-1.3	1.2	-0.2	-0.5					
Q2	102.8	0.5	-0.7	2.0	0.6	1.0	-0.4	-0.9					
Q3	103.2	0.8	0.0	2.2	2.0	1.4	-0.5	-0.7					

#### 5. Gross domestic product deflators

	Total (s.a. index	Total		Domest	ic demand		Exports <sup>3)</sup>	Imports <sup>3)</sup>
	2000 = 100)		Total	Private consumption	Government consumption	Gross fixed capital formation		
	1	2	3	4	5	6	7	8
2001	102.4	2.4	2.2	2.4	2.5	1.5	1.2	0.8
2002	104.9	2.5	2.0	1.9	2.7	1.3	-0.4	-2.1
2003	107.0	2.0	1.8	1.9	2.2	1.1	-1.2	-1.8
2004	109.0	1.8	2.0	2.0	2.1	2.4	1.2	1.4
2004 Q3	109.3	1.7	2.1	2.0	1.0	3.1	1.9	3.1
Ò4	109.7	1.8	2.3	2.1	2.1	3.3	1.8	3.6
2005 Q1	110.1	1.9	2.4	2.0	2.0	3.5	2.5	4.1
Q2	110.6	1.6	2.1	1.8	1.4	2.6	1.7	3.5
Ò3	110.9	1.5	2.2	2.1	1.6	2.5	2.2	4.3

Sources: ECB calculations based on Eurostat data.
Compensation (at current prices) per employee divided by value added (at constant prices) per person employed.
Value added (at constant prices) per person employed.
Deflators for exports and imports refer to goods and services and include cross-border trade within the euro area.



Prices, output, demand and labour markets

#### 5.2 Output and demand

#### 1. GDP and expenditure components

					GDP				
	Total		D	omestic demand		Exter	rnal balance 1)		
		Total	Private consumption	Government consumption	Gross fixed capital formation	Changes in inventories 2)	Total	Exports <sup>1)</sup>	Imports <sup>1)</sup>
	1	2	3	4	5	6	7	8	9
			Curre	nt prices (EUR billi	ons, seasonally ad	justed)			
2001	6,971.0	6,859.1	4,021.2	1,381.1	1,453.4	3.4	112.0	2,595.4	2,483.5
2002 2003	7,213.6 7,411.8	7,023.7 7,248.5	4,138.0 4,264.6	1,455.3 1,511.5	1,443.6 1,471.3	-13.3 1.2	189.9 163.3	2,630.6 2,633.2	2,440.7 2,469.9
2003	7,686.1	7,522.7	4,409.6	1,560.7	1,532.8	19.6	163.4	2,821.9	2,658.5
2004 Q3	1,930.4	1,891.8	1,104.5	391.2	386.5	9.6	38.7	717.0	678.3
Q4	1,941.3	1,911.2	1,118.9	392.3	391.0	8.9	30.1	720.0	689.9
2005 Q1	1,955.3	1,922.0	1,124.5	396.8	392.1	8.6	33.4	720.0	686.7
Q2 Q3	1,972.6 1,988.9	1,946.4 1,962.9	1,133.8 1,145.2	401.3 402.8	398.6 408.5	12.7 6.4	26.3 26.1	736.8 770.8	710.5 744.7
	-,,	-,,		percentag					
2004	100.0	97.9	57.4	20.3	19.9	0.3	2.1	-	-
			Chain-linked volu	umes (prices of the	previous year, seas	sonally adjusted 3) )			
				quarter-on-quarter	percentage change	es			
2004 Q3	0.3	0.6	0.2	0.3	0.4	-	-	1.2	2.0
Q4	0.2	0.6	0.8	-0.1	0.6	-	-	0.3	1.3
2005 Q1	0.3	0.1	0.1	0.3	0.1	-	-	-0.8	-1.5
Q2 03	0.4 0.6	0.5 0.3	0.2 0.3	0.6 0.6	0.8 1.6	-	-	2.2 3.4	2.6 2.8
	0.0	0.5	0.5		ntage changes			5.1	2.0
2001	1.9	1.2	1.9	2.2	0.4	-	-	3.7	1.8
2002	0.9	0.4	0.9	2.6	-2.0	-	-	1.7	0.2
2003	0.7	1.4	1.1	1.6	0.8	-	-	1.2	3.0
2004	2.1	2.1	1.5	1.2	2.2	-	-	6.5	6.6
2004 Q3 O4	1.9 1.6	2.3 2.0	1.1 1.8	1.2 0.7	2.0 1.6	-	-	6.3 5.9	7.6 7.1
2005 Q1	1.0	1.6	1.8	0.9	1.3	-	-	3.2	4.2
2005 Q2	1.2	1.8	1.4	1.1	1.9	-	-	2.9	4.5
Q3	1.6	1.5	1.5	1.4	3.1	-	-	5.2	5.2
		con	tributions to quarte	r-on-quarter percen	tage changes of G	DP in percentage p	oints		
2004 Q3	0.3	0.5	0.1	0.1	0.1	0.3	-0.3	-	-
Q4	0.2	0.5	0.5	0.0	0.1	0.0	-0.4	-	-
2005 Q1 Q2	0.3 0.4	0.1 0.5	0.1 0.1	0.1 0.1	0.0 0.2	-0.1 0.1	0.2 -0.1	-	-
Q2 Q3	0.6	0.3	0.1	0.1	0.2	-0.3	0.3	-	_
			contributions to a	unnual percentage c	hanges of GDP in	percentage points			
2001	1.9	1.2	1.1	0.4	0.1	-0.5	0.7	-	-
2002	0.9	0.4	0.5	0.5	-0.4	-0.2	0.5	-	-
2003 2004	0.7 2.1	1.3	0.6	0.3	0.2	0.2	-0.6 0.1	-	-
		2.0	0.9	0.2	0.4	0.4		-	-
2004 Q3 04	1.9 1.6	2.2 1.9	0.6 1.0	0.2 0.1	0.4 0.3	0.9 0.4	-0.3 -0.3	-	-
2005 Q4	1.0	1.5	0.8	0.1	0.3	0.4	-0.3	-	-
Q2	1.2	1.7	0.8	0.2	0.4	0.3	-0.5	-	-
Q3	1.6	1.5	0.9	0.3	0.6	-0.3	0.1	-	-

Sources: Eurostat and ECB calculations.

Exports and incred calculations.
 Exports and imports cover goods and services and include cross-border intra-euro area trade. They are not fully consistent with Table 1 in Section 7.3.
 Including acquisitions less disposals of valuables.
 Annual data are not adjusted for the variations in the number of working days.



### 5.2 Output and demand

#### 2. Value added by economic activity

			Gross	value added (basic p	rices)			Taxes less subsidies on			
	Total	Agriculture, hunting, forestry and fishing activities	Mining, manufacturing and energy	Construction	Trade, repairs, hotels and restaurants, transport and communication	Financial, real estate, renting and business activities	Public administration, education, health and other services	products			
	1	2	3 Current prices	4 (EUR billions, seasor	5	6	7	8			
2001	( )5( )	156.0		· · · · · · · · · · · · · · · · · · ·		1 (01 1	1 204 2	714.2			
2001 2002 2003 2004	6,256.8 6,476.6 6,651.6 6,891.0	156.8 151.9 151.6 152.1	1,354.8 1,366.8 1,375.6 1,418.0	352.8 365.5 382.6 406.6	1,327.0 1,377.6 1,408.8 1,456.8	1,681.1 1,758.3 1,821.7 1,892.2	1,384.3 1,456.6 1,511.3 1,565.3	714.2 736.9 760.3 795.0			
2004 Q3 Q4 2005 Q1 Q2	1,730.7 1,739.1 1,753.5 1,769.5	37.7 38.1 37.3 37.1	357.5 356.6 359.9 363.8	101.9 103.7 104.1 106.7	366.5 367.6 369.7 372.0	476.2 479.0 484.2 490.4	390.8 394.0 398.4 399.5	199.8 202.2 201.8 203.1			
Q3	1,779.3	37.2	364.8	108.9	374.9	493.7	399.8	209.7			
				rcentage of value add							
2004	100.0	2.2	20.6	5.9	21.1	27.5	22.7	-			
	Chain-linked volumes (prices of the previous year, seasonally adjusted <sup>1)</sup> )										
	quarter-on-quarter percentage changes										
2004 Q3 Q4 2005 Q1 Q2	0.2 0.2 0.3 0.4	-0.2 0.8 -1.8 -1.0	0.2 -0.6 0.3 0.7	-0.4 0.6 -0.3 1.9	0.4 0.4 0.4 0.6	0.4 0.2 0.7 0.4	0.1 0.4 0.2 -0.2	0.6 0.5 0.4 0.2			
Q3	0.5	-0.3	0.7	0.3	0.7	0.4	0.6	1.4			
				nual percentage chang							
2001 2002 2003 2004	2.0 1.0 0.7 2.2	-1.4 -0.3 -4.1 7.3	1.2 -0.1 0.2 2.4	1.2 0.0 1.0 2.0	2.8 1.2 0.5 2.3	2.7 1.4 1.3 1.9	1.8 2.0 1.1 1.7	0.8 0.1 0.9 1.4			
2004 Q3 Q4 2005 Q1 Q2 Q3	2.0 1.7 1.4 1.2 1.4	9.2 6.8 0.3 -2.2 -2.3	2.2 0.8 0.7 0.7 1.2	1.2 1.3 -0.1 1.7 2.4	1.8 2.3 2.1 1.8 2.0	1.8 1.7 2.0 1.8 1.7	1.6 1.5 0.9 0.5 0.9	1.2 0.9 0.2 1.7 2.5			
		contributions to	auarter-on-auarter	percentage changes		entage points					
2004 Q3 Q4 2005 Q1 Q2 Q3	0.2 0.2 0.3 0.4 0.5	0.0 0.0 0.0 0.0 0.0	0.0 -0.1 0.1 0.1 0.1 0.1	0.0 0.0 0.0 0.1 0.0	0.1 0.1 0.1 0.1 0.1	0.1 0.1 0.2 0.1 0.1	0.0 0.1 0.0 0.0 0.1	-			
		contributi	ons to annual perce	ntage changes of valu	e added in percentag	e points					
2001 2002 2003 2004	2.0 1.0 0.7 2.2	0.0 0.0 -0.1 0.2	0.3 0.0 0.0 0.5	0.1 0.0 0.1 0.1	0.6 0.2 0.1 0.5	0.7 0.4 0.4 0.5	0.4 0.4 0.2 0.4				
2004 Q3 Q4 2005 Q1 Q2 Q3	2.0 1.7 1.4 1.2 1.4	0.2 0.2 0.0 -0.1 -0.1	0.4 0.2 0.1 0.1 0.2	0.1 0.1 0.0 0.1 0.1	0.4 0.5 0.4 0.4 0.4	0.5 0.5 0.6 0.5 0.5	0.4 0.4 0.2 0.1 0.2	- - - - -			

Sources: Eurostat and ECB calculations. 1) Annual data are not adjusted for the variations in the number of working days.



Prices, output, demand and labour markets

### (annual percentage changes, unless otherwise indicated)

#### **3. Industrial production**

	Total		Industry excluding construction         Industry excluding construction and energy         Energy												
		Total (s.a. index	Т	otal		Industry e	xcluding cor	struction a	nd energy		Energy				
		2000 = 100		Manu- facturing	Total	Intermediate goods	Capital goods		Consumer go	ods					
				racturing		goods	goous	Total	Durable	Non-durable					
% of total <sup>1)</sup>	100.0	82.9	82.9	75.0	74.0	30.0	22.4	21.5	3.6	17.9	8.9	17.1			
	1	2	3	4	5	6	7	8	9	10	11	12			
2002	-0.2	99.9	-0.5	-0.8	-0.7	-0.1	-1.7	-0.3	-5.5	0.7	1.1	1.4			
2003	0.2	100.2	0.3	0.0	0.0	0.4	-0.2	-0.5	-4.6	0.2	2.9	-0.1			
2004	2.1	102.1	2.0	2.0	1.9	1.8	3.0	0.5	0.0	0.6	2.5	-0.2			
2004 Q4	1.1	102.4	1.1	0.7	0.5	1.0	1.7	-0.2	-3.5	0.4	3.0	-0.8			
2005 Q1	-0.4	102.3	0.6	0.4	0.2	0.8	2.1	-1.0	-3.9	-0.5	1.1	-4.2			
Q2	1.1 1.1	102.9 103.7	0.7 1.4	0.8 1.4	0.3 1.3	-0.4 1.2	2.1 2.6	0.5 1.7	-1.8 -0.2	0.9 2.0	1.2 0.1	-0.1 0.8			
Q3															
2005 May	0.3	102.7	0.1	-0.3	-0.8	-0.4	0.5	0.4	-4.2	1.2	0.8	-0.3			
June	0.8	103.1 103.2	0.7 0.6	0.5 0.0	-0.1 -0.1	-1.2 -0.5	2.4	0.5 0.0	-0.5	0.7	2.6 2.0	0.6			
July Aug.	0.6 1.8	103.2	2.6	0.0 3.0	-0.1	-0.5 3.8	2.6 2.3	0.0 3.5	-1.9 2.6	0.4 3.6	-0.5	0.6 2.3			
Sep.	1.0	104.1	1.2	1.6	1.5	1.0	2.3	1.9	-0.3	2.3	-0.3	-0.3			
Oct.	1.1	103.0	0.2	0.7	0.5	1.2	0.2	0.9	-0.8	1.2	-2.2				
				month-o	on-month p	ercentage chang	es (s.a.)								
2005 May	-0.6	-	-0.4	-1.0	-1.2	-0.4	-0.3	-0.3	-2.1	0.0	1.8	0.4			
June	0.1	-	0.5	0.6	0.5	-0.3	1.3	0.1	1.5	-0.1	2.1	0.9			
July	0.0	-	0.1	-0.1	0.1	0.9	0.1	0.0	-0.4	0.0	-0.3	-0.1			
Aug.	0.6	-	0.8	1.4	1.5	2.1	-0.1	1.2	1.0	1.3	-2.4	0.2			
Sep.	-0.2	-	-0.2	-0.3	-0.3	-1.2	0.9	-0.5	-1.3	-0.4	-0.1	-1.5			
Oct.		-	-0.7	-0.8	-0.8	0.0	-1.3	-0.6	-0.6	-0.6	-1.7				

#### 4. Industrial new orders and turnover, retail sales and new passenger car registrations

	Industrial ne								New passen registrat				
	Manufactu (current p		Manufac (current p		Current prices			Constan	t prices			, ighter an	
	Total (s.a. index 2000 = 100)	Total	Total (s.a. index 2000 = 100)	Total	Total	Total (s.a. index 2000 = 100)	Total	Food, beverages, tobacco		Non-food Textiles, clothing, footwear	Household equipment	Total (s.a., thousands) <sup>3)</sup>	Total
% of total 1)	100.0	100.0	100.0	100.0	100.0	100.0	100.0	43.7	56.3	10.6	14.8		
	1	2	3	4	5	6	7	8	9	10	11	12	13
2002	98.4	-0.4	101.4	-0.6	1.9	101.7	0.3	1.2	-0.4	-1.9	-1.9	925	-4.4
2003 2004	98.4	0.1	101.0	-0.3 4.9	1.8	102.1	0.4 0.9	1.1	-0.3	-2.7	0.2	911 922	-1.5
	105.1	7.1	105.9		1.7	102.9		0.8	0.8	0.9	2.8		1.1
2004 Q4	109.3	8.2	107.3	5.1	1.7	103.3	1.0	1.2	0.7	1.2	2.6	942	3.6
2005 Q1 Q2	106.0 108.6	3.1 3.3	106.7 110.3	2.5 4.2	2.0 1.6	104.0 103.6	1.3 0.6	1.2 0.3	1.2 0.6	0.5 0.8	0.8 0.6	921 938	$0.4 \\ 1.0$
Q3	109.8	4.9	110.8	3.9	2.0	103.8	1.0	0.3	1.4	1.3	1.7	942	4.6
2005 June	110.8	6.2	112.3	3.6	1.3	104.0	0.4	0.7	0.0	-0.3	0.6	981	6.1
July	109.2	1.7	106.7	0.3	0.5	103.3	-0.2	-1.0	0.3	0.1	0.5	940	3.0
Aug.	109.2	7.8	112.6	7.3	3.2	104.4	2.2	0.9	3.3	4.4	3.3	933	7.4
Sep.	110.9	6.0	113.1	4.8	2.3	103.6	1.0	1.0	0.8	0.0	1.6	953	4.5
Oct. Nov.	110.2	4.2	106.4	1.4	1.4 1.4	103.8 103.7	0.2 0.3	0.4 -0.2	0.0 0.7	0.8	0.8	942 934	0.1 -2.0
1107.	•	•	•	•		nonth percentag			0.7	•		754	-2.0
2005 1									0.0	1.0	0.5		0.6
2005 June July	-	3.2 -1.5	-	4.2 -5.0	0.0 -0.5	-	-0.1 -0.6	-0.2 -0.8	0.0 -0.3	1.0 0.1	0.5 -0.4	-	9.6 -4.1
Aug.		0.0		5.6	-0.5		-0.0	0.5	1.3	1.9	1.3		-0.8
Sep.	_	1.6	_	0.5	-0.5	_	-0.7	0.0	-1.2	-3.6	-0.7	_	2.2
Oct.	-	-0.7	-	-6.0	0.3	-	0.2	0.4	-0.1	1.1	-0.3	-	-1.1
Nov.	-		-		-0.1	-	-0.1	-0.4	0.2			-	-0.9

Sources: Eurostat, except columns 12 and 13 in Table 4 in Section 5.2 (ECB calculations based on data from the ACEA, European Automobile Manufacturers' Association).
In 2000.
Includes manufacturing industries working mainly on the basis of orders, representing 62.6% of total manufacturing in 2000.
Annual and quarterly figures are averages of monthly figures in the period concerned.



#### 5. Business and Consumer Surveys

	Economic sentiment		Manu	ifacturing inc	lustry			Consum	er confidence i	ndicator <sup>3)</sup>	
	indicator <sup>2)</sup> (long-term			lence indicator		Capacity utilisation <sup>4)</sup>	Total <sup>5)</sup>	Financial situation	Economic situation	Unemployment situation	Savings over next
	average = 100)	Total <sup>5</sup>	books finished expectations products		(percentages)		over next 12 months	over next 12 months	over next 12 months	12 months	
	1	2	2 3 4 -11 -25 11				7	8	9	10	11
2002 2003	94.4 93.5				3	81.2 81.0	-11 -18	-1 -5	-12 -21	27 38	-3 -9
2003	99.5		-11 $-25$ $10$ $3-5$ $-16$ $8$ $10$			81.6	-18	-4	-21	30	-9
2005	98.2	-7	-17	11	6		-14	-4	-15	28	-8
2004 Q4	100.5	-3	-12	8	9	82.0	-13	-3	-14	29	-6
2005 Q1	98.7	-6	-15	11	6	81.5	-13	-3	-13	30	-8
Q2	96.1	-10	-20	13	3	81.0	-14	-3	-16	31	-7
Q3	97.8	-8	-18	11	6	81.1	-15	-4	-17	29	-8
Q4	100.2	-6	-15	10	/	•	-12	-4	-15	22	-9
2005 July	97.3	-8	-18	11	4	80.9	-15	-4	-19	30	-9
Aug.	97.6	-8	-18	11	6	-	-15	-4	-17	29	-8
Sep.	98.4	-7	-16	10	7	-	-14	-5	-16	28	-9
Oct.	100.2	-6 -7	-16	10	8	81.2	-13	-5	-15	23	-9
Nov.	99.9 100.5	-7 -5	-16 -13	9 10	6	-	-13 -11	-5	-17 -12	23 19	-8
Dec.	100.5	-3	-13	10	ð	-	-11	-4	-12	19	-9

	Constructio	n confidence	indicator	Reta	ail trade confid	ence indicator		Ser	vices confide	nce indicator	
	Total <sup>5)</sup>	Order books	Employment expectations	Total <sup>5)</sup>	Present business situation	Volume of stocks	Expected business situation	Total <sup>5)</sup>	Business climate	Demand in recent months	Demand in the months ahead
	12	13	14	15	16	17	18	19	20	21	22
2002	-18	-25	-11	-17	-20	18	-12	2	-3	-4	14
2003	-19	-26	-13	-12	-16	17	-3	4	-5	3	14
2004	-15	-23	-7	-8	-12	14	1	11	7	8	17
2005	-10	-15	-4	-7	-12	14	3	11	6	10	17
2004 Q4	-12	-19	-5	-8	-14	13	3	11	9	9	16
2005 Q1	-12	-16	-7	-8	-12	12	1	11	8	7	17
Q2	-12	-18	-6	-8	-13	13	1	9	0	9	17
Q3	-9	-16	-3	-9	-13	14	1	11	6	10	17
Q4	-5	-11	0	-5	-8	15	10	14	10	13	18
2005 July	-12	-18	-5	-10	-14	15	-3	12	5	11	19
Aug.	-9	-16	-2	-9	-14	13	1	10	3	10	15
Sep.	-8	-14	-1	-8	-12	15	5	11	8	10	15
Oct.	-8	-14	-1	-4	-9	13	12	14	10	13	19
Nov.	-3	-7	1	-6	-11	16	9	14	11	13	18
Dec.	-5	-10	0	-4	-5	16	9	13	9	13	18

Source: European Commission (Economic and Financial Affairs DG).

1) Difference between the percentages of respondents giving positive and negative replies.

2) The economic sentiment indicator is composed of the industrial, services, consumer, construction and retail trade confidence indicators; the industrial confidence indicator has a weight of 40%, the services confidence indicator a weight of 30%, the consumer confidence indicator a weight of 20% and the two other indicators a weight of 30%, the consumer confidence indicator a weight of 20% and the two other indicators a weight of 5% each. Values of the economic sentiment indicator above (below) 100 indicate above-average (below-average) economic sentiment, calculated for the period from January 1985. Owing to changes in the questionnaire used for the French survey, euro area results from January 2004 onwards are not fully comparable with previous results. Data are collected in January, April, July and October each year. The quarterly figures shown are averages of two successive surveys. Annual data are derived from quarterly 3)

4)

averages.

5) The confidence indicators are calculated as simple averages of the components shown; the assessments of stocks (columns 4 and 17) and unemployment (column 10) are used with inverted signs for the calculation of confidence indicators.



### 5.3 Labour markets 1)

#### 1. Employment

	Whole eco	onomy	By employ	ment status			By ec	onomic activity		
	Millions (s.a.)		Employees	Self- employed	Agriculture, hunting, forestry and fishing	Mining, manufacturing and energy	Construction	Trade, repairs, hotels and restaurants, transport and communication	Financial, real estate, renting and business services	Public administration, education, health and other services
% of total 2)	100.0	100.0	84.4	15.6	4.5	18.1	7.2	24.9	15.3	30.0
	1	2	3	4	5	6	7	8	9	10
2001 2002 2003 2004	134.494 135.466 135.842 136.817	1.5 0.7 0.3 0.7	1.7 0.8 0.3 0.6	0.2 0.1 0.3 1.2	-0.8 -1.6 -2.0 -0.8	0.1 -1.6 -1.5 -1.6	0.7 0.1 0.2 1.0	1.8 0.5 0.3 0.9	4.2 2.6 1.3 2.6	1.3 2.0 1.3 1.2
2004 Q3 Q4 2005 Q1 Q2 Q3	136.888 137.225 137.293 137.499 137.890	0.8 1.0 0.9 0.8 0.8	0.6 0.9 0.8 0.8 0.8	1.5 1.5 1.0 1.0 0.5	-0.4 -0.7 -1.5 -1.3 -1.9	-1.8 -0.9 -1.1 -1.3 -1.0	2.0 1.5 1.5 1.4 1.5	0.9 1.1 1.0 0.8 0.4	2.4 2.5 2.2 2.3 2.3	1.3 1.4 1.4 1.4 1.4 1.6
				quarter	on-quarter per	centage changes (	(s.a.)			
2004 Q3 Q4 2005 Q1 Q2 Q3	0.360 0.337 0.068 0.206 0.391	0.3 0.2 0.0 0.2 0.3	0.1 0.3 0.3 0.0 0.2	1.4 0.0 -1.1 0.7 0.6	0.1 -0.4 -1.1 -0.2 -0.7	-0.5 0.1 -0.8 -0.1 -0.1	1.1 -0.3 0.0 0.3 0.4	0.4 0.2 0.1 0.2 0.2	0.6 0.5 0.7 0.3 0.6	0.3 0.5 0.4 0.2 0.5

#### 2. Unemployment

(	seasonal	ly	ad	just	ed)	

	Total			B	y age <sup>3)</sup>			By	gender <sup>4)</sup>	
	Millions	% of labour force	A	dult	Yo	outh	١	Aale	F	emale
			Millions	% of labour force	Millions	% of labour force	Millions	% of labour force	Millions	% of labour force
% of total 2)	100.0		75.6		24.4		48.0		52.0	
	1	2	3	4	5	6	7	8	9	10
2001 2002 2003 2004	11.017 11.747 12.534 12.869	7.9 8.3 8.7 8.9	8.093 8.729 9.413 9.732	6.6 7.0 7.5 7.6	2.924 3.018 3.120 3.137	16.1 16.8 17.6 17.9	5.037 5.513 5.971 6.174	6.3 6.9 7.4 7.6	5.980 6.233 6.562 6.695	9.9 10.1 10.5 10.5
2004 Q3 Q4 2005 Q1 Q2 Q3	12.902 12.865 12.818 12.622 12.279	8.9 8.8 8.8 8.6 8.4	9.762 9.741 9.616 9.571 9.343	7.6 7.6 7.5 7.4 7.2	3.140 3.124 3.202 3.050 2.935	17.9 18.0 18.3 17.6 17.2	6.181 6.259 6.205 6.136 6.010	7.6 7.6 7.6 7.5 7.3	6.721 6.606 6.613 6.486 6.269	10.5 10.3 10.3 10.1 9.8
2005 June July Aug. Sep. Oct. Nov.	12.515 12.384 12.294 12.158 12.093 12.100	8.6 8.5 8.4 8.3 8.3 8.3	9.535 9.455 9.362 9.212 9.129 9.121	7.4 7.3 7.3 7.1 7.1 7.1	2.981 2.928 2.932 2.945 2.964 2.980	17.3 17.1 17.1 17.2 17.4 17.4	6.091 6.046 6.025 5.958 5.899 5.874	7.4 7.4 7.3 7.2 7.2	6.425 6.338 6.269 6.200 6.194 6.226	10.0 9.9 9.8 9.7 9.7 9.7

Sources: ECB calculations based on Eurostat data (in Table 1 in Section 5.3) and Eurostat (Table 2 in Section 5.3). 1) Data for employment refer to persons and are based on the ESA 95. Data for unemployment refer to persons and follow ILO recommendations.

In 2004.
 In 2004.
 Adult: 25 years of age and over; youth: below 25 years of age; rates are expressed as a percentage of the labour force for the relevant age group.
 Rates are expressed as a percentage of the labour force for the relevant gender.





### **GOVERNMENT FINANCE**

### 6.1 Revenue, expenditure and deficit/surplus <sup>1)</sup>

#### 1. Euro area - revenue

	Total					Curre	ent revenue					Capital	revenue	Memo: fiscal
		Г	Direct			Indirect		Social			Sales		Capital	burden <sup>2)</sup>
			taxes	Households	Corporations	taxes	Received by EU		Employers	Employees			taxes	
							institutions							
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
1996	47.0	46.6	11.7	9.1	2.3	13.2	0.7	17.3	8.5	5.5	2.4	0.3	0.2	42.5
1997	47.2	46.7	11.9	9.1	2.5	13.4	0.7	17.3	8.6	5.5	2.3	0.5	0.3	42.9
1998	46.7	46.4	12.2	9.6	2.3	14.0	0.6	16.2	8.4	4.9	2.3	0.3	0.3	42.7
1999	47.2	47.0	12.6	9.7	2.5	14.2	0.6	16.2	8.4	4.9	2.3	0.3	0.3	43.3
2000	46.9	46.6	12.8	9.8	2.6	14.0	0.6	16.0	8.3	4.8	2.2	0.3	0.3	43.0
2001	46.1	45.8	12.4	9.6	2.4	13.6	0.6	15.7	8.2	4.7	2.2	0.2	0.3	42.0
2002	45.5	45.2	11.9	9.4	2.2	13.6	0.4	15.7	8.2	4.6	2.2	0.3	0.3	41.5
2003	45.5	44.8	11.6	9.2	2.1	13.6	0.4	15.9	8.3	4.7	2.2	0.7	0.5	41.6
2004	45.0	44.5	11.5	8.8	2.3	13.7	0.3	15.7	8.2	4.6	2.1	0.5	0.4	41.2

#### 2. Euro area - expenditure

	Total				Current e	expenditure				Capital ex	penditure		Memo: primary	
		Total	Compensation		Interest	Current	Seciel	Subsidies			Investment		Paid by EU	expenditure 3)
			of employees	consumption		transfers	payments		Paid by EU			transfers	institutions	
	1	2	3	4	5	6	7	8	institutions 9	10	11	12	13	14
1996	51.3	47.4	11.1	4.8	5.6	25.9	22.9	2.2	0.6	3.8	2.6	1.3	0.0	45.6
1997	49.9	46.3	10.9	4.7	5.0	25.6	22.8	2.1	0.6	3.6	2.4	1.2	0.1	44.9
1998	49.0	45.2	10.6	4.6	4.6	25.4	22.3	2.1	0.5	3.8	2.4	1.4	0.1	44.4
1999	48.6	44.7	10.6	4.7	4.1	25.3	22.3	2.1	0.5	3.9	2.5	1.4	0.1	44.5
2000	47.9	44.1	10.5	4.7	3.9	25.0	21.9	2.0	0.5	3.8	2.5	1.3	0.0	44.0
2001	48.0	44.0	10.4	4.8	3.8	25.0	21.9	1.9	0.5	4.0	2.5	1.4	0.0	44.1
2002	48.1	44.2	10.5	4.9	3.6	25.3	22.4	1.9	0.5	3.8	2.4	1.4	0.0	44.5
2003	48.5	44.5	10.6	4.9	3.4	25.7	22.8	1.9	0.5	4.0	2.6	1.4	0.1	45.2
2004	47.7	43.9	10.5	4.9	3.2	25.3	22.6	1.8	0.5	3.8	2.5	1.4	0.0	44.6

#### 3. Euro area - deficit/surplus, primary deficit/surplus and government consumption

	Deficit (-)/surplus (+)					Primary deficit (-)/			(	Government	consumption <sup>4)</sup>			
	Total	Central	State	Local	Social	surplus (+)	Total						Collective	Individual
		gov.	gov.	gov.	security	- · · ·		Compensation	Intermediate	Transfers	Consumption		consumption	consumption
					funds			of employees	consumption	in kind via market		(minus)		
										producers	capitai			
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
1996	-4.3	-3.7	-0.4	0.0	-0.1	1.3	20.3	11.1	4.8	4.9	1.9	2.4	8.5	11.9
1997	-2.7	-2.4	-0.4	0.1	0.1	2.4	20.1	10.9	4.7	4.9	1.9	2.3	8.4	11.7
1998	-2.3	-2.2	-0.2	0.1	0.1	2.4	19.7	10.6	4.6	4.9	1.8	2.3	8.1	11.6
1999	-1.3	-1.7	-0.1	0.1	0.4	2.7	19.9	10.6	4.7	4.9	1.8	2.3	8.2	11.6
2000	-1.0	-1.4	-0.1	0.1	0.5	2.9	19.8	10.5	4.7	4.9	1.8	2.2	8.1	11.7
2001	-1.9	-1.7	-0.4	0.0	0.3	2.0	19.8	10.4	4.8	5.0	1.8	2.2	8.0	11.8
2002	-2.6	-2.1	-0.5	-0.2	0.2	1.0	20.2	10.5	4.9	5.1	1.8	2.2	8.1	12.1
2003	-3.0	-2.3	-0.4	-0.2	0.0	0.3	20.4	10.6	4.9	5.2	1.8	2.2	8.1	12.3
2004	-2.7	-2.3	-0.3	-0.3	0.1	0.4	20.3	10.5	4.9	5.2	1.8	2.1	8.0	12.3
4. Euro a	rea cou	ntries -	– defic	:it (-)/s	urplus	(+) <sup>5)</sup>								
	1	<b>BE</b> 1	<b>DE</b> 2		GR 3	<b>ES</b> 4	<b>FR</b> 5	<b>IE</b> 6	<b>IT</b> 7	<b>LU</b> 8	<b>NL</b> 9	<b>AT</b> 10	<b>PT</b> 11	<b>FI</b> 12
2001	(	0.6	-2.9		-6.1	-0.5	-1.5	0.8	-3.2	6.5	-0.2	0.1	-4.2	5.2
2002		0.0	-3.8		-4.9	-0.3	-3.2	-0.4	-2.7	2.1	-2.0	-0.4	-2.8	4.3
2002		0.1	-4.1		-5.7	0.0	-4.1	0.2	-3.2	0.2	-3.2	-1.2	-2.9	2.5
2005		0.0	-3.7		-6.6	-0.1	-3.7	1.4	-3.2	-1.2	-2.1	-1.0	-3.0	2.1
2001			5.7		0.0	5.1	5.7	1	5.2	1.2	2.1	1.0	5.0	2.1

 Sources: ECB for euro area aggregated data; European Commission for data relating to countries' deficit/surplus.
 Revenue, expenditure and deficit/surplus are based on the ESA 95, but the figures exclude proceeds from the sale of UMTS licences in 2000 (the euro area deficit/surplus including those proceeds is equal to 0.0% of GDP). Transactions involving the EU budget are included and consolidated. Transactions among Member States' governments are not consolidated.

2) The fiscal burden comprises taxes and social contributions.

Comprises total expenditure minus interest expenditure.

3) 4) 5) Corresponds to final consumption expenditure (P.3) of general government in the ESA 95. Ratios are computed using GDP excluding financial intermediation services indirectly measured (FISIM). Includes proceeds from the sale of UMTS licences and settlements under swaps and forward rate agreements.



#### 6.2 Debt <sup>1)</sup> (as a percentage of GDP)

#### 1. Euro area - by financial instrument and sector of the holder

	Total		Financial in	struments				Holders		
		Coins and	Loans	Short-term securities	Long-term securities		Domestic c	reditors <sup>2)</sup>		Other creditors 3)
		deposits				Total	MFIs	Other financial corporations	Other sectors	
	1	2	3	4	5	6	7	8	9	10
1995	74.0	2.8	17.6	8.0	45.7	58.6	30.3	10.3	18.0	15.4
1996	75.4	2.8	17.1	7.9	47.5	59.1	30.9	12.0	16.1	16.3
1997	74.5	2.8	16.1	6.6	49.1	56.7	29.4	13.3	14.0	17.8
1998	73.1	2.7	15.1	5.6	49.6	53.3	27.6	14.2	11.5	19.7
1999	72.4	2.9	14.3	4.3	50.9	49.4	26.5	11.4	11.6	22.9
2000	69.9	2.7	13.2	3.7	50.3	44.8	23.3	10.3	11.1	25.1
2001	68.6	2.7	12.5	3.9	49.5	42.6	21.9	9.8	11.0	25.9
2002	68.5	2.7	11.8	4.5	49.5	40.2	20.4	8.9	10.9	28.3
2003	69.8	2.0	12.4	4.9	50.4	39.7	21.0	9.4	9.3	30.1
2004	70.2	2.2	11.9	4.7	51.4	39.3	20.1	9.8	9.4	30.9

#### 2. Euro area – by issuer, maturity and currency denomination

	Total	Issued by <sup>4</sup> )				0	riginal matu	rity	R	esidual maturi	ty	Currenci	ies
		Central gov.	State gov.	Local gov.	Social security funds	Up to 1 year	Over 1 year	Variable interest rate	Up to 1 year	Over 1 year and up to 5 years	Over 5 years	Euro or participating currencies 5)	Other currencies
	1	2	3	4	5	6	7	8	9	10	11	12	13
1995	74.0	61.8	5.6	5.9	0.8	11.5	62.5	7.5	18.6	26.7	28.8	71.8	2.2
1996	75.4	63.1	5.9	5.8	0.5	11.2	64.2	7.0	20.0	26.1	29.2	73.1	2.2
1997	74.5	62.4	6.1	5.4	0.6	9.7	64.8	6.8	19.4	25.9	29.3	72.3	2.2
1998	73.1	61.3	6.1	5.3	0.4	8.5	64.6	6.4	16.7	27.0	29.4	71.0	2.1
1999	72.4	60.9	6.1	5.1	0.3	7.3	65.0	5.7	15.1	27.9	29.3	70.4	1.9
2000	69.9	58.7	5.9	4.9	0.3	6.5	63.4	5.0	15.0	28.4	26.5	68.1	1.8
2001	68.6	57.4	6.1	4.8	0.3	6.8	61.7	3.6	15.6	26.4	26.6	67.1	1.5
2002	68.5	57.0	6.3	4.8	0.3	7.6	60.9	3.4	16.4	25.2	26.8	67.2	1.3
2003	69.8	57.4	6.6	5.2	0.6	7.7	62.1	3.5	15.3	26.4	28.1	68.8	1.0
2004	70.2	57.9	6.7	5.2	0.4	7.5	62.7	3.4	15.5	26.8	27.9	69.3	0.9

#### 3. Euro area countries<sup>6)</sup>

	BE	DE	GR	ES	FR	IE	IT	LU	NL	AT	РТ	FI
	1	2	3	4	5	6	7	8	9	10	11	12
2001 2002 2003 2004	108.3 105.8 100.4 96.2	59.6 61.2 64.8 66.4	114.4 111.6 108.8 109.3	56.3 53.2 49.4 46.9	56.8 58.8 63.2 65.1	35.9 32.4 31.5 29.8	110.9 108.3 106.8 106.5	6.7 6.8 6.7 6.6	51.5 51.3 52.6 53.1	67.0 66.7 65.1 64.3	53.6 56.1 57.7 59.4	43.6 42.3 45.2 45.1

Sources: ECB for euro area aggregated data; European Commission for data relating to countries' debt.
Gross general government debt at nominal value and consolidated between sub-sectors of government. Holdings by non-resident governments are not consolidated. Data are partially estimated.

2) Holders resident in the country whose government has issued the debt.

3) Includes residents of euro area countries other than the country whose government has issued the debt.

4) Excludes debt held by general government in the country whose government has issued it.

Before 1999, this comprises debt in ECU, in domestic currency and in the currencies of other Member States which have adopted the euro.
Ratios are computed using GDP excluding financial intermediation services indirectly measured (FISIM).



#### 6.3 Change in debt <sup>1)</sup>

#### (as a percentage of GDP)

#### 1. Euro area - by source, financial instrument and sector of the holder

	Total				Financial	instruments	s		Ho	ders			
		Borrowing requirement <sup>2)</sup>	Valuation effects 3)	Other changes in volume <sup>4)</sup>	Aggregation effect <sup>5)</sup>	Coins and deposits	Loans	Short-term securities	Long-term securities	Domestic creditors <sup>6)</sup>	MFIs	Other financial corporations	Other creditors <sup>7)</sup>
	1	2	3	4	5	6	7	8	9	10	11	12	13
1996	3.9	4.2	0.0	0.0	-0.4	0.1	0.1	0.2	3.4	2.5	1.7	2.0	1.4
1997	2.0	2.2	0.4	-0.4	-0.1	0.0	-0.3	-1.1	3.3	-0.1	-0.4	1.8	2.1
1998	1.8	2.0	0.0	0.0	-0.1	0.1	-0.3	-0.6	2.7	-1.0	-0.5	1.5	2.8
1999	2.0	1.6	0.5	0.0	-0.1	0.2	-0.2	-1.2	3.1	-1.9	-0.1	-2.3	3.9
2000	1.0	0.9	0.2	0.0	0.0	0.0	-0.4	-0.4	1.9	-2.2	-1.8	-0.5	3.3
2001	1.8	1.6	0.0	0.1	0.0	0.2	-0.1	0.4	1.4	-0.2	-0.5	-0.1	2.0
2002	2.2	2.5	-0.4	0.1	0.0	0.1	-0.2	0.8	1.6	-1.0	-0.7	-0.6	3.2
2003	3.1	3.3	-0.1	0.0	0.0	-0.6	0.9	0.5	2.3	0.6	1.1	0.8	2.5
2004	3.1	3.2	0.0	-0.1	0.0	0.2	0.0	0.0	2.9	1.1	-0.1	0.7	2.0

#### 2. Euro area - deficit-debt adjustment

#### Change in debt Surplus (+)<sup>8)</sup> Deficit-debt adjustment<sup>9)</sup> Total Transactions in main financial assets held by general government Other Other 10) Valuation changes in volume Exchange effects Total Securities 11) Loans Currency Shares and rate effects other Privatisations Equity and deposits equity injections 10 14 9 11 12 13 -4.3 -2.7 -2.3 -1.3 0.0 -0.4 -0.7 -0.5 0.7 1.1 -0.2 -0.7 -0.6 0.2 0.2 0.2 -0.3 -0.2 0.0 1996 3.9 2.0 1.8 2.0 1.0 1.8 2.2 3.1 3.1 -0.1 -0.5 -0.4 0.0 -0.1 0.0 0.0 0.0-0.1 -0.1-0.1 0.0 0.0 0.1 1997 1998 0.1 0.1 0.0 -0.5 -0.4 0.4 0.0 0.2 0.0 -0.4 0.0 1999 2000 0.0 1.0 0.5 0.7 0.1 0.2 -0.6 0.0 -0.7 -0.4 0.5 0.2 0.3 0.1 0.0 0.0 0.2 -0.1 $\begin{array}{c} 0.0\\ 0.2\\ 0.1\\ 0.1\\ 0.1\\ 0.1\end{array}$ -1.9 -2.5 -3.0 -2.7 -0.1 -0.3 0.1 0.4 -0.5 0.1 0.1 -0.6 0.0 0.0 0.2 0.1 0.0 0.0 0.1 0.0 -0.4 -0.1 0.0 0.2 -0.1 0.1 0.2 2001 2002 0.1 0.1 -0.1 0.0 -0.3 -0.3 $0.0 \\ 0.0$ 0.1 0.1 0.1 0.0 0.2 0.1 -0.1 2003 -0.4 -0.1 0.0 2004 0.3 -0.3 0.0 -0.1

Source: ECB.

1) Data are partially estimated. Annual change in gross nominal consolidated debt is expressed as a percentage of GDP, i.e. [debt(t) - debt(t-1)] + GDP(t).

2) The borrowing requirement is by definition equal to transactions in debt.

3) Includes, in addition to the impact of foreign exchange movements, effects arising from measurement at nominal value (e.g. premia or discounts on securities issued).

Includes, in particular, the impact of the reclassification of units and certain types of debt assumption.
 The difference between the changes in the aggregated debt, resulting from the aggregation of countries' debt, and the aggregation of countries' change in debt is due to variations in the exchange rates used for aggregation before 1999.

6) Holders resident in the country whose government has issued the debt.

7) Includes residents of euro area countries other than the country whose government has issued the debt.

8) Including proceeds from sales of UMTS licences.

9) The difference between the annual change in gross nominal consolidated debt and the deficit as a percentage of GDP.

10) Mainly composed of transactions in other assets and liabilities (trade credits, other receivables/payables and financial derivatives).

11) Excluding financial derivatives.



### 6.4 Quarterly revenue, expenditure and deficit/surplus 1)

#### 1. Euro area - quarterly revenue

	Direct taxes	Indirect taxes	Social					
			contributions	Sales	Property income		Capital taxes	fiscal burden <sup>2)</sup>
1	2 3	4	5	6	7	8	9	10
1999 Q2 47.8	47.2 13.4	13.5	16.0	2.1	1.4	0.6	0.3	43.1
Q3 44.9 Q4 50.8	44.3 11.8 50.1 14.2	13.0 14.5	16.0 16.7	2.0 2.8	0.8 0.9	0.5 0.7	0.3 0.3	41.0 45.8
2000 Q1 43.7	43.1 11.1	14.5	15.5	1.9	0.7	0.6	0.3	39.9
Q2 47.8	47.2 13.9	13.4	15.8	2.1	1.2	0.6	0.3	43.3
Q3 44.5	44.0 12.0	12.7	15.8	2.0	0.8	0.5	0.2	40.7
Q4 50.0	49.4 13.9	14.1	16.6	2.9	0.9	0.6	0.3	44.9
2001 Q1 42.6	42.1 10.5	12.8	15.3	1.8	0.9	0.5	0.2	38.9
Q2 47.1	46.7 13.5	13.0	15.6	2.0	1.7	0.5	0.2	42.4
Q3 43.8	43.4 11.7	12.4	15.6	1.9	0.9	0.4	0.3	40.0
Q4 49.3	48.8 13.6	13.9	16.3	3.0	1.1	0.5	0.3	44.1
2002 Q1 42.3	41.8 10.2	12.8	15.5	1.7	0.8	0.5	0.2	38.8
Q2 45.9	45.3 12.6	12.8	15.5	2.0	1.6	0.6	0.3	41.2
Q3 43.9	43.4 11.3	12.9	15.5	2.0	0.8	0.5	0.3	40.0
Q4 49.3	48.7 13.5	14.1	16.3	3.0	0.9	0.6	0.3	44.2
2003 Q1 42.3	41.8 9.9	12.9	15.6	1.8	0.7	0.5	0.2	38.7
Q2 46.4	44.8 12.2	12.8	15.8	2.0	1.3	1.6	1.3	42.0
Q3 43.3	42.7 10.9	12.8	15.6	1.9	0.7	0.5	0.2	39.6
Q4 49.6	48.5 13.2	14.3	16.3	2.9	0.8	1.1	0.3	44.1
2004 Q1 41.9	41.3 9.7	13.0	15.4	1.7	0.7	0.5	0.3	38.4
Q2 45.2	44.3 12.2	13.0	15.4	2.0	0.9	0.9	0.6	41.3
Q3 43.0	42.5 10.8	12.8	15.5	1.9	0.7	0.5	0.3	39.3
Q4 49.7	48.6 13.1	14.6	16.3	2.9	0.8	1.1	0.4	44.3
2005 Q1 42.6	42.0 10.1	13.1	15.5	1.7	0.7	0.6	0.2	38.9
Q2 44.6	43.9 11.9	13.0	15.3	2.0	0.9	0.7	0.3	40.5

#### 2. Euro area - quarterly expenditure and deficit/surplus

	Total			Current	expenditu	ire			Cap	ital expendit	ure	Deficit (-)/ surplus (+)	Primary deficit (-)/
		Total 2	Compensation of employees	Intermediate consumption 4	Interest 5	Current transfers	Social benefits 7	Subsidies 8	9	Investment	Capital transfers	12	surplus (+)
1999 Q2	47.4	43.8	10.5	4.6	4.1	24.5	21.3	1.5	3.6	2.4	1.2	0.4	4.5
Q3	47.4	43.7	10.3	4.6	4.0	24.9	21.2	1.6	3.7	2.5	1.2	-2.5	1.5
Q4	50.5	45.8	11.0	5.3	3.7	25.8	22.2	1.7	4.7	3.1	1.7	0.3	4.0
2000 Q1	46.2	42.9	10.2	4.5	4.1	24.1	20.9	1.2	3.4	2.0	1.4	-2.6	1.5
Q2	46.6	43.2	10.4	4.6	3.9	24.3	21.0	1.4	3.5	2.4	1.2	1.2	5.0
Q3	43.3	42.9	10.1	4.5	4.0	24.3	20.9	1.5	0.4	2.5	1.1	1.2	5.2
Q4	49.6	45.8	11.0	5.2	3.7	25.8	21.9	1.6	3.9	3.1	1.6	0.3	4.1
2001 Q1	45.8	42.4	10.1	4.1	4.0	24.1	21.0	1.2	3.4	2.0	1.5	-3.2	$0.8 \\ 4.3 \\ 1.1 \\ 1.8$
Q2	46.7	43.1	10.3	4.6	3.9	24.3	20.9	1.4	3.5	2.4	1.2	0.4	
Q3	46.7	42.9	10.1	4.6	3.9	24.3	21.0	1.5	3.8	2.5	1.3	-2.8	
Q4	51.1	46.0	10.9	5.5	3.6	25.9	22.1	1.6	5.0	3.2	1.8	-1.8	
2002 Q1	46.3	42.8	10.4	4.2	3.7	24.5	21.3	1.2	3.5	1.9	1.5	-4.0	-0.3
Q2	47.0	43.5	10.4	4.9	3.6	24.6	21.3	1.4	3.5	2.3	1.2	-1.1	2.5
Q3	47.3	43.5	10.1	4.7	3.6	25.1	21.5	1.4	3.8	2.5	1.3	-3.4	0.2
Q4	51.0	46.5	11.0	5.5	3.4	26.5	22.7	1.6	4.5	2.8	1.7	-1.7	1.7
2003 Q1	47.0	43.5	10.4	4.4	3.6	25.1	21.7	1.2	3.5	1.9	1.6	-4.7	-1.1
Q2	47.8	44.1	10.5	4.7	3.5	25.5	21.9	1.4	3.6	2.4	1.2	-1.4	2.1
Q3	47.5	43.7	10.3	4.7	3.3	25.4	21.8	1.4	3.8	2.6	1.2	-4.2	-0.9
Q4	51.4	46.5	11.0	5.7	3.1	26.7	22.9	1.5	4.9	3.3	1.7	-1.8	1.3
2004 Q1	46.7	43.3	10.4	4.4	3.3	25.1	21.6	1.1	3.4	2.0	1.4	-4.8	-1.5
Q2	46.9	43.5	10.5	4.7	3.2	25.0	21.6	1.3	3.4	2.4	1.1	-1.7	1.5
Q3	46.5	43.0	10.1	4.5	3.2	25.1	21.6	1.3	3.5	2.5	1.0	-3.5	-0.2
Q4	50.9	45.9	10.9	5.6	3.0	26.3	22.6	1.4	5.0	3.1	1.9	-1.2	1.9
2005 Q1	46.8	43.3	10.3	4.4	3.3	25.3	21.6	1.1	3.5	1.9	1.6	-4.2	-0.9
Q2	46.5	43.0	10.3	4.8	3.1	24.8	21.4	1.3	3.4	2.3	1.1	-1.9	1.3

Source: ECB calculations based on Eurostat and national data.
 Revenue, expenditure and deficit/surplus are based on the ESA 95. Transactions involving the EU budget are not included. Including these transactions would increase both revenue and expenditure by, on average, about 0.2% of GDP. Otherwise, and except for different data transmission deadlines, the quarterly data are consistent with the annual data.

The data are not seasonally adjusted.

2) The fiscal burden comprises taxes and social contributions.



## 6.5 Quarterly debt and change in debt (as a percentage of GDP)

2. Euro area - deficit-debt adjustment

#### 1. Euro area – Maastricht debt by financial instrument<sup>1)</sup>

	Total		Financial in	struments	
	1	Coins and deposits	Loans 3	Short-term securities	Long-term securities 5
2002 Q3	69.2	2.6	11.9	4.5	50.3
Q4	68.5	2.7	11.8	4.5	49.4
2003 Q1	69.7	2.6	11.9	5.2	50.1
Q2	70.2	2.6	11.7	5.7	50.2
Q3	70.5	2.7	11.7	5.5	50.6
Q4	69.8	2.0	12.4	4.9	50.4
2004 Q1 Q2 Q3 Q4	71.4 71.8 71.8 71.8 70.2	2.0 2.1 2.2 2.2	12.5 12.3 12.1 11.9	5.4 5.7 5.5 4.7	51.5 51.7 51.9 51.4
2005 Q1	71.7	2.1	12.0	4.9	52.7
Q2	72.3	2.3	11.7	5.1	53.2

De		Change in debt	
Transactions in main financial a	Total		

	Change in debt	Deficit (-)/ surplus (+)				Deficit-debt	adjustment				Memo: Borrowing	
			Total	Transacti	ions in main fina	ncial assets held	by general go	overnment	Valuation effects and other changes	Other	requirement	
				Total         Currency and deposits         Securities         Loans         Shares and other equity         in volume								
	1	2	3	4	5	6	7	8	9	10	11	
2002 Q3	2.1	-3.4	-1.2	-2.4	-2.5	-0.1	0.0	0.2	0.7	0.5	1.5	
Q4	-0.7	-1.7	-2.4	0.6	0.3	0.0	0.4	-0.1	-2.8	-0.2	2.1	
2003 Q1	7.1	-4.7	2.4	2.8	1.9	0.2	0.5	0.3	-0.7	0.3	7.8	
Q2	3.5	-1.4	2.2	3.1	2.2	0.0	0.1	0.9	-0.7	-0.3	4.3	
Q3	3.0	-4.2	-1.2	-1.6	-1.6	-0.1	0.0	0.1	0.0	0.4	3.0	
Q4	-0.8	-1.8	-2.6	-3.6	-2.2	-0.1	-0.6	-0.7	0.8	0.1	-1.6	
2004 Q1	9.0	-4.8	4.2	1.9	1.2	0.2	0.5	0.0	-0.1	2.4	9.1	
Q2	5.0	-1.7	3.3	3.8	3.4	0.3	0.0	0.1	-0.3	-0.2	5.3	
Q3	2.4	-3.5	-1.0	-0.8	-1.2	0.1	0.1	0.1	-0.5	0.3	2.9	
Q4	-3.4	-1.2	-4.6	-3.5	-2.6	-0.2	0.1	-0.7	0.4	-1.5	-3.8	
2005 Q1	7.6	-4.2	3.5	2.2	1.7	0.4	0.1	0.0	1.1	0.2	6.6	
Q2	4.7	-1.9	2.8	3.3	2.7	0.4	0.1	0.1	-0.1	-0.4	4.7	

-3.0

-4.0

2001

2002

### C26 Deficit, borrowing requirement and change in debt (four-quarter moving sum as a percentage of GDP)







2003

2004

-3.0

-4.0

Source: ECB calculations based on Eurostat and national data. 1) The stock data in quarter t are expressed as a percentage of the sum of GDP in t and the previous three quarters.





### EXTERNAL TRANSACTIONS AND POSITIONS

### 7.1 Balance of payments

#### 1. Summary balance of payments

		Cu	rrent accou	unt		Capital	Net lending/			Financial	account			Errors and
	Total	Goods	Services	Income	Current transfers	account	borrowing to/from rest of the world (columns 1+6)	Total	Direct investment	Portfolio investment	Financial derivatives		Reserve assets	omissions
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
2002 2003 2004	57.3 33.9 45.6	130.2 108.2 106.7	13.7 19.5 28.3	-37.6 -37.4 -33.1	-49.0 -56.4 -56.3	10.2 12.9 17.4	67.6 46.8 63.0	-15.2 4.1 -8.3	21.9 -1.7 -46.8	136.3 68.9 71.2	-11.0 -11.2 -4.8	-159.9 -80.2 -40.4	-2.6 28.2 12.5	-52.3 -50.9 -54.7
2004 Q3 Q4	11.9 15.5	24.5 21.6	9.2 6.3	-4.0 1.9	-17.8 -14.3	4.1 5.9	15.9 21.4	1.0 2.3	8.4 -19.9	8.8 30.7	-2.3 -4.1	-17.4 -6.8	3.5 2.4	-16.9 -23.7
2005 Q1	3.5	15.6 18.8	3.2 9.5	-3.0 -25.5	-12.2 -16.1	1.1	4.6	25.9 46.0	-20.0	3.8	-7.2 3.2	44.5 -55.3	4.8 3.1	-30.5 -36.6
Q2 Q3	-13.3 -3.9	18.8	9.5 8.4	-25.5	-16.1	3.9 2.9	-9.4 -1.0	46.0 18.8	-11.6 -94.0	106.7 70.2	3.2 -7.6	-55.5 48.2	3.1 2.0	-30.0
2004 Oct. Nov.	3.2 4.6	8.9 5.2	4.1 0.8	-4.1 3.1	-5.7 -4.5	0.7 1.1	3.8 5.7	-16.4 13.2	-10.4 -9.1	9.8 -22.4	-3.7 1.8	-13.1 42.9	0.9 0.0	12.6 -18.9
Dec.	7.7	7.5	1.4	3.0	-4.1	4.1	11.9	5.6	-0.4	43.3	-2.1	-36.6	1.5	-17.4
2005 Jan.	-5.8	1.4	0.5	-4.1	-3.6	-0.8	-6.6	18.3	-10.5	-17.3	-3.4	51.0	-1.6	-11.7
Feb. Mar.	5.9 3.5	5.9 8.3	1.2 1.5	1.4 -0.4	-2.6 -6.0	1.1 0.8	6.9 4.3	27.1 -19.4	-2.7 -6.8	23.3 -2.1	1.3 -5.0	0.4 -7.0	4.9 1.5	-34.0 15.1
Apr.	-10.7	8.3 4.0	2.6	-0.4	-0.0	0.8	-10.5	-19.4	-0.8	-2.1	-0.4	-7.0	-0.8	22.0
May	-2.8	6.1	3.0	-6.7	-5.3	1.6	-1.2	39.5	8.2	19.7	0.6	8.5	2.6	-38.3
June	0.2	8.7	3.9	-5.8	-6.5	2.1	2.3	18.0	-8.2	101.5	3.0	-79.7	1.3	-20.3
July	1.3	9.9	3.8	-6.8	-5.6	0.9	2.2	-1.9	-81.2	75.3	0.7	0.8	2.4	-0.3
Aug.	-1.7	2.1	1.8	0.0	-5.6	0.8	-0.9	-5.3	-15.0	-19.5	-6.3	35.6	-0.1	6.2
Sep.	-3.5	4.7	2.9	-3.6	-7.5	1.2	-2.4	26.0	2.2	14.3	-2.0	11.9	-0.3	-23.7
Oct.	-9.2	2.3	3.5	-9.5	-5.5	0.8	-8.4	-6.6	-0.9	-5.1	-1.4	0.9	0.1	15.0
						12-moi	nth cumulated	transaction	S					
2005 Oct.	-10.6	66.1	26.8	-42.4	-61.1	13.9	3.4	102.8	-136.1	196.4	-13.4	44.5	11.5	-106.2





### C29 B.o.p. net direct and portfolio investment (EUR billions)



Source: ECB.

Monthly Bulletin January 2006 S 55

## 7.1 Balance of payments (EUR billions; transactions)

#### 2. Current and capital accounts

					Cu	irrent accour	ıt					Capital acc	count
		Total		Goods		Service	es	Income	e	Current trai	nsfers		
	Credit	Debit	Net	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit
	1	2	3	4	5	6	7	8	9	10	11	12	13
2002 2003 2004	1,723.5 1,691.0 1,840.7	1,666.2 1,657.1 1,795.1	57.3 33.9 45.6	1,061.6 1,041.2 1,133.1	931.4 933.0 1,026.4	331.1 331.9 360.3	317.4 312.3 332.0	245.7 236.3 266.2	283.3 273.6 299.3	85.1 81.7 81.0	134.1 138.1 137.3	19.1 23.7 24.0	8.9 10.8 6.6
2004 Q3 Q4 2005 Q1 Q2 Q3	458.4 485.0 460.5 496.7 497.0	446.5 469.5 457.0 510.0 500.9	11.9 15.5 3.5 -13.3 -3.9	280.7 299.9 278.9 304.3 309.4	256.2 278.3 263.3 285.5 292.7	98.0 92.6 83.9 95.0 101.7	88.8 86.2 80.7 85.5 93.3	63.7 74.5 65.3 80.8 70.7	67.8 72.6 68.4 106.3 81.0	15.9 18.0 32.3 16.6 15.2	33.7 32.3 44.5 32.7 33.9	5.8 7.6 4.8 5.7 4.4	1.7 1.7 3.7 1.8 1.5
2005 Aug. Sep. Oct.	160.0 169.8 168.4	161.7 173.3 177.6	-1.7 -3.5 -9.2	97.9 108.0 108.0	95.7 103.3 105.7	33.2 33.2 33.2	31.4 30.3 29.7	23.4 24.0 23.1	23.4 27.5 32.6	5.6 4.6 4.2	11.1 12.1 9.7	1.4 1.7 1.7	0.6 0.5 0.9
						easonally adju							
2004 Q3 Q4 2005 Q1 Q2 Q3	460.3 470.3 473.7 485.9 501.3	454.8 460.4 469.2 485.4 512.9	5.4 9.9 4.5 0.5 -11.6	282.5 287.8 289.0 297.4 313.2	262.7 268.0 267.8 278.3 301.6	91.1 91.0 93.2 93.3 94.8	83.4 83.8 85.8 87.0 88.0	66.1 71.4 70.7 73.7 73.7	73.5 74.8 77.4 85.6 88.1	20.6 20.1 20.8 21.5 19.7	35.3 33.9 38.2 34.6 35.2		
2005 Feb. Mar. May June July Aug. Sep. Oct.	$\begin{array}{c} 157.1 \\ 160.1 \\ 161.2 \\ 162.3 \\ 162.4 \\ 165.3 \\ 168.3 \\ 167.8 \\ 166.8 \end{array}$	152.8 158.8 161.5 161.7 162.2 168.5 173.3 171.1 175.8	4.3 1.3 -0.3 0.6 0.2 -3.2 -5.0 -3.3 -9.0	95.8 97.3 98.0 99.6 99.9 102.4 105.0 105.8 103.0	88.8 90.4 91.8 93.8 92.8 98.4 102.7 100.5 102.7	31.1 31.2 30.9 31.0 31.4 31.7 31.4 31.7 32.1	28.7 28.8 29.4 29.3 29.6 29.2 29.2 29.6	23.3 24.6 24.9 24.7 24.1 24.3 24.9 24.5 25.2	24.6 27.0 30.0 27.1 28.6 29.5 29.2 29.4 32.6	6.9 7.0 7.5 6.9 7.1 6.8 7.0 5.9 6.5	10.7 12.7 11.4 11.6 11.5 11.0 12.2 12.1 10.9		- - - - - - - - - - - - - -

### C30 B (EUR bi











#### EURO AREA STATISTICS

External transactions and positions

## 7.1 Balance of payments (EUR billions)

#### 3. Income account

(transactions)

	Compens of emplo							Investr	nent income					
			Tota	ıl		Direct inv	estment			Portfolio i	nvestment		Other inve	stment
					Equit	у	Debt		Equit	/	Debt			
	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit	Credit	Debit
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
2002 2003	15.1 14.8	6.2 6.1	230.6 221.4	277.1 267.5	54.9 59.4	55.2 52.6	$7.5 \\ 10.0$	7.1 9.7	19.9 18.6	52.1 53.5	65.4 65.7	71.9 79.1	83.0 67.8	90.8 72.6
2005	15.2	6.2	251.0	293.1	77.8	67.5	11.7	12.3	24.0	57.3	74.6	84.3	63.0	71.6
2004 Q2	3.7	1.6	64.8	90.7	20.2	20.0	3.1	3.1	8.2	26.2	17.8	23.9	15.4	17.5
Q3	3.8 4.0	1.8 1.6	59.9 70.5		16.5 25.5	16.3 15.1	2.5 3.0	2.9 3.5	5.7 5.4	11.3 10.4	19.3 19.5	17.6 22.7	15.9 17.1	17.8 19.4
Q4 2005 Q1 Q2	4.0 3.7 3.7	1.6 1.4 1.7	70.5 61.7 77.1	67.0 104.5	25.5 15.6 24.2	13.1 13.2 23.8	2.8 3.1	3.5 2.9 3.5	5.4 6.1 9.6	10.4 11.2 30.2	19.5 19.2 21.7	19.1 24.1	17.1 18.0 18.5	20.5 22.8

#### 4. Direct investment

(net transactions)

			By resid	ent units a	ibroad					By non-reside	nt units in	the euro a	rea	
	Total		Equity capital einvested earni	ngs	(mostly	Other capital inter-company	v loans)	Total		Equity capital einvested earn	ings	(mostly	Other capital inter-company	loans)
		Total	MFIs excluding Eurosystem	Non- MFIs	Total	MFIs excluding Eurosystem	Non- MFIs		Total	MFIs excluding Eurosystem	Non- MFIs	Total	MFIs excluding Eurosystem	Non- MFIs
	1	2	3	4	5	6	7	8	9	10	11	12	13	14
2002 2003 2004	-170.1 -139.7 -130.8	-175.9 -122.6 -141.6	-22.8 -2.3 -18.5	-153.2 -120.3 -123.0	5.9 -17.1 10.8	0.0 0.0 0.1	5.9 -17.1 10.7	192.0 138.0 84.0	141.4 120.4 76.6	2.1 3.1 1.2	139.3 117.4 75.5	50.5 17.6 7.4	0.5 0.1 0.7	50.1 17.5 6.7
2004 Q3 Q4 2005 Q1 Q2 Q3	-15.4 -65.2 -32.3 -27.0 -104.0	-26.9 -68.6 -15.5 -21.4 -80.5	-1.2 -8.6 -2.4 -2.1 -3.8	-25.7 -60.0 -13.0 -19.3 -76.7	11.4 3.4 -16.9 -5.7 -23.5	0.0 0.1 0.1 0.0 0.0	11.4 3.3 -16.9 -5.7 -23.5	23.9 45.4 12.3 15.4 10.0	19.6 36.4 12.2 4.6 -2.7	0.6 1.0 0.3 0.6 0.5	19.0 35.4 11.8 4.0 -3.2	4.3 9.0 0.1 10.9 12.7	0.4 -0.1 0.3 -0.1 0.1	3.9 9.1 -0.2 11.0 12.5
2004 Oct. Nov. Dec.	-29.6 -29.9 -5.7	-23.0 -25.7 -20.0	-0.2 -13.2 4.8	-22.7 -12.5 -24.8	-6.6 -4.2 14.3	0.0 0.1 0.0	-6.7 -4.3 14.3	19.3 20.8 5.3	11.0 11.4 14.0	0.3 0.2 0.6	10.7 11.2 13.5	8.3 9.5 -8.8	-0.1 0.2 -0.2	8.4 9.2 -8.6
2005 Jan. Feb. Mar. Apr. May June July Aug. Sep. Oct.	-13.9 -5.9 -12.5 -16.0 8.5 -19.6 -82.6 -11.7 -9.6 -4.3	-7.7 -2.1 -5.7 1.5 -4.3 -18.5 -74.8 -4.0 -1.7 7.0	-0.5 -1.2 -0.8 -1.9 -0.6 0.4 -2.7 -0.5 -0.6 0.6	-7.2 -0.9 -4.9 3.4 -3.7 -18.9 -72.0 -3.5 -1.1 6.5	-6.2 -3.9 -6.8 -17.5 12.9 -1.1 -7.8 -7.8 -7.9 -11.4	$\begin{array}{c} 0.0\\ 0.0\\ 0.1\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\ 0.0\\$	-6.2 -3.9 -6.9 -17.5 12.9 -1.1 -7.8 -7.7 -7.9 -11.4	3.4 3.2 5.7 4.4 -0.4 11.4 -3.3 11.8 3.4	4.7 2.9 4.5 7.8 -2.6 -0.6 3.6 -3.2 -3.1 6.3	$\begin{array}{c} 0.1 \\ 0.3 \\ -0.1 \\ -0.1 \\ 0.3 \\ 0.4 \\ 0.2 \\ 0.1 \\ 0.2 \\ 0.1 \end{array}$	4.7 2.6 4.6 7.9 -2.9 -1.0 3.4 -3.3 -3.3 6.2	-1.3 0.3 1.2 -3.4 2.2 12.0 -2.2 -0.1 14.9 -2.9	$\begin{array}{c} 0.1\\ 0.1\\ 0.2\\ 0.1\\ 0.0\\ -0.3\\ 0.0\\ -0.1\\ 0.2\\ 0.0\\ \end{array}$	-1.4 0.2 1.0 -3.5 2.2 12.3 -2.2 0.0 14.7 -2.9



#### 7.1 Balance of payments (EUR billions; transactions)

#### 5. Portfolio investment by instrument and sector of holder

		E	quity							Debt ins	struments				
							Bonds	and note	s			Money mar	rket instru	iments	
		Assets			Liabilities		Assets			Liabilities		Assets			Liabilities
	Eurosystem	MFIs excluding Eurosystem	Non-M	IFIs General gov.		Eurosystem	MFIs excluding Eurosystem	Non-	MFIs General gov.		Eurosystem	MFIs excluding Eurosystem	Non-	MFIs General gov.	
	1	-0.3 -13.9 -63.1 -2.6 11				6	7	8	9	10	11	12	13	14	15
2002 2003 2004					85.4 111.6 137.5	-0.6 -2.4 1.2	-17.4 -45.0 -81.4	-70.7 -129.4 -94.7	-0.9 -0.2 -2.1	167.4 197.5 255.8	2.0 0.2 -0.1	-31.9 -45.9 -43.3	-18.8 21.6 -9.7	-1.1 0.6 0.1	59.8 38.0 8.6
2004 Q3 Q4 2005 Q1 Q2 Q3	0.0 0.0 0.0 0.0 -0.1	-2.5 -0.9 -27.5 21.7 -5.8	-7.3 -23.8 -20.6 -22.4 -15.6	-1.0 -0.2 -0.9 -0.6	39.5 82.9 36.5 27.6 137.9	0.7 0.6 -0.1 -0.7 -0.4	-23.3 -20.9 -35.4 -40.5 -19.6	-20.4 -27.2 -38.9 -33.1 -55.7	-0.4 -0.5 -0.3 -0.1	51.7 41.3 45.3 155.7 24.3	0.0 -0.1 0.3 -0.4 0.1	-14.7 -12.2 5.9 -9.7 -7.4	-4.8 5.6 -6.6 -4.6 -8.7	-0.7 4.3 -3.7 -2.3	-10.1 -14.6 45.1 13.1 21.2
2004 Oct. Nov Dec.	0.0	-3.7 -9.1 11.9	-10.7 -7.2 -5.9	-	22.1 22.7 38.2	0.2 0.3 0.1	-13.4 -5.9 -1.5	-6.8 -12.5 -7.9	-	18.0 8.0 15.3	-0.1 0.2 -0.2	0.7 -14.3 1.5	0.0 4.7 0.9	- - -	3.6 -9.2 -9.0
2005 Jan. Feb. Mar Apr. May June July Aug	. 0.0 0.0 0.0 0.0 -0.1 . 0.0	-9.2 -16.5 -1.8 9.9 6.7 5.1 -3.5 2.0 -4.2	-7.7 -3.7 -9.3 -5.3 -15.5 -1.6 -11.1 -4.8 0.3		10.5 9.2 16.8 -47.5 22.8 52.3 105.0 23.0 9.8	-0.1 -0.2 -0.9 -0.1 0.2 0.3 -0.5 -0.2	-27.0 -4.0 -4.5 -13.3 -16.1 -11.1 -1.8 -5.2 -12.6	-2.1 -16.4 -20.4 -10.6 -5.3 -17.2 -17.6 -15.9 -22.1		4.9 37.5 2.9 51.9 27.8 75.9 2.6 -10.7 32.4	0.2 0.1 0.0 -0.3 0.0 0.0 -0.6 0.3 0.4	-4.1 17.2 -7.3 -10.6 -2.4 3.4 0.4 -11.6 3.8	-5.9 -1.9 1.2 2.0 -6.2 -0.4 -0.1 -1.6 -7.0		23.1 1.9 20.1 10.1 8.0 -5.1 1.8 5.6 13.8
Sep. Oct.	0.0	-4.2 4.9	-7.4	-	-10.0	-0.2	-12.6	-22.1	-	32.4 20.0	0.4	5.8 7.0	-7.0	-	4.1

#### 6. Other investment by sector

	Т	otal	Euro	osystem		General governme			MFIs	(excludi	ing Eurosyst	tem)			Other sect	ors
								Т	otal	Lon	g-term	Shor	rt-term			
	Assets	Liabilities	Assets	Liabilities	Assets		Liabilities	Assets	Liabilities	Assets	Liabilities	Assets	Liabilities	Assets		Liabilities
						Currency and deposits									Currency and deposits	
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
2002 2003 2004	-225.4 -254.1 -314.3	65.5 173.9 273.9	-0.9 -0.8 -0.2	19.3 10.0 7.1	-0.1 -0.4 -2.2	-2.0	-8.0 -3.4 -2.6	-168.0 -152.4 -259.6	25.8 134.8 246.9	-34.9 -50.7 -20.0	52.5 52.3 -3.3	-133.1 -101.7 -239.6	-26.7 82.5 250.2	-56.4 -100.5 -52.4	-13.2	28.4 32.6 22.5
2004 Q3 Q4 2005 Q1 Q2 Q3	-32.8 -71.8 -170.6 -160.4 -81.9	15.4 65.0 215.1 105.1 130.0	-1.5 1.4 0.5 -1.3 0.4	3.2 3.5 4.7 0.4 4.3	0.2 3.1 4.0 -7.4 7.9	-0.2 3.7 2.7 -8.4 5.3	2.3 -1.6 0.3 -1.8 1.1	-23.8 -73.8 -126.8 -97.4 -75.1	6.5 58.8 195.8 45.2 118.7	-7.9 0.9 -21.5 -18.7 -37.2	-6.0 -0.9 10.3 22.2 15.1	-15.9 -74.6 -105.2 -78.7 -37.9	12.4 59.7 185.6 23.0 103.6	-7.7 -2.5 -48.3 -54.3 -15.0	-12.9 10.3 -19.0 13.2 -12.0	3.4 4.4 14.2 61.3 5.9
2004 Oct. Nov. Dec.	-15.9 -66.1 10.2	2.9 109.0 -46.9	0.0 0.8 0.7	1.2 2.0 0.3	2.0 -0.7 1.8	2.3 -0.8 2.2	0.1 1.0 -2.7	-9.6 -59.9 -4.3	-0.5 97.1 -37.9	8.4 -0.3 -7.2	7.0 3.2 -11.2	-18.0 -59.5 2.9	-7.6 94.0 -26.7	-8.3 -6.3 12.1	-1.3 -0.9 12.5	2.0 9.0 -6.7
2005 Jan. Feb. Mar. Apr. May June	-50.4 -65.5 -54.7 -120.3 -13.0 -27.1 -40.4	101.4 65.9 47.8 136.1 21.5 -52.6 41.2	0.7 0.1 -0.2 0.1 -0.8 -0.7 0.0	3.9 -3.5 4.3 -0.2 -0.2 0.8 -1.1	0.2 -1.8 5.5 -5.1 0.2 -2.5 -0.6	-1.3 0.3 3.7 -5.2 2.3 -5.5 -4.3	2.6 -4.3 2.0 -2.1 0.5 -0.2 0.8	-33.8 -60.8 -32.1 -98.0 17.5 -16.8 -34.4	95.4 61.8 38.6 94.5 20.9 -70.2 42.0	-9.0 -8.2 -4.4 -9.1 -3.4 -6.2 -10.3	12.9 4.6 -7.2 0.0 10.9 11.3 3.4	-24.9 -52.6 -27.8 -89.0 20.9 -10.6 -24.1	82.5 57.2 45.8 94.5 10.0 -81.5 38.6	-17.4 -3.0 -27.9 -17.2 -30.0 -7.1 -5.4	-16.1 5.2 -8.0 11.6 -4.2 5.8 -5.5	-0.5 11.9 2.9 43.9 0.4 17.0 -0.5
July Aug. Sep. Oct.	-40.4 28.3 -69.7 -55.1	41.2 7.3 81.5 56.0	0.0 0.2 0.2 0.0	-1.1 0.8 4.5 -1.0	-0.6 6.3 2.2 0.3	-4.3 8.6 1.0 -0.6	0.8 0.5 -0.2 2.0	-34.4 24.1 -64.8 -49.1	42.0 0.7 76.0 50.7	-10.3 -6.4 -20.4 -5.0	3.4 4.1 7.6 0.6	-24.1 30.6 -44.4 -44.1	-3.4 68.4 50.1	-5.4 -2.3 -7.3 -6.3	-5.5 -1.9 -4.6 -0.4	-0.5 5.3 1.2 4.3



External transactions and positions

#### 7.1 Balance of payments (EUR billions; transactions)

#### 7. Other investment by sector and instrument

		Eu	rosystem					General	l governme	nt		
	Assets		Liabilitie	es			Assets	3			Liabilities	
	Loans/currency and	Other assets	Loans/currency and	Other liabilities	Trade credits	Loans	currency a	nd deposits	Other assets	Trade credits	Loans	Other liabilities
	deposits		deposits			Total	Loans	Currency and deposits				
	1	2	3	4	5	6	7	8	9	10	11	12
2002	-0.9	0.0	19.3	0.0	1.5	-0.4	-	-	-1.1	0.0	-7.8	-0.3
2003	-0.8	0.0	10.0	0.0	-0.1	0.7	-	-	-1.0	0.0	-3.7	0.3
2004	0.1	-0.3	7.1	0.1	0.0	-0.3	1.8	-2.0	-2.0	0.0	-2.6	0.0
2004 Q2	0.8	0.0	1.5	0.2	0.0	-4.6	0.3	-4.9	-0.4	0.0	2.9	0.2
Q3	-1.5	0.0	3.3	-0.1	0.0	0.5	0.7	-0.2	-0.3	0.0	2.2	0.1
Q4	1.7	-0.3	3.5	0.0	0.0	3.6	-0.1	3.7	-0.5	0.0	-1.6	-0.1
2005 Q1	0.5	0.0	4.7	0.0	0.0	4.4	1.7	2.7	-0.5	0.0	0.6	-0.2
Q2	-1.2	-0.1	0.4	0.0	0.0	-6.8	1.6	-8.4	-0.5	0.0	-1.8	0.0

	M	FIs (exclu	ding Eurosystem)					Othe	er sectors			
	Assets		Liabiliti	es			Assets	8			Liabilities	
	Loans/currency and	Other assets	Loans/currency and	Other liabilities	Trade credits	Loans	currency a	nd deposits	Other assets	Trade credits	Loans	Other liabilities
	deposits		deposits			Total	Loans	Currency and deposits				
	13	14	15	16	17	18	19	20	21	22	23	24
2002	-163.0	-5.0	27.9	-2.1	-1.9	-51.0	-	-	-3.5	-3.7	25.5	6.6
2003	-151.9	-0.5	134.8	-0.1	-1.2	-97.1	-	-	-2.3	4.1	28.3	0.1
2004	-256.5	-3.1	244.0	2.9	-6.0	-41.3	-28.1	-13.2	-5.0	8.6	11.7	2.2
2004 Q2	-4.8	-0.6	22.5	0.1	-4.9	-8.9	-16.9	8.0	-0.9	3.3	-6.4	-2.2
Q3	-22.1	-1.7	4.9	1.5	1.8	-8.7	4.2	-12.9	-0.8	0.0	-0.3	3.7
Q4	-75.6	1.8	59.1	-0.3	-0.1	-0.8	-11.2	10.3	-1.6	2.5	2.4	-0.5
2005 Q1	-124.8	-1.9	193.0	2.8	-2.8	-42.7	-23.8	-19.0	-2.8	2.9	6.2	5.2
Q2	-97.2	-0.2	44.0	1.2	-5.0	-47.3	-60.5	13.2	-2.0	1.1	58.6	1.6

#### 8. Reserve assets

	Total	Monetary gold	Special drawing	Reserve position in			For	eign exchang	e			Other claims
		<b>8</b> * "	rights	the IMF	Total	Currency and	deposits		Securities		Financial derivatives	
						With monetary authorities and the BIS	With banks		Bonds and notes	Money market instruments		
	1	2	3	4	5	6	7	8	9	10	11	12
2002	-2.6	0.7	0.2	-2.0	-1.5	-1.7	-17.1	0.0	8.5	8.9	-0.2	0.0
2003	28.2	1.7	0.0	-1.6	28.1	-2.5	1.9	-0.1	22.1	6.7	0.1	0.0
2004	12.5	1.2	0.5	4.0	6.9	-3.8	4.0	0.3	18.7	-12.2	-0.1	0.0
2004 Q2	-2.8	0.5	0.1	0.6	-4.0	-3.3	2.2	0.0	5.4	-8.3	0.1	0.0
Q3	3.5	0.0	-0.1	1.5	2.1	2.6	-3.6	0.1	1.4	1.7	0.0	0.0
Q4	2.4	0.8	0.5	1.1	0.0	-3.9	3.4	-0.1	3.4	-2.8	-0.1	0.0
2005 Q1	4.8	0.8	0.0	1.6	2.4	5.2	-1.1	0.0	1.1	-2.7	0.0	0.0
Q2	3.1	1.3	0.0	1.3	0.5	-4.4	1.1	0.0	0.8	3.0	0.0	0.0



7.2	Monetary	presentation	of th	e ba	lance	o f	payments
	(EUR billions:						

				•		actions in the ex		•				Memo: Transactions
	Current and capital	Direct inv		Po	ortfolio inves		Other in	ivestment	Financial derivatives	Errors and	Total of	in the external
	accounts balance	By resident units	By non- resident units	Assets	Lia	bilities	Assets	Liabilities		omissions	columns 1 to 10	counterpart of M3
		abroad (non-MFIs)	in the euro area	Non-MFIs	Equity 1)	Debt instruments <sup>2)</sup>	Non-MFIs	Non-MFIs				
	1	2	3	4	5	6	7	8	9	10	11	12
2002	67.6	-147.3	191.5	-120.6	49.1	217.5	-56.5	20.3	-11.0	-52.3	158.2	170.4
2003	46.8	-137.4	137.9	-170.9	115.9	235.8	-100.9	29.2	-11.2	-50.9	94.3	94.1
2004	63.0	-112.4	83.3	-184.8	127.5	244.5	-54.6	19.9	-4.8	-54.7	127.0	160.7
2004 Q3	15.9	-14.3	23.5	-32.5	38.6	53.2	-7.5	5.8	-2.3	-16.9	63.4	63.7
Q4	21.4	-56.7	45.4	-45.4	90.3	13.3	0.6	2.7	-4.1	-23.7	43.9	57.2
2005 Q1	4.6	-30.0	12.0	-66.2	29.3	71.2	-44.4	14.6	-7.2	-30.5	-46.6	-24.8
Q2	-9.4	-25.0	15.6	-60.2 -79.9	1.8	174.4	-61.7	59.5	3.2	-36.6	61.6	65.0
Q3	-1.0	-100.1	9.9	- /9.9	142.4	39.6	-7.1	7.0	-7.6	-17.8	-14.6	-17.2
2004 Oct.	3.8	-29.4	19.4	-17.5	20.8	18.7	-6.3	2.1	-3.7	12.6	20.5	20.5
Nov.	5.7	-16.8	20.6	-15.0	26.5	-3.4	-7.0	10.0	1.8	-18.9	3.5	10.3
Dec.	11.9	-10.5	5.4	-12.9	43.0	-2.0	13.9	-9.4	-2.1	-17.4	19.8	26.4
2005 Jan.	-6.6	-13.4	3.3	-15.7	7.4	22.9	-17.2	2.1	-3.4	-11.7	-32.2	-15.8
Feb.	6.9	-4.8	3.1	-22.0	16.4	35.8	-4.8	7.6	1.3	-34.0	5.5	13.7
Mar.	4.3	-11.8	5.5	-28.6	5.5	12.5	-22.4	4.9	-5.0	15.1	-19.9	-22.7
Apr.	-10.5	-14.1	4.3	-13.9	-57.9	59.2	-22.4	41.8	-0.4	22.0	8.1	2.3
May	-1.2	9.1	-0.4	-27.0	11.1	33.8	-29.7	0.8	0.6	-38.3	-41.2	-39.5
June	2.3	-20.0	11.7	-19.3	48.6	81.4	-9.6	16.8	3.0	-20.3	94.7	102.2
July	2.2	-79.9 -11.2	1.4 -3.2	-28.8 -22.3	112.0 24.7	2.6 -6.2	-6.0 3.9	0.3	0.7 -6.3	-0.3 6.2	4.3 -9.6	0.6 1.7
Aug.	-0.9	-11.2	-3.2	-22.3 -28.8	24.7 5.8	-6.2	-5.1	5.7 1.0	-0.5	-23.7	-9.6	-19.5
Sep. Oct.	-2.4	-9.0	3.4	-28.8 -15.7	-10.7	43.2	-5.1 -6.0	6.2	-2.0	-23.7	-9.3	-19.5 -9.5
001.	-0.4		5.4	-15.7		h cumulated trar		0.2	-1.4	15.0	-+.5	-9.5
2005 Oct.	3.4	-187.3	66.9	-249.9	232.4	298.0	-112.4	87.8	-13.4	-106.2	19.3	50.2

C32 Main b.o.p. transactions underlying the developments in MFI net external assets (EUR billions; 12-month cumulated transactions)

MFI net external assets

- . . . . current and capital accounts balance
- direct and portfolio equity investment abroad by non-MFIs



- portfolio investment liabilities in the form of debt instruments<sup>2)</sup>

Source: ECB.

1)

Excluding money market fund shares/units. Excluding debt securities with a maturity of up to two years issued by euro area MFIs. 2)



## 7.3 Geographical breakdown of the balance of payments and international investment position (EUR billions)

## **1. Balance of payments: current and capital accounts** *(cumulated transactions)*

	Total		Europ	ean Union (	outside the e	uro area)		Canada	Japan	Switzerland	United States	Other
		Total	Denmark	Sweden	United	Other EU	EU					
					Kingdom	countries	institutions					
2004 Q3 to 2005 Q2	1	2	3	4	5	6	7	8	9	10	11	12
						Credits						
Current account	1,900.5	701.5	39.0	62.8	378.7	161.3	59.6	25.5	49.5	129.6	324.4	670.1
Goods	1,163.8	406.9	26.8	43.4	205.1	131.5	0.2	14.9	33.3	67.3	174.9	466.4
Services	369.5	133.8	7.3	10.3	94.3	17.7	4.2	5.1	10.5	35.6	74.0	110.5
Income	284.4	100.4	4.6	8.7	70.5	10.8	5.8	5.0	5.4	20.5	69.2	84.0
of which: investment income	269.2	95.3	4.5	8.6	69.0	10.6	2.5	4.9	5.3	14.4	67.6	81.7
Current transfers	82.9	60.4	0.4	0.5	8.7	1.3	49.4	0.5	0.2	6.2	6.3	9.2
Capital account	23.9	21.3	0.0	0.1	0.8	0.1	20.3	0.0	0.1	0.3	1.1	1.1
						Debits						
Current account	1,883.0	615.2	34.0	59.1	296.8	137.3	88.0	18.7	79.4	120.8	285.2	763.7
Goods	1,083.3	321.2	25.4	39.4	147.0	109.4	0.0	8.6	51.1	53.9	113.7	534.8
Services	341.3	106.6	6.0	7.9	71.0	21.6	0.1	5.1	7.3	29.5	73.4	119.4
Income	315.1	93.8	2.3	10.9	71.3	4.7	4.6	3.3	20.7	32.2	89.7	75.3
of which: investment income	308.5	90.6	2.2	10.9	70.3	2.6	4.5	3.2	20.6	31.7	88.8	73.6
Current transfers	143.3	93.6	0.3	0.9	7.6	1.6	83.3	1.6	0.3	5.2	8.5	34.2
Capital account	8.9	1.0	0.0	0.1	0.6	0.2	0.1	0.1	0.0	0.4	0.6	6.8
						Net						
Current account	17.5	86.3	5.0	3.7	81.9	24.0	-28.4	6.8	-29.9	8.8	39.2	-93.7
Goods	80.4	85.7	1.3	4.0	58.1	22.1	0.2	6.3	-17.8	13.4	61.3	-68.4
Services	28.2	27.2	1.3	2.3	23.4	-3.9	4.1	0.0	3.2	6.1	0.6	-8.9
Income	-30.7	6.6	2.3	-2.2	-0.8	6.1	1.2	1.7	-15.3	-11.7	-20.5	8.7
of which: investment income	-39.3	4.7	2.3	-2.3	-1.3	8.0	-2.0	1.7	-15.3	-17.3	-21.2	8.2
Current transfers	-60.4	-33.2	0.1	-0.3	1.2	-0.3	-33.8	-1.1	0.0	1.0	-2.1	-25.0
Capital account	15.0	20.3	0.0	0.0	0.3	-0.1	20.2	-0.1	0.1	-0.1	0.5	-5.7

#### 2. Balance of payments: direct investment

(cumulated transactions)

	Total		Europ	ean Union	(outside the	euro area)		Canada	Japan	Switzerland		Offshore financial	Other
		Total	Denmark	Sweden	United	Other EU	EU					centres	
					Kingdom	countries	institutions						
2004 Q3 to 2005 Q2	1	2	3	4	5	6	7	8	9	10	11	12	13
Direct investment	-43.1	-27.0	5.1	-0.7	-31.7	0.4	0.0	-2.0	1.3	20.3	10.1	-14.2	-31.6
Abroad	-140.0	-77.9	-0.1	-5.8	-52.3	-19.7	0.0	1.5	-1.5	7.5	-10.3	-20.7	-38.6
Equity/reinvested earnings	-132.3	-65.1	-2.6	-2.4	-36.9	-23.2	0.0	2.1	-1.3	1.1	-22.0	-22.1	-25.2
Other capital	-7.7	-12.8	2.6	-3.4	-15.4	3.4	0.0	-0.6	-0.3	6.4	11.7	1.4	-13.4
In the euro area	96.9	50.9	5.2	5.1	20.5	20.1	0.0	-3.5	2.8	12.8	20.4	6.5	7.0
Equity/reinvested earnings	72.7	41.8	-1.1	4.0	36.5	2.4	0.0	-4.5	1.1	7.5	15.1	10.1	1.6
Other capital	24.2	9.1	6.2	1.1	-15.9	17.7	0.0	1.0	1.7	5.3	5.3	-3.5	5.4



## 7.3 Geographical breakdown of the balance of payments and international investment position (EUR billions)

## **3.** Balance of payments: portfolio investment assets by instrument *(cumulated transactions)*

	Total		Europe	an Union (	(outside the	euro area)		Canada	Japan	Switzerland		Offshore financial	Other
		Total	Denmark	Sweden	United	Other EU	EU					centres	
					Kingdom	countries	institutions						
2004 Q3 to 2005 Q2	1	2	3	4	5	6	7	8	9	10	11	12	13
Portfolio investment assets	-364.0	-157.1	-5.3	-6.0	-116.4	-20.2	-9.2	-3.6	-36.9	0.2	-54.6	-57.6	-54.3
Equity	-83.4	-23.9	0.5	-1.4	-20.4	-2.5	-0.1	-2.3	-11.7	-0.1	-7.9	-19.6	-17.8
Debt instruments	-280.6	-133.2	-5.8	-4.5	-96.1	-17.7	-9.1	-1.3	-25.3	0.3	-46.7	-38.0	-36.4
Bonds and notes	-239.4	-109.1	-4.0	-6.6	-71.4	-18.1	-8.9	-1.7	-16.6	0.5	-55.9	-23.3	-33.4
Money market instruments	-41.2	-24.1	-1.7	2.1	-24.6	0.4	-0.2	0.3	-8.6	-0.2	9.2	-14.8	-3.1

4. Balance of payments: other investment by sector

(cumulated transactions)

	Total		Europe	an Union	(outside th	e euro area	)	Canada	Japan	Switzerland	United States		Internat. organisa-	
		Total	Denmark	Sweden	United	Other EU	EU					centres	tions	
					Kingdom	countries	institutions							
2004 Q3 to 2005 Q2	1	2	3	4	5	6	7	8	9	10	11	12	13	14
Other investment	-35.0	-73.1	3.6	-0.4	-71.9	-11.1	6.7	-2.5	12.8	-2.0	33.7	-27.5	1.9	21.6
Assets	-435.6	-320.1	-9.6	-4.3	-282.7	-21.4	-2.1	-3.5	7.3	-10.1	-32.3	-46.4	-3.2	-27.4
General government	-0.1	-1.1	-1.3	-0.1	-0.4	1.4	-0.6	-0.3	0.0	-0.1	0.0	0.0	-1.8	3.2
MFIs	-322.7	-232.1	-6.2	-1.8	-200.4	-22.0	-1.5	-2.8	8.8	-8.3	-35.8	-30.2	-1.4	-20.9
Other sectors	-112.9	-86.9	-2.0	-2.3	-81.8	-0.8	0.0	-0.3	-1.5	-1.8	3.5	-16.2	0.0	-9.7
Liabilities	400.6	247.0	13.2	3.8	210.8	10.3	8.8	0.9	5.5	8.2	66.0	18.9	5.1	49.0
General government	-0.8	0.6	0.0	0.1	-0.6	0.0	1.1	0.0	-0.5	0.0	-0.8	0.0	-0.4	0.3
MFIs	318.1	176.1	13.0	3.0	147.0	8.4	4.7	0.6	5.1	8.1	49.0	15.6	5.6	58.0
Other sectors	83.3	70.3	0.2	0.8	64.4	1.9	3.0	0.3	1.0	0.1	17.8	3.2	-0.1	-9.3

#### 5. International investment position

(end-of-period outstanding amounts)

	Total		Europe	an Union	(outside the	e euro area	)	Canada	Japan	Switzerland	United States	Offshore financial	Internat. organisa-	Other
		Total	Denmark	Sweden		Other EU	EU					centres	tions	
					Kingdom	countries	institutions							
2004	1	2	3	4	5	6	7	8	9	10	11	12	13	14
Direct investment	33.1	-273.1	-10.4	-11.8	-361.5	110.8	-0.2	22.7	-4.0	35.3	-63.5	-30.9	0.0	346.7
Abroad	2,265.1	759.8	26.1	71.1	537.6	125.1	0.0	66.8	55.9	220.3	486.6	272.2	0.0	403.5
Equity/reinvested earnings	1,825.7	608.4	22.9	43.8	432.7	108.9	0.0	58.3	50.5	171.0	377.2	255.7	0.0	304.8
Other capital	439.3	151.4	3.1	27.2	104.9	16.2	0.0	8.5	5.4	49.4	109.4	16.5	0.0	98.7
In the euro area	2,231.9	1,032.9	36.5	82.8	899.1	14.3	0.2	44.1	59.8	185.1	550.2	303.0	0.1	56.8
Equity/reinvested earnings	1,642.1	814.3	23.0	67.4	719.4	4.4	0.1	40.4	48.8	129.6	387.7	177.0	0.0	44.2
Other capital	589.9	218.6	13.4	15.4	179.8	9.9	0.1	3.7	11.1	55.4	162.4	126.1	0.0	12.6
Portfolio investment assets	2,984.0	941.1	45.1	100.8	680.8	56.8	57.6	63.4	174.3	91.9	1,050.2	310.3	28.4	324.4
Equity	1,238.7	315.3	6.6	32.9	261.4	14.4	0.0	12.6	109.5	82.3	483.3	106.8	0.9	128.0
Debt instruments	1,745.3	625.8	38.5	67.9	419.4	42.4	57.6	50.8	64.8	9.7	566.9	203.5	27.5	196.3
Bonds and notes	1,458.6	513.8	34.4	58.7	322.5	41.1	57.1	48.7	39.9	8.5	463.5	185.9	27.1	171.2
Money market instruments	286.7	112.1	4.1	9.2	96.9	1.3	0.5	2.1	25.0	1.2	103.4	17.6	0.3	25.1
Other investment	-196.1	34.7	26.1	30.2	90.7	20.8	-133.0	3.6	20.0	-68.9	-42.6	-232.8	-13.4	103.3
Assets	2,940.3	1,472.4	53.8	67.1	1,261.0	85.5	5.0	14.5	85.0	174.1	415.3	258.2	39.8	481.0
General government	98.6	10.4	1.1	0.0	4.1	2.2	3.1	0.0	0.2	0.1	2.8	1.2	34.3	49.6
MFIs	2,004.7	1,136.1	45.0	54.2	971.8	64.0	1.1	7.4	67.1	106.8	244.4	171.5	4.8	266.7
Other sectors	837.0	325.9	7.8	12.9	285.2	19.3	0.8	7.1	17.7	67.2	168.1	85.6	0.7	164.6
Liabilities	3,136.4	1,437.6	27.7	36.9	1,170.3	64.8	138.0	10.9	65.0	243.0	457.9	491.1	53.2	377.7
General government	43.6	24.0	0.0	0.2	5.3	0.0	18.5	0.0	0.9	0.1	4.1	0.3	2.9	11.3
MFIs	2,539.6	1,143.3	23.9	20.5	955.2	52.2	91.6	6.9	44.5	207.0	355.4	449.5	48.7	284.4
Other sectors	553.2	270.2	3.8	16.2	209.8	12.5	27.9	4.0	19.6	35.9	98.4	41.3	1.6	82.1
Source: ECB.														



#### 7.4 International investment position (including international reserves) (EUR billions, unless otherwise indicated; end-of-period outstanding amounts)

#### 1. Summary international investment position

	Total	Total as a % of GDP	Direct investment	Portfolio investment	Financial derivatives	Other investment	Reserve assets
	1	2	3	4	5	6	7
			Net international inve	stment position			
2001	-389.0	-5.6	422.9	-834.8	2.5	-372.3	392.7
2002	-703.6	-9.8	184.5	-937.6	-12.0	-304.6	366.1
2003	-809.3	-10.9	43.1	-914.0	-8.3	-236.8	306.6
2004	-946.4	-12.3	33.1	-1,049.4	-14.8	-196.1	280.8
2005 Q1	-956.7	-12.1	89.2	-1,071.1	-21.3	-238.5	285.0
Q2	-1,009.2	-12.7	113.5	-1,226.1	-13.5	-185.4	302.3
			Outstanding	assets			
2001	7,758.3	111.3	2,086.0	2,513.0	129.9	2,636.7	392.7
2002	7,429.3	103.1	2,008.7	2,292.7	136.0	2,625.9	366.1
2003	7,934.3	107.1	2,152.0	2,634.6	158.0	2,683.1	306.6
2004	8,632.6	112.1	2,265.1	2,984.0	162.3	2,940.3	280.8
2005 Q1	9,118.1	114.9	2,324.0	3,141.0	174.3	3,193.8	285.0
Q2	9,684.6	122.0	2,378.1	3,355.3	197.7	3,451.2	302.3
			Outstanding li	abilities			
2001	8,147.3	116.9	1,663.1	3,347.8	127.4	3,009.0	-
2002	8,132.9	112.8	1,824.3	3,230.2	147.9	2,930.5	-
2003	8,743.6	118.1	2,108.9	3,548.6	166.3	2,919.8	-
2004	9,579.0	124.4	2,231.9	4,033.4	177.2	3,136.4	-
2005 Q1	10,074.8	126.9	2,234.8	4,212.2	195.6	3,432.3	-
Q2	10,693.8	134.7	2,264.6	4,581.4	211.2	3,636.6	-

#### 2. Direct investment

		:	By resident (	units abroad				By not	n-resident un	its in the eur	o area	
		Equity capital einvested earnin	ngs	(mostly	Other capital inter-company	loans)		Equity capital einvested earni	ngs	(mostly	Other capital inter-company	y loans)
	Total	MFIs excluding Eurosystem	Non- MFIs	Total	MFIs excluding Eurosystem	Non- MFIs	Total	MFIs excluding Eurosystem	Non- MFIs	Total	MFIs excluding Eurosystem	Non- MFIs
	1	2	3	4	5	6	7	8	9	10	11	12
2001	1,557.6	124.1	1,433.5	528.4	2.1	526.3	1,165.5	43.9	1,121.6	497.6	2.8	494.8
2002	1,547.4	133.3	1,414.1	461.4	1.6	459.7	1,293.1	42.1	1,251.0	531.2	2.9	528.3
2003	1,702.8	125.9	1,577.0	449.2	1.4	447.8	1,526.9	46.6	1,480.3	582.0	2.9	579.1
2004	1,825.7	139.9	1,685.9	439.3	1.2	438.1	1,642.1	46.1	1,596.0	589.9	3.4	586.5
2005 Q1 Q2	1,858.6 1,908.5	145.3 152.1	1,713.3 1,756.4	465.3 469.6	1.2 1.2	464.2 468.4	1,640.0 1,655.1	46.1 47.2	1,593.9 1,608.0	594.7 609.5	3.7 3.7	591.0 605.8

#### 3. Portfolio investment assets by instrument and sector of holder

		1	Equity							Debt ins	truments				
							Bonds	s and note	s			Money ma	ırket instru	ments	
		Assets			Liabilities		Assets			Liabilities		Asset	s		Liabilities
	Eurosystem	MFIs excluding	Non-l	MFIs		Eurosystem	MFIs excluding	Non-l	MFIs		Eurosystem	MFIs excluding	Non-	MFIs	
		Eurosystem	General gov.	Other sectors			Eurosystem	General gov.	Other sectors			Eurosystem	General gov.	Other sectors	
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
2001	0.6	38.5	6.7	1,068.8	1,643.9	2.0	424.8	8.2	783.5	1,517.4	2.8	135.1	0.2	41.8	186.5
2002	0.7	43.6	8.3	799.2	1,364.3	6.4	402.9	8.0	784.6	1,654.4	1.2	189.4	1.3	47.1	211.5
2003	1.7	53.6	11.5	1,008.2	1,555.0	8.3	459.2	8.0	842.5	1,744.1	1.1	191.5	0.6	48.4	249.5
2004	2.1	74.1	15.8	1,146.7	1,782.6	6.2	538.4	9.7	904.3	2,011.2	1.0	231.6	0.5	53.7	239.6
2005 Q1 Q2	2.1 2.5	104.2 88.1	17.1 18.8	1,180.5 1,266.5	1,862.5 2,002.9	6.1 6.9	582.4 643.4	10.1 10.2	950.3 1,006.5	2,078.1 2,283.8	0.5 0.9	223.9 242.0	4.1 6.5	59.7 63.0	271.6 294.6

#### 7.4 International investment position (including international reserves) (EUR billions, unless stated otherwise; end-of-period outstanding amounts)

#### 4. Other investment by instrument

. ..

		Eu	osystem					Genera	l governmen	it		
	Assets		Liabilitie	es			Assets			]	Liabilities	
	Loans/currency and	Other assets	Loans/currency and	Other liabilities	Trade credits	Loans	currency a	nd deposits	Other assets	Trade credits	Loans	Other liabilities
	deposits		deposits			Total	Loans	Currency and deposits				
	1	2	3	4	5	6	7	8	9	10	11	12
2001 2002 2003 2004	2.3 3.6 4.4 4.5	0.8 0.1 0.6 0.1	40.2 57.2 65.3 73.2	0.2 0.2 0.2 0.2	3.1 1.3 1.4 1.4	70.1 59.4 54.2 57.6	50.1 51.0	4.1 6.7	55.8 54.5 39.1 39.6	0.2 0.1 0.0 0.0	44.7 42.2 40.2 40.1	12.3 13.8 3.8 3.5
2005 Q1 Q2	3.8 5.0	0.1 0.2	77.5 78.9	0.2 0.2	1.4 1.4	55.3 62.2	49.1 47.7	6.2 14.5	40.8 42.3	$\begin{array}{c} 0.0\\ 0.0\end{array}$	43.7 42.0	2.4 3.1

	MI	FIs (exclu	ding Eurosystem)	Other sectors										
	Assets		Liabilitie	3			Assets			I	Liabilities			
	Loans/currency Other and assets		Loans/currency and	Other liabilities	Trade credits	Loans	currency an	d deposits	Other assets	Trade credits	Loans	Other liabilities		
	deposits		deposits			Total	Loans	Currency and deposits						
	13	14	15	16	17	18	19	20	21	22	23	24		
2001	1,666.4	48.8	2,362.1	49.3	176.4	511.7	-	-	101.2	109.7	349.7	40.7		
2002	1,686.3	60.8	2,251.1	48.5	174.5	492.6	-	-	92.7	104.4	365.2	47.8		
2003	1,739.6	38.4	2,242.9	30.9	170.3	538.4	208.7	329.8	96.7	106.6	383.5	46.3		
2004	1,955.8	44.3	2,424.3	42.0	172.3	558.6	227.5	331.1	106.2	109.5	394.7	48.9		
2005 Q1	2,117.3	58.5	2,662.9	60.8	177.6	627.5	259.7	367.9	111.4	116.7	411.6	56.4		
Q2	2,277.6	66.4	2,780.1	72.5	184.4	683.0	325.8	357.1	128.6	116.4	480.5	62.8		

#### 5. International reserves

	Reserve assets												N	Iemo		
														Assets	Liabilities	
	Total Monetary gold			Special drawing	Reserve position	Foreign exchange Othe claim									Claims on euro	Predetermined short-term
		In EUR billions	In fine troy ounces		in the IMF	Total	Cotal Currency and deposits		Securities Financial derivatives				area residents in	net drains in		
			(millions)				With monetary authorities and the BIS	With banks	Total	Equity	and	Money market instruments			foreign currency	foreign currency
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15	16
	Eurosystem															
2002 2003 2004	366.1 306.6 280.8	130.4 130.0 125.4	399.022 393.543 389.998	4.8 4.4 3.9	25.0 23.3 18.6	205.8 148.9 132.9	10.3 10.0 12.5	35.3 30.4 25.5	159.8 107.7 94.7	1.0 1.0 0.5	120.2 80.2 58.5	38.5 26.5 35.6	0.4 0.9 0.2	$0.0 \\ 0.0 \\ 0.0$	22.4 20.3 19.1	-26.3 -16.3 -12.8
2005 Q1 Q2	285.0 302.3	127.7 138.2	387.359 382.323	4.0 4.2	17.4 16.5	135.8 143.4	7.7 12.4	27.8 28.3	$100.4 \\ 103.0$	0.5 0.5	59.8 62.8	40.1 39.7	-0.1 -0.4	$\begin{array}{c} 0.0\\ 0.0\end{array}$	21.4 23.4	-15.1 -17.7
2005 Sep. Oct. Nov.	311.6 310.5 322.7	149.4 148.1 158.8	380.258 378.357 377.023	4.2 4.2 4.3	13.8 13.6 13.4	144.2 144.5 146.2	10.8 8.7 8.8	27.3 28.8 27.8	106.3 107.1 109.7	-	-	-	-0.2 -0.1 -0.1	$0.0 \\ 0.0 \\ 0.0$	24.0 23.6 23.6	-19.5 -20.4 -19.4
						of w	hich held by t	he Europ	ean Cent	ral Bank						
2002 2003 2004	45.5 36.9 35.1	8.1 8.1 7.9	24.656 24.656 24.656	0.2 0.2 0.2	0.0 0.0 0.0	37.3 28.6 27.0	1.2 1.4 2.7	9.9 5.0 3.3	26.1 22.2 21.1	$0.0 \\ 0.0 \\ 0.0$	19.5 14.9 9.7	6.7 7.3 11.3	0.0 0.0 0.0	0.0 0.0 0.0	3.0 2.8 2.6	-5.2 -1.5 -1.3
2005 Q1 Q2	36.2 39.7	8.1 8.4	24.656 23.145	0.2	0.0	27.9 31.2	1.1 3.8	4.2 5.1	22.6 22.3	0.0	7.7 8.2	14.9 14.1	0.0 0.0	0.0	2.7 2.6	-0.9 -1.4
2005 Sep. Oct. Nov.	41.0 41.0 42.3	9.1 9.1 9.7	23.145 23.145 23.145	0.2 0.2 0.2	0.0 0.0 0.0	31.8 31.7 32.3	4.7 2.3 2.2	5.1 6.9 6.4	22.0 22.6 23.8	-	-	- - -	0.0 0.0 0.0	$\begin{array}{c} 0.0 \\ 0.0 \\ 0.0 \end{array}$	2.3 2.3 3.1	-1.5 -1.5 -2.3



#### EURO AREA STATISTICS

External transactions and positions

#### 7.5 Trade in goods (seasonally adjusted, unless otherwise indicated)

#### 1. Values, volumes and unit values by product group

	Total (n.s.a.) Exports (f.o.b.)							Imports (c.i.f.)							
				Tota	1		Memo:		Tota	ıl		Memo:			
	Exports	Imports	Г	Intermediate	Capital	Consumption	Manufactures		Intermediate	Capital (	Consumption	Manufactures	Oil		
	1	2	3	4	5	6	7	8	9	10	11	12	13		
							centage changes								
2001 2002	6.1 2.0	-0.7 -3.0	1,062.7 1,083.7	506.0 512.4	235.1 227.8	289.2 309.5	932.5 949.3	1,016.8 984.6	579.2 559.4	178.9 163.2	228.4 234.3	740.9 717.4	107.7 105.2		
2003 2004	-2.3 8.9	0.5 9.3	1,059.8	500.9	222.8	300.4	925.2	989.9	553.8	164.2	240.9	716.4	109.0		
2004 2004 Q2	12.6	9.3	1,147.1	548.0 137.3	247.0 62.0	313.6	997.1 249.5	1,075.1	604.1	183.7 46.6	256.2 63.7	769.7 189.8	129.4 29.8		
Q3 Q4	9.0	14.5	287.9	138.7	61.7	78.6	250.0	275.6	157.1	46.1	64.0	195.2	36.6		
	8.8	12.5	291.7	139.5	62.8	77.9	254.0	279.1	158.5	47.3	65.1	199.7	36.6		
2005 Q1 Q2	3.4 5.9	8.5 10.5	292.1 301.6	137.7 143.0	62.0 63.7	77.3 80.2	255.7 258.4	279.3 292.1	155.3 165.1	45.0 49.0	63.2 64.5	197.8 200.8	36.1 40.5		
Q3	9.5	14.0	317.3	148.2	69.4	83.4	273.1	314.2	179.2	51.4	68.0	215.9	51.0		
2005 May June	6.8 6.4	13.1 7.8	$101.1 \\ 100.9$	47.7 47.7	21.1 21.5	26.6 27.2	86.8 85.8	98.1 97.2	55.4 54.8	15.9 16.7	21.8 21.7	68.0 66.5	13.2 13.7		
July	2.8	9.4	103.3	47.7	22.3	26.9	88.8	102.2	56.8	16.5	22.0	70.9	14.7		
Aug. Sep.	14.1 12.4	19.7 13.3	106.7 107.3	50.0 50.5	23.4 23.7	28.1 28.4	91.9 92.3	106.8 105.1	62.4 60.1	18.0 16.9	23.0 23.0	73.3 71.7	18.1 18.2		
Oct.	6.9	12.3	104.9	48.7	22.6	27.7	88.8	106.0	59.2	17.3	22.7	71.3	14.8		
							percentage char	-							
2001 2002	5.1 2.9	-0.8 -0.7	104.9 107.9	102.1 105.0	108.6 106.2	107.9 115.1	105.4 108.2	99.1 98.3	99.3 98.8	96.3 89.5	100.5 104.1	97.8 96.3	99.3 101.4		
2003	1.0	-0.7	107.9	105.0	108.2	113.1	108.2	102.0	98.8 100.4	89.3 95.2	110.4	100.0	101.4		
2004	8.8	6.6	117.9	115.4	121.0	120.0	118.2	108.0	103.8	108.3	118.4	107.3	105.6		
2004 Q2 Q3	12.1 7.7	6.8 8.2	118.4 117.6	115.9 116.0	121.0 120.2	122.0 120.1	118.3 118.0	$107.1 \\ 108.8$	103.0 105.1	109.1 108.2	117.9 117.8	105.8 108.1	101.3 114.4		
Q4	7.5	6.0	119.6	116.1	123.6	119.3	120.1	109.5	104.2	112.9	120.0	110.7	105.0		
2005 Q1	1.3	2.4	119.0	113.6 117.0	122.3	117.9	120.1 121.0	110.1	102.6	108.9	116.3 118.3	109.6	105.8		
Q2 Q3	4.2 6.8	4.3 4.3	121.8 126.5	117.0	125.0 135.3	121.4 124.7	121.0	111.4 113.5	103.8 104.5	116.9 119.8	118.3	110.5 117.3	103.6 108.9		
2005 May	5.6	8.0	123.3	117.8	124.8	121.1	122.5	113.9	106.6	114.3	119.9	112.6	107.3		
June July	4.4 0.3	1.4 0.9	121.5 123.8	116.2 115.5	126.2 130.6	122.7 121.4	120.0 124.0	108.9 111.8	100.8 100.6	117.5 116.1	118.2 119.3	108.7 115.8	99.3 97.7		
Aug.	11.5	9.1	123.8	121.5	136.4	125.6	128.2	115.3	108.8	123.7	124.1	119.1	117.3		
Sep. Oct.	9.4	3.4	128.2	122.2	138.8	127.0	128.7	113.5	104.1	119.4	124.1	117.1	111.7		
	•	•	•	Unit value ir	dices (20	00 = 100 annua	l percentage cha	inges for c	olumns 1 and 2)	•	•	•	· ·		
2001	1.1	0.3 -2.3	101.0	100.7			100.9	-		101.4	102.8 101.9	101.7	88.6		
2001 2002	-0.9	-2.3	100.1	99.1	100.1 99.2	102.1 102.4	100.1	100.2 97.8	98.7 95.8	99.6	101.9	100.0	84.6		
2003 2004	-3.2 0.1	-3.1 2.5	96.9 96.9	96.1 96.4	95.4 94.4	99.5 99.5	96.6 96.2	94.8 97.2	93.3 98.4	94.2 92.6	98.8 97.9	96.1 96.2	85.0 99.6		
2004 Q2	0.4	3.0	97.1	96.3	94.7	100.0	96.3	96.7	97.4	93.4	97.9	96.2	95.7		
Q3 Q4	1.3 1.2	5.8 6.1	97.6 97.3	97.1 97.6	94.8 93.9	99.7 99.5	96.7 96.5	99.0 99.6	101.1 103.0	93.1 91.5	98.4 98.1	96.9 96.8	104.0 113.6		
2005 Q1	2.1	6.0	97.3	97.0	93.9	99.9	90.3	99.0	103.0	90.2	98.1	96.9	111.2		
Q2	1.6	5.9	98.7	99.4	94.3	100.7	97.5	102.5	107.7	91.6	98.8	97.5	127.5		
Q3	2.5	9.3	100.0	100.6	94.9	101.9	98.1	108.1	116.0	93.8	100.4	98.8	152.3		
2005 May June	1.1 2.0	4.7 6.3	98.1 99.4	98.7 100.1	93.9 94.4	100.3 101.4	97.0 97.8	$101.0 \\ 104.6$	105.4 110.4	91.1 92.9	98.8 99.5	97.3 98.4	120.2 134.8		
July	2.5	8.4	99.8	100.6	94.8	101.3	98.1	107.1	114.5	93.3	100.0	98.6	146.7		
Aug. Sep.	2.2 2.8	9.8 9.6	100.1 100.1	100.4 100.7	95.0 94.8	102.4 102.0	98.1 98.1	108.6 108.6	116.3 117.2	95.4 92.6	100.5 100.7	99.1 98.6	151.3 159.0		
Oct.		•	•	•		•		•		•	•	•	•		

Sources: Eurostat and ECB calculations based on Eurostat data (volume indices and seasonal adjustment of unit value indices).



7.5 Trade in goods (EUR billions, unless otherwise indicated; seasonally adjusted)

#### 2. Geographical breakdown

	- Total	European	Union (ou	itside the e	uro area)	Russia	Switzer- land	Turkey	United States		Asia		Africa	Latin America	Other countries
		Denmark	Sweden	United Kingdom	Other EU countries		ianu		States	China	Japan	Other Asian countries		America	countries
	1	2	3	4	5	6	7	8	9	10	11	12	13	14	15
							Exports								
2001 2002 2003 2004	1,062.7 1,083.7 1,059.8 1,147.1	24.4 25.3 24.9 25.7	37.0 37.1 38.7 41.8	202.5 205.8 194.9 203.7	105.8 112.1 117.6 128.0	24.7 27.1 29.2 35.6	66.4 64.0 63.4 66.1	17.9 21.4 24.9 31.8	180.2 184.1 166.3 173.8	25.2 29.9 35.2 40.3	34.5 33.1 31.3 33.1	140.3 140.5 135.5 149.9	60.4 59.5 59.5 63.8	49.9 43.4 37.9 40.3	93.4 100.4 100.5 113.3
2004 Q2 Q3 Q4	288.4 287.9 291.7	6.3 6.5 6.7	10.4 10.5 10.7	51.0 51.4 51.1	32.3 31.7 32.8	9.0 9.2 9.2	16.3 17.2 17.1	8.3 7.9 7.7	44.1 43.3 43.8	10.7 9.7 10.1	8.0 8.4 8.2	36.9 38.3 37.8	16.0 16.5 16.0	9.9 10.2 10.5	29.2 26.9 30.2
2005 Q1 Q2 Q3	292.1 301.6 317.3	6.7 7.0 7.3	10.9 11.2 11.4	49.8 49.9 51.3	33.3 34.0 35.8	9.8 10.6 11.3	17.4 16.8 17.8	8.0 8.1 8.9	43.5 45.4 47.0	10.3 10.0 11.3	8.5 8.4 8.5	39.0 40.4 43.7	17.1 17.1 19.1	11.0 11.2 12.2	26.9 31.6 31.6
2005 May June July Aug. Sep. Oct.	101.1 100.9 103.3 106.7 107.3 104.9	2.4 2.4 2.5 2.5	3.7 3.7 3.8 3.8 3.8	16.9 16.5 16.8 17.5 16.9	11.2 11.5 11.7 11.8 12.3	3.5 3.6 3.4 4.0 3.8 3.6	5.7 5.5 5.7 6.2 5.9 5.8	2.7 2.7 2.8 3.1 3.1 3.0	15.1 15.3 15.2 15.9 15.9 15.5	3.3 3.4 3.7 3.7 3.9 3.7	2.7 2.8 2.8 2.9 2.8 2.8 2.8	13.3 13.3 14.5 14.4 14.8 13.8	$     \begin{array}{r}       6.1 \\       5.5 \\       6.0 \\       6.6 \\       6.5 \\       6.0 \\     \end{array} $	3.7 3.6 4.1 4.0 4.1 4.0	10.8 11.3 10.5 10.1 11.0
						9	6 share of to	otal exports							
2004	100.0	2.2	3.6	17.8	11.2	3.1	5.8	2.8	15.1	3.5	2.9	13.1	5.6	3.5	9.9
							Imports								
2001 2002 2003 2004	1,016.8 984.6 989.9 1,075.1	22.0 23.0 23.7 25.3	35.6 35.6 36.9 39.7	154.6 149.7 138.9 143.7	88.9 93.5 102.1 107.1	42.8 42.0 47.4 56.4	52.9 52.1 50.4 53.4	16.7 17.7 19.3 22.8	138.7 125.6 110.3 113.8	57.5 61.8 74.4 92.1	58.6 52.7 52.2 53.9	150.5 142.7 141.4 163.1	74.0 67.9 68.9 72.8	41.0 39.4 39.8 45.1	83.0 80.9 84.2 86.0
2004 Q2 Q3 Q4	265.1 275.6 279.1	6.1 6.4 6.5	9.9 10.1 10.2	35.2 37.4 36.4	26.8 26.1 27.3	13.6 14.5 15.9	13.2 13.6 13.8	5.7 6.0 6.1	29.8 28.7 28.8	22.5 23.6 25.3	13.1 13.7 13.5	40.7 42.2 43.2	17.2 19.1 19.9	11.0 11.6 11.7	20.3 22.6 20.7
2005 Q1 Q2 Q3	279.3 292.1 314.2	6.2 6.4 6.3	10.0 10.2 10.5	35.7 36.6 38.6	27.0 28.8 29.8	16.7 17.5 18.7	13.4 14.4 15.0	6.2 5.8 6.1	29.1 29.9 30.6	26.6 27.8 30.4	13.0 12.4 13.4	41.0 45.8 49.2	20.2 21.8 26.6	12.1 11.9 13.7	22.1 22.8 25.3
2005 May June July Aug. Sep. Oct.	98.1 97.2 102.2 106.8 105.1 106.0	2.3 2.0 2.0 2.2 2.1	3.4 3.4 3.5 3.5 3.5	12.2 11.9 12.8 13.0 12.9	9.9 9.6 9.7 9.9 10.1	6.0 5.8 5.7 6.7 6.4 6.3	4.8 4.9 5.0 5.0 5.0 5.0 5.0	2.1 1.8 2.0 2.1 2.0 2.1	9.9 10.0 10.1 10.3 10.2 10.2	9.5 9.4 9.9 10.2 10.3 10.4	4.2 4.1 4.3 4.7 4.4 4.2	14.9 16.1 15.6 17.3 16.3 15.1	7.6 7.2 7.8 9.8 9.1 8.2	3.9 4.0 4.1 4.9 4.7 4.7	7.5 7.0 9.8 7.4 8.2
2004	100.0			10.4	10.0		6 share of to	•	10.6			15.0			
2004	100.0	2.4	3.7	13.4	10.0	5.2	5.0 Balar	2.1	10.6	8.6	5.0	15.2	6.8	4.2	8.0
2001 2002 2003 2004	45.9 99.2 69.9 72.0	2.3 2.3 1.1 0.5	1.4 1.5 1.7 2.1	47.9 56.1 56.0 60.0	17.0 18.6 15.5 20.9	-18.1 -14.9 -18.2 -20.7	13.5 12.0 12.9 12.6	1.2 3.8 5.5 8.9	41.5 58.4 56.0 60.0	-32.3 -31.9 -39.1 -51.8	-24.1 -19.7 -20.9 -20.8	-10.2 -2.2 -5.9 -13.2	-13.6 -8.3 -9.4 -9.0	8.9 4.0 -1.8 -4.8	10.4 19.5 16.3 27.3
2004 Q2 Q3 Q4	23.3 12.3 12.6	0.3 0.1 0.2	0.6 0.4 0.5	15.7 14.0 14.7	5.5 5.7 5.5	-4.5 -5.3 -6.6	3.1 3.5 3.3	2.6 1.9 1.6	14.3 14.6 15.0	-11.8 -13.8 -15.2	-5.1 -5.3 -5.3	-3.8 -3.9 -5.5	-1.3 -2.5 -3.9	-1.1 -1.4 -1.2	8.8 4.3 9.5
2005 Q1 Q2 Q3	12.8 9.5 3.1	0.5 0.6 1.1	0.9 1.0 0.9	14.1 13.3 12.7	6.3 5.2 6.1	-6.9 -6.9 -7.4	4.0 2.5 2.9	1.7 2.3 2.9	14.4 15.5 16.4	-16.3 -17.8 -19.1	-4.5 -4.0 -4.9	-2.1 -5.4 -5.5	-3.2 -4.7 -7.6	-1.1 -0.7 -1.5	4.8 8.8 6.3
2005 May June July Aug. Sep. Oct.	3.0 3.8 1.1 -0.2 2.2 -1.2	0.1 0.3 0.3 0.3 0.4	0.3 0.3 0.3 0.3 0.3	4.7 4.6 4.0 4.6 4.1	1.3 1.8 1.9 1.9 2.2	-2.4 -2.1 -2.3 -2.7 -2.5 -2.7	0.9 0.6 0.8 1.2 0.9 0.8	$\begin{array}{c} 0.7\\ 0.9\\ 0.9\\ 1.0\\ 1.0\\ 1.0\end{array}$	5.1 5.3 5.1 5.6 5.7 5.3	-6.2 -6.0 -6.2 -6.4 -6.4 -6.7	-1.5 -1.3 -1.5 -1.8 -1.6 -1.4	-1.6 -2.8 -1.1 -2.9 -1.5 -1.4	-1.5 -1.7 -1.8 -3.2 -2.6 -2.2	-0.2 -0.4 0.0 -0.9 -0.6 -0.7	3.4 4.3 0.7 2.8 2.8

Sources: Eurostat and ECB calculations based on Eurostat data (balance and columns 5, 12 and 15).





### **EXCHANGE RATES**

8.1 Effective exchange rates <sup>1)</sup>

		EER-23									
	Nominal	Real CPI	Real PPI	Real GDP deflator	Real ULCM	Real ULCT	Nominal	Real CPI			
	1	2	3	4	5	6	7	8			
2003 2004 2005	99.9 103.8 102.9	101.7 105.9 105.2	102.2 105.2 103.6	101.2 105.0	97.8 103.2	98.6 102.8	106.6 111.0 109.5	101.6 105.4 103.6			
2004 Q4 2005 Q1 Q2	105.7 105.7 103.4	107.7 107.9 105.6	106.6 106.9 104.2	106.6 106.8 104.5	105.2 103.9 100.8	104.2 103.6 101.6	113.0 112.6 110.1	107.1 106.6 104.1			
Q3 Q4	101.9 100.9	104.2 103.2	102.4 100.9	102.7	99.5	100.2	108.3 107.2	102.4 101.2			
2004 Dec.	107.1	109.2	108.1	-	-	-	114.4	108.4			
2005 Jan. Feb.	105.8 105.1	108.0 107.2	107.1 106.4	-	-	-	112.9 111.9	106.9 105.9			
Mar. Apr.	106.0 105.1	108.3 107.3	107.3 105.9	-	-	-	112.9 111.9	106.9 105.8			
May June	104.0 101.2	106.2 103.4	104.6 102.1	-	-	-	110.6 107.6	104.6 101.8			
July Aug.	101.7 102.3 101.8	103.9 104.6 104.1	102.3 102.9 102.0	-	-	-	108.0 108.7 108.2	102.1 102.8 102.4			
Sep. Oct. Nov.	101.8 101.4 100.7	104.1 103.7 103.0	102.0 101.4 100.7	-	-	-	108.2 107.8 106.9	102.4 101.8 100.8			
Dec.	100.7	103.0	100.6	-			106.9	100.8			
			0	is previous month							
2005 Dec.	0.0	0.1	-0.1	-	-	-	0.0	0.0			
			0	us previous year							
2005 Dec.	-6.0	-5.6	-7.0	-	-	-	-6.6	-7.0			

## C33 Effective exchange rates (monthly averages; index 1999 Q1=100)



## C34 Bilateral exchange rates (monthly averages; index 1999 Q1=100)



Source: ECB.

1) For the definition of the trading partner groups and other information, please refer to the General notes.


#### 8.2 Bilateral exchange rates

Japanese Hong Kong dollar Danish Swedish Pound US Swiss South Korean Singapore dollar Canadian Norwegian Australian krona dollar franc dollar dollar krone sterling yen won krone 10 11 12 7.4307 7.4399 9.1242 9.1243 1.1312 1.2439 130.97 134.44 1.5212 1.5438 1,346.90 1,422.62 8.8079 9.6881 1.9703 2.1016 1.5817 1.6167 8.0033 8.3697 1 7379 2003 0.69199 2004 0.67866 1.6905 2005 7.4518 9.2822 0.68380 1.2441 136.85 1.5483 1.273.61 9.6768 2.0702 1.5087 8.0092 1.6320 2005 Q2 7.4463 9.2083 0.67856 1.2594 135.42 1.5437 1,269.53 9.8090 2.0885 1.5677 8.0483 1.6389 Q3 Q4 7.4588 7.4586 9.3658 9.4731 0.68344 0.67996 9.4782 9.2157 1.4668 1.3956 7.8817 7.8785 1 2199 135 62 1 5533 1 255 21 2.0436 1 6054 139.41 1.5472 1,231.69 2.00450 1.5983 1.1884 7.4448 7.4584 7.4596 7.4584 2005 June 9.2628 9.4276 9.3398 0.66895 1.2165 1.2037 1.2292 132.22 1.5391 1,231.12 9.4597 2.0342 2.0257 1.5111 7.8932 1 5875 134.75 135.98 1,248.53 9.3590 9.5529 1.5578 7.9200 0.68756 1.4730 1.6002 Julv 1.255.33 1.4819 1.4452 Aug. 0.68527 1 5528 2 0439 7.9165 1.6144 9.3342 1.2256 1,261.46 9.5138 7.8087 0.67760 136.06 1.5496 2.0603 1.6009 Sep. Oct. 7.4620 7.4596 9.4223 9.5614 0.68137 0.67933 1.2015 1.1786 1.5490 1.5449 1,256.66 1,226.38 9.3191 9.1390 2.0326 2.0017 1.4149 1.3944 7.8347 7.8295 138.05 1 5037 139.59 1.6030 Nov. Dec 7.4541 9.4316 0.67922 1.1856 140.58 1.5479 1,212.30 9.1927 1.9855 1.3778 7.9737 1.5979 % change versus previous month 2005 Dec. -0.1 -1.4 0.0 0.6 0.7 0.2 -1.1 0.6 -0.8 -1.2 1.8 -0.3 % change versus previous year 2005 Dec. 0.3 5.0 -2.3 -11.6 1.0 0.7 -13.9 -11.8 -9.8 -15.6 -3.0 -8.5 New Roma-nian leu<sup>1)</sup> Czech Estonian Cyprus pound Latvian Lithuanian Hungarian Maltese Polish Slovenian Slovak Bulgarian lats forint lira koruna kroor litas zloty tola koruna le 13 14 15 16 18 19 20 21 22 24 17 23 2003 31.846 15.6466 0.58409 0.6407 3.4527 253.62 0.4261 4.3996 233.85 41.489 1.9490 37.551 2004 31 801 15 6466 0 58185 0.6652 3.4529 251 66 0.4280 4 5268 230 00 40 022 1 0533 40 510 4.0230 239.57 2005 248.05 0.4299 38.599 29.782 15.6466 0.57683 0.6962 3.4528 1.9558 3.6209 2005 Q2 30.129 15.6466 0.57824 0.6960 3.4528 249.75 0.4295 4.1301 239.54 38.919 1.9558 36,195 Q3 Q4  $0.57328 \\ 0.57339$  $3.4528 \\ 3.4528$ 245.57 251.84 29.688 15.6466 0.6960 0.4293 4.0186 239.49 38.672 1.9558 3.5250 29.304 15.6466 0.6965 0.4293 3.9152 239.51 38.494 1.9558 3.6379 30.034 0.6960 3.4528 249.04 0.4293 239.47 38.535 1.9558 36,136 15.6466 0.57405 4.0606 2005 June 3.4528 3.4528 1.9558 3.5647 July 30.180 29.594 15.6466 0.57367 0.6961 246.47 0.4293 4.0986 239.48 38.886 0 57321 244 49 0 4293 4 0436 239 51 Aug. 15 6466 0.6960 38 681 Sep. 29.317 15.6466 0.57296 0.6961 3.4528 245.83 0.4293 3.9160 239.47 38.459 1.9558 3.5097 0.57319 0.57351 3.4528 3.4528 Oct. 29.675 15.6466 0.6965 251.85 0.4293 3.9229 239.53 239.51 38.923 1.9559 3.5997 Nov 29.266 15.6466 0.6963 251.04 0.4293 3.9701 38.678 1 9557 3.6543 28.972 0.57346 239.51 Dec. 15.6466 0.6967 3.4528 252.68 0.4293 3.8501 37.872 1.9558 3.6589 % change versus previous month 0.0 2005 Dec. 0.7 0.0 -2.1 0.1 -1.0 0.0 0.0 0.1 -3.0 0.0 0.0 % change versus previous year 2005 Dec -0.1 -54 0.0 -1.01.0 0.0 2.8 -0.7 -6.9 -2.6 0.0 \_ Chinese yuan renminbi<sup>2)</sup> Croatian kuna<sup>2)</sup> Icelandic krona Indonesian Philippine Malaysian New Zealand Russian South African Thai New Turkish peso<sup>2)</sup> rupiah<sup>2)</sup> ringgit<sup>2)</sup> dollar rouble <sup>2</sup> rand baht lira

	25	26	27	28	29	30	31	32	33	34	35
2003 2004	9.3626 10.2967	7.5688 7.4967	86.65 87.14	9,685.54 11,127.34	4.2983 4.7273	1.9438 1.8731	61.336 69.727	34.6699 35.8192	8.5317 8.0092	46.923 50.077	1,694,851 1,777,052
2005	10.1955	7.4008	78.23	12,072.83	4.7119	1.7660	68.494	35.1884	7.9183	50.068	1.6771
2005 Q2 Q3 Q4	10.4232 9.9250 9.6057	7.3443 7.3728 7.3831	80.79 77.64 73.86	12,032.61 12,216.99 11.875.37	4.7858 4.6008 4.4881	1.7597 1.7640 1.7124	68.847 68.335 64.821	35.3733 34.7864 34.1294	8.0799 7.9392 7.7706	50.497 50.375 48.780	1.7193 1.6372 1.6132
2005 June July Aug.	10.0683 9.8954 9.9589	7.3169 7.3090 7.3684	79.30 78.40 78.37	11,716.31 11,803.89 12,283.08	4.6234 4.5590 4.6216	1.7175 1.7732 1.7675	67.214 67.394 68.768	34.6951 34.5513 35.0119	8.2194 8.0790 7.9508	49.793 50.199 50.604	1.6560 1.6133 1.6534
Sep. Oct. Nov.	9.9177 9.7189 9.5273	7.4384 7.3822 7.3791	76.15 73.29 72.98	12,542.23 12,118.09 11.834.55	4.6190 4.5330 4.4534	1.7515 1.7212 1.7088	68.782 66.777 64.258	34.7750 34.3262 33.9184	7.7936 7.9139 7.8502	50.305 49.153 48.469	1.6430 1.6331 1.6033
Dec.	9.5746	7.3882	75.36	11,675.40	4.4796	1.7072	63.454	34.1538	7.5439	48.731	1.6038
	% change versus previous month										
2005 Dec.	0.5	0.1	3.3	-1.3	0.6	-0.1	-1.3	0.7	-3.9	0.5	0.0
	% change versus previous year										
2005 Dec.	-13.7	-2.3	-10.3	-5.7	-12.1	-8.9	-15.8	-8.7	-1.8	-7.3	-

Source: ECB.

1) Data prior to July 2005 refer to the Romanian leu; 1 new Romanian leu is equivalent to 10,000 old Romanian lei.

2) For these currencies the ECB computes and publishes euro reference exchange rates as from 1 April 2005. Previous data are indicative.

3) Data prior to January 2005 refer to the Turkish lira; 1 new Turkish lira is equivalent to 1,000,000 old Turkish liras.





# DEVELOPMENTS OUTSIDE THE EURO AREA

### 9.1 In other EU Member States

#### 1. Economic and financial developments

1. Economic	Czech Republic	Denmark	Estonia	Cyprus	Latvia	Lithuania	Hungary	Malta	Poland	Slovenia	Slovakia	Sweden	United Kingdom
	1	2	3	4	5	6 HICI	7	8	9	10	11	12	13
2004 2005	2.6 1.6	0.9 1.7	3.0 4.1	1.9 2.0	6.2	1.1	6.8	2.7	3.6	3.6 2.5	7.5	1.0	1.3
2005 Q2 Q3 Q4	1.2 1.6 2.2	1.6 2.2 2.0	3.6 4.3 4.0	2.1 1.7 1.9	6.7 6.7	2.4 2.2	3.6 3.5	2.2 2.1	2.2 1.7	2.2 2.3 2.6	2.6 2.2	0.5 0.9	1.9 2.4
2005 Aug. Sep. Oct.	1.4 2.0 2.5	2.3 2.4 1.9	4.2 4.9 4.5	1.5 2.1 2.2	6.3 7.4 7.7	2.3 2.6 3.0	3.5 3.6 3.1	2.5 2.0 3.0	1.8 1.9 1.6	1.8 3.2 3.2	2.1 2.3 3.5	1.1 1.1 0.9	2.4 2.5 2.3
Nov. Dec.	2.2 2.2 1.9	1.9 1.9 2.2	4.0 3.6	2.0 1.4	7.5	2.8	3.3	4.3	1.0	2.1 2.4	3.6	1.2	2.1
							surplus (+) as						
2002 2003 2004	-6.8 -12.5 -3.0	1.4 1.2 2.9	1.5 2.6 1.7	-4.5 -6.3 -4.1	-2.3 -1.2 -0.9	-1.4 -1.2 -1.4	-8.5 -6.5 -5.4	-5.8 -10.4 -5.1	-3.3 -4.8 -3.9	-2.7 -2.7 -2.1	-7.8 -3.8 -3.1	-0.3 0.2 1.6	-1.7 -3.3 -3.2
							debt as a % o						
2002 2003 2004	29.8 36.8 36.8	47.6 45.0 43.2	5.8 6.0 5.5	65.2 69.8 72.0	14.2 14.6 14.7	22.4 21.4 19.6	55.5 57.4 57.4	63.2 72.8 75.9	41.2 45.3 43.6	29.8 29.4 29.8	43.7 43.1 42.5	52.4 52.0 51.1	38.2 39.7 41.5
							s a % per ann						
2005 June July Aug. Sep.	3.31 3.35 3.37 3.26	3.16 3.21 3.24 3.05	-	5.13 4.84 4.84 4.81	3.87 3.87 3.87 3.87 3.87	3.78 3.61 3.50 3.50	6.59 6.13 5.85 5.64	4.56 4.55 4.43 4.41	4.91 4.72 4.88 4.57	3.90 3.78 3.79 3.74	3.36 3.22 3.24 3.13	3.11 3.06 3.14 2.98	4.31 4.31 4.34 4.25
Oct. Nov.	3.46 3.76	3.22 3.46	-	4.22 4.22	3.87 3.56	3.50 3.64	6.49 6.81	4.41 4.39	4.91 5.38	3.62 3.62	3.25 3.70	3.17 3.39	4.40 4.37
2005 June	1.75	2.16	2.34	4.11	2.81	2.36	er annum, per	3.27	5.22	4.05	2.88	1.82	4.84
July Aug. Sep. Oct.	1.78 1.79 1.80 1.91	2.16 2.17 2.18 2.22	2.33 2.33 2.32 2.32	3.92 3.85 3.80 3.59	2.71 2.76 2.82 2.78	2.32 2.33 2.32 2.31	6.50 6.35 5.65 6.15	3.28 3.26 3.26 3.24	4.68 4.67 4.51 4.55	4.04 4.02 4.03 4.01	2.89 2.94 2.93 3.03	1.64 1.67 1.67 1.72	4.66 4.59 4.60 4.59
Nov.	2.24	2.39	2.32	3.51	2.84	2.42 Real G	6.20	3.19	4.64	4.01	3.19	1.72	4.62
2003	3.2	0.6	6.7	1.9	7.2	10.4	3.4	-1.7	3.8	2.7	4.5	1.7	2.5
2004 2005 Q1	4.7 5.0	2.1	7.8	3.8	9.8 7.3	7.0 6.3	4.6	0.1	5.4 2.1	4.2	5.5 5.1	3.7 2.3	3.2
Q2 Q3	5.2 4.9	3.1 4.7	10.2 10.4	3.7 3.9	11.4 11.4	7.6 8.5	4.3 4.4	1.9 2.9	2.8 3.7	5.0 4.2	5.1 6.2	2.3 2.3 2.8	1.6 1.7
2002	(2)		11.6				balance as a		2.1	1.0	0.5	7.4	
2003 2004	-6.3 -5.7	3.3 2.4	-11.6 -11.9	-2.4 -4.9	-7.5 -11.8	-6.4 -6.4	-8.7 -8.5	-5.5 -8.7	-2.1 -3.8	-1.0 -2.5	-0.5 -3.1	7.4 8.2	-1.3 -1.8
2005 Q1 Q2 Q3	3.0 -4.1 -4.4	3.4 5.3 5.2	-9.5 -10.9 -6.6	-13.5 1.5 3.1	-9.3 -9.8 -11.1	-3.7 -6.7 -6.7	-6.8 -7.2 -7.1	-7.9 -10.4 -0.6	-1.0 -0.8 -1.4	-1.7 0.4 0.6	-2.2 -11.9 -4.7	8.3 6.2 7.8	-1.7 -1.1 -3.1
						Unit labou						1.0	
2003 2004	7.6 1.1	1.6 1.1	4.9 3.0	-	5.6 7.3	1.5	7.2 4.2	-	•	4.7 3.8	3.5 2.1	1.0 -0.6	3.2 2.0
2005 Q1 Q2 Q3	-0.3 0.4 2.0	1.9 2.6 -1.1	3.1 2.4 4.1	- -	- -	4.0 1.2 3.4	- -	-			5.7 3.6 4.1	2.6 0.1 0.9	4.3 3.3
							as a % of lab		,				
2004 2005	8.3 7.9	5.4 ·	9.2 ·	5.2 6.0	9.8	10.9 8.2	6.0	7.7	18.8	6.0	18.2	6.4	4.7
2005 Q2 Q3 Q4	8.0 7.8 7.9	5.0 4.8	7.8 7.3	5.8 6.1 5.9	9.1 8.8	8.6 7.9 7.0	7.1 7.2	8.2 7.8	18.0 17.7	5.8 5.8	16.5 16.3		4.6 4.7
2005 Aug. Sep. Oct. Nov.	7.8 7.9 7.9 7.9	4.8 4.7 4.6 4.5	7.3 7.1 6.8 6.6	6.1 6.2 6.1 5.9	8.8 8.8 8.7 8.6	7.9 7.6 7.3 7.0	7.2 7.2 7.2 7.3	7.8 7.8 7.7 7.7	17.7 17.7 17.6 17.4	5.8 5.8 5.8 5.9	16.2 16.2 16.2 16.0		4.6 4.8
Dec.	7.8	4.5		5.8		6.8						:	•

Sources: European Commission (Economic and Financial Affairs DG and Eurostat), national data, Reuters and ECB calculations. 1) Ratios are computed using GDP excluding financial intermediation services indirectly measured (FISIM).



#### 9.2 In the United States and Japan

#### 1. Economic and financial developments

	Consumer price index	Unit labour costs <sup>1)</sup> (manufacturing)	Real GDP	Industrial production index (manufacturing)	Unemployment rate as a % of labour force (s.a.)	Broad money <sup>2)</sup>	3-month interbank deposit rate <sup>3)</sup> as a % per annum	10-year government bond yield <sup>3)</sup> as a % per annum	Exchange rate <sup>4)</sup> as national currency per euro	Fiscal deficit (-)/ surplus (+) as a % of GDP	Gross public debt <sup>5)</sup> as a % of GDP
	1	2	3	4	5	6	7	8	9	10	11
					United States						
2002	1.6	-0.6	1.6	0.3	5.8	8.0	1.80	4.60	0.9456	-3.8	45.2
2003	2.3	2.2	2.7	0.7	6.0	6.4	1.22	4.00	1.1312	-5.0	47.9
2004	2.7	-3.1	4.2	5.0	5.5	5.1	1.62	4.26	1.2439	-4.7	48.6
2005		•			5.1	•	3.56	4.28	1.2441		•
2004 Q4	3.3	-1.6	3.8	5.2	5.4	5.8	2.30	4.17	1.2977	-4.3	48.6
2005 Q1	3.0	2.3	3.6	4.8	5.2	5.8	2.84	4.30	1.3113	-3.7	49.5
Q2	2.9	3.1	3.6	3.4	5.1	5.0	3.28	4.16	1.2594	-3.4	48.6
Q3	3.8	1.7	3.6	3.0	5.0	6.1	3.77	4.21	1.2199	•	•
Q4				•	4.9		4.34	4.48	1.1884		•
2005 Aug.	3.6	-	-	3.1	4.9	6.1	3.80	4.26	1.2292	-	-
Sep.	4.7	-	-	2.7	5.1	6.6	3.91	4.19	1.2256	-	-
Oct.	4.3	-	-	3.7	4.9	7.3	4.17	4.45	1.2015	-	-
Nov.	3.5	-	-	4.1	5.0	7.4	4.35	4.53	1.1786	-	-
Dec.		-	-		4.9	•	4.49	4.46	1.1856	-	-
					Japan						
2002	-0.9	-3.2	-0.3	-1.2	5.4	3.3	0.08	1.27	118.06	-7.9	141.5
2003	-0.3	-3.8	1.4	3.2	5.2	1.7	0.06	0.99	130.97	-7.7	149.2
2004	0.0	-5.2	2.7	5.5	4.7	1.9	0.05	1.50	134.44		
2005			•	•	•	•	0.06	1.39	136.85	•	•
2004 Q4	0.5	-1.5	0.9	1.8	4.6	2.0	0.05	1.45	137.11		
2005 Q1	-0.2	-1.0	1.0	1.4	4.6	2.0	0.05	1.41	137.01		
Q2	-0.1	0.9	2.2	0.3	4.4	1.7	0.05	1.28	135.42		
Q3	-0.3	0.3	2.8	0.1	4.3	1.8	0.06	1.36	135.62		
Q4					•		0.06	1.53	139.41	•	· .
2005 Aug.	-0.3	-1.1	-	1.5	4.3	1.7	0.06	1.43	135.98	-	-
Sep.	-0.3	-0.6	-	1.1	4.2	2.0	0.06	1.38	136.06	-	-
Oct.	-0.7		-	3.0	4.5	2.0	0.06	1.54	138.05	-	-
Nov.	-0.8		-	3.3	4.6	2.2	0.06	1.52	139.59	-	-
Dec.			-				0.07	1.54	140.58	-	-



5

4

3

2

1

0

-1

-2

2005



Sources: National data (columns 1, 2 (United States), 3, 4, 5 (United States), 6, 9 and 10); OECD (column 2 (Japan)); Eurostat (column 5 (Japan), euro area chart data); Reuters (columns 7 and 8); ECB calculations (column 11).

Ì) Data for the United States are seasonally adjusted.

Average-of-period values; M3 for US, M2+CDs for Japan.

2) 3) 4) 5)

For more information, see Sections 4.6 and 4.7. For more information, see Section 8.2. Gross consolidated general government debt (end of period).



### 9.2 In the United States and Japan

#### 2. Saving, investment and financing

	National saving and investment			Investment and financing of non-financial corporations							Investment and financing of households 1)				
	Gross saving	Gross capital formation	Net lending to the rest of the world	Gross capital formation	Gross fixed capital formation	Net acquisition of financial assets	Gross saving	Net incurrence of liabilities	Securities and shares	Capital expend- itures <sup>2)</sup>	Net acquisition of financial assets	Gross saving <sup>3)</sup>	Net incurrence of liabilities		
	1	2	3	4	5	6	7	8	9	10	11	12	13		
	United States														
2001 2002 2003 2004	16.4 14.2 13.4 13.4	19.1 18.4 18.5 19.6	-3.7 -4.4 -4.6 -5.6	7.9 7.0 6.8 7.3	8.3 7.0 6.8 7.0	1.8 1.2 0.8 4.0	7.5 7.7 8.0 8.0	0.9 0.8 0.3 3.0	1.7 -0.2 0.8 0.9	12.8 13.0 13.3 13.5	4.9 4.1 7.7 6.7	10.8 11.4 11.3 11.0	5.5 6.6 7.9 9.4		
2003 Q4	13.9	18.8	-4.3	7.0	6.9	1.2	8.3	0.4	0.0	13.5	5.6	11.2	4.0		
2004 Q1 Q2 Q3 Q4	13.4 13.3 13.5 13.5	19.0 19.8 19.8 19.9	-5.0 -5.6 -5.5 -6.2	7.1 7.4 7.3 7.5	6.8 7.0 7.1 7.2	5.0 3.3 3.3 4.6	8.2 8.1 8.4 7.3	3.7 2.0 1.9 4.3	1.1 -0.2 0.5 2.1	13.3 13.6 13.6 13.6	6.9 5.2 6.7 7.8	11.0 10.7 10.9 11.4	9.7 8.9 8.9 10.2		
2005 Q1 Q2 Q3	13.4 13.2 13.3	20.2 19.8 19.9	-6.4 -6.0 -6.0	7.6 7.2 7.2	7.2 7.3 7.4	3.0 2.7 2.5	7.7 8.1 8.5	2.5 1.3 0.7	0.8 0.6 -0.7	13.7 13.9 13.9	5.1 3.7 5.5	10.0 9.4 9.7	7.8 9.3 9.6		
						Japar	ı								
2001 2002 2003 2004	26.6 25.7 26.4	25.8 24.2 23.9 23.9	2.0 2.8 3.1	15.3 13.8 14.3	15.3 14.1 14.4	-2.8 -1.7 2.3 4.6	14.4 15.4 16.1	-6.4 -7.4 -5.3 0.8	0.2 -0.8 0.2 0.8	4.9 4.8 4.6	2.8 -0.2 0.3 1.9	8.6 9.1 9.2	0.2 -2.1 -0.6 -0.7		
2003 Q4	27.9	24.8	2.9			10.5		5.5	1.1		9.5		-1.4		
2004 Q1 Q2 Q3 Q4 2005 Q1	31.0	24.0 23.0 23.8 24.6 24.4	3.9			12.5 -13.7 7.1 12.1 8.6	· · ·	-1.9 -11.2 0.7 14.6 -2.3	-0.6 0.6 0.2 2.8 -2.9		-7.2 8.0 -2.1 8.3 -8.1		2.6 -6.2 1.5 -0.5 3.3		
Q2 Q3	:	23.7 23.5			:	-17.0 5.7		-16.4 4.3	0.9 -1.5	:	7.6 -4.0		-6.7 3.3		

#### C37 Net lending of non-financial corporations

#### C38 Net lending of households <sup>1)</sup>



Sources: ECB, Federal Reserve Board, Bank of Japan and Economic and Social Research Institute.

Including non-profit institutions serving households.
 Gross capital formation in Japan. Capital expenditures in the United States include purchases of consumer durable goods.
 Gross saving in the United States is increased by expenditures on consumer durable goods.



8

6

2

-2

-4

-6

2003

2000

2001

2002



# LIST OF CHARTS

C1	Monetary aggregates	S12
C2	Counterparts	<b>SI2</b>
C3	Components of monetary aggregates	\$13
C4	Components of longer-term financial liabilities	\$13
C5	Loans to financial intermediaries and non-financial corporations	S14
C6	Loans to households	S15
C7	Loans to government and non-euro area residents	S16
C8	Deposits by financial intermediaries	S17
С9	Deposits by non-financial corporations and households	<b>SI8</b>
C10	Deposits by government and non-euro area residents	S19
C11	MFI holdings of securities	<b>S20</b>
C12	Total assets of investment funds	<b>S24</b>
C13	Total outstanding amounts and gross issues of securities, other than shares, issued by euro area residents	<b>S30</b>
C14	Net issues of securities, other than shares, seasonally adjusted and non-seasonally adjusted	<b>S32</b>
C15	Annual growth rates of long-term debt securities, by sector of the issuer, in all currencies combined	\$33
C16	Annual growth rates of short-term debt securities, by sector of the issuer, in all currencies combined	<b>S34</b>
C17	Annual growth rates for quoted shares issued by euro area residents	<b>S35</b>
C18	Gross issues of quoted shares by sector of the issuer	<b>S36</b>
C19	New deposits with agreed maturity	<b>S38</b>
C20	New loans at floating rate and up to 1 year initial rate fixation	<b>S38</b>
C21	Euro area money market rates	<b>S39</b>
C22	3-month money market rates	S39
C23	Euro area government bond yields	<b>S40</b>
C24	10-year government bond yields	<b>S40</b>
C25	Dow Jones EURO STOXX Broad, Standard & Poor's 500 and Nikkei 225	<b>S4</b> I
C26	Deficit, borrowing requirement and change in debt	<b>S54</b>
C27	Maastricht debt	<b>S54</b>
C28	B.o.p. current account balance	<b>S</b> 55
C29	B.o.p. net direct and portfolio investment	<b>S</b> 55
C30	B.o.p. goods	<b>S56</b>
C31	B.o.p. services	<b>S56</b>
C32	Main b.o.p. transactions underlying the developments in MFI net external assets	<b>S60</b>
C33	Effective exchange rates	S67
C34	Bilateral exchange rates	S67
C35	Real gross domestic product	<b>S70</b>
C36	Consumer price indices	<b>S70</b>
C37	Net lending of non-financial corporations	<b>S7 I</b>
C38	Net lending of households	S7 I





### **TECHNICAL NOTES**

#### **RELATING TO THE EURO AREA OVERVIEW**

# CALCULATION OF GROWTH RATES FOR MONETARY DEVELOPMENTS

The average growth rate for the quarter ending in month t is calculated as:

a) 
$$\left(\frac{0.5I_{t} + \sum_{i=1}^{2} I_{t-i} + 0.5I_{t-3}}{0.5I_{t-12} + \sum_{i=1}^{2} I_{t-i-12} + 0.5I_{t-15}} - 1\right) \times 100$$

where  $I_t$  is the index of adjusted outstanding amounts as at month t (see also below). Likewise, for the year ending in month t, the average growth rate is calculated as:

b) 
$$\left(\frac{0.5I_{t} + \sum_{i=1}^{11}I_{t-i} + 0.5I_{t-12}}{0.5I_{t-12} + \sum_{i=1}^{11}I_{t-i-12} + 0.5I_{t-24}} - 1\right) \times 100$$

#### **RELATING TO SECTIONS 2.1 TO 2.6**

#### **CALCULATION OF TRANSACTIONS**

Monthly transactions are calculated from monthly differences in outstanding amounts adjusted for reclassifications, other revaluations, exchange rate variations and any other changes which do not arise from transactions.

If  $L_t$  represents the outstanding amount at the end of month t,  $C_t^M$  the reclassification adjustment in month t,  $E_t^M$  the exchange rate adjustment and  $V_t^M$  the other revaluation adjustments, the transactions  $F_t^M$  in month t are defined as:

c) 
$$F_{t}^{M} = (L_{t} - L_{t-1}) - C_{t}^{M} - E_{t}^{M} - V_{t}^{M}$$

Similarly, the quarterly transactions  $F_t^Q$  for the quarter ending in month t are defined as:

d) 
$$F_{t}^{Q} = (L_{t} - L_{t-3}) - C_{t}^{Q} - E_{t}^{Q} - V_{t}^{Q}$$

where  $L_{t-3}$  is the amount outstanding at the end of month t-3 (the end of the previous quarter)

and, for example,  $C_t^Q$  is the reclassification adjustment in the quarter ending in month t.

For those quarterly series for which monthly observations are now available (see below), the quarterly transactions can be derived as the sum of the three monthly transactions in the quarter.

# CALCULATION OF GROWTH RATES FOR MONTHLY SERIES

Growth rates may be calculated from transactions or from the index of adjusted outstanding amounts. If  $F_t^M$  and  $L_t$  are defined as above, the index  $I_t$  of adjusted outstanding amounts in month t is defined as:

e) 
$$I_t = I_{t-1} \times \left(1 + \frac{F_t}{L_{t-1}}\right)$$

The base of the index (of the non-seasonally adjusted series) is currently set as December 2001 = 100. Time series of the index of adjusted outstanding amounts are available on the ECB's website (www.ecb.int) under the "Money, banking and financial markets" sub-section of the "Statistics" section.

The annual growth rate  $a_t$  for month t - i.e.the change in the 12 months ending in month t - may be calculated using either of the following two formulae:

f) 
$$a_t = \left[\prod_{i=0}^{11} \left(1 + \frac{F_{t-i}^M}{L_{t-1-i}}\right) - 1\right] \times 100$$

g) 
$$a_t = \left( \frac{I_t}{I_{t-12}} - 1 \right) \times 100$$

Unless otherwise indicated, the annual growth rates refer to the end of the indicated period. For example, the annual percentage change for the year 2002 is calculated in g) by dividing the index of December 2002 by the index of December 2001.



Growth rates for intra-annual periods may be derived by adapting formula g). For example, the month-on-month growth rate  $a_t^M$  may be calculated as:

h) 
$$a_t^M = \left( \frac{I_t}{I_{t-1}} - 1 \right) \times 100$$

Finally, the three-month moving average (centred) for the annual growth rate of M3 is obtained as  $(a_{t+1} + a_t + a_{t-1})/3$ , where  $a_t$  is defined as in f) or g) above.

# CALCULATION OF GROWTH RATES FOR QUARTERLY SERIES

If  $F_t^Q$  and  $L_{t-3}$  are defined as above, the index  $I_t$  of adjusted outstanding amounts for the quarter ending in month t is defined as:

i) 
$$I_t = I_{t-3} \times \left(1 + \frac{F_t^Q}{L_{t-3}}\right)$$

The annual growth rate in the four quarters ending in month t, i.e.  $a_t$ , may be calculated using formula g).

#### SEASONAL ADJUSTMENT OF THE EURO AREA MONETARY STATISTICS'

The approach used relies on a multiplicative decomposition through X-12-ARIMA.<sup>2</sup> The seasonal adjustment may include a day-of-the-week adjustment, and for some series is carried out indirectly by means of a linear combination of components. In particular, this is the case for M3, derived by aggregating the seasonally adjusted series for M1, M2 less M1, and M3 less M2.

The seasonal adjustment procedures are first applied to the index of adjusted outstanding amounts.<sup>3</sup> The resulting estimates of the seasonal factors are then applied to the levels and to the adjustments arising from reclassifications and revaluations, in turn yielding seasonally adjusted transactions. Seasonal (and trading day) factors are revised at annual intervals or as required.

#### **RELATING TO SECTIONS 3.1 TO 3.3**

#### **CALCULATION OF GROWTH RATES**

Growth rates are calculated on the basis of financial transactions and therefore exclude reclassifications, revaluations, exchange rate variations and any other changes which do not arise from transactions.

If  $T_t$  represents the transactions in quarter t and  $L_t$  represents the outstanding amount at the end of quarter t, then the growth rate for the quarter t is calculated as:

j) 
$$\frac{\sum_{i=0}^{3} T_{t-i}}{L_{t-4}} \times 100$$

#### **RELATING TO SECTION 4.3 AND 4.4**

#### CALCULATION OF GROWTH RATES FOR DEBT SECURITIES AND QUOTED SHARES

Growth rates are calculated on the basis of financial transactions and therefore exclude reclassifications, revaluations, exchange rate variations and any other changes which do not arise from transactions. They may be calculated from transactions or from the index of notional stocks. If  $N^{\rm M}_{+}$  represents the transactions (net

<sup>1</sup> For details, see "Seasonal adjustment of monetary aggregates and HICP for the euro area", ECB (August 2000) and the "Statistics" section of the ECB's website (www.ecb.int), under the "Money, banking and financial markets" sub-section.

<sup>2</sup> For details, see Findley, D., Monsell, B., Bell, W., Otto, M., and Chen, B. C. (1998), "New Capabilities and Methods of the X-12-ARIMA Seasonal Adjustment Program", Journal of Business and Economic Statistics, 16, 2, pp.127-152, or "X-12-ARIMA Reference Manual", Time Series Staff, Bureau of the Census, Washington, D.C.

For internal purposes, the model-based approach of TRAMO-SEATS is also used. For details on TRAMO-SEATS, see Gomez, V. and Maravall, A. (1996), "Programs TRAMO and SEATS: Instructions for the User", Banco de España, Working Paper No. 9628, Madrid.

<sup>3</sup> It follows that for the seasonally adjusted series, the level of the index for the base period, i.e. December 2001, generally differs from 100, reflecting the seasonality of that month.

issues) in month t and  $L_t$  the level outstanding at the end of the month t, the index  $I_t$  of notional stocks in month t is defined as:

k) 
$$I_t = I_{t-1} \times \left(1 + \frac{N_t}{L_{t-1}}\right)$$

As a base, the index is set equal to 100 on December 2001. The growth rate  $a_t$  for month t corresponding to the change in the 12 months ending in month t, may be calculated using either of the following two formulae:

l) 
$$a_{t} = \left[\prod_{i=0}^{11} \left(1 + \frac{N_{t-i}^{M}}{L_{t-1-i}}\right) - 1\right] \times 100$$

m) 
$$a_t = \begin{pmatrix} I_t \\ I_{t-12} \end{pmatrix} \times 100$$

The method used to calculate the growth rates for securities other than shares is the same as that used for the monetary aggregates, the only difference being that an "N" is used rather than an "F". The reason for this is to distinguish between the different ways of obtaining "net issues" for securities issues statistics and the equivalent "transactions" calculated used for the monetary aggregates.

The average growth rate for the quarter ending in month t is calculated as:

n) 
$$\left(\frac{0.5I_{t} + \sum_{i=1}^{2}I_{t-i} + 0.5I_{t-3}}{0.5I_{t-12} + \sum_{i=1}^{2}I_{t-i-12} + 0.5I_{t-15}} - 1\right) \times 100$$

where  $I_t$  is the index of notional stocks as at month t. Likewise, for the year ending in month t, the average growth rate is calculated as:

o) 
$$\left(\frac{0.5I_{t} + \sum_{i=1}^{11} I_{t-i} + 0.5I_{t-12}}{0.5I_{t-12} + \sum_{i=1}^{11} I_{t-i-12} + 0.5I_{t-24}} - 1\right) \times 100$$

The calculation formula used for Section 4.3 is also used for Section 4.4 and is likewise based on that used for the monetary aggregates. Section 4.4 is based on market values and the basis for the calculation are financial transactions, which exclude reclassifications, revaluations or any other changes that do not arise from transactions. Exchange rate variations are not included as all quoted shares covered are denominated in euro.

#### SEASONAL ADJUSTMENT OF SECURITIES ISSUES STATISTICS<sup>4</sup>

The approach used relies on a multiplicative decomposition through X-12-ARIMA. The seasonal adjustment for the securities issues total is carried out indirectly by means of a linear combination of sector and maturity component breakdowns.

The seasonal adjustment procedures are applied to the index of notional stocks. The resulting estimates of the seasonal factors are then applied to the outstanding amounts, from which seasonally adjusted net issues are derived. Seasonal factors are revised at annual intervals or as required.

Similar as depicted in formula 1) and m), the growth rate a, for month t corresponding to the change in the 6 months ending in month t, may be calculated using either of the following two formulae:

p) 
$$a_t = \left[\prod_{i=0}^{5} \left(1 + \frac{N_{t-i}^M}{L_{t-1-i}}\right) - 1\right] x 100$$

q) 
$$a_t = \left( \frac{I_t}{I_{t-6}} - 1 \right) x 100$$

4 For details, see "Seasonal adjustment of monetary aggregates and HICP for the euro area", ECB (August 2000) and the "Statistics" section of the ECB's website (www.ecb.int), under the "Money, banking and financial markets" sub-section.



#### **RELATING TO TABLE I IN SECTION 5.1**

#### **SEASONAL ADJUSTMENT OF THE HICP<sup>4</sup>**

The approach used relies on multiplicative decomposition through X-12-ARIMA (see footnote 2 on page S74). The seasonal adjustment of the overall HICP for the euro area is carried out indirectly by aggregating the seasonally adjusted euro area series for processed food, unprocessed food, industrial goods excluding energy, and services. Energy is added without adjustment since there is no statistical evidence of seasonality. Seasonal factors are revised at annual intervals or as required.

#### RELATING TO TABLE 2 IN SECTION 7.1

# SEASONAL ADJUSTMENT OF THE BALANCE OF PAYMENTS CURRENT ACCOUNT

The approach relies on multiplicative decomposition through X-12-ARIMA (see footnote 2 on page S74). The raw data for goods, services, income and current transfers are pre-adjusted to take a working-day effect into account. For goods, services and income, the working-day adjustment is corrected for national public holidays. Data on goods credits are also pre-adjusted for Easter. The seasonal adjustment for these items is carried out using these pre-adjusted series. The seasonal adjustment of the total current account is carried out by aggregating the seasonally adjusted euro area series for goods, services, income and current transfers. Seasonal (and trading day) factors are revised at semi-annual intervals or as required.





### **GENERAL NOTES**

The "Euro area statistics" section of the Monthly Bulletin focuses on statistics for the euro area as a whole. More detailed and longer runs of data, with further explanatory notes, are available in the "Statistics" section of the ECB's website (www.ecb.int). Services available under the "Data services" sub-section include a browser interface with search facilities, subscription to different datasets and a facility for downloading data directly as compressed Comma Separated Value (CSV) files. For further information, please contact us at: statistics@ecb.int.

In general, the cut-off date for the statistics included in the Monthly Bulletin is the day preceding the first meeting in the month of the Governing Council. For this issue, the cut-off date was 11 January 2006.

All data relate to the Euro 12, unless otherwise indicated. For the monetary data, the Harmonised Index of Consumer Prices (HICP), investment fund and financial market statistics, the statistical series relating to the euro area cover the EU Member States that had adopted the euro at the time to which the statistics relate. Where applicable, this is shown in the tables by means of a footnote; in the charts, the break is indicated by a dotted line. In these cases, where underlying data are available, absolute and percentage changes for 2001, calculated from a base in 2000, use a series which takes into account the impact of Greece's entry into the euro area.

Given that the composition of the ECU does not coincide with the former currencies of the countries which have adopted the single currency, pre-1999 amounts converted from the participating currencies into ECU at current ECU exchange rates are affected by movements in the currencies of EU Member States which have not adopted the euro. To avoid this effect on the monetary statistics, the pre-1999 data in Sections 2.1 to 2.8 are expressed in units converted from national currencies at the irrevocable euro exchange rates established on 31 December 1998. Unless otherwise indicated, price and cost statistics before 1999 are based on data expressed in national currency terms.

Methods of aggregation and/or consolidation (including cross-country consolidation) have been used where appropriate.

Recent data are often provisional and may be revised. Discrepancies between totals and their components may arise from rounding.

The group "Other EU Member States" comprises the Czech Republic, Denmark, Estonia, Cyprus, Latvia, Lithuania, Hungary, Malta, Poland, Slovenia, Slovakia, Sweden and United Kingdom.

In most cases, the terminology used within the tables follows international standards, such as those contained in the European System of Accounts 1995 (ESA 95) and the IMF Balance of Payments Manual. Transactions refer to voluntary exchanges (measured directly or derived), while flows also encompass changes in outstanding amounts owing to price and exchange rate changes, write-offs, and other changes.

In the tables, the term "up to (x) years" means "up to *and including* (x) years".

#### OVERVIEW

Developments in key indicators for the euro area are summarised in an overview table.

#### MONETARY POLICY STATISTICS

Section 1.4 shows statistics on minimum reserve and liquidity factors. Annual and quarterly observations refer to averages of the last reserve maintenance period of the year/quarter. Until December 2003, the maintenance periods started on the 24th calendar day of a month and ran to the 23rd of the following month. On 23 January 2003 the ECB announced changes to the operational



framework, which were implemented on 10 March 2004. As a result of these changes, maintenance periods start on the settlement day of the main refinancing operation (MRO) following the Governing Council meeting at which the monthly assessment of the monetary policy stance is scheduled. A transitional maintenance period was defined to cover the period from 24 January to 9 March 2004.

Table 1 in Section 1.4 shows the components of the reserve base of credit institutions subject to reserve requirements. The liabilities vis-à-vis other credit institutions subject to the ESCB's minimum reserve system, the ECB and participating national central banks are excluded from the reserve base. When a credit institution cannot provide evidence of the amount of its issues of debt securities with a maturity of up to two years held by the institutions mentioned above, it may deduct a certain percentage of these liabilities from its reserve base. The percentage for calculating the reserve base was 10% until November 1999 and 30% thereafter.

Table 2 in Section 1.4 contains average data for completed maintenance periods. The amount of the reserve requirement of each individual credit institution is first calculated by applying the reserve ratio for the corresponding categories of liabilities to the eligible liabilities, using the balance sheet data from the end of each calendar month. Subsequently, each credit institution deducts from this figure a lump-sum allowance of €100,000. The resulting required reserves are then aggregated at the euro area level (column 1). The current account holdings (column 2) are the aggregate average daily current account holdings of credit institutions, including those that serve the fulfilment of reserve requirements. The excess reserves (column 3) are the average current account holdings over the maintenance period in excess of the required reserves. The deficiencies (column 4) are defined as the average shortfalls of current account holdings from required reserves over the maintenance period, computed

on the basis of those credit institutions that have not fulfilled their reserve requirement. The interest rate on minimum reserves (column 5) is equal to the average, over the maintenance period, of the ECB's rate (weighted according to the number of calendar days) on the Eurosystem's main refinancing operations (see Section 1.3).

Table 3 in Section 1.4 shows the banking system's liquidity position, which is defined as the current account holdings in euro of credit institutions in the euro area with the Eurosystem. All amounts are derived from the consolidated financial statement of the Eurosystem. The other liquidity-absorbing operations (column 7) exclude the issuance of debt certificates initiated by national central banks in Stage Two of EMU. The net other factors (column 10) represent the netted remaining items in the consolidated financial statement of the Eurosystem. The credit institutions' current accounts (column 11) are equal to the difference between the sum of liquidity-providing factors (columns 1 to 5) and the sum of liquidity-absorbing factors (columns 6 to 10). The base money (column 12) is calculated as the sum of the deposit facility (column 6), the banknotes in circulation (column 8) and the credit institutions' current account holdings (column 11).

#### MONEY, BANKING AND INVESTMENT FUNDS

Section 2.1 shows the aggregated balance sheet of the monetary financial institution (MFI) sector, i.e. the sum of the harmonised balance sheets of all MFIs resident in the euro area. MFIs are central banks, credit institutions as defined under Community law, money market funds and other institutions whose business it is to receive deposits and/or close substitutes for deposits from entities other than MFIs and, for their own account (at least in economic terms), to grant credits and/or make investments in securities. A complete list of MFIs is published on the ECB's website.

**S** 78 Monthly Bulletin January 2006 Section 2.2 shows the consolidated balance sheet of the MFI sector, which is obtained by netting the aggregated balance sheet positions between MFIs in the euro area. Due to limited heterogeneity in recording practices, the sum of the inter-MFI positions is not necessarily zero; the balance is shown in column 10 of the liabilities side of Section 2.2. Section 2.3 sets out the euro area monetary aggregates and counterparts. These are derived from the consolidated MFI balance sheet, and include positions of non-MFIs resident in the euro area held with MFIs resident in the euro area; they also take account of some monetary assets/ liabilities of central government. Statistics on monetary aggregates and counterparts are adjusted for seasonal and trading-day effects. The external liabilities item of Sections 2.1 and 2.2 shows the holdings by non-euro area residents of i) shares/units issued by money market funds located in the euro area and ii) debt securities issued with a maturity of up to two years by MFIs located in the euro area. In Section 2.3, however, these holdings are excluded from the monetary aggregates and contribute to the item "net external assets".

Section 2.4 provides an analysis by sector, type and original maturity of loans granted by MFIs other than the Eurosystem (the banking system) resident in the euro area. Section 2.5 shows a sectoral and instrument analysis of deposits held with the euro area banking system. Section 2.6 shows the securities held by the euro area banking system, by type of issuer.

Sections 2.2 to 2.6 include transactions, which are derived as differences in outstanding amounts adjusted for reclassifications, revaluations, exchange rate variations and any other changes which do not arise from transactions. Section 2.7 shows selected revaluations which are used in the derivation of transactions. Sections 2.2 to 2.6 also provide growth rates in terms of annual percentage changes based on the transactions. Section 2.8 shows a quarterly currency breakdown of selected MFI balance sheet items. Details of the sector definitions are set out in the "Money and Banking Statistics Sector Manual – Guidance for the statistical classification of customers" (ECB, November 1999). The "Guidance Notes to the Regulation ECB/2001/13 on the MFI Balance Sheet Statistics" (ECB, November 2002) explains practices recommended to be followed by the NCBs. Since 1 January 1999 the statistical information has been collected and compiled on the basis of Regulation ECB/1998/16 of 1 December 1998 concerning the consolidated balance sheet of the Monetary Financial Institutions sector<sup>1</sup>, as last amended by Regulation ECB/2003/10<sup>2</sup>.

In line with this Regulation, the balance sheet item "money market paper" has been merged with the item "debt securities" on both the assets and liabilities side of the MFI balance sheet.

Section 2.9 shows end-of-quarter outstanding amounts for the balance sheet of the euro area investment funds (other than money market funds). The balance sheet is aggregated and therefore includes, among the liabilities, holdings by investment funds of shares/units issued by other investment funds. Total assets/ liabilities are also broken down by investment policy (equity funds, bond funds, mixed funds, real estate funds and other funds) and by type of investor (general public funds and special investors' funds). Section 2.10 shows the aggregated balance sheet for each investment fund sector as identified by investment policy and type of investor.

#### FINANCIAL AND NON-FINANCIAL ACCOUNTS

Sections 3.1 and 3.2 show quarterly data on financial accounts for non-financial sectors in the euro area, comprising general government (S.13 in the ESA 95), non-financial

1 OJL 356, 30.12.1998, p. 7.

 $<sup>2 \</sup> OJ\,L\,250, 2.10.2003, p.\,19.$ 

corporations (S.11 in the ESA 95), and households (S.14 in the ESA 95) including nonprofit institutions serving households (S.15 in the ESA 95). The data cover non-seasonally adjusted amounts outstanding and financial transactions classified according to the ESA 95 and show the main financial investment and financing activities of the non-financial sectors. On the financing side (liabilities), the data are presented by ESA 95 sector and original maturity ("short-term" refers to an original maturity of up to one year; "long-term" refers to an original maturity of over one year). Whenever possible, the financing taken from MFIs is presented separately. The information on financial investment (assets) is currently less detailed than that on financing, especially since a breakdown by sector is not possible.

Section 3.3 shows quarterly data on financial accounts for insurance corporations and pension funds (S.125 in the ESA 95) in the euro area. As in Sections 3.1 and 3.2, the data cover non-seasonally adjusted amounts outstanding and financial transactions, and show the main financial investment and financing activities of this sector.

The quarterly data in these three sections are based on quarterly national financial accounts data and MFI balance sheet and securities issues statistics. Sections 3.1 and 3.2 also refer to data taken from the BIS international banking statistics. Although all euro area countries contribute to the MFI balance sheet and securities issues statistics, Ireland and Luxembourg do not yet provide quarterly national financial accounts data.

Section 3.4 shows annual data on saving, investment (financial and non-financial) and financing for the euro area as a whole, and separately for non-financial corporations and households. These annual data provide, in particular, fuller sectoral information on the acquisition of financial assets and are consistent with the quarterly data in the two previous sections.

#### FINANCIAL MARKETS

The series on financial market statistics for the euro area cover the EU Member States that had adopted the euro at the time to which the statistics relate.

Statistics on securities other than shares and quoted shares (Sections 4.1 to 4.4) are produced by the ECB using data from the ESCB and the BIS. Section 4.5 presents MFI interest rates on euro-denominated deposits and loans by euro area residents. Statistics on money market interest rates, long-term government bond yields and stock market indices (Sections 4.6 to 4.8) are produced by the ECB using data from wire services.

Statistics on securities issues cover securities other than shares (debt securities), which are presented in Sections 4.1, 4.2 and 4.3, and quoted shares, which are presented in Section 4.4. Debt securities are broken down into shortterm and long-term securities. "Short-term" means securities with an original maturity of one year or less (in exceptional cases two years or less). Securities with a longer maturity, or with optional maturity dates, the latest of which is more than one year away, or with indefinite maturity dates, are classified as "long-term". Long-term debt securities issued by euro area residents are further broken down into fixed and variable rate issues. Fixed rate issues consist of issues where the coupon rate does not change during the life of the issues. Variable rate issues include all issues where the coupon is periodically refixed by reference to an independent interest rate or index. The statistics on debt securities are estimated to cover approximately 95% of total issues by euro area residents. Euro-denominated securities indicated in Sections 4.1, 4.2 and 4.3 also include items expressed in national denominations of the euro.

Section 4.1 shows securities other than shares, by original maturity, residency of the issuer and currency. The section presents outstanding amounts, gross issues and net issues of



securities other than shares denominated in euro and securities other than shares issued by euro area residents in euro and in all currencies for total and long-term debt securities. Net issues differ from the changes in outstanding amounts owing to valuation changes, reclassifications and other adjustments. This section also presents seasonally adjusted statistics including annualised six-month seasonally adjusted growth rates for total and long-term debt securities. The latter are calculated from the seasonally adjusted index of notional stocks from which the seasonal effects have been removed. See the Technical notes for details.

Section 4.2 contains a sectoral breakdown of outstanding amounts, gross issues and net issues for issuers resident in the euro area in line with the ESA 95. The ECB is included in the Eurosystem.

The total outstanding amounts for total and long-term debt securities in column 1 of table 1 in Section 4.2, corresponds to the data on outstanding amounts for total and long-term debt securities issued by euro area residents in column 7 of Section 4.1. The outstanding amounts for total and long-term debt securities issued by MFIs in column 2 of table 1, Section 4.2 are broadly comparable with data for debt securities issued as shown on the liabilities side of the aggregated MFI balance sheet in column 8 of table 2. Section 2.1. The total net issues for total debt securities in column 1 of table 2 in Section 4.2 correspond to the data on total net issues by euro area residents in column 9 of Section 4.1. The residual difference between long-term debt securities and total fixed and variable rate long-term debt securities in table 1, Section 4.2 consists of zero coupon bonds and revaluation effects.

Section 4.3 shows non-seasonally and seasonally adjusted growth rates for debt securities issued by euro area residents (broken down by maturity, type of instrument, sector of the issuer and currency), which are based on financial transactions that occur when an institutional unit incurs or redeems liabilities. The growth rates therefore exclude reclassifications, revaluations, exchange rate variations and any other changes which do not arise from transactions. The seasonally adjusted growth rates have been annualised for presentational purposes. See the Technical notes for details.

Section 4.4, columns 1, 4, 6 and 8, show the outstanding amounts of quoted shares issued by euro area residents broken down by issuing sector. The monthly data for quoted shares issued by non-financial corporations correspond to the quarterly series shown in Section 3.2 (main liabilities, column 21).

Section 4.4, columns 3, 5, 7 and 9, show annual growth rates for quoted shares issued by euro area residents (broken down by the sector of the issuer), which are based on financial transactions that occur when an issuer sells or redeems shares for cash excluding investments in the issuers' own shares. Transactions include the quotation of an issuer on a stock exchange for the first time and the creation or deletion of new instruments. The calculation of annual growth rates excludes reclassifications, revaluations and any other changes which do not arise from transactions.

Section 4.5 presents statistics on all the interest rates that MFIs resident in the euro area apply to euro-denominated deposits and loans vis-àvis households and non-financial corporations resident in the euro area. Euro area MFI interest rates are calculated as a weighted average (by corresponding business volume) of the euro area countries' interest rates for each category.

MFI interest rate statistics are broken down by type of business coverage, sector, instrument category and maturity, period of notice or initial period of interest rate fixation. The new MFI interest rate statistics replace the ten transitional statistical series on euro area retail interest rates that have been published in the ECB's Monthly Bulletin since January 1999. Section 4.6 presents money market interest rates for the euro area, the United States and Japan. For the euro area, a broad spectrum of money market interest rates is covered spanning from interest rates on overnight deposits to those on twelve-month deposits. Before January 1999 synthetic euro area interest rates were calculated on the basis of national rates weighted by GDP. With the exception of the overnight rate to December 1998, monthly, quarterly and yearly values are period averages. Overnight deposits are represented by interbank deposit bid rates up to December 1998. From January 1999 column 1 of Section 4.6 shows the euro overnight index average (EONIA). These are end-of-period rates up to December 1998 and period averages thereafter. From January 1999 interest rates on one-, three-, sixand twelve-month deposits are euro interbank offered rates (EURIBOR); until December 1998, London interbank offered rates (LIBOR) where available. For the United States and Japan, interest rates on three-month deposits are represented by LIBOR.

Section 4.7 presents government bond yields for the euro area, the United States and Japan. Until December 1998, two-, three-, five- and seven-year euro area yields were end-of-period values and ten-year yields period averages. Thereafter, all yields are period averages. Until December 1998, euro area yields were calculated on the basis of harmonised national government bond yields weighted by GDP; thereafter, the weights are the nominal outstanding amounts of government bonds in each maturity band. For the United States and Japan, ten-year yields are period averages.

Section 4.8 shows stock market indices for the euro area, the United States and Japan.

# PRICES, OUTPUT, DEMAND AND LABOUR MARKETS

Most of the data described in this section are produced by the European Commission (mainly Eurostat) and national statistical authorities. Euro area results are obtained by aggregating data for individual countries. As far as possible, the data are harmonised and comparable. Statistics on hourly labour costs, GDP and expenditure components, value added by economic activity, industrial production, retail sales and passenger car registrations are adjusted for the variations in the number of working days.

The Harmonised Index of Consumer Prices (HICP) for the euro area (Section 5.1) is available from 1995 onwards. It is based on national HICPs, which follow the same methodology in all euro area countries. The breakdown by goods and services components is derived from the Classification of individual consumption by purpose (Coicop/HICP). The HICP covers monetary expenditure on final consumption by households on the economic territory of the euro area. The table includes seasonally adjusted HICP data which are compiled by the ECB.

Industrial producer prices (Table 2 in Section 5.1), industrial production, industrial new orders, industrial turnover and retail sales (Section 5.2) are covered by Council Regulation (EC) No 1165/98 of 19 May 1998 concerning short-term statistics<sup>3</sup>. The breakdown by enduse of products for industrial producer prices and industrial production is the harmonised sub-division of industry excluding construction (NACE sections C to E) into Main Industrial Groupings (MIGs) as defined by Commission Regulation (EC) No 586/2001 of 26 March 2001<sup>4</sup>. Industrial producer prices reflect the exfactory gate prices of producers. They include indirect taxes except VAT and other deductible taxes. Industrial production reflects the value added of the industries concerned.

World market prices of raw materials (Table 2 in Section 5.1) measures price changes of eurodenominated euro area imports compared with the base period.

- 3 OJL 162, 5.6.1998, p. 1.
- 4 OJL 86, 27.3.2001, p. 11.

The labour cost indices (Table 3 in Section 5.1) measure the changes in labour costs per hour worked in industry (including construction) and market services. Their methodology is laid down in Regulation (EC) No 450/2003 of the European Parliament and of the Council of 27 February 2003 concerning the labour cost index<sup>5</sup> and in the implementing Commission Regulation (EC) No 1216/2003 of 7 July 20036. A breakdown of hourly labour costs for the euro area is available by labour cost component (wages and salaries, and employers' social contributions plus employment-related taxes paid by the employer less subsidies received by the employer) and by economic activity. The ECB calculates the indicator of negotiated wages (memo item in Table 3 of Section 5.1) on the basis of non-harmonised, nationaldefinition data.

Unit labour cost components (Table 4 in Section 5.1), GDP and its components (Tables 1 and 2 in Section 5.2), GDP deflators (Table 5 in Section 5.1) and employment statistics (Table 1 in Section 5.3) are results of the ESA 95 quarterly national accounts.

Industrial new orders (Table 4 in Section 5.2) measure the orders received during the reference period and cover industries working mainly on the basis of orders – in particular textile, pulp and paper, chemical, metal, capital goods and durable consumer goods industries. The data are calculated on the basis of current prices.

Indices for turnover in industry and for the retail trade (Table 4 in Section 5.2) measure the turnover, including all duties and taxes with the exception of VAT, invoiced during the reference period. Retail trade turnover covers all retail trade excluding sales of motor vehicles and motorcycles, and except repairs. New passenger car registrations covers registrations of both private and commercial passenger cars.

Qualitative business and consumer survey data (Table 5 in Section 5.2) draw on the European Commission Business and Consumer Surveys. Unemployment rates (Table 2 in Section 5.3) conform to International Labour Organisation (ILO) guidelines. They refer to persons actively seeking work as a share of the labour force, using harmonised criteria and definitions. The labour force estimates underlying the unemployment rate are different from the sum of the employment and unemployment levels published in Section 5.3.

#### **GOVERNMENT FINANCE**

Sections 6.1 to 6.5 show the general government fiscal position in the euro area. The data are mainly consolidated and are based on the ESA 95 methodology. The annual euro area aggregates in Sections 6.1 to 6.3 are compiled by the ECB from harmonised data provided by the NCBs, which are regularly updated. The deficit and debt data for the euro area countries may therefore differ from those used by the European Commission within the excessive deficit procedure. The quarterly euro area aggregates in Sections 6.4 and 6.5 are compiled by the ECB on the basis of Eurostat and national data.

Section 6.1 presents annual figures on general government revenue and expenditure on the basis of definitions laid down in Commission Regulation (EC) No 1500/2000 of 10 July 20007 amending the ESA 95. Section 6.2 shows of general government gross details consolidated debt at nominal value in line with the Treaty provisions on the excessive deficit procedure. Sections 6.1 and 6.2 include summary data for the individual euro area countries owing to their importance in the framework of the Stability and Growth Pact. The deficits/surpluses presented for the individual euro area countries correspond to EDP B.9 as defined by Commission Regulation (EC) No 351/2002 of 25 February 2002

5 OJL 69, 13.3.2003, p. 1.

6 OJL 169, 8.7.2003, p. 37.



<sup>7</sup> OJL 172, 12.7.2000, p. 3.

amending Council Regulation (EC) No 3605/93 as regards references to the ESA 95. Section 6.3 presents changes in general government debt. The difference between the change in the government debt and the government deficit the deficit-debt adjustment - is mainly explained by government transactions in financial assets and by foreign exchange valuation effects. Section 6.4 presents quarterly figures on general government revenue and expenditure on the basis of definitions laid down in the Regulation (EC) No 1221/2002 of the European Parliament and of the Council of 10 June 20028 on quarterly nonfinancial accounts for general government. Section 6.5 presents quarterly figures on gross consolidated government debt, the deficit-debt adjustment and the government borrowing requirement. These figures are compiled using data provided by the Member States under Regulations (EC) No 501/2004 and 1222/2004 and data provided by the National Central Banks.

#### **EXTERNAL TRANSACTIONS AND POSITIONS**

The concepts and definitions used in balance of payments (b.o.p.) and international investment position (i.i.p.) statistics (Sections 7.1 to 7.4) are generally in line with the IMF Balance of Payments Manual (fifth edition, October 1993), the ECB Guideline of 16 July 2004 on the statistical reporting requirements of the ECB  $(ECB/2004/15)^9$ , and Eurostat documents. Additional references about the methodologies and sources used in the euro area b.o.p. and i.i.p. statistics can be found in the ECB publication entitled "European Union balance of payments/international investment position statistical methods" (November 2004), and in the following task force reports: "Portfolio investment collection systems" (June 2002), "Portfolio investment income" (August 2003) and "Foreign direct investment" (March 2004), which can be downloaded from the ECB's website. In addition, the report of the ECB/ Commission (Eurostat) Task Force on Quality of balance of payments and international investment position statistics (June 2004) is available on the website of the Committee on Monetary, Financial and Balance of Payments Statistics (www.cmfb.org). The first annual quality report on the euro area b.o.p./i.i.p. (January 2005), which is based on the Task Force's recommendations, is available on the ECB's website.

The presentation of net transactions in the financial account follows the sign convention of the IMF Balance of Payments Manual: an increase of assets appears with a minus sign, while an increase of liabilities appears with a plus sign. In the current account and capital account, both credit and debit transactions are presented with a plus sign.

The euro area b.o.p. is compiled by the ECB. The recent monthly figures should be regarded as provisional. Data are revised when figures for the following month and/or the detailed quarterly b.o.p. are published. Earlier data are revised periodically or as a result of methodological changes in the compilation of the source data.

In Section 7.1, Table 2 contains seasonally adjusted data for the current account. Where appropriate, the adjustment covers also working-day, leap year and/or Easter effects. Table 5 provides a sectoral breakdown of euro area purchasers of securities issued by nonresidents of the euro area. It is not yet possible to show a sectoral breakdown of euro area issuers of securities acquired by non-residents. In Tables 6 and 7 the breakdown between "loans" and "currency and deposits" is based on the sector of the non-resident counterpart, i.e. assets vis-à-vis non-resident banks are classified as deposits, whereas assets vis-à-vis other non-resident sectors are classified as loans. This breakdown follows the distinction made in other statistics, such as the MFI consolidated balance sheet, and conforms to the IMF Balance of Payments Manual.

8 OJ L 179, 9.7.2002, p. 1.

<sup>9</sup> OJ L 354, 30.11.2004, p. 34.

Section 7.2 contains a monetary presentation of the b.o.p.: the b.o.p. transactions mirroring the transactions in the external counterpart of M3. The data follow the sign conventions of the b.o.p., except for the transactions in the external counterpart of M3 taken from money and banking statistics (column 12), where a positive sign denotes an increase of assets or a decrease of liabilities. In portfolio investment liabilities (columns 5 and 6), the b.o.p. transactions include sales and purchases of equity and debt securities issued by MFIs in the euro area, apart from shares of money market funds and debt securities with a maturity of up to two years. A methodological note on the monetary presentation of the euro area b.o.p. is available in the "Statistics" section of the ECB's website. See also Box 1 in the June 2003 issue of the Monthly Bulletin.

Section 7.3 presents a geographical breakdown of the euro area b.o.p. (Tables 1 to 4) and i.i.p. (Table 5) vis-à-vis main partner countries individually or as a group, distinguishing between EU Member States outside the euro area and countries or areas outside the European Union. The breakdown also shows transactions and positions vis-à-vis EU institutions (which, apart from the ECB, are treated statistically as outside the euro area, regardless of their physical location) and for some purposes also offshore centres and international organisations. Tables 1 to 4 show cumulative b.o.p. transactions in the latest four quarters; Table 5 shows a geographical breakdown of the i.i.p. for the latest end-year. The breakdown does not cover transactions or positions in portfolio investment liabilities, financial derivatives and international reserves. The geographical breakdown is described in the article entitled "Euro area balance of payments and international investment position vis-à-vis main counterparts" in the February 2005 issue of the Monthly Bulletin.

The data on the euro area i.i.p. in Section 7.4 are based on positions vis-à-vis non-residents of the euro area, considering the euro area as a single economic entity (see also Box 9 in the December 2002 issue of the Monthly Bulletin). The i.i.p. is valued at current market prices, with the exception of direct investment, where book values are used to a large extent. The quarterly i.i.p. is compiled on the basis of the same methodological framework as the annual i.i.p. As some data sources are not available on a quarterly basis (or are available with a delay), the quarterly i.i.p. is partly estimated on the basis of financial transactions and asset prices and foreign exchange developments.

The outstanding amounts of the Eurosystem's international reserves and related assets and liabilities are shown in Section 7.4, Table 5, together with the part held by the ECB. These figures are not fully comparable with those of the Eurosystem's weekly financial statement owing to differences in coverage and valuation. The data in Table 5 are in line with the recommendations for the IMF/BIS template on international reserves and foreign currency liquidity. Changes in the gold holdings of the Eurosystem (column 3) are due to transactions in gold within the terms of the Central Bank Gold Agreement of 26 September 1999, updated on 8 March 2004. More information on the statistical treatment of the Eurosystem's international reserves can be found in a publication entitled "Statistical treatment of the Eurosystem's international reserves" (October 2000), which can be downloaded from the ECB's website. The website also contains more comprehensive data in accordance with the template on international reserves and foreign currency liquidity.

Section 7.5 shows data on euro area external trade in goods. The main source is Eurostat. The ECB derives volume indices from Eurostat value and unit value indices, and performs seasonal adjustment of unit value indices, while value data are seasonally and working-day adjusted by Eurostat.

The breakdown by product group in columns 4 to 6 and 9 to 11 of Table 1 in Section 7.5 is in line with the classification by Broad Economic Categories. Manufactured goods (columns 7

and 12) and oil (column 13) are in line with the SITC Rev. 3 definition. The geographical breakdown (Table 2 in Section 7.5) shows main trading partners individually or in regional groups. Mainland China excludes Hong Kong.

Owing to differences in definitions, classification, coverage and time of recording, external trade data, in particular for imports, are not fully comparable with the goods item in the balance of payments statistics (Sections 7.1 to 7.3). The difference for imports has been around 5% in recent years (ECB estimate), a significant part of which relates to the inclusion of insurance and freight services in the external trade data (c.i.f. basis).

#### **EXCHANGE RATES**

Section 8.1 shows nominal and real effective exchange rate (EER) indices for the euro calculated by the ECB on the basis of weighted averages of bilateral exchange rates of the euro against the currencies of the euro area's trading partners. A positive change denotes an appreciation of the euro. Weights are based on trade in manufactured goods with the trading partners in the periods 1995-1997 and 1999-2001, and are calculated to account for thirdmarket effects. The EER indices result from the linking at the beginning of 1999 of the indices based on 1995-1997 weights to those based on 1999-2001 weights. The EER-23 group of trading partners is composed of the 13 non-euro area EU Member States, Australia, Canada, China, Hong Kong, Japan, Norway, Singapore, South Korea, Switzerland and the United States. The EER-42 group includes, in addition to the EER-23, the following countries: Algeria, Argentina, Brazil, Bulgaria, Croatia, India, Indonesia, Israel, Malaysia, Mexico, Morocco, New Zealand, the Philippines, Romania, Russia, South Africa, Taiwan, Thailand and Turkey. Real EERs are calculated using consumer price indices, producer price indices, gross domestic product deflators, unit labour costs in manufacturing and unit labour costs in the total economy.

For more detailed information on the calculation of the EERs, see Box 10 entitled "Update of the overall trade weights for the effective exchange rates of the euro and computation of a new set of euro indicators" in the September 2004 issue of the Monthly Bulletin and the ECB's Occasional Paper No 2 ("The effective exchange rates of the euro" by Luca Buldorini, Stelios Makrydakis and Christian Thimann, February 2002), which can be downloaded from the ECB's website.

The bilateral rates shown in Section 8.2 are monthly averages of those published daily as reference rates for these currencies.

#### **DEVELOPMENTS OUTSIDE THE EURO AREA**

Statistics on other EU Member States (Section 9.1) follow the same principles as those for data relating to the euro area. Data for the United States and Japan contained in Section 9.2 are obtained from national sources.



### ANNEXES

### CHRONOLOGY OF MONETARY POLICY MEASURES OF THE EUROSYSTEM'

#### 8 JANUARY 2004

The Governing Council of the ECB decides that the minimum bid rate on the main refinancing operations and the interest rates on the marginal lending facility and the deposit facility will remain unchanged at 2.0%, 3.0% and 1.0% respectively.

#### **12 JANUARY 2004**

The Governing Council of the ECB decides to increase the allotment amount for each of the longer-term refinancing operations to be conducted in the year 2004 from  $\in 15$  billion to  $\in 25$  billion. This increased amount takes into consideration the higher liquidity needs of the euro area banking system anticipated for the year 2004. The Eurosystem will, however, continue to provide the bulk of liquidity through its main refinancing operations. The Governing Council may decide to adjust the allotment amount again at the beginning of 2005.

#### 5 FEBRUARY, 4 MARCH 2004

The Governing Council of the ECB decides that the minimum bid rate on the main refinancing operations and the interest rates on the marginal lending facility and the deposit facility will remain unchanged at 2.0%, 3.0% and 1.0% respectively.

#### 10 MARCH 2004

In accordance with the Governing Council's decision of 23 January 2003, the maturity of the Eurosystem's main refinancing operations is reduced from two weeks to one week and the maintenance period for the Eurosystem's required reserve system is redefined to start on the settlement day of the main refinancing operation following the Governing Council meeting at which the monthly assessment of the monetary policy

stance is pre-scheduled, rather than on the 24th day of the month.

#### I APRIL, 6 MAY, 3 JUNE, I JULY, 5 AUGUST, 2 SEPTEMBER, 7 OCTOBER, 4 NOVEMBER, 2 DECEMBER 2004 AND 13 JANUARY 2005

The Governing Council of the ECB decides that the minimum bid rate on the main refinancing operations and the interest rates on the marginal lending facility and the deposit facility will remain unchanged at 2.0%, 3.0% and 1.0% respectively.

#### **14 JANUARY 2005**

The Governing Council of the ECB decides to increase the allotment amount for each of the longer-term refinancing operations to be conducted in the year 2005 from €25 billion to €30 billion. This increased amount takes into consideration the higher liquidity needs of the euro area banking system anticipated in 2005. The Eurosystem will however continue to provide the bulk of liquidity through its main refinancing operations. The Governing Council may decide to adjust the allotment amount again at the beginning of 2006.

#### 3 FEBRUARY, 3 MARCH, 7 APRIL, 4 MAY, 2 JUNE, 7 JULY, 4 AUGUST, 1 SEPTEMBER, 6 OCTOBER AND 3 NOVEMBER 2005

The Governing Council of the ECB decides that the minimum bid rate on the main refinancing operations and the interest rates on the marginal lending facility and the deposit facility will



<sup>1</sup> The chronology of monetary policy measures of the Eurosystem taken between 1999 and 2003 can be found on pages 176 to 180 of the ECB's Annual report 1999, on pages 205 to 208 of the ECB's Annual report 2000, on pages 219 to 220 of the ECB's Annual Report 2001, on pages 234 to 235 of the ECB's Annual Report 2002 and on pages 217 to 218 of the ECB's Annual Report 2003 respectively.

remain unchanged at 2.0%, 3.0% and 1.0% respectively.

#### I DECEMBER 2005

The Governing Council of the ECB decides to increase the minimum bid rate on the main refinancing operations by 0.25 percentage point to 2.25%, starting from the operation to be settled on 6 December 2005. In addition, it decides to increase the interest rates on both the marginal lending facility and the deposit facility by 0.25 percentage point, to 3.25% and 1.25% respectively, both with effect from 6 December 2005.

#### I6 DECEMBER 2005

The Governing Council of the ECB decides to increase the allotment amount for each of the longer-term refinancing operations to be conducted in the year 2006 from €30 billion to €40 billion. This increased amount takes two aspects into consideration. First, the liquidity needs of the euro area banking system are expected to increase further in the year 2006. Second, the Eurosystem has decided to increase slightly the share of the liquidity needs satisfied by the longer-term refinancing operations. The Eurosystem will, however, continue to provide the bulk of liquidity through its main refinancing operations. The Governing Council may decide to adjust the allotment amount again at the beginning of 2007.

#### **12 JANUARY 2006**

The Governing Council of the ECB decides that the minimum bid rate on the main refinancing operations and the interest rates on the marginal lending facility and the deposit facility will remain unchanged at 2.25%, 3.25% and 1.25% respectively.





### DOCUMENTS PUBLISHED BY THE EUROPEAN CENTRAL BANK SINCE 2005

This list is designed to inform readers about selected documents published by the European Central Bank since January 2005. For Working Papers, the list only refers to publications released between October and December 2005. The publications are available to interested parties free of charge from the Press and Information Division. Please submit orders in writing to the postal address given on the back of the title page.

For a complete list of documents published by the European Central Bank and by the European Monetary Institute, please visit the ECB's website (http://www.ecb.int).

#### **ANNUAL REPORT**

"Annual Report 2004", April 2005.

#### **MONTHLY BULLETIN ARTICLES**

"The new Basel Capital Accord: main features and implications", January 2005. "Financial flows to emerging market economies: changing patterns and recent developments", January 2005.

"Bank market discipline", February 2005.

"Initial experience with the changes to the Eurosystem's operational framework for monetary policy implementation", February 2005.

"Euro area balance of payments and international investment position vis-à-vis main counterparts", February 2005.

"Asset price bubbles and monetary policy", April 2005.

"Comparability of statistics for the euro area, the United States and Japan", April 2005.

"The ESCB-CESR standards for securities clearing and settlement in the European Union", April 2005.

"Monetary policy and inflation differentials in a heterogeneous currency area", May 2005.

"Consolidation and diversification in the euro area banking sector", May 2005.

"The evolving framework for corporate governance", May 2005.

"The Harmonised Index of Consumer Prices: concept, properties and experience to date", July 2005.

"The Lisbon strategy - five years on", July 2005

"The use of harmonised MFI interest rate statistics", July 2005.

"The reform of the Stability and Growth Pact", August 2005.

"The role of 'Emerging Asia' in the global economy", August 2005.

"The euro banknotes: developments and future challenges", August 2005.

"Money demand and uncertainty", October 2005.

"Assessing the performance of financial systems", October 2005.

"Price-setting behaviour in the euro area", November 2005.

"Developments in corporate finance in the euro area", November 2005.

"Economic and financial relations between the euro area and Russia", November 2005.

"The predictability of the ECB's monetary policy", January 2006.

"Hedge funds: developments and policy implications", January 2006.

#### **STATISTICS POCKET BOOK**

Available monthly since August 2003.

#### **OCCASIONAL PAPER SERIES**

- 22 "Assessing potential output growth in the euro area a growth accounting perspective" by A. Musso and T. Westermann, January 2005.
- 23 "The bank lending survey for the euro area" by J. Berg, A. Van Rixtel, A. Ferrando, G. de Bondt and S. Scopel, February 2005.
- 24 "Wage diversity in the euro area an overview of labour cost differentials across industries" by V. Genre, D. Momferatou and G. Mourre, February 2005.
- 25 "Government debt management in the euro area: recent theoretical developments and changes in practices" by G. Wolswijk and J. de Haan, March 2005.
- 26 "Analysing banking sector conditions: how to use macro-prudential indicators" by L. Mörttinen, P. Poloni, P. Sandars and J. Vesala, April 2005.
- 27 "The EU budget: how much scope for institutional reform?" by H. Enderlein, J. Lindner, O. Calvo-Gonzalez and R. Ritter, April 2005.
- 28 "Regulatory reforms in selected EU network industries" by R. Martin, M. Roma and I. Vansteenkiste, April 2005.
- 29 "Wealth and asset price effects on economic activity" by F. Altissimo, E. Georgiou, T. Sastre, M. T. Valderrama, G. Sterne, M. Stocker, M. Weth, K. Whelan, A. Willman, June 2005.
- 30 "Competitiveness and the export performance of the euro area" by a task force of the Monetary Policy Committee of the European System of Central Banks, June 2005.
- 31 "Regional monetary integration in the member states of the Gulf Cooperation Council" by M. Sturm and N. Siegfried, June 2005.
- 32 "Managing financial crises in emerging market economies experience with the involvement of private sector creditors" by an International Relations Committee Task Force, June 2005.
- 33 "Integration of securities market infrastructures in the euro area" by H. Schmiedel and A. Schönenberger, July 2005.
- 34 "Hedge funds and their implications for financial stability" by T. Garbaravicius and F. Dierick, August 2005.
- 35 "The institutional framework for financial market policy in the USA seen from an EU perspective" by R. Petschnigg, September 2005.
- 36 "Economic and monetary integration of the new Member States: helping to chart the route" by I. Angeloni, M. Flad and F. P. Mongelli, September 2005.
- 37 "Financing conditions in the euro area" by L. Bê Duc, G. de Bondt, A. Calza, D. Marqués Ibáñez, A. van Rixtel and S. Scopel, October 2005.
- 38 "Economic reactions to public finance consolidation: a survey of the literature" by M. G. Briotti, October 2005.
- 39 "Labour productivity in the Nordic EU countries: a comparative overview and explanatory factors 1980-2004" by A. Annenkov and C. Madaschi, October 2005.
- 40 "What does European institutional integration tell us about trade integration?" by F. P. Mongelli, E. Dorrucci and I. Agur, December 2005.
- 41 "Trends and patterns in working time across euro area countries 1970-2004: causes and consequences" by N. Leiner-Killinger, C. Madaschi and M. Ward-Warmedinger, December 2005.
- 42 "The New Basel Capital Framework and its implementation in the European Union" by F. Dierick, F. Pires, M. Scheicher and K. G. Spitzer, December 2005.

#### **WORKING PAPER SERIES**

531 "Market power, innovative activity and exchange rate pass-through in the euro area" by S. N. Brissimis and T. S. Kosma, October 2005.



- 532 "Intra- and extra-euro area import demand for manufactures" by R. Anderton,B. H. Baltagi, F. Skudelny and N. Sousa, October 2005.
- 533 "Discretionary policy, multiple equilibria and monetary instruments" by A. Schabert, October 2005.
- 534 "Time-dependent or state-dependent price-setting? Micro-evidence from German metal-working industries" by H. Stahl, October 2005.
- 535 "The pricing behaviour of firms in the euro area: new survey evidence" by S. Fabiani,
  M. Druant, I. Hernando, C. Kwapil, B. Landau, C. Loupias, F. Martins, T. Y. Mathä,
  R. Sabbatini, H. Stahl and A. C. J. Stokman, October 2005.
- 536 "Heterogeneity in consumer price stickiness: a microeconometric investigation" by D. Fougère, H. Le Bihan and P. Sevestre, October 2005.
- 537 "Global inflation" by M. Ciccarelli and B. Mojon, October 2005.
- 538 "The price-setting behaviour of Spanish firms: evidence from survey data" by L. J. Álvarez and I. Hernando, October 2005.
- 539 "Inflation persistence and monetary policy design: an overview" by A. T. Levin and R. Moessner, November 2005.
- 540 "Optimal discretionary policy and uncertainty about inflation persistence" by R. Moessner, November 2005.
- 541 "Consumer price behaviour in Luxembourg: evidence from micro CPI data" by P. Lünnemann and T. Y. Mathä, November 2005.
- 542 "Liquidity and real equilibrium interest rates: a framework of analysis" by L. Stracca, November 2005.
- 543 "Lending booms in the new EU Member States: will euro adoption matter?" by M. Brzoza-Brzezina, November 2005.
- 544 "Forecasting the yield curve in a data-rich environment: a no-arbitrage factor-augmented VAR approach" by E. Mönch, November 2005.
- 545 "Trade integration of central and eastern European countries: lessons from a gravity model" by M. Bussière, J. Fidrmuc and B. Schnatz, November 2005.
- 546 "The natural real interest rate and the output gap in the euro area: a joint estimation" by J. Garnier and B.-R. Wilhelmsen, November 2005.
- 547 "Bank finance versus bond finance: what explains the differences between the US and Europe?" by F. de Fiore and H. Uhlig, November 2005.
- 548 "The link between interest rates and exchange rates: do contractionary depreciations make a difference?" by M. Sánchez, November 2005.
- 549 "Eigenvalue filtering in VAR models with application to the Czech business cycle" by J. Beneš and D. Vávra, November 2005.
- 550 "Underwriter competition and gross spreads in the eurobond market" by M. G. Kollo, November 2005.
- 551 "Technological diversification" by M. Koren and S. Tenreyro, November 2005.
- 552 "European Union enlargement and equity markets in accession countries" by T. Dvorak and R. Podpiera, November 2005.
- 553 "Global bond portfolios and EMU" by P. R. Lane, November 2005.
- 554 "Equilibrium and inefficiency in fixed rate tenders" by C. Ewerhart, N. Cassola and N. Valla, November 2005.
- 555 "Near-rational exuberance" by J. Bullard, G. W. Evans and S. Honkapohja, November 2005.
- 556 "The role of real wage rigidity and labour market frictions for unemployment and inflation dynamics" by K. Christoffel and T. Linzert, November 2005.

- 557 "How should central banks communicate?" by M. Ehrmann and M. Fratzscher, November 2005.
- 558 "Ricardian fiscal regimes in the European Union" by A. Afonso, November 2005.
- 559 "When did unsystematic monetary policy have an effect on inflation?" by B. Mojon, December 2005.
- 560 "The determinants of 'domestic' original sin in emerging market economies" by A. Mehl and J. Reynaud, December 2005.
- 561 "Price setting in German manufacturing: new evidence from new survey data" by H. Stahl, December 2005.
- 562 "The price setting behaviour of Portuguese firms: evidence from survey data" by F. Martins, December 2005.
- 563 "Sticky prices in the euro area: a summary of new micro evidence" by L. J. Álvarez,
  E. Dhyne, M. M. Hoeberichts, C. Kwapil, H. Le Bihan, P. Lünnemann, F. Martins,
  R. Sabbatini, H. Stahl, P. Vermeulen and J. Vilmunen, December 2005.
- 564 "Forecasting the central bank's inflation objective is a good rule of thumb" by M. Diron and B. Mojon, December 2005.
- 565 "The timing of central bank communication" by M. Ehrmann and M. Fratzscher, December 2005.
- 566 "Real versus financial frictions to capital investment" by N. Bayraktar, P. Sakellaris and P. Vermeulen, December 2005.
- 567 "Is the time ripe for a currency union in emerging East Asia? The role of monetary stabilisation" by M. Sánchez, December 2005.
- 568 "Exploring the international linkages of the euro area: a global VAR analysis" by S. Dées,F. di Mauro, M. H. Pesaran and L. V. Smith, December 2005.
- 569 "Towards European monetary integration: the evolution of currency risk premium as a measure for monetary convergence prior to the implementation of currency unions" by F. González and S. Launonen, December 2005.

#### **OTHER PUBLICATIONS**

"Recycling of euro banknotes: framework for the detection of counterfeits and fitness sorting by credit institutions and other professional cash handlers", January 2005.

"Review of the international role of the euro", January 2005.

"Euro area balance of payments and international investment position statistics – Annual quality report", January 2005.

"Banking structures in the new EU Member States", January 2005.

"Progress Report on Target2", February 2005.

"The implementation of monetary policy in the euro area: General documentation on Eurosystem monetary policy instruments and procedures", February 2005.

"Review of the application of the Lamfalussy framework to EU securities markets legislation", February 2005.

"Payment and securities settlement systems in the accession countries – Addendum incorporating 2003 figures", February 2005.

"Statistics and their use for monetary and economic policy-making", March 2005.

"Letter from the ECB President to the Chairman of the International Accounting Standards Board of 13 April 2005: in support of the current proposal to amendments to IAS 39 – The fair value option", April 2005.

"Euro money market study 2004", May 2005.



"Correspondent central banking model (CCBM) – procedure for Eurosystem counterparties", May 2005.

"Regional economic integration in a global framework – proceedings of the G20 Workshop held in Beijing, 22-23 September 2004", May 2005.

"TARGET Annual Report 2004", May 2005.

"The New EU Member States: Convergence and Stability", May 2005.

"Financial stability review", June 2005.

"Letter from the ECB President to Mr Nikolaos Vakalis, Member of the European Parliament", June 2005.

"Guide to consultation of the European Central Bank by national authorities regarding draft legislative provisions", June 2005.

"Assessment of SORBNET-EURO and BIREL against the Core Principles: connection of SORBNET-EURO to TARGET via the Banca d'Italia and its national RTGS system BIREL", June 2005.

"Information guide for credit institutions using TARGET", June 2005.

"Statistical classification of financial markets instruments", July 2005.

"Reply of the ECB to the public consultation by the CEBS on the consolidated financial reporting framework for credit institutions", July 2005.

"Payment and securities settlement systems in the European Union – Addendum incorporating 2003 figures" (Blue Book, August 2005), August 2005.

"Eurosystem contribution to the public consultation by the European Commission on the Green Paper on Financial Services Policy (2005-2010)", August 2005.

"Central banks' provision of retail payment services in euro to credit institutions – policy statement", August 2005.

"ECB statistics: a brief overview", August 2005.

"Assessment of euro retail payment systems against the applicable core principles", August 2005.

"Indicators of financial integration in the euro area", September 2005.

"EU banking structures", October 2005.

"EU banking sector stability", October 2005.

"Second progress report on TARGET2", October 2005.

"Legal aspects of the European System of Central Banks", October 2005.

"European Union balance of payments/international investment position statistical methods", November 2005.

"Large EU banks' exposures to hedge funds", November 2005.

"Green paper on the enhancement of the EU framework for investment funds. Eurosystem contribution to the Commission's public consultation", November 2005.

"The European Commission's Green Paper on mortgage credit in the EU. Eurosystem

contribution to the public consultation", December 2005.

"Financial stability review", December 2005.

"Review of the international role of the euro", December 2005.

"The Eurosystem, the Union and beyond", December 2005.

#### **INFORMATION BROCHURES**

"The current TARGET system", August 2005.

"TARGET2 – innovation and transformation", August 2005.

"The euro area at a glance", August 2005.





## **GLOSSARY**

This glossary contains selected items that are frequently used in the Monthly Bulletin. A more comprehensive and detailed glossary can be found on the ECB's website (www.ecb.int/home/glossary/html/index.en.html).

**Autonomous liquidity factors:** liquidity factors that do not normally stem from the use of monetary policy instruments. Such factors are, for example, banknotes in circulation, government deposits with the central bank and the net foreign assets of the central bank.

**Bank lending survey (BLS):** a quarterly survey on lending policies that has been conducted by the Eurosystem since January 2003. It addresses qualitative questions on developments in credit standards, terms and conditions of loans and loan demand for both enterprises and households to a predefined sample group of banks in the euro area.

**Borrowing requirement (general government):** net incurrence of debt by general government.

**Central parity (or central rate):** the exchange rate of each ERM II member currency vis-à-vis the euro, around which the ERM II fluctuation margins are defined.

**Compensation per employee:** the total remuneration, in cash or in kind, that is payable by employers to employees, i.e. gross wages and salaries, as well as bonuses, overtime payments and employers' social security contributions, divided by the total number of employees.

**Consolidated balance sheet of the MFI sector:** a balance sheet obtained by netting out inter-MFI positions (e.g. inter-MFI loans and deposits) in the aggregated MFI balance sheet. It provides statistical information on the MFI sector's assets and liabilities vis-à-vis residents of the euro area not belonging to this sector (i.e. general government and other euro area residents) and vis-à-vis non-euro area residents. It is the main statistical source for the calculation of monetary aggregates, and it provides the basis for the regular analysis of the counterparts of M3.

**Debt (financial accounts):** loans, deposit liabilities, debt securities issued and pension fund reserves of non-financial corporations (resulting from employers' direct pension commitments on behalf of their employees), valued at market value at the end of the period. However, due to data limitations, the debt given in the quarterly financial accounts does not include loans granted by non-financial sectors (e.g. inter-company loans) or by banks outside the euro area, whereas these components are included in the annual financial accounts.

**Debt (general government):** the gross debt (deposits, loans and debt securities excluding financial derivatives) at nominal value outstanding at the end of the year and consolidated between and within the sectors of general government.

**Debt security:** a promise on the part of the issuer (i.e. the borrower) to make one or more payment(s) to the holder (the lender) at a specified future date or dates. Such securities usually carry a specific rate of interest (the coupon) and/or are sold at a discount to the amount that will be repaid at maturity. Debt securities issued with an original maturity of more than one year are classified as long-term.

**Debt-to-GDP ratio (general government):** the ratio of general government debt to GDP at current market prices. It is the subject of one of the fiscal criteria laid down in Article 104 (2) of the Treaty establishing the European Community to define the existence of an excessive deficit.

**Deficit (general government):** the general government's net borrowing, i.e. the difference between total government revenue and total government expenditure.

**Deficit-debt adjustment (general government):** the difference between the general government deficit and the change in general government debt.

**Deficit ratio (general government):** the ratio of the general government deficit to GDP at current market prices. It is the subject of one of the fiscal criteria laid down in Article 104 (2) of the Treaty establishing the European Community to define the existence of an excessive deficit. It is also referred to as the budget deficit ratio or the fiscal deficit ratio.

**Deflation:** a decline in the general price level, e.g. in the consumer price index.

**Deposit facility:** a standing facility of the Eurosystem which counterparties may use to make overnight deposits, remunerated at a pre-specified interest rate, at a national central bank.

**Direct investment:** cross-border investment for the purpose of obtaining a lasting interest in an enterprise resident in another economy (assumed, in practice, for ownership of at least 10% of the ordinary shares or voting power). Included are equity capital, reinvested earnings and other capital associated with inter-company operations. The direct investment account records net transactions/positions in assets abroad by euro area residents (as "direct investment abroad") and net transactions/positions in euro area assets by non-residents (as "direct investment in the euro area").

**Effective exchange rates (EERs) of the euro (nominal/real):** weighted averages of bilateral euro exchange rates against the currencies of the euro area's main trading partners. The ECB publishes nominal EER indices for the euro against two groups of trading partners: the EER-23 (comprising the 13 non-euro area EU Member States and the 10 main trading partners outside the EU) and the EER-42 (composed of the EER-23 and 19 additional countries). The weights used reflect the share of each partner country in euro area trade and account for competition in third markets. Real EERs are nominal EERs deflated by a weighted average of foreign, relative to domestic, prices or costs. They are thus measures of price and cost competitiveness.

**EONIA** (euro overnight index average): a measure of the effective interest rate prevailing in the euro interbank overnight market. It is calculated as a weighted average of the interest rates on unsecured overnight lending transactions denominated in euro, as reported by a panel of contributing banks.

**Equities:** securities representing ownership of a stake in a corporation. They comprise shares traded on stock exchanges (quoted shares), unquoted shares and other forms of equity. Equities usually produce income in the form of dividends.

**ERM II (exchange rate mechanism II):** the exchange rate arrangement that provides the framework for exchange rate policy cooperation between the euro area countries and the EU Member States not participating in Stage Three of EMU.



EURIBOR (euro interbank offered rate): the rate at which a prime bank is willing to lend funds in euro to another prime bank, computed daily for interbank deposits with different maturities of up to 12 months.

Euro area: the area formed by those EU Member States in which the euro has been adopted as the single currency in accordance with the Treaty.

European Commission surveys: harmonised surveys of business and/or consumer sentiment conducted on behalf of the European Commission in each of the EU Member States. Such questionnaire-based surveys are addressed to managers in the manufacturing, construction, retail and services industries, as well as to consumers. From each monthly survey, composite indicators are calculated that summarise the replies to a number of different questions in a single indicator (confidence indicators).

Eurosystem: the central banking system made up of the European Central Bank and the national central banks of those EU Member States that have already adopted the euro.

Eurozone Purchasing Managers' Surveys: surveys of business conditions in manufacturing and in services industries conducted for a number of countries in the euro area and used to compile indices. The Eurozone Manufacturing Purchasing Managers' Index (PMI) is a weighted indicator calculated from indices of output, new orders, employment, suppliers' delivery times and stocks of purchases. The services sector survey asks questions on business activity, expectations of future business activity, the amount of business outstanding, incoming new business, employment, input prices and prices charged. The Eurozone Composite Index is calculated by combining the results from the manufacturing and services sector surveys.

**External trade in goods:** exports and imports of goods with countries outside the euro area, measured in terms of value and as indices of volume and unit value. External trade statistics are not comparable with the exports and imports recorded in the national accounts, as the latter include both intra-euro area and extra-euro area transactions, and also combine goods and services. Nor are they fully comparable with the goods item in b.o.p. statistics. Besides methodological adjustments, the main difference is to be found in the fact that imports in external trade statistics are recorded including insurance and freight services, whereas they are recorded free on board in the goods item in the b.o.p. statistics.

Fixed rate tender: a tender procedure in which the interest rate is specified in advance by the central bank and in which participating counterparties bid the amount of money they wish to transact at the fixed interest rate.

General government: a sector defined in the ESA 95 as comprising resident entities that are engaged primarily in the production of non-market goods and services intended for individual and collective consumption and/or in the redistribution of national income and wealth. Included are central, regional and local government authorities as well as social security funds. Excluded are government-owned entities that conduct commercial operations, such as public enterprises.

Gross domestic product (GDP): the value of an economy's total output of goods and services less intermediate consumption, plus net taxes on products and imports. GDP can be broken down by output, expenditure or income components. The main expenditure aggregates that make up GDP are household final consumption, government final consumption, gross fixed capital



formation, changes in inventories, and imports and exports of goods and services (including intra-euro area trade).

**Harmonised Index of Consumer Prices (HICP):** a measure of consumer prices that is compiled by Eurostat and harmonised for all EU Member States.

**Hourly labour cost index:** a measure of labour costs, including gross wages and salaries (in cash and in kind, including bonuses) and other labour costs (employers' social contributions plus employment-related taxes paid by the employer minus subsidies received by the employer), per hour actually worked (including overtime).

**Implied volatility:** the expected volatility (i.e. standard deviation) in the rates of change of the price of an asset (e.g. a share or a bond). It can be derived from the asset's price, maturity date and exercise price of its options, as well as from a riskless rate of return, using an option pricing model such as the Black-Scholes model.

**Index of negotiated wages:** a measure of the direct outcome of collective bargaining in terms of basic pay (i.e. excluding bonuses) at the euro area level. It refers to the implied average change in monthly wages and salaries.

**Industrial producer prices:** factory-gate prices (transportation costs are not included) of all products sold by industry excluding construction on the domestic markets of the euro area countries, excluding imports.

Industrial production: the gross value added created by industry at constant prices.

Inflation: an increase in the general price level, e.g. in the consumer price index.

**Inflation-indexed government bonds:** debt securities issued by the general government, the coupon payments and principal of which are linked to a specific consumer price index.

**International reserves:** external assets readily available to and controlled by monetary authorities for directly financing or regulating the magnitude of payments imbalances through intervention in exchange markets. The international reserves of the euro area comprise non-euro denominated claims on non-euro area residents, gold, special drawing rights (SDRs) and the reserve positions in the IMF which are held by the Eurosystem.

**International investment position (i.i.p.):** the value and composition of an economy's outstanding net financial claims on (or financial liabilities to) the rest of the world.

**Job vacancies:** a collective term covering newly created jobs, unoccupied jobs or jobs about to become vacant in the near future, for which the employer has taken recent active steps to find a suitable candidate.

**Key ECB interest rates:** the interest rates, set by the Governing Council, which reflect the monetary policy stance of the ECB. They are the minimum bid rate on the main refinancing operations, the interest rate on the marginal lending facility and the interest rate on the deposit facility.



Labour force: the sum total of persons in employment and the number of unemployed.

**Labour productivity:** the output that can be produced with a given input of labour. It can be measured in several ways, but is commonly measured as GDP at constant prices divided by either total employment or total hours worked.

**Longer-term refinancing operation:** a regular open market operation executed by the Eurosystem in the form of reverse transactions. Such operations are carried out through a monthly standard tender and normally have a maturity of three months.

**M1:** a narrow monetary aggregate that comprises currency in circulation plus overnight deposits held with MFIs and central government (e.g. at the post office or treasury).

M2: an intermediate monetary aggregate that comprises M1 plus deposits redeemable at a period of notice of up to and including three months (i.e. short-term savings deposits) and deposits with an agreed maturity of up to and including two years (i.e. short-term time deposits) held with MFIs and central government.

M3: a broad monetary aggregate that comprises M2 plus marketable instruments, in particular repurchase agreements, money market fund shares and units, and debt securities with a maturity of up to and including two years issued by MFIs.

**Main refinancing operation:** a regular open market operation executed by the Eurosystem in the form of reverse transactions. Such operations are carried out through a weekly standard tender and normally have a maturity of one week.

**Marginal lending facility:** a standing facility of the Eurosystem which counterparties may use to receive overnight credit from a national central bank at a pre-specified interest rate against eligible assets.

**MFI credit to euro area residents:** MFI loans granted to non-MFI euro area residents (including general government and the private sector) and MFI holdings of securities (shares, other equity and debt securities) issued by non-MFI euro area residents.

**MFI interest rates:** the interest rates that are applied by resident credit institutions and other MFIs, excluding central banks and money market funds, to euro-denominated deposits and loans vis-à-vis households and non-financial corporations resident in the euro area.

**MFI longer-term financial liabilities:** deposits with an agreed maturity of over two years, deposits redeemable at a period of notice of over three months, debt securities issued by euro area MFIs with an original maturity of more than two years and the capital and reserves of the euro area MFI sector.

**MFI net external assets:** the external assets of the euro area MFI sector (such as gold, foreign currency banknotes and coins, securities issued by non-euro area residents and loans granted to non-euro area residents) minus the external liabilities of the euro area MFI sector (such as non-euro area residents' deposits and repurchase agreements, as well as their holdings of money market fund shares/units and debt securities issued by MFIs with a maturity of up to and including two years).

**MFIs (monetary financial institutions):** financial institutions which together form the money-issuing sector of the euro area. These include the Eurosystem, resident credit institutions (as defined in Community law) and all other resident financial institutions whose business is to receive deposits and/or close substitutes for deposits from entities other than MFIs and, for their own account (at least in economic terms), to grant credit and/or invest in securities. The latter group consists predominantly of money market funds.

**Portfolio investment:** euro area residents' net transactions and/or positions in securities issued by non-residents of the euro area ("assets") and non-residents' net transactions and/or positions in securities issued by euro area residents ("liabilities"). Included are equity securities and debt securities (bonds and notes, and money market instruments). Transactions are recorded at the effective price paid or received, less commissions and expenses. To be regarded as a portfolio asset, ownership in an enterprise must be equivalent to less than 10% of the ordinary shares or voting power.

**Price stability:** the maintenance of price stability is the primary objective of the Eurosystem. The Governing Council defines price stability as a year-on-year increase in the Harmonised Index of Consumer Prices (HICP) for the euro area of below 2%. The Governing Council has also made it clear that, in the pursuit of price stability, it aims to maintain inflation rates below, but close to, 2% over the medium term.

**Reference value for M3 growth:** the annual growth rate of M3 over the medium term that is consistent with the maintenance of price stability. At present, the reference value for annual M3 growth is  $4\frac{1}{2}$ %.

**Reserve requirement:** the minimum amount of reserves a credit institution is required to hold with the Eurosystem. Compliance is determined on the basis of the average of the daily balances over a maintenance period of around one month.

**Survey of Professional Forecasters (SPF):** a quarterly survey that has been conducted by the ECB since 1999 to collect macroeconomic forecasts on euro area inflation, real GDP growth and unemployment from a panel of experts affiliated to financial and non-financial organisations based in the EU.

**Unit labour costs:** a measure of total labour costs per unit of output calculated for the euro area as the ratio of total compensation per employee to labour productivity (defined as GDP at constant prices per person employed).

**Variable rate tender:** a tender procedure where the counterparties bid both the amount of money they wish to transact with the central bank and the interest rate at which they wish to enter into the transaction.

**Yield curve:** a graphical representation of the relationship between the interest rate or yield and the maturity at a given point in time for debt securities with the same credit risk but different maturity dates. The slope of the yield curve can be measured as the difference between the interest rates at two selected maturities.



